UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 6-K

REPORT OF FOREIGN PRIVATE ISSUER PURSUANT TO RULE 13a-16 OR 15d-16 UNDER THE SECURITIES EXCHANGE ACT OF 1934

For the month of August 2019

Commission File Number: 001-37384

GALAPAGOS NV

(Translation of registrant's name into English)

Generaal De Wittelaan L11 A3 2800 Mechelen, Belgium (Address of principal executive office)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F.

Form 20-F ⊠ Form 40-F □

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1):

Note: Regulation S-T Rule 101(b)(1) only permits the submission in paper of a Form 6-K if submitted solely to provide an attached annual report to security holders.

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7):

Note: Regulation S-T Rule 101(b)(7) only permits the submission in paper of a Form 6-K if submitted to furnish a report or other document that the registrant foreign private issuer must furnish and make public under the laws of the jurisdiction in which the registrant is incorporated, domiciled or legally organized (the registrant's "home country"), or under the rules of the home country exchange on which the registrant's securities are traded, as long as the report or other document is not a press release, is not required to be and has not been distributed to the registrant's security holders, and, if discussing a material event, has already been the subject of a Form 6-K submission or other Commission filing on EDGAR.

The information contained in this report on Form 6-K, including the exhibits but excluding the quotes of Daniel O'Day and Onno van de Stolpe contained in Exhibit 99.1, is hereby incorporated by reference into the Company's Registration Statements on Forms F-3 (File No. 333-230639) and S-8 (File Nos. 333-204567, 333-208697, 333-211834, 333-215783, 333-218160, 333-225263 and 333-231765).

Exhibit

Exhibit 99.1.	Press release dated August 23, 2019
Exhibit 99.2#.	Option, License and Collaboration Agreement dated as of July_14, 2019 by and between the registrant and Gilead Sciences, Inc.
Exhibit 99.3#.	Amended and Restated License and Collaboration Agreement dated as of August 23, 2019 by and between the registrant and Gilead Biopharmaceutics Ireland UC
Exhibit 99.4.	Subscription Agreement relating to ordinary shares in the registrant dated as of July 14, 2019 by and between the registrant and Gilead Therapeutics A1 Unlimited Company

Confidential treatment has been requested for portions of this exhibit. These portions have been omitted from this Form 6-K.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

GALAPAGOS NV

(Registrant)

/s/ Xavier Maes

Xavier Maes Company Secretary

Date: August 29, 2019





GILEAD AND GALAPAGOS COMPLETE CLOSING OF THEIR TRANSFORMATIVE RESEARCH AND DEVELOPMENT COLLABORATION

Foster City, Calif. and Mechelen, Belgium; 23 August 2019; 22.01 CET; regulated information – Gilead Sciences, Inc. (NASDAQ: GILD) and Galapagos NV (Euronext & NASDAQ: GLPG) today announced the closing of the global research and development collaboration agreement signed on 14 July 2019.

This agreement has received clearance from the U.S. Federal Trade Commission under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 and merger control approval from the Austrian Federal Competition Authority.

Under the terms of the agreement, the closing of this transaction triggers an upfront license fee payment of \$3.95 billion by Gilead to Galapagos. In addition, Gilead has made an equity investment in Galapagos of approximately \$1.1 billion (or approximately €960 million) by subscribing for new shares at a price of €140.59 per share, including issuance premium. As a result, Gilead now owns 13,589,686 ordinary shares of Galapagos, representing approximately 22 percent of the currently outstanding share capital of Galapagos.

"We are excited to close this unique agreement, which will generate both long-term strategic value and mutual, immediate benefits," said Daniel O'Day, Chairman and Chief Executive Officer of Gilead. "The collaboration reflects Gilead's intent to grow our innovation network through diverse and creative partnerships."

"This agreement is about maximizing innovation based on developing new mode of action medicines. With the capital provided by Gilead, we aim to progress innovation to patients," said Onno van de Stolpe, Chief Executive Officer of Galapagos.

In accordance with Belgian transparency legislation¹, Galapagos notes that its total share capital currently amounts to \leq 333,479,569.76; the total number of securities conferring voting rights is 61,652,086, which is also the total number of voting rights (the "denominator"), and all securities conferring voting rights and all voting rights are of the same category. The total number of rights (warrants) to subscribe to not yet issued securities conferring voting rights is 5,958,292, which equals the total number of voting rights that may result from the exercise of these warrants. Galapagos does not have any convertible bonds or shares without voting rights outstanding.

¹ Belgian Act of 2 May 2007 on the disclosure of major shareholdings in issuers whose shares are admitted to trading on a regulated market

August 23, 2019

About Galapagos

Galapagos (Euronext & NASDAQ: GLPG) discovers and develops small molecule medicines with novel modes of action, three of which show promising patient results and are currently in late-stage development in multiple diseases. The company's pipeline comprises Phase 3 through to discovery programs in inflammation, fibrosis, osteoarthritis and other indications. Galapagos' ambition is to become a leading global biopharmaceutical company focused on the discovery, development and commercialization of innovative medicines. More information at www.glpg.com.

About Gilead Sciences

Gilead Sciences, Inc. is a research-based biopharmaceutical company that discovers, develops and commercializes innovative medicines in areas of unmet medical need. The company strives to transform and simplify care for people with life-threatening illnesses around the world. Gilead has operations in more than 35 countries worldwide, with headquarters in Foster City, California. For more information on Gilead Sciences, please visit the company's website at <u>www.gilead.com</u>.

<u>Galapagos Contacts</u> Investors: Elizabeth Goodwin VP IR +1 781 460 1784	Media: Carmen Vroonen Senior Director Communications +32 473 824 874
Sofie Van Gijsel Director IR +32 485 19 14 15 ir@glpg.com	Evelyn Fox Director Communications +31 6 53 591 999 communications@glpg.com
<u>Gilead Contact</u> Investors: Sung Lee +1 (650) 524-7792	Media: Amy Flood +1 (650) 522-5643

Galapagos Forward-Looking Statements

This release may contain forward-looking statements with respect to Galapagos, including statements regarding Galapagos' strategic ambitions, the implementation of the global collaboration agreement, and the amount and timing of any payments by Gilead to Galapagos. Galapagos cautions the reader that forward-looking statements are not guarantees of future performance. Forward-looking statements involve known and unknown risks, uncertainties and other factors which might cause the actual results, financial condition and liquidity, performance or achievements of Galapagos, or industry results, to be materially different from any historic or future results, financial conditions and liquidity, performance or achievements expressed or implied by such forward-looking statements. In addition, even if Galapagos' results, performance, financial condition and liquidity, and the development of the industry in which it operates are consistent with such forward-looking statements, they may not be predictive of results or developments, clinical trial and product development activities and regulatory approval requirements (including that data from the ongoing and planned clinical research programs may not support registration or further development of its candidate products due to safety, efficacy or other reasons),

Galapagos' reliance on collaborations with third parties (including its collaboration partner Gilead), and estimating the commercial potential of its candidate products. A further list and description of these risks, uncertainties and other risks can be found in Galapagos' Securities and Exchange Commission (SEC) filings and reports, including in Galapagos' most recent annual report on Form 20-F filed with the SEC and subsequent filings and reports filed by Galapagos with the SEC. Given these uncertainties, the reader is advised not to place any undue reliance on such forward-looking statements. These forward-looking statements speak only as of the date of publication of this document. Galapagos expressly disclaims any obligation to update any such forward-looking statements in this document to reflect any change in its expectations with regard thereto or any change in events, conditions or circumstances on which any such statement is based or that may affect the likelihood that actual results will differ from those set forth in the forward-looking statements, unless specifically required by law or regulation.

Gilead Forward-Looking Statements

This press release includes forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, related to Gilead, Galapagos and the global research and development collaboration agreement that are subject to risks, uncertainties and other factors. All statements other than statements of historical fact are statements that could be deemed forward-looking statements, including all statements regarding the intent, belief or current expectation of the companies and members of their senior management team. Forward-looking statements include, without limitation, the risk that Gilead may not realize any benefits from the global collaboration agreement; its potential effects on Gilead's revenues and earnings; Gilead may fail to discover, develop and commercialize any of Galapagos' pipeline products under the agreement; the filing of the new drug applications for approval of filgotinib in the currently anticipated timeframe; approval of filgotinib by regulatory authorities, including any approvals, if granted, may have significant limitations on its use; the anticipated timing of clinical data of Galapagos' pipeline products; the possibility of unfavorable results from these clinical trials; and the accuracy of any assumptions underlying any of the foregoing. Investors are cautioned that any such forward-looking statements are not guarantees of future performance and involve risks and uncertainties and are cautioned not to place undue reliance on these forwardlooking statements. Actual results may differ materially from those currently anticipated due to a number of risks and uncertainties. Risks and uncertainties that could cause the actual results to differ from expectations contemplated by forward-looking statements include: the occurrence of any event, change or other circumstance that could give rise to the termination of the collaboration agreement; the effects of the transaction on relationships with employees, customers, other business partners or governmental entities; transaction costs; the risk Galapagos' stockholders do not approve Gilead's board nominees or issuance of the warrants, as the case may be. These and other risks are described in detail in Gilead's Quarterly Report on Form 10-Q for the quarter ended June 30, 2019, as filed with the U.S. Securities and Exchange Commission. All forward-looking statements are based on information currently available to Gilead, and Gilead assumes no obligation to update any such forward-looking statements.

CERTAIN CONFIDENTIAL INFORMATION IN THIS EXHIBIT HAS BEEN OMITTED AND REPLACED WITH "[...***...]" BECAUSE IT IS NOT MATERIAL AND WOULD BE COMPETITIVELY HARMFUL IF PUBLICLY DISCLOSED.

OPTION, LICENSE AND COLLABORATION AGREEMENT

by and between

Galapagos NV

and

Gilead Sciences, Inc.

dated as of July 14, 2019

TABLE OF CONTENTS

Page
2
13
16
19
23
25
27
29
39
45
60
68
72
77
85
88
90

i

List of Appendices

Appendix A	Definitions
List of Schedules	
Schedule A:	Access Territory
Schedule B:	Excluded Programs
Schedule C:	Existing Galapagos Third Party Obligations Schedule
Schedule D:	Structure of GLPG1690
Schedule E:	Structure of GLPG1972
Schedule F:	Galapagos Option Exercise Representations
Schedule G:	Galapagos Programs Existing as of the Execution Date and their Collaboration Targets
Schedule H:	Galapagos Territory
Schedule I:	Qualifying Data Package
Schedule J:	Form of Security Agreement
Schedule 3.1(b):	Initial R&D Plan and Budget for Autotaxin Program
Schedule 8.10-1	Shared Program Process Definitions
Schedule 8.10-2	Shared Program Process
Schedule 8.10-3	Baseball Arbitration Bookends
Schedule 11.2:	Galapagos Schedule of Exceptions
[***]	
Schedule 13.3(a):	Form of Press Release

i

OPTION, LICENSE AND COLLABORATION AGREEMENT

THIS OPTION, LICENSE AND COLLABORATION AGREEMENT (this "Agreement") is entered into as of July 14, 2019 (the "Execution Date") by and between GALAPAGOS NV, a corporation organized under the laws of Belgium and having its principal place of business at Generaal De Wittelaan L11 A3, 2800 Mechelen, Belgium ("Galapagos"), and GILEAD SCIENCES, INC., a Delaware corporation having its principal place of business at 333 Lakeside Drive, Foster City, CA, 94404, USA ("Gilead"). Galapagos and Gilead are sometimes referred to herein individually as a "Party" and collectively as the "Parties." For purposes of this Agreement, the capitalized terms defined in <u>Appendix A</u> or in other provisions of this Agreement shall have the meanings provided in <u>Appendix A</u> or such other provisions.

BACKGROUND

WHEREAS, Galapagos is in the business of discovering, researching and developing biopharmaceutical products, including conducting Galapagos Programs (as defined below);

WHEREAS, Galapagos Controls (as defined below) and may in the future Control certain intellectual property and other rights relating to the discovery, research and development of biopharmaceutical products, including the Galapagos Programs;

WHEREAS, Galapagos wishes to grant, and Gilead wishes to accept, effective as of the Effective Date, a research and development license under the Galapagos IP as contemplated herein;

WHEREAS, Galapagos wishes to grant, and Gilead wishes to accept, an Option to collaborate with respect to each Galapagos Program (each, as defined below) as contemplated herein, which Option will become exercisable upon the completion of the first Phase 2 Clinical Trial (or if there is no Phase 2 Clinical Trial, Phase 3 Clinical Trial) for all Galapagos Products in such Galapagos Program;

WHEREAS, the Parties will co-develop any Optioned Molecules and Optioned Products (each, as defined below) with respect to each Galapagos Program that becomes an Optioned Program and, subject to global coordination and under the oversight of the JCC (as defined below) as contemplated hereunder, each Party will have the right to Commercialize the Optioned Products in its Respective Territory (each, as defined below);

WHEREAS, Galapagos and Gilead have agreed to collaborate on the Autotaxin Program (as defined below), including GLPG1690, on and after the Effective Date, as contemplated herein; and

WHEREAS, simultaneous with the execution and delivery of this Agreement, the Parties have entered into that certain Subscription Agreement by and between Galapagos and an Affiliate of Gilead (including the warrants attached thereto, the "**Subscription Agreement**"), which provides for the issuance by Galapagos, and the subscription by Gilead, of a number of Galapagos' ordinary shares and of warrants to purchase shares.

NOW THEREFORE, in consideration of the foregoing premises and the mutual promises, covenants and conditions contained in this Agreement, the Parties agree as follows:

ARTICLE I

GOVERNANCE

1.1 Joint Steering Committee.

(a) Formation; Composition. The Parties hereby establish a joint steering committee (the "Joint Steering Committee" or "JSC"), which shall have the responsibilities set forth in Section 1.1(b). Each Party shall initially appoint four (4) representatives to the JSC, all of whom will have sufficient seniority within such Party to make decisions arising within the scope of the JSC's responsibilities and each of whom shall have experience in one of the following functions: research, development, commercialization and manufacturing. The Parties shall notify each other of their respective initial representatives to the JSC within [...***...] Business Days after the Effective Date. The JSC may change its size from time to time if agreed by consensus among its members; *provided that* the JSC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JSC representatives at any time upon written notice to the other Party. Either Party may invite non-members to participate in the discussions and meetings of the JSC provided notice is given to the other Party, and such non-members shall (i) be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII and (ii) have no voting authority at the JSC. The JSC shall have a chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos or Gilead. The initial chairperson shall be to convene and preside at meetings of the JSC. The chairperson shall have no additional powers or rights beyond those held by the other JSC representatives.

(b) Responsibilities.

(i) Generally. The JSC shall have the following general responsibilities:

- (A) discuss the activities of the Parties under this Agreement;
- (B) serve as a forum for the sharing of information with respect to each Galapagos Program;
- (C) review, discuss and, if applicable, provide comments on any materials or information delivered to the JSC pursuant to

this Agreement;

(D) review and approve press releases submitted to the JSC by the JCRC; and

(E) attempt to resolve disputes within the JSC or presented to the JSC by the JDC, JCC, JCRC or any other Committee.

(ii) <u>For Galapagos Programs</u>. Without limitation to the foregoing, with respect to each Galapagos Program, the JSC shall determine whether a Clinical Trial constitutes a Triggering Clinical Trial and whether and when a Qualifying Data Package has been delivered, in each case, for such Galapagos Program as further described in Sections 2.3(c) and 2.3(d).

(iii) For Optioned Programs. Without limitation to the foregoing, the JSC shall have the following responsibilities with respect to each Optioned Program:

(A) monitor and provide strategic oversight of such Optioned Program and facilitate communications between the Parties with respect to the Development, Manufacture and Commercialization of Optioned Molecules and Optioned Products included in such Optioned Program;

(B) review and approve the R&D Plan and Budget, Global Manufacturing Plan and Budget and Global Commercialization Plan and Budget for such Optioned Program and, in each case, any amendments thereto proposed by the applicable Committee, *provided* that the final initial draft R&D Plan and Budget for the Autotaxin Program agreed by the Parties prior to the Effective Date shall be deemed approved by the JSC;

(C) review and approve the design of each Phase 2 Clinical Trial and Phase 3 Clinical Trial and any Phase 4 Clinical Trials in each case included in the R&D Plan and Budget and Mutual Post-Approval Commitments that are Initiated with respect to such Optioned Program after the Initial Option Closing for such Optioned Program;

(D) discuss strategies regarding intellectual property and new indications for such Optioned Program;

Program; and

(E) in accordance with Section 7.2(d), review and discuss any Independent Activities Plan with respect to such Optioned

(F) allocate responsibility between the Parties with respect to any Mutual Post-Approval Commitments; which, unless mutually agreed, shall provide for Galapagos conduct of activities in the Galapagos Territory and Gilead conduct of activities in the Gilead Territory.

(c) Meetings. The JSC shall meet at least once per calendar quarter during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JSC, the Alliance Manager of the Party whose representative is the chairperson, in collaboration with the chairperson of the JSC, shall prepare and circulate an agenda for such meeting; provided, however, that either Party may propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JSC by providing at least [...***...] Business Days (or, in the case of meetings to determine if a Clinical Trial constitutes a Triggering Clinical Trial or, if or when a Qualifying Data Package has been delivered, [...***...] Business Days) prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JSC and Alliance Managers of each Party to provide the members of the JSC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The Parties may mutually agree to additional meetings on shorter or longer notice in the event that matters arise requiring JSC consideration. The JSC may meet in-person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one (1) meeting per calendar year shall be in-person unless the Parties mutually agree in writing to waive such requirement. In-person JSC meetings will be held at locations alternately selected by Galapagos and by Gilead, with [...***...] selecting the location of the first in-person JSC meeting, or at any other location mutually agreed by the members of the JSC. Each Party shall bear the cost and expense of its respective JSC members' participation in JSC meetings. Meetings of the JSC shall be effective only if at least two (2) representatives of each Party (which representatives are not such Party's Alliance Manager) are present or participating in such meeting. The Alliance Manager of the Party whose representative is the chairperson shall be responsible for preparing reasonably detailed written minutes of all JSC meetings that reflect material decisions made and action items identified at such meetings. Such Alliance Manager shall send draft meeting minutes to each member of the JSC for review and approval within [...***...] Business Days after each JSC meeting. Such minutes shall be deemed

approved unless one (1) or more members of the JSC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [... ***...] Business Days of receipt, in which case such Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JSC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines as set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. If agreed upon by the JSC, the minutes shall be promptly signed by the Alliance Managers.

(d) <u>Decision-Making</u>. The representatives from each Party on the JSC shall have, collectively, one (1) vote on behalf of that Party, and, subject to Section 3.7(c)(vii), all decisions shall be made by consensus. Disputes at the JSC shall be handled in accordance with Section 1.5.

1.2 Joint Development Committee.

(a) Formation; Composition. Within [...***...] days after the Effective Date, the Parties shall establish a joint development committee (the "Joint Development Committee" or "JDC"), which shall have the responsibilities set forth in Section 1.2(b). Each Party shall initially appoint three (3) representatives to the JDC, with each representative having knowledge and expertise in the development of compounds, Molecules and products similar to the Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products and having sufficient seniority within such Party to make decisions arising within the scope of the JDC's responsibilities. The JDC may change its size from time to time if agreed by consensus among its members; *provided that* the JDC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JDC representatives at any time upon written notice to the other Party. Either Party may invite non-members to participate in the discussions and meetings of the JDC provided notice is given to the other Party and such non-members shall (i) be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII and (ii) have no voting authority at the JDC. The JDC shall have a chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos or Gilead. The initial chairperson shall be selected by [...***...]. The role of the chairperson shall be to convene and preside at meetings of the JDC, but the chairperson shall have no additional powers or rights beyond those held by the other JDC representatives.

(b) Responsibilities.

(i) <u>Generally</u>. The JDC shall have the following general responsibilities:

(A) serve as a forum for the sharing of information with respect to the Development and Manufacture of Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products included in each Galapagos Program and Optioned Program (as applicable); and

(B) review, discuss and, if applicable, provide comments on any materials or information delivered to the JDC pursuant to

this Agreement.

(ii) <u>For Pre-Program Activities and Galapagos Programs</u>. Without limitation to the foregoing, the JDC shall have the following responsibilities with respect to Pre-Program Activities and each Galapagos Program:

(A) serve as a forum for the sharing of information with respect to Galapagos R&D Activities, including to discuss the results of any Galapagos R&D Activities provided to the JDC pursuant to Section 2.3, the results of any target screens requested by Gilead pursuant to Section 2.5 or any Gilead Contributions; and

(B) with respect to each Galapagos Program, during the period commencing on delivery of the Option Exercise Notice for such Galapagos Program and ending on the applicable Option Exercise Closing with respect to each country in the Territory for such Galapagos Program, to the extent permitted by Applicable Law, the JDC shall plan for the transition to Gilead of any Development and Manufacturing activities that will be assigned to Gilead by the JSC under the R&D Plan and Budget or the Global Manufacturing Plan and Budget, in each case, for such Galapagos Program once it becomes an Optioned Program.

(iii) <u>For Optioned Programs</u>. Without limitation to the foregoing, the JDC shall have the following responsibilities with respect to each Optioned Program:

(A) monitor and provide strategic oversight of the Development activities of the Parties with respect to Optioned Molecules and Optioned Products included in such Optioned Program (including the conduct of the R&D Plan and Budget) and facilitate communications between the Parties with respect to such Development activities, including with respect to any discussions with, and commitments to or agreements with, Regulatory Authorities (including post-approval commitments) with respect to any such Optioned Product;

(B) monitor and provide strategic oversight of the Manufacturing activities of the Parties with respect to Optioned Molecules and Optioned Products included in such Optioned Program (including the conduct of the Global Manufacturing Plan and Budget and formulation, validation, scale up, and other activities to maintain supply for Development), and facilitate communications between the Parties with respect to such Manufacturing activities;

(C) prepare the R&D Plan and Budget and Global Manufacturing Plan and Budget for such Optioned Program, in each case, for review and approval by the JSC;

(D) review and approve the final substance and form of each Publication covering activities (or the results thereof) in any R&D Plan and Budget or Independent Activities Plan;

(E) review, at least annually, the R&D Plan and Budget and Global Manufacturing Plan and Budget for such Optioned Program, and, in each case, suggest any applicable amendments thereto for review and approval by the JSC;

(F) discuss when to initiate or discontinue any Clinical Trial or Nonclinical Study under the R&D Plan and Budget for such Optioned Program; *provided that* nothing herein is intended to limit a Party's ability to comply with Applicable Law or manage subject safety;

(G) review the conduct of all Clinical Trials and Nonclinical Studies conducted under the R&D Plan and Budget for such Optioned Program, including any Requested or Required Phase 4 Clinical Trials and any Mutual Post-Approval Commitments, in each case, included in such R&D Plan and Budget;

(H) agree upon a Technology Transfer Plan for such Optioned Program;

(I) in accordance with Section 7.2(b), review and discuss any Independent Activities Plan with respect to such Optioned Program, and present any such Independent Activities Plan and any comments with respect thereto to the JSC for its review and discussion;

(J) review the progress of any Independent Activities with respect to such Optioned Program; and

(K) serve as a forum to discuss interactions with key opinion leaders and patient advocacy groups.

(c) Meetings. The JDC shall meet at least once per calendar quarter during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JDC, the Alliance Manager of the Party whose representative is the chairperson, in collaboration with the chairperson of the JDC, shall prepare and circulate an agenda for such meeting; provided, however, that either Party may propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JDC by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JDC and Alliance Managers of each Party to provide the members of the JDC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The Parties may mutually agree to additional meetings on shorter or longer notice in the event that matters arise requiring JDC attention. The JDC may meet in-person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one (1) meeting per calendar year shall be in-person unless the Parties mutually agree in writing to waive such requirement. In-person JDC meetings will be held at locations alternately selected by Galapagos and by Gilead, with Gilead selecting the location of the first in-person JDC meeting, or at any other location mutually agreed by the members of the JDC. Each Party shall bear the cost and expense of its respective JDC members' participation in JDC meetings. Meetings of the JDC shall be effective only if at least two (2) representatives of each Party (which representatives are not such Party's Alliance Manager) are present or participating in such meeting. The Alliance Manager of the Party whose representative is the chairperson shall be responsible for preparing reasonably detailed written minutes of all JDC meetings that reflect material decisions made and action items identified at such meetings. Such Alliance Manager shall send draft meeting minutes to each member of the JDC for review and approval within [...***...] Business Days after each JDC meeting. Such minutes shall be deemed approved unless one (1) or more members of the JDC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case such Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JDC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines as set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. If agreed upon by the JDC, the minutes shall be promptly signed by the Alliance Managers.

(d) <u>Galapagos Program Updates</u>. At least once per calendar quarter Galapagos shall, through the JDC, provide an update regarding the Galapagos Programs, [...***...]. At least once per calendar quarter Gilead shall, through the JDC, provide an update regarding the Gilead Contributions per Section 2.3(f).

(e) <u>Decision-Making</u>. Subject to the remainder of this Section 1.2(e), Section 1.5 and Section 3.7(c)(vii), the JDC shall act by consensus on matters within its jurisdiction. The representatives from each Party on the JDC shall have, collectively, one (1) vote on behalf of that Party. If the JDC cannot reach consensus on an issue over which it has decision-making authority within [...***...] days after the first meeting in which such issue was raised, then either Party may refer such matter to the JSC for resolution in accordance with Sections 1.1(d) and 1.5, except for any dispute regarding the final substance and form of any Publication covering activities (or the results thereof) in any R&D Plan and Budget or Independent Activities, which dispute shall be escalated pursuant Section 1.5(b)(ii)(B)(4) without escalation to the JSC.

1.3 Joint Commercialization Committee.

(a) <u>Formation; Composition</u>. Within [...***...] days after the Effective Date, the Parties shall establish a joint commercialization committee (the "Joint Commercialization Committee" or "JCC") to oversee Commercialization of Optioned Products. Each Party shall initially appoint three (3) representatives to the JCC, with each representative having knowledge and expertise in the commercialization of products similar to the Optioned Products and having sufficient seniority within such Party to make decisions arising within the scope of the JCC's responsibilities. The JCC may change its size from time to time if agreed by consensus among its members; *provided that* the JCC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JCC representatives at any time upon written notice to the other Party. Either Party may invite non-members to participate in the discussions and meetings of the JCC provided notice is given to the other Party and such non-members shall (i) be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII and (ii) have no voting authority at the JCC. The JCC shall have a chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos or Gilead. The initial chairperson shall be selected by [...***...]. The role of the chairperson shall be to convene and preside at meetings of the JCC, but the chairperson shall have no additional powers or rights beyond those held by the other JCC representatives.

(b) <u>Responsibilities</u>. The JCC shall have the following responsibilities with respect to each Optioned Program:

(i) prepare the Global Commercialization Plan and Budget for such Optioned Program and amendments thereto, in each case, for review and approval by the JSC;

(ii) review, at least once every [...***...] months, the Global Commercialization Plan and Budget for such Optioned Program and suggest any applicable amendments thereto for review and approval by the JSC;

(iii) (A) oversee the establishment of a Global Optioned Product Trademark throughout the Territory in accordance with Sections 5.2(b) and 10.9(a), (B) determine the allocation between the Parties of the ownership of and the control of obtaining, maintaining, enforcing and defending the Global Optioned Product Trademarks and (C) establish each Party's strategy to use the Global Optioned Product Trademarks in connection with the Commercialization of the applicable Optioned Products;

(iv) review any Global Promotional Materials;

(v) review and discuss uses of the Global Optioned Product Trademarks with respect to Optioned Products included in such Optioned Program in accordance with Sections 5.2(b) and 10.9(a); review and approve any proposed Regional Optioned Product Trademarks with respect to Optioned Products included in such Optioned Program in accordance with Section 10.9(b);

(vi) discuss and agree upon any issues set forth in Section 5.3 with respect to such Optioned Program, and agree upon any methods to address such issues; and

(vii) serve as a forum to discuss plans for appointing Distributors by either Party or any of its Affiliates for one (1) or more countries in such Party's Respective Territory.

(c) Meetings. The JCC shall meet at least once per calendar quarter during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JCC, the Alliance Manager of the Party whose representative is the chairperson, in collaboration with the chairperson of the JCC, shall prepare and circulate an agenda for such meeting; provided, however, that either Party shall be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JCC by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JCC and Alliance Managers of each Party to provide the members of the JCC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The Parties may mutually agree to additional meetings on shorter or longer notice in the event that matters arise requiring JCC consideration. The JCC may meet in-person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one (1) meeting per calendar year shall be in-person unless the Parties mutually agree in writing to waive such requirement. In-person JCC meetings will be held at locations alternately selected by Galapagos and by Gilead, with [...***...] selecting the location of the first in-person JCC meeting, or at any other location mutually agreed by the members of the JCC. Each Party shall bear the cost and expense of its respective JCC members' participation in JCC meetings. Meetings of the JCC shall be effective only if at least two (2) representatives of each Party (which representatives are not such Party's Alliance Manager) are present or participating in such meeting. The Alliance Manager of the Party whose representative is the chairperson shall be responsible for preparing reasonably detailed written minutes of all JCC meetings that reflect material decisions made and action items identified at such meetings. Such Alliance Manager shall send draft meeting minutes to each member of the JCC for review and approval within [...***...] Business Days after each JCC meeting. Such minutes shall be deemed approved unless one (1) or more members of the JCC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case such Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JCC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines as set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. If agreed upon by the JCC, the minutes shall be promptly signed by the Alliance Managers.

(d) <u>Decision-Making</u>. Subject to the remainder of this Section 1.3(d), Section 1.5 and Section 3.7(c)(vii), the JCC shall act by consensus on matters within its jurisdiction. The representatives from each Party on the JCC shall have, collectively, one (1) vote on behalf of that Party. If the JCC cannot reach consensus on an issue over which it has decision-making authority within [...***...] days after the first meeting in which such issue was raised, then either Party may refer such matter to the JSC for resolution in accordance with Sections 1.1(d) and 1.5.

1.4 Joint Communication Review Committee.

(a) <u>Formation; Composition</u>. Within [...***...] days after the Effective Date, the Parties shall establish a joint communications review committee (the "Joint Communication Review Committee" or "JCRC") to oversee publications and other public communications related to Optioned Programs (including Optioned Molecules and Optioned Products). Each Party shall initially appoint three (3) representatives to the JCRC, with each representative having knowledge and expertise in the communications related to products similar to the Optioned Products and having sufficient seniority within such Party to make decisions arising within the scope of the JCRC's responsibilities. The JCRC may change its size from time to time if agreed by consensus among its members; *provided that* the JCRC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JCRC representatives at any time upon written notice to the other Party. Either Party may invite non-members to participate in the discussions and meetings of the JCRC provided notice is given to the other Party and such non-members shall (i) be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII and (ii) have no voting authority at the JCRC. The JCRC shall have a

chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos or Gilead. The initial chairperson shall be selected by [...***...]. The role of the chairperson shall be to convene and preside at meetings of the JCRC, but the chairperson shall have no additional powers or rights beyond those held by the other JCRC representatives.

(b) <u>Responsibilities</u>. The JCRC shall have the following responsibilities with respect to each Optioned Program:

(i) serve as a forum to discuss long term strategies and policies regarding press releases, Publications, conferences, and other public communications for Optioned Programs;

(ii) serve as a forum to discuss conference attendance in accordance with Section 5.5;

(iii) serve as a forum to discuss interactions with the medical community and patient advocacy groups;

(iv) review and, where required by Section 13.4(a), approve each Party's Publication plan;

(v) review Publications for Optioned Programs;

(vi) conduct initial review of draft press releases and submit the same to the JSC for final approval; and

(vii) coordinate communications with the medical community regarding such Optioned Program, including what information will be disclosed regarding such Optioned Program.

(c) Meetings. The JCRC shall meet at least once per calendar quarter during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JCRC, the Alliance Manager of the Party whose representative is the chairperson, in collaboration with the chairperson of the JCRC, shall prepare and circulate an agenda for such meeting; provided, however, that either Party shall be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JCRC by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a Publication, press release, or other communication or significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JCRC and Alliance Managers of each Party to provide the members of JCRC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The Parties may mutually agree to additional meetings on shorter or longer notice in the event that matters arise requiring JCRC consideration. The JCRC may meet in-person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one (1) meeting per calendar year shall be in-person unless the Parties mutually agree in writing to waive such requirement. In-person JCRC meetings will be held at locations alternately selected by Galapagos and by Gilead, with [...***...] selecting the location of the first in-person JCRC meeting, or at any other location mutually agreed by the members of the JCRC. Each Party shall bear the cost and expense of its respective JCRC members' participation in JCRC meetings. Meetings of the JCRC shall be effective only if at least two (2) representatives of each Party (which representatives are not such Party's Alliance Manager) are present or participating in such meeting. The Alliance Manager of the Party whose representative is the chairperson shall be responsible for preparing reasonably detailed written minutes of all JCRC meetings that reflect material decisions made and action items identified at

such meetings. Such Alliance Manager shall send draft meeting minutes to each member of the JCRC for review and approval within [...***...] Business Days after each JCRC meeting. Such minutes shall be deemed approved unless one (1) or more members of the JCRC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case such Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JCRC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines as set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. If agreed upon by the JCRC, the minutes shall be promptly signed by the Alliance Managers. The JCRC may establish *ad hoc* procedures for approval of Publications outside of the JCRC meetings.

(d) <u>Decision-Making</u>. Subject to the remainder of this Section 1.4(d), Section 1.5 and Section 3.7(c)(vii), the JCRC shall act by consensus on matters within its jurisdiction. The representatives from each Party on the JCRC shall have, collectively, one (1) vote on behalf of that Party. If the JCRC cannot reach consensus on an issue over which it has decision-making authority within [...***...] days after the first meeting in which such issue was raised, then either Party may refer such matter to the JSC for resolution in accordance with Sections 1.1(d) and 1.5.

1.5 Resolution of Committee Disputes.

(a) <u>Within Committees</u>. Subject to Section 3.7(c)(vii), if a dispute arises with respect to a matter within the decision-making jurisdiction of a Committee other than the JSC that cannot be resolved within the applicable Committee, then either Party may refer such dispute to the JSC for resolution in accordance with Section 1.1(d) and this Section 1.5. For clarity, any dispute with respect to a matter that is outside the jurisdiction of a Committee, including any dispute with respect to any alleged failure to perform, or breach of, this Agreement, or any issue relating to the interpretation or application of this Agreement shall be resolved pursuant to Section 15.1(b), and not pursuant to this Section 1.5.

(b) Within the JSC.

(i) Pre-Program Activities and Galapagos Programs.

(A) <u>Galapagos Programs Generally</u>. For each Galapagos Program, if the JSC cannot reach consensus on a matter within its jurisdiction (including any matter referred to it by any other Committee) within [...***...] Business Days (or [...***...] Business Days for Qualifying Data Package disputes) after a Party affirmatively states in writing that a decision must be made, then such dispute shall be decided in accordance with Section 1.5(b)(i)(B), 1.5(b)(i)(C) or Section 1.5(b)(i)(D), as applicable.

(B) <u>Disputes Regarding Pre-Program Activities</u>. If such dispute relates to any Pre-Program Activities, then [...***...] shall have final decision-making authority; *provided that* [...***...] shall not have the right to exercise such final decision-making authority in a manner that would require [...***...].

(C) <u>Disputes Regarding Triggering Clinical Trial or Qualifying Data Package Determination</u>. If such dispute is regarding whether a Clinical Trial constitutes a Triggering Clinical Trial or whether or when a Qualifying Data Package has been delivered, in each case, for any Galapagos Program, then either Party may refer such dispute to the Parties' Subject Matter Experts for Development after the [...***...] Business Day after a Party requests a meeting of the JSC to make such a determination. Following such referral, such Subject Matter Experts shall attempt to reach consensus on such dispute during a period of [...***...] Business Days, and any final decision agreed to in writing by

the Subject Matter Experts with respect to such dispute shall be binding on the Parties. If such Subject Matter Experts cannot reach consensus on such dispute within such period, then either Party may refer such dispute to the Executive Officers of the Parties for resolution. Following referral to the Executive Officers, the Executive Officers shall attempt to reach consensus on such dispute during a period of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers with respect to such dispute shall be binding on the Parties. If the Executive Officers cannot reach consensus on such dispute reason such dispute is unresolved for more than [...***...] Business Days after the initial referral of such dispute to the Parties' Subject Matter Experts, then either Party may refer such dispute for resolution [...***...].

(D) <u>Other Disputes</u>. Except as provided in Section 1.5(b)(i)(A), 1.5(b)(i)(B) or 1.5(b)(i)(C), for any other dispute within the decision-making jurisdiction of the JSC with respect to a Galapagos Program, either Party may refer such dispute to the Executive Officers for resolution. Following referral to the Executive Officers, the Executive Officers shall attempt to reach consensus on such dispute during a period of [... ***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers with respect to such dispute shall be binding on the Parties. If the Executive Officers cannot reach consensus on such dispute within such period, then [...***...] shall have final decision-making authority; *provided that* [...***...] shall not have the right to exercise such final decision-making authority in a manner that would require [...***...]

(ii) Optioned Programs.

(A) <u>Generally</u>. Subject to Section 10.3, for each Optioned Program, if the JSC cannot reach consensus on any matter within its jurisdiction (including any matter referred to it by any other Committee) within [...***...] Business Days after a Party affirmatively states in writing that a decision must be made, then such dispute shall be decided in accordance with [...***...], as applicable.

(B) Escalation Procedure. Either Party may refer such dispute to the applicable Subject Matter Experts for each Party. Following such referral, such Subject Matter Experts shall attempt to reach consensus on such dispute during a period of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Subject Matter Experts with respect to such dispute shall be binding on the Parties. If such Subject Matter Experts cannot reach consensus on any such dispute within such period, then either Party shall have the right to refer such dispute to the Executive Officers of the Parties for resolution. Following referral to the Executive Officers, the Executive Officers shall attempt to reach consensus on such dispute during a period of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers with respect to such dispute shall be binding on the Parties. If the Executive Officers cannot reach consensus on any such dispute within such period, then such dispute shall be binding on the Parties. If the Executive Officers cannot reach consensus on any such dispute within such period, then such dispute shall be binding on the Parties. If the Executive Officers cannot reach consensus on any such dispute within such period, then such dispute shall be decided in accordance with the remainder of this Section 1.5(b)(ii)(B).

(1) <u>Disputes Regarding [...***...</u>]. If such dispute is regarding the [...***...], then (I) [...***...] and (II) [...***...].

(2) <u>Disputes Regarding [...***...</u>]. If such dispute is regarding [...***...], then [...***...].

(3) <u>Disputes Regarding [...***...]</u>. If such dispute is regarding [...***...], then either Party may refer such dispute for

resolution [...***...].

(4) [...***...]. If such dispute is regarding [...***...], then [...***...]. The provisions of Section 1.4 shall be without prejudice to the right and ability of each Party to comply with its disclosure obligations pursuant to Applicable Law or the rules of a stock exchange on which the securities of the disclosing Party (or its parent entity) are listed (or to which an application for listing has been submitted), taking into account the provisions of Section 13.4(b).

(C) <u>Other Disputes</u>. If such dispute is regarding any matter not covered by clauses (1) through (4) above, then either Party may refer such dispute to [...***...].

1.6 <u>Good Faith</u>. In conducting themselves on Committees, and in exercising their rights under this ARTICLE I, all representatives of each Party shall consider reasonably and in good faith all input received from the other Party.

1.7 <u>General Committee Authority</u>. Each Committee shall have solely the powers expressly assigned to it in this ARTICLE I and elsewhere in this Agreement. No Committee shall have any power to amend, modify, or waive compliance with this Agreement, or to require a Party to share any information other than as obligated pursuant to this Agreement. It is expressly understood and agreed that the control of decision-making authority by either Party pursuant to this ARTICLE I, so as to resolve a disagreement or deadlock on a Committee for any matter, shall not authorize either Party to perform any function or exercise any decision-making right not delegated to a Committee or such Party, and that neither Galapagos nor Gilead shall have any right to unilaterally modify or amend, or waive its own compliance with, the terms of this Agreement. Provisions that require a Committee hereunder to "agree," "consent" or "approve" or the like shall require that such agreement, consent, approval or the like be specific and reflected in approved minutes of the Committee. For clarity, no Committee has any power, decision-making authority, or right to receive any information, for any Excluded Programs.

1.8 Additional Committees and Working Groups.

(a) The Parties may agree in writing to establish such additional committees (*e.g.*, a joint manufacturing committee, joint technology transfer committee or a separate JSC, JDC, JCC or JCRC, or other committee that is specific to one or more Galapagos Programs or Optioned Programs) as they mutually deem necessary to achieve the objectives of this Agreement.

(b) Each Committee may establish and delegate duties to directed teams ("**Working Groups**") as needed to oversee particular projects or activities (*e.g.*, to prepare initial drafts of the R&D Plan and Budget, Global Manufacturing Plan and Budget, Global Commercialization Plan and Budget, and Technology Transfer Plan). Each such Working Group shall (i) have equal representation from each Party, unless otherwise mutually agreed, (ii) be subject to the approval of, oversight of, and shall report to, the Committee that formed such Working Group, and (iii) have no greater authority than the Committee that formed such Working Group. All decisions of a Working Group shall be by consensus. Any disagreement between the designees of Parties on a Working Group shall be referred to the Committee that formed the Working Group for resolution.

1.9 <u>Appointment of Alliance Managers</u>. Each Party shall appoint an appropriately qualified employee who is not a representative on the JSC to have alliance management responsibility under this Agreement (such employee, an "**Alliance Manager**") and who shall attend all Committee meetings as an observer. Such persons shall endeavor to assure clear and responsive communication between the Parties and the effective exchange of information, and may serve as a single point of contact for any matters arising under this Agreement. The Alliance Managers shall not have any authority under this Agreement. Each Alliance Manager may, in his/her discretion, appoint one (1) or more assistant alliance managers and delegate any obligation of such Alliance Manager to any such assistant alliance manager.

1.10 <u>Disbandment</u>. Any Committee or Working Group under this Agreement may be dissolved upon the mutual written agreement of the Parties and as set forth in Section 3.7(c)(vi).

ARTICLE II

GALAPAGOS R&D ACTIVITIES AND OTHER TERMS

2.1 <u>Galapagos R&D Activities</u>. Galapagos shall be solely responsible, in its discretion and at its sole cost and expense, for conducting all (a) Pre-Program Activities and (b) Development activities with respect to Galapagos Programs, including, in each case ((a) and (b)) (as applicable), all applicable pre-clinical and non-clinical research activities, regulatory activities (including filing INDs) and clinical activities (collectively, the "Galapagos R&D Activities"), except to the extent otherwise agreed by the Parties in writing or set forth in this Agreement.

2.2 <u>Gilead Contributions to Galapagos R&D Activities</u>. If mutually agreed by the Parties in writing, Gilead may provide, [...***...], support for the Galapagos R&D Activities (collectively, the "**Gilead Contributions**") and, for clarity, unless otherwise agreed by the Parties in writing, the terms of ARTICLE X shall apply with respect to any Gilead Contributions, [...***...]. If the Parties do not reach agreement on [...***...]. For clarity, pursuant to Section 8.5, Galapagos shall retain rights under the Galapagos IP and Joint Collaboration IP to perform any Gilead Contribution that Gilead fails to conduct.

2.3 Information Sharing for Galapagos R&D Activities.

(a) <u>Generally</u>. Galapagos shall keep Gilead reasonably apprised, via the JDC, of any Galapagos R&D Activities by providing: (i) quarterly (A) a report [...***...]; (B) any clinical Development plans with respect to each Galapagos Program; (C) any updated versions of any such clinical Development plans, as such updates occur; and (ii) at Gilead's reasonable request (made no more than quarterly), copies of [...***...] with respect to such Significant Pre-Program Activities and each Galapagos Program.

(b) <u>Phase 2 Clinical Trials and Phase 3 Clinical Trials</u>. At least [...***...] days prior to Initiating any Phase 2 Clinical Trial or Phase 3 Clinical Trial for a Galapagos Product, Galapagos shall provide to the JDC for review the [...***...] any Regulatory Materials Controlled by Galapagos or its Affiliates with respect to such Clinical Trial. No later than [...***...] days after the Completion Date of any Phase 2 Clinical Trial or Phase 3 Clinical Trial for a Galapagos Product, Galapagos shall notify the JDC of such Completion Date. Galapagos shall provide Gilead with a copy of the [...***...] for such Clinical Trial as soon as reasonably practicable (and in no event later than [...***...] days after) completion of the final study report with respect thereto.

(c) Information Provided in Connection with a Triggering Clinical Trial Determination.

(i) In addition to the information shared pursuant to Section 2.3(b), if Galapagos in good faith believes such Clinical Trial is reasonably likely to constitute the Triggering Clinical Trial with respect to the applicable Galapagos Program, then no later than [...***...] days after the Initiation of such Clinical Trial (or [...***...] days after the Effective Date with respect to any Clinical Trial Initiated prior to the Effective Date that has not reached its Completion Date as of the Effective Date), Galapagos shall (A) establish a Data Room containing, [...***...] any Regulatory Materials (including the regulatory correspondence and complete IND for such Clinical Trial) for such Galapagos Program Controlled by Galapagos or its Affiliates with respect to such Clinical Trial (such information with respect to a given Galapagos Program, a "**TCT Determination Package**") and (B) notify the JSC and Gilead of such Initiation, which notice shall also include instructions and credentials with which each JSC member and Gilead may access the TCT Determination Package for such Galapagos Program.

(ii) No later than [...***...] days after delivery of the information set forth in Section 2.3(c)(i), the JSC shall notify Galapagos that the JSC either agrees or disagrees that such Clinical Trial constitutes a Triggering Clinical Trial. Any dispute regarding the Triggering Clinical Trial status of a Clinical Trial shall be subject to resolution [...***...].

(iii) With respect to any Acquired Galapagos Program that constitutes a Galapagos Program upon acquisition or any Excluded Program that becomes an Acquired Galapagos Program and, in either case, [...***...] Galapagos shall promptly, and in no case later than [...***...] days after the applicable acquisition has closed or the change in status to a Galapagos Program has occurred, as applicable, notify the JSC of such acquisition or change in status, as applicable, and the existence of a potential or deemed Triggering Clinical Trial, such notice to include instructions and credentials with which each JSC member and Gilead may access the TCT Determination Package with respect to such potential or deemed Triggering Clinical Trial. In the case of foregoing clause (A), the JSC shall promptly (in no case later than [...***...] days after such notice from Galapagos) make a good faith determination of whether such Clinical Trial constitutes a Triggering Clinical Trial. In the case of foregoing clause (B) or (C), a Triggering Clinical Trial will be deemed to have occurred as of the closing of the applicable acquisition or the change in status to a Galapagos Program, as applicable.

(d) Information Provided in Connection with a Qualifying Data Package Determination. Once a Clinical Trial is determined by the JSC [... ***...] to be a Triggering Clinical Trial, or a Triggering Clinical Trial is deemed to have occurred, then, no later than [...***...] days after such Clinical Trial reaches its Completion Date, Galapagos shall deliver to Gilead [...***...] instructions and credentials with which Gilead may access a Data Room containing the data and information required to be included in a Qualifying Data Package with respect to the Galapagos Program that includes such Triggering Clinical Trial; provided that [...***...] and Gilead delivers an Option Exercise Notice for such Galapagos Program prior to the Completion Date of the applicable Triggering Clinical Trial, Galapagos shall deliver the documents required to be included in the Qualifying Data Package that are existing and available at such time (and prior to the applicable Initial Option Closing, supplement such Qualifying Data Package with information required to be included promptly upon such information becoming available) for such Galapagos Program no later than [...***...] days after the delivery of such Option Exercise Notice by Gilead; provided further that in the event Galapagos applies the Final Term Extension to such Galapagos Program pursuant to Section 14.1, Galapagos shall provide the documents required to be included in the a Qualifying Data Package that are existing and available at such time (and prior to the applicable Initial Option Closing, supplement such Qualifying Data Package with information required to be included promptly upon such information becoming available) at the time that Galapagos notifies Gilead that the Final Term Extension shall apply, solely to the extent such information is existing and available prior to expiration of the Final Term Extension. The JSC shall promptly (and in no case later than [...***...] Business Days after such notice from Galapagos) make a good faith determination of whether the information in the Data Room constitutes a complete Qualifying Data Package with respect to such Clinical Trial and Galapagos Program, and the provisions of Section 8.2(b) shall apply.

(e) <u>Data Room Access Generally</u>. Gilead may not provide any Person with access to the Data Room except its Affiliates, and its and its Affiliates' officers, directors, employees, agents and advisors, in each case, that (A) have a need-to-know in connection with Gilead's determination of whether it wishes to exercise its Option with respect to the applicable Galapagos Program and (B) are bound (prior to receiving access) by obligations of confidentiality and non-use at least equivalent in scope as those set forth in Section 13.1 and Section 13.2 (except for advisors who must be bound by obligations of confidentiality and non-use that are commercially reasonable).

(f) <u>Gilead Contributions</u>. Gilead shall keep Galapagos reasonably apprised, via the JDC, of any Gilead Contributions, including by (i) promptly disclosing to Galapagos any inventions in the Galapagos Program Period Know-How in accordance with Section 10.1(a), (ii) providing a [...***...] report in reasonable detail summarizing the progress with respect to any Gilead Contributions, and (iii) at Galapagos' reasonable request, providing copies of [...***...] that is conceived, discovered, developed, generated or otherwise made by or on behalf of Gilead or its Affiliates in performing such Gilead Contributions.

2.4 <u>Suspension or Termination</u>. If Galapagos determines to Suspend or Terminate all Pre-Program Activities relating to an Active Galapagos Target (as defined on **Schedule 8.10-1**) (such activities relating to any such Target, "**Significant Pre-Program Activities**") or any Galapagos Program, in each case in its entirety, then Galapagos shall promptly notify Gilead of such Suspension or Termination. [...***...]. If Galapagos elects to restart or continue such Significant Pre-Program Activities or Galapagos Program, as applicable, then it shall conduct such Significant Pre-Program Activities or Galapagos Program, as applicable, in good faith in accordance with this Agreement. [...***...]. For clarity, any Suspension or Termination shall not be deemed a termination of this Agreement or a breach by Galapagos of this Agreement.

2.5 <u>Prior Target Screening Activities</u>. Subject to any obligations Galapagos or its Affiliates may have pursuant to Existing Galapagos Third Party Agreements as of the Execution Date with respect to providing such Third Parties results of target screening activities, within [...***...] days following Gilead's reasonable request through the JDC, to be made no more frequently than once per calendar quarter, Galapagos shall share with Gilead through the JDC, in reasonable detail, the results of any target screening activities of Galapagos or any of its Affiliates that were conducted prior to the Effective Date and that are Controlled by Galapagos or any of its Affiliates; *provided that* Galapagos shall not disclose to Gilead or any of its Affiliates any [... ***...] and shall have no obligation to make any disclosures related to any Excluded Program. [...***...]. For clarity, nothing in this Section 2.5 is intended to cover any independent program of Gilead or its Affiliates that does not use the results of any such target screening activities. Any requests for results under this Section 2.5, and the results provided in response to such request, shall be documented in the applicable JDC minutes.

2.6 Certain Terms Applicable to Both Galapagos Programs and Optioned Programs.

(a) <u>Clinical Trial Reporting</u>. Each Party agrees that (i) each Clinical Trial conducted with respect to a Galapagos Product or an Optioned Product that is required to be posted pursuant to Applicable Law or applicable industry codes, including the PhRMA Code, on clinicaltrials.gov or any other similar registry shall be so posted, and (ii) all results of such Clinical Trials that are necessary for obtaining a Regulatory Approval for a Galapagos Product or an Optioned Product shall be posted on clinicaltrials.gov and on any other registry with requirements consistent with the registration and publication guidelines of the International Committee of Medical Journal Editors, to the extent required. The Party responsible for conducting any Clinical Trial shall be responsible for the activities described in the preceding sentence with respect to such Clinical Trial.

(b) <u>Development Records</u>. Each Party shall and shall cause its Affiliates to maintain complete and accurate records (in the form of technical notebooks or electronic files where appropriate) of all work conducted by it or on its behalf that constitute Significant Pre-Program Activities or that relate to any Galapagos Program or Optioned Program, and all Information resulting from such work. Such records shall fully and properly reflect all work done and results achieved in the performance of the activities under this Agreement in sufficient detail and in good scientific manner appropriate for patent and regulatory purposes. Such records shall be maintained in a manner that permits access to, and facilitates the transfer of, such records with respect to each applicable Galapagos Program or Optioned Program on a Galapagos Program-by-Galapagos Program or Optioned Program basis to the extent reasonably practicable. Each Party shall have the right to access such records maintained by the other Party and its Affiliates to the extent reasonably necessary to perform obligations or exercise rights under this Agreement. The JDC shall determine the means by which such access will be provided.

(c) Combination Products.

(i) <u>Covenant Regarding Filgotinib and Gilead Combination Products</u>. Each Party hereby covenants to the other Party, on behalf of itself and its Affiliates, that, except under a Combination Product Activities Agreement or the Filgotinib Agreement, it and its Affiliates shall not [... ***...].

(ii) <u>Gilead Covenant Regarding Combinations involving Excluded Programs and Other Non-Optioned Products</u>. Gilead hereby covenants to Galapagos, on behalf of itself and its Affiliates, that it and its Affiliates shall not [...***...].

(iii) <u>Combination Product Activities Agreements</u>. If the Parties mutually desire to conduct Combination Product Activities, then [...

***...].

***...].

(iv) For clarity, this Section 2.6(c) shall not restrict Galapagos from conducting any activities in or for its Respective Territory [...

(d) <u>Materials</u>. Except as otherwise agreed by the Parties in writing, including in any Ancillary Agreement, any biological or chemical materials, including samples, provided by or on behalf of a Party or any of its Affiliates to the other Party or any of its Affiliates in connection with this Agreement shall remain the property of the Party providing such materials, and shall not be provided to any Third Party without the prior written consent of the Party providing such materials (or as otherwise set forth in an applicable Ancillary Agreement, R&D Plan and Budget, Global Manufacturing Plan and Budget or Global Commercialization Plan and Budget) and shall be used by such other Party and its Affiliates solely for the conduct of activities pursuant to this Agreement. Any materials provided under this Agreement are provided "as is" and the Party receiving such materials acknowledges and agrees that such materials are experimental in nature. Upon the request of the Party providing any such materials, the Party receiving such materials shall either return or destroy such materials, in each case, at the request of the providing Party (unless the receiving Party has the right or obligation to use such materials hereunder). Without limiting any of the foregoing, reasonably prior to transferring any materials to the other Party in connection with Pre-Program Activities or Galapagos Programs, each Party may require the other Party to execute a material transfer agreement with respect to such materials, which material transfer agreement shall not be inconsistent with the terms of this Agreement.

ARTICLE III R&D FOR OPTIONED PROGRAMS

3.1 Initial R&D Plan and Budget.

(a) For each Optioned Program, no later than [...***...] days after the Initial Option Closing, the JDC shall (i) review and discuss the high-level principles for the worldwide Development of such Optioned Program, (ii) review and discuss the Development activities each Party proposes to conduct or have conducted for such Optioned Program and (iii) prepare and submit to the JSC for review and approval the R&D Plan and Budget for such Optioned Program; [...***...]. Such proposed R&D Plan and Budget shall include any Development activities for such Optioned Program proposed by one Party which the other Party agrees to include in the R&D Plan and Budget for such Optioned Program and may include any other Development activities that the Parties mutually agree to include (including Development activities to be conducted on a global basis or in both Parties' Respective Territories). Each Party shall

[...***...]. Either Party may propose amendments to an R&D Plan and Budget from time-to-time through its members on the JDC. Unless otherwise mutually agreed by the Parties in writing, each R&D Plan shall provide that each Party shall be responsible for the portion of included Development activities occurring in its Respective Territory; *provided that*, pursuant to Section 4.2(c), Galapagos shall have the right, but not the obligation, to continue to conduct any Clinical Trials that were being conducted by Galapagos in Gilead Territory at the time the applicable Galapagos Program became an Optioned Program, and such Clinical Trials shall be included in the applicable R&D Plan and Budget.

(b) With respect to each Galapagos Program for which Gilead provides a timely Option Exercise Notice, upon Galapagos' request at any time during the period beginning upon Gilead's delivery of such Option Exercise Notice and ending on the Initial Option Closing, the Parties shall work together through the JDC to prepare the initial R&D Plan and Budget that would apply with respect to such Galapagos Program if such Galapagos Program were to become an Optioned Program. A preliminary draft of the initial R&D Plan and Budget for the Autotaxin Program is attached hereto as **Schedule 3.1(b)** as of the Execution Date. The Parties agree that Galapagos shall continue to conduct its ongoing Clinical Trials as of the Effective Date, including those being conducted as of the Execution Date by Galapagos for GLPG1690, and such Clinical Trial(s) shall be included in the initial R&D Plan and Budget for the Autotaxin Program.

3.2 <u>Optioned Program R&D Activities</u>. For each Optioned Program, each Party shall conduct the activities assigned to it under the applicable R&D Plan and Budget (such activities, collectively, the "**Optioned Program R&D** Activities") and use Commercially Reasonable Efforts to achieve the timelines set forth therein. For any Clinical Trials included in the Optioned Program R&D Activities, at the request of either Party, the Parties will enter into a specific agreement to govern the conduct of such Clinical Trial. The Parties may develop and agree upon a form to cover Clinical Trials conducted hereunder. Without limiting the foregoing, for each Optioned Program, Gilead shall use Commercially Reasonable Efforts to Develop and obtain Regulatory Approval for [...***...]. Notwithstanding the foregoing, if the [...***...], then Gilead shall (a) deliver a written notice to Galapagos within [...***...] days of [...***...] notifying Galapagos of the same, (b) use Commercially Reasonable Efforts to (x) [...***...] and (y) make a final determination regarding [...***...] provided that Gilead shall make such final determination no later than [...***...] days after Gilead has made such final determination, Gilead shall keep Galapagos reasonably apprised, via the JDC, of [...***...]. Within [...***...]. In the case of an election under clause (ii), [...***...].

3.3 <u>Covenant with Respect to R&D Activities</u>. Each Party, on behalf of itself and its Affiliates, hereby covenants that neither it nor any of its Affiliates shall conduct any Development activity with respect to any Optioned Molecule or Optioned Product unless [...***...].

3.4 <u>Research and Development Reports</u>. Each Party shall keep the JDC reasonably informed regarding the progress and results of Development activities for Optioned Molecules and Optioned Products, including by providing (a) [...***...] reports in reasonable detail of results versus goals (as such goals are set forth in the applicable R&D Plan and Budget) as, in the case of each Party, is typically generated by such Party with respect to its product research and development efforts and (b) any information with respect to Independent Activities required to be provided pursuant to Section 7.3(c).

3.5 <u>Clinical Data</u>. Each Party will disclose the Clinical Data from each Clinical Trial that it conducts with respect to any Optioned Programs and any applicable Optioned Product to the other Party no later than [...***...] days after the [...***...] first become available, subject to Section 13.4(a).

3.6 <u>Development Records</u>. With respect to each Optioned Program, as promptly as reasonably practicable following the Initial Option Closing, Galapagos shall make available to Gilead copies of all records with respect to such Optioned Program maintained by Galapagos pursuant to Section 2.6(b).

3.7 Galapagos' Right to Opt-Out.

(a) Subject to Section 3.7(d), with respect to each Optioned Program, if Galapagos desires to cease (i) performance of its obligations under Section 3.2 with respect to the R&D Plan and Budget for such Optioned Program and (ii) sharing of Research and Development Costs with respect to such Optioned Program, then, in each case ((i) and (ii)), Galapagos shall have the right to opt-out of performing such obligations and sharing such Research and Development Costs upon [...***...] days' written notice to Gilead stating that Galapagos is opting-out of such Optioned Program (such notice, with respect to any Optioned Program, an "**Opt-Out Notice**").

(b) If Gilead desires to take over any of Galapagos' ongoing obligations (e.g., Clinical Trials or Manufacturing activities), then Gilead shall provide Galapagos notice (a "**Gilead' Opt-In Notice**") of its intention to do so within [...***...] days after receipt of the Opt-Out Notice. In such a case, in anticipation of the effective date of the opt-out, Galapagos shall cooperate with Gilead to transfer to Gilead in an orderly manner those activities that Gilead desires to take over and such activities, once transferred will not be continued by Galapagos pursuant to Section 3.7(d) and reasonably assist Gilead as necessary to enable an orderly assumption of Development, Manufacturing and Commercialization activities with respect to such Optioned Program (and associated Optioned Product(s) and Optioned Molecule(s)). Galapagos may wind down any activities for which Gilead does not timely provide a Gilead' Opt-In Notice, to the extent not inconsistent with Applicable Law.

(c) Whether or not Gilead is assuming any activities under clause (b), effective as of the effective date of the opt-out:

(i) [...***...];
(ii) [...***...];
(iii) [...***...];
(iv) [...***...];
(vi) [...***...];
(vii) [...***...];
(viii) [...***...];
(xi) [...***...];
(xi) [...***...]; and
(xii) [...***...].

(d) Notwithstanding the foregoing terms of this Section 3.7, with respect to each Optioned Program for which Galapagos delivers an Opt-Out Notice, and for which Gilead timely delivers aa Gilead Opt-In Notice:

(i) with respect to any Clinical Trial for any applicable Optioned Product being conducted by or on behalf of Galapagos that was Initiated prior to the date on which Galapagos provides Gilead the applicable Opt-Out Notice, Galapagos shall, at Gilead's election, (A) [...***...], (B) [...***...] or (C) [...***...]; and

(ii) with respect to any Clinical Trial for any applicable Optioned Product that was Initiated prior to the date on which Galapagos provides Gilead the applicable Opt-Out Notice and is not an Independent Activity of Galapagos:

(A) the costs and expenses with respect to [...***...] shall continue to be treated as Research and Development Costs and shared by the Parties in accordance with the terms of this Agreement; and

(B) Gilead's diligence obligations under Section 3.2 shall continue with respect to any such Clinical Trial for so long as such costs and expenses with respect thereto are treated as Research and Development Costs and shared by the Parties in accordance with the terms of this Agreement.

If Galapagos delivers an Opt-Out Notice, [...***...].

3.8 <u>Secondment</u>. Prior to the secondment of an employee of one Party or any of its Affiliates to the other Party or any of its Affiliates, the Parties shall negotiate in good faith the terms to apply to such secondment.

ARTICLE IV

REGULATORY MATTERS

4.1 Pre-Program Activities and Galapagos Programs.

(a) <u>Ownership of Regulatory Materials</u>. Unless otherwise agreed by the Parties in writing, Galapagos shall solely own all Regulatory Materials and Regulatory Approvals with respect to any Pre-Program Activities or any Galapagos Product.

(b) <u>Responsibility for Regulatory Matters</u>. Unless otherwise agreed by the Parties in writing, Galapagos shall be solely responsible for all regulatory matters relating to any Pre-Program Activities or Galapagos Product.

(c) <u>Costs for Regulatory Affairs</u>. With respect to any Pre-Program Activities and each Galapagos Program, any costs and expenses incurred in connection with Regulatory Materials (including INDs) with respect thereto, Regulatory Approvals with respect thereto, and regulatory affairs activities with respect to any of the foregoing, shall be borne by Galapagos.

(d) <u>Global Safety Database</u>. With respect to each Galapagos Program, Galapagos shall be responsible for establishing, holding and maintaining the global safety database for the applicable Galapagos Molecules and Galapagos Products, at its sole cost and expense.

4.2 Optioned Programs.

(a) Regulatory Transition.

(i) Assignment and Transfer. For each Optioned Program, subject to Section 4.2(c):

(A) as soon as reasonably practicable after the applicable Initial Option Closing but not later than [...***...] months thereafter, Galapagos shall assign and transfer to Gilead or its designated Affiliate the following to the extent Controlled by Galapagos or its Affiliates: [...***...]; and

(B) solely with respect to any Regulatory Materials, Regulatory Approvals or safety or clinical database with respect to any applicable Optioned Product in a country that becomes part of the Gilead Territory for such Optioned Program after the Initial Option Closing for such Optioned Program, as soon as reasonably practicable after the Option Exercise Closing for such country for such Optioned Program, but not later than [...***...] months thereafter, Galapagos shall assign and transfer to Gilead the following to the extent Controlled by Galapagos or its Affiliates: [...***...].

Each Party shall submit all filings, letters and other documentation to the applicable Regulatory Authorities as necessary to effect such assignments and transfers, and the Parties shall cooperate in good faith to expedite such assignments and transfers and shall take or cause to be taken such steps as necessary to minimize any delay with respect thereto and in the event that such an assignment or transfer has not occurred and this causes a delay in the performance of activities hereunder, then Galapagos will cooperate and take such reasonable actions as needed to mitigate the effects of such delay and to expedite the performance of such activities. With respect to each Optioned Program and each item set forth in this Section 4.2(a)(i) to be assigned and transferred to Gilead, until the date that such assignment and transfer becomes effective (the "**Regulatory Transition Date**"), Galapagos shall, under the direction of Gilead, handle all matters involving interactions with a Regulatory Authority with respect to such item. [...***...].

(ii) <u>Copies of Documentation and Databases</u>. Except to the extent such information has already been assigned and transferred to Gilead pursuant to Section 4.2(a)(i), for each Optioned Program, as soon as reasonably practicable but not later than [...***...] months after the applicable Initial Option Closing, or earlier as required by Applicable Law, Galapagos shall provide to Gilead: (A) copies of all Regulatory Materials and Regulatory Approvals in the Territory with respect to any applicable Optioned Product and (B) copies of any safety or clinical databases with respect to any applicable Optioned Product in the Territory, in each case ((A) and (B)), Controlled by Galapagos.

(b) <u>Ownership of Regulatory Materials</u>. Subject to Section 7.4(b), for each Optioned Program, unless otherwise agreed by the Parties in writing, the Party conducting any Clinical Trial for any applicable Optioned Product shall solely own Regulatory Materials with respect to such Clinical Trial. Subject to the preceding sentence and Section 7.4(b), for each Optioned Program, unless otherwise agreed by the Parties in writing, each Party shall solely own all Regulatory Materials and Regulatory Approvals for any Optioned Product in its Respective Territory.

(c) <u>Responsibility for Regulatory Matters and Clinical Trials</u>. Subject to ARTICLE VII, for each Optioned Program, unless otherwise agreed by the Parties in writing, following the Regulatory Transition Date, each Party shall be solely responsible for all regulatory matters (including applications for Regulatory Approval and communications with Regulatory Authorities) and the conduct of all Clinical Trials in its Respective Territory. Notwithstanding the foregoing, Galapagos shall continue to conduct (and be responsible for all associated regulatory matters for) any Clinical Trials that were being conducted by Galapagos in the Gilead Territory at the time the applicable Galapagos Program became an Optioned Program.

(d) <u>Communications with Regulatory Authorities</u>. With respect to each Optioned Program, within [...***...] Business Days after receipt of any Major Market Material Regulatory Communication with respect to any applicable Optioned Product, the Party receiving such Major Market Material Regulatory Communication shall provide the other Party, a copy of such Major Market Material Regulatory Communication. With respect to any Major Market Material Regulatory Communication, the Party receiving such Major Market Material Regulatory Communication shall allow the other Party at least [...***...] Business Days (or such shorter period as may be required by Applicable Law) to review and comment on the proposed response to such Major Market Material Regulatory Communication in advance of the transmission of such response, and shall reasonably consider all comments timely provided by such other Party in connection therewith.

(e) <u>Meetings with Regulatory Authorities</u>. With respect to each Optioned Program, each Party shall provide the other Party with as much advance notice as practicable under the circumstances of all formal meetings and teleconferences with the FDA and EMA pertaining to any applicable Optioned Product. Each Party shall permit the other Party to have, at such other Party's cost and expense, [...***...] representatives of such other Party attend, as observers, such formal meetings and teleconferences with the FDA and EMA pertaining to any applicable Optioned Products.

(f) <u>Submissions and Other Events</u>. With respect to (i) the filing of any IND (or supplement thereto) for an Optioned Product and (ii) the submission of any filings or applications for Regulatory Approval (including orphan drug applications and designations) for an Optioned Product to any Regulatory Authority, in each case ((i) and (ii)), each Party shall notify the other Party reasonably in advance, and in no event less than [...***...] days prior to such filing or submission, of the filing Party's plan for such filing or submission, including timelines that highlight time periods allocated for the non-filing Party's review of such filing or submission. Such time period for such non-filing Party's review shall be no less than [...***...] Business Days, and such non-filing Party shall be entitled to comment on any portion of such filing or submission, and the filing Party shall reasonably consider all comments provided by the non-filing Party in connection therewith. With respect to each Optioned Program, each Party shall notify the other Party promptly and, in any event, within [...***...] days after the occurrence of any of the following events in any Major Market: (i) the filing of any IND (or supplement thereto) for an applicable Optioned Product and the acceptance or other action by a Regulatory Authority with respect thereto; (ii) the submission of any filings or applications for Regulatory Approval (including orphan drug applications and designations) for an applicable Optioned Product to any Regulatory Authority; and (iii) receipt or denial of Regulatory Approval for an applicable Optioned Product. Notwithstanding the foregoing, each Party shall inform the other Party of any such proposed filing, submission, receipt or denial reasonably in advance of, and in no event less than [...***...] days prior to, any public disclosure of such event.

(g) Pharmacovigilance; Global Safety Database.

(i) <u>Pharmacovigilance Agreement</u>. With respect to each Optioned Program, the Parties shall negotiate in good faith and enter into a Pharmacovigilance Agreement ("**PV Agreement**") in a form to be agreed to by the Parties not later than [...***...] days following the date of the applicable Initial Option Closing (and in any event prior to the date on which Gilead first conducts Development of any applicable Optioned Product), which PV Agreement shall define the pharmacovigilance responsibilities of the Parties and include safety data exchange procedures governing the coordination of collection, investigation, reporting and exchange of information concerning any adverse experiences, and any product complaints associated with adverse experiences, related to any applicable Optioned Product sufficient to enable each Party to comply with its obligations under Applicable Law.

(ii) <u>Global Safety Database</u>. With respect to each Optioned Program, [...***...] shall have the right, but not the obligation, to be responsible for establishing (if applicable), holding and maintaining the global safety database for the applicable Optioned Products. If requested by Gilead at the [...***...] or at any time thereafter, Galapagos shall transfer to Gilead any global safety database that has been established by Galapagos with respect to the applicable Optioned Products. Gilead shall provide Galapagos with, and Galapagos shall have the right to access as to be further defined in the PV Agreement, any such global safety database. Without limitation to the foregoing, in the event that Gilead does not exercise its right to establish any such global safety database (if applicable), or request transfer to Gilead of any such global safety database that has been established by Galapagos, Galapagos shall: (A) upon Gilead's request (1) cooperate with Gilead to provide Gilead with access to such global safety database established by Galapagos and (2) provide Gilead with copies of any such global safety database pursuant to Section 4.2(a)(i) or (B) be responsible for establishing (if applicable), holding and maintaining such global safety database. The Parties agree that at the [...***...] following the Effective Date, the JDC shall discuss whether the global safety database for [...***...] shall be transferred to Gilead.

(iii) <u>Product Withdrawals and Recalls</u>. With respect to each Optioned Program, if (A) any Regulatory Authority threatens, initiates or advises any action to remove any applicable Optioned Product from the market in the Territory or requires or advises Galapagos, Gilead, or any of their respective Affiliates or Sublicensees to distribute a "Dear Doctor" letter or its equivalent regarding use of any such Optioned Product in the Territory or (B) either Party determines that an event, incident, or circumstance has occurred that may result in the need for a recall or market withdrawal in the Territory, then, in each case ((A) or (B)), Galapagos or Gilead, as applicable, shall, to the extent practicable, notify the other Party of such event or determination immediately, and in any event within [...***...] (or sooner if required by Applicable Law) after such Party becomes aware of the event or makes such determination. Each Party shall, to the extent practicable, endeavor to discuss and agree with the other Party whether to recall or withdraw such Optioned Product in the Territory; *provided, however, that* if such discussion is not practicable or if the Parties fail to agree within an appropriate time period (recognizing the exigencies of the situation), then each Party shall decide whether to recall or withdraw such Optioned Product in its Respective Territory. Each Party shall be responsible, at its sole cost and expense, for conducting any recalls or withdrawals or taking such other necessary remedial actions with respect to Optioned Products in its Respective Territory, except (1) in each Party or any of its Affiliates or subcontractors, in which event such other Party shall bear such costs and expenses to the extent of its or its Affiliate's or subcontractor's responsibility or (2) in each Party's respective Territory, solely to the extent set forth in any applicable supply agreement, if any, between such Party (or any of its Affiliates) and the other Party (or any of its Affiliates) with respect to an applicab

4.3 <u>Ownership of Information Contained in Regulatory Materials and Regulatory Approvals</u>. For clarity, ownership of Regulatory Materials and Regulatory Approvals is not intended, and shall not be construed, to alter ownership of any information contained in such Regulatory Materials and Regulatory Approvals, including any Galapagos Know-How, Gilead Collaboration Know-How or Joint Collaboration Know-How contained therein.

ARTICLE V

COMMERCIALIZATION; MEDICAL AFFAIRS

5.1 Commercialization Responsibility and Diligence.

(a) Each Party shall be solely responsible for all Commercialization activities and costs for Optioned Products in its Respective Territory, except as otherwise set forth in the Global Commercialization Plan and Budget and subject to the oversight of the JCC. For each Optioned Program for which there is an approved Optioned Product, Gilead shall use Commercially Reasonable Efforts to [...***...]. If Gilead determines not to launch an approved Optioned Product, fails to launch an approved Optioned Product within [...***...] after receipt of approval, including any required pricing approvals, or withdraws the Optioned Product, in each case for reasons other than (a) manufacturing or CMC-related issues, (b) regulatory issues or (c) other activities outside of its control, in each case in [...***...] (and there are no other Optioned Products in [...***...] or [...***...] (and there are no other Optioned Products in [...***...] with respect to the applicable Optioned Program [...***...], as applicable, by [...***...] days' written notice to Gilead.

(b) If, during such notice period, Gilead provides Galapagos with written notice [...***...] or claiming that it has launched or relaunched, as applicable, the applicable Optioned Product in [...***...], as applicable, and invokes and continues to pursue in good faith the dispute resolution procedures set forth in Section 15.1(b), [...***...]. During the pendency of any such dispute, all of the terms and conditions of this Agreement shall remain in effect.

(c) With respect to each Optioned Program, if Gilead decides, in its sole discretion, not to Commercialize any Optioned Product for such Optioned Program in [...***...], then Gilead shall notify Galapagos, and upon Galapagos' written request, the Parties shall enter into a separate agreement that sets forth the terms pursuant to which [...***...] shall become part of the Galapagos Territory for such Optioned Program (and, for clarity, [...***...] shall no longer be part of the Gilead Territory for such Optioned Program) and [...***...] shall be deemed a Terminated Region; *provided that*, such separate agreement [...***...]. If the Parties do not reach agreement on the terms of any such agreement, such terms shall be determined pursuant to Section 15.1(b) and Section 15.2(a).

5.2 Commercialization of Optioned Products.

(a) <u>Global Commercialization Plan and Budget</u>. For each Optioned Product, certain global Commercialization principles, strategies and activities shall be set forth in the global commercialization plan and budget for such Optioned Product, which plan, as well as any applicable amendments thereto, shall be prepared by the JCC (each, a "**Global Commercialization Plan and Budget**"), and then reviewed and approved by the JSC. For clarity, the Global Commercialization Plan and Budget shall not include any activities specific to a Party's Respective Territory unless mutually agreed by the Parties in writing. The JCC shall review, at least [...***...], the Global Commercialization Plan and Budget for each Optioned Program and suggest any applicable amendments thereto for review and approval by the JSC. The Global Commercialization Plan and Budget for an Optioned Product shall include any agreed global commercial strategy, process for selection of global trademarks, if any, global branding principles, high-level pricing strategy, certain Medical Affairs Activities and a budget for any costs and expenses that the Parties agree to share in connection with the activities set forth in the Global Commercialization Plan and Budget. Each Party shall conduct the activities assigned to it under the Global Commercialization Plan and Budget for the applicable Optioned Program. Notwithstanding the foregoing, each Party shall set the selling prices for each Optioned Product in its sole discretion with respect to its Respective Territory.

(b) <u>Trademarks</u>. For each Optioned Program, the JCC shall conduct the activities set forth in Section 10.9(a) with respect to the Global Optioned Product Trademarks.

(c) <u>Coordination on Global Promotional Materials</u>. If the Parties desire to coordinate on any mutually agreed core marketing, promotional or advertising materials ("**Global Promotional Materials**"), then the Global Commercialization Plan and Budget shall set forth the Parties' responsibilities for producing all promotional materials and any updates to those materials. If there are Global Promotional Materials, each Party shall be entitled to an electronic copy of such Global Promotional Materials, but each Party shall be responsible for localization of such Global Promotional Materials, but each Party shall be responsible for localization of such Global Promotional Materials for its Respective Territory and for producing copies of its own promotional materials. Each Party and its Affiliates shall conduct its Commercialization activities in a manner consistent with the Global Promotional Materials, but in no event shall a Party or any of its Affiliate be required to violate Applicable Law in its Respective Territory. If a Party or any of its Affiliates desires to create additional marketing, promotional or advertising materials, then it may do so for use in its Respective Territory in its discretion, but shall use reasonable efforts to do so in a manner consistent with any key messages set forth in the Global Commercialization Plan and Budget and any Global Promotional Materials.

(d) <u>Coordination on Training Materials</u>. Each Party and its Affiliates shall be responsible for its own product training, business conduct, compliance and safety training materials, except to the extent included in the Global Commercialization Plan and Budget.

(e) Corporate Compliance Program.

(i) Gilead shall maintain its existing corporate compliance program or a variant thereof. Galapagos shall establish and maintain a corporate compliance program, including at least one (1) full-time employee whose sole area of responsibility is compliance and who primarily reports to someone not in any Commercial function and is responsible for ensuring that all employees of Galapagos and any of its Affiliates comply with Applicable Law, national and international pharmaceutical industry codes of practice and guidelines and Galapagos' business conduct rules and regulations, which, subject to the foregoing, shall be substantially consistent with Gilead's business conduct rules and regulations. Galapagos shall report to Gilead no less than once every [...***...] months as to the compliance by Galapagos with establishing and maintaining a compliance program in accordance with this Section 5.2(e). Each compliance program, (B) a periodic risk assessment that guides development of policies, training and monitoring activities, (C) appropriate corporate compliance policies, (D) regular compliance training and communication to applicable employees as selected on a risk-based approach, (E) auditing or monitoring or other risk-evaluation processes for applicable activities and (F) mechanisms, compliant with all Applicable Laws, to receive complaints or questions and investigate and remediate potential noncompliance, including a disciplinary component to handle compliance violations.

(ii) Each Party shall [...***...]. The Parties shall coordinate as each of the Parties deems to be in its own respective best interest in order to ensure compliance in connection with its activities under this Agreement. Nothing in this Section 5.2(e) requires any Party to [...***...].

(f) <u>Authority over Personnel; Personnel Costs</u>. Nothing in this Agreement shall be construed to conclude that any of either Party's or its Affiliates' agents or employees are agents or employees of the other Party or subject to the other Party's direction and control. Each Party shall have sole authority over the terms and conditions of employment of its agents and employees, including their selection, management, compensation (including incentive plans) and discharge. Each Party shall be responsible for all costs and expenses in connection with their agents or employees (except to the extent reimbursable as FTEs at the FTE Rate under this Agreement (*e.g.*, in the case of Research and Development Costs)), including salaries, incentive compensation, travel expenses and other expenses, providing benefits, deducting federal, state and local payroll Taxes, Federal Insurance Contribution Act Taxes, unemployment insurance Taxes, and any similar Taxes and paying workers' compensation premiums, unemployment insurance contributions and any other payments required by Applicable Law to be made on behalf of employees.

5.3 <u>Commercialization Reports</u>. Each Party shall keep the JCC informed regarding the progress of Commercialization activities for Optioned Products in its Respective Territory, including appointment of Distributors. Each Party shall report to the JCC on all material issues relating to the Commercialization of Optioned Products in such Party's Respective Territory promptly after such issues arise. Without limiting the foregoing, in light of Galapagos' economic interest in Gilead's Commercialization activities for Optioned Products in the Gilead Territory, Gilead shall provide to Galapagos [...***...].

5.4 <u>Medical Affairs</u>. Each Party shall be responsible for conducting its and its Affiliates' own Medical Affairs Activities for Optioned Products in its Respective Territory at its sole cost and expense except to the extent any such costs and expenses are Shared Commercialization Costs.

5.5 <u>Scientific and Medical Conferences</u>. The Parties shall coordinate, through the JCRC, on behalf of themselves and their Affiliates, regarding their activities at scientific or medical conferences, including regarding which Party's or its Affiliate's or Sublicensee's employees shall staff any promotional and medical information booths that include any Optioned Product. If both Parties participate in any such conference, then each Party shall maintain its own promotional and medical information booths at such conference unless otherwise agreed between the Parties.

ARTICLE VI

MANUFACTURE AND SUPPLY

6.1 <u>Pre-Program Activities and Galapagos Programs</u>. Except as otherwise agreed by the Parties in writing, Galapagos shall be solely responsible, in its discretion and at its sole cost and expense, for conducting all Manufacturing activities with respect to all Pre-Program Activities and each Galapagos Program.

6.2 Optioned Programs.

(a) <u>Global Manufacturing Strategy</u>; <u>Plan and Budget</u>. The goal of the Parties is to have a coordinated and integrated supply chain for the Territory as a whole. For each Optioned Program, within [...***...] days after the applicable Initial Option Closing, the JDC shall prepare and submit to the JSC for review and approval the global manufacturing strategy plan and budget (the "**Global Manufacturing Plan and Budget**") that shall include the global manufacturing strategy, the division of responsibility between the Parties for Manufacturing activities (including overseeing CMOs), any Manufacturing-related activities with respect to which the Parties are sharing Development Manufacturing Costs and a corresponding budget therefor (to the extent not budgeted in the R&D Plan and Budget), any capital expenditures that will be shared by the Parties and any CMOs for particular Manufacturing activities; *provided that* the Parties agree to take into account the Manufacturing and supply needs of both Parties in their Respective Territories for the applicable Optioned Program. The Global Manufacturing Plan and Budget may include agreed capital improvements and capital costs; *provided that*, [...***...]. The Global Manufacturing Plan and Budget may include agreed capital improvements and capital costs, including with respect to (i) cost sharing and financing associated therewith and (ii) ownership of such capital improvements. Within [...***...] days after the Effective Date, the JSC shall approve the initial Global Manufacturing Plan and Budget for the Autotaxin Program. Either Party may propose amendments the Global Manufacturing Plan and Budget from time to time through its members on the JDC.

(b) <u>Efforts</u>. Each Party shall conduct the Manufacturing activities assigned to it in the Global Manufacturing Plan and Budget. No Party shall be required to expand capacity or incur any capital expenditures except as set forth in the Global Manufacturing Plan and Budget.

(c) <u>Independent Supply</u>. So long as the conditions set forth in Section 7.1 for permitting independent Manufacturing activities are met, each Party shall have the right to develop capacity or capabilities beyond those set forth in the Global Manufacturing Plan and Budget but the costs and expenses associated therewith shall be at such Party's sole expense and such Party shall have no obligation to make such capacity available to the other Party. Such Manufacturing shall be conducted by such Party, its Affiliates or a CMO.

(d) <u>Compliance</u>. Notwithstanding anything herein or in any Ancillary Agreement, in no event shall a Party or its Affiliates be required to supply the other Party or its Affiliates with any Optioned Products (including product samples if applicable) if the purchasing Party is conducting its activities hereunder in violation of this Agreement or Applicable Law.

(e) <u>Development Supplies</u>. The Parties shall in good faith negotiate the terms of one (1) or more supply agreements and corresponding quality agreement(s) pursuant to which the Party assigned a given Manufacturing activity under the Global Manufacturing Plan and Budget shall supply Optioned Molecules and Optioned Products to the other Party (to the extent contemplated in the Global Manufacturing Plan and Budget) for Development activities (the "**Clinical Supply Agreement**"); *provided that* the price for such Optioned Molecules and Optioned Products under any such Clinical Supply Agreement (other than a Clinical Supply Agreement supporting Independent Activities) shall reflect [...***...]. In the case of Independent Activities, the supplying Party may charge a mark-up equal to the lowest of (i) [...***...] percent [...***...] of the Development Manufacturing Costs, (ii) such Party's mark-up used for [...***...] or (iii) the mark-up agreed by the Parties in the applicable Clinical Supply Agreement required to support the Autotaxin Program shall be agreed and signed by the Parties or their applicable Affiliates no later than [...***...] days after the Effective Date.

(f) <u>Commercial Supplies</u>. The Parties shall in good faith negotiate the terms of one (1) or more supply agreements and corresponding quality agreement(s) pursuant to which the Party assigned a given Manufacturing activity under the Global Manufacturing Plan and Budget shall supply Optioned Molecules and Optioned Products to the other Party for Commercial activities and such terms shall be set forth in a commercial supply agreement to be entered into between the Parties on terms to be negotiated in good faith (the "**Commercial Supply Agreement**"). The supplying Party may charge a mark-up equal to the lowest of (i) [...***...] percent [...***...], (ii) such Party's markup used [...***...] or (iii) such lesser mark-up agreed by the Parties in the applicable Clinical Supply Agreement.

6.3 Manufacturing Technology Transfer.

(a) <u>Galapagos Programs</u>. In order to plan for any anticipated Technology Transfers, if Gilead in good faith anticipates that it might exercise the Option for a Galapagos Program and desires to begin discussions regarding the global supply chain for the applicable Galapagos Product(s), then the Parties, through the JDC, shall commence such discussions, including the anticipated timing and process for the corresponding Technology Transfer.

(b) <u>Optioned Programs</u>. With respect to each Optioned Program, if Gilead is or will be obligated to or otherwise intends to Manufacture an Optioned Molecule or Optioned Product, upon Gilead's request, Galapagos shall conduct and complete a customary technology transfer to Gilead or its designee for the Manufacture of such Optioned Molecule or Optioned Product for such Optioned Program (each, a "**Technology Transfer**"), or cause its CMO to conduct and complete such Technology Transfer, including providing supplies and other support to validate Gilead's or its CMO's Manufacturing facility for the Optioned Product. The Technology Transfer shall include the transfer to Gilead of all Information Controlled by Galapagos during the Term that is necessary or useful to enable the Manufacture of Optioned Molecules and Optioned Products, and not previously transferred to Gilead under this Agreement, by providing copies or samples of relevant documentation, materials and other embodiments of such Information, and by making available its qualified technical personnel on a reasonable basis to consult with Gilead with respect to such Information. For each Technology Transfer, the Parties, through the JDC, shall mutually agree upon a plan and timeline for Technology Transfer Plan"). Each Party shall perform its obligations under each Technology Transfer Plan. [...***...]. The Parties shall mutually agree on a Technology Transfer Plan with respect to the Autotaxin Program within [...***...] days following the Effective Date.

ARTICLE VII INDEPENDENT ACTIVITIES FOR OPTIONED PROGRAMS

7.1 <u>Conditions for Permitting Independent Activities</u>. For each Optioned Program, if, subject to Section 3.1(a) and 6.2(a), either Party desires to conduct any Development activities in, or Manufacturing activities for, its Respective Territory that (a) the other Party does not agree to include in the applicable R&D Plan and Budget or the applicable Global Manufacturing Plan and Budget, as applicable, in either case, following completion of the dispute resolution process set forth in Section 1.5(b)(ii)(B) and (b) would not be reasonably expected to have a material adverse effect on the Development or Commercialization of any Optioned Molecule or Optioned Product in the Respective Territory of the other Party; *provided that* in no event shall any Clinical Trial that is required for Regulatory Approval in the Independent Activities Party's Respective Territory be reasonably expected to have a material adverse effect on the Development or Commercialization of any Optioned of any Optioned Molecules or Optioned Product in the Respective Territory be reasonably expected to have a material adverse effect on the Development or Commercialization of any Optioned Molecules or Optioned Product in the Respective Territory be reasonably expected to have a material adverse effect on the Development or Commercialization of any Optioned Molecules or Optioned Product in the Respective Territory of the other Party (any such activities, "**Independent Activities**"), then Section 7.2 shall apply.

7.2 Independent Activity Plans. For each Optioned Program, if the conditions set forth in Section 7.1 apply, then:

(a) the Party that desires to conduct the Independent Activities (the "**Independent Activities Party**") shall provide the JDC with a written plan for conducting such Independent Activities ("**Independent Activities Plan**"), which Independent Activities Plan shall describe the applicable Independent Activities in reasonable detail, including for Development activities, a description of the applicable Clinical Trial and expected data to be generated therefrom;

(b) within [...***...] Business Days after receipt of such Independent Activities Plan, the JDC shall meet to discuss any comments of the other Party with respect to such Independent Activities Plan, and the Independent Activities Party shall consider any such comments in good faith;

(c) within [...***...] Business Days after the occurrence of such meeting of the JDC, the JDC shall refer the applicable Independent Activities Plan to the JSC, along with any comments of either Party with respect thereto;

(d) within [...***...] Business Days after such referral, the JSC shall meet to discuss such Independent Activities Plan and any comments of either Party with respect thereto; and

(e) following such meeting of the JSC, such Independent Activities Party shall have the right to conduct Clinical Trials as Independent Activities set forth in such Independent Activities Plan, in each case, in accordance with Section 7.3.

7.3 Conduct of Independent Activities. Any Independent Activities shall be conducted in accordance with this Section 7.3.

(a) Any Independent Activities shall be carried out in accordance with the applicable Independent Activities Plan that was last reviewed by the JSC in accordance with Section 7.2(d); *provided that* non-substantive changes do not have to be reviewed by the JSC. For clarity, if the Independent Activities Party desires to carry out additional substantive Independent Activities that are not set forth in an Independent Activities Plan that was reviewed by the JSC in accordance with Section 7.2(d), then such Independent Activities Party shall follow the procedures set forth in Section 7.2 with respect to such additional Independent Activities.

(b) Subject to Section 7.5 or in the case of Manufacturing-related activities, as otherwise set forth in a Supply Agreement, the Independent Activities Party shall be solely responsible for all costs and expenses for Independent Activities carried out by or on behalf of such Independent Activities Party.

(c) The applicable Independent Activities Party shall keep the JDC reasonably informed regarding the progress of Independent Activities, including by providing a [...***...] report in reasonable detail of results as, in the case of each Party, is typically generated by such Party with respect to its product research and development efforts. Upon completion of any Independent Activities, the Independent Activities Party shall provide the other Party with a report in reasonable detail of the Independent Activities Data generated in connection with such Independent Activities.

(d) Any Independent Activities conducted by Gilead with respect to an Optioned Program shall be performed in accordance with any Existing Galapagos Third Party Obligations applicable to such Optioned Program.

7.4 Independent Activities Data and Regulatory Documentation.

(a) <u>Independent Activities Data</u>. Notwithstanding Section 10.1, as between the Parties, the Independent Activities Party shall solely own all right, title and interest in and to all Collaboration Know-How consisting of data generated by or on behalf of such Independent Activities Party or its Affiliates solely from performing the applicable Independent Activities ("**Independent Activities Data**"). Any Independent Activities Data generated by or on behalf of Gilead shall be deemed to be Gilead Collaboration Know-How and, notwithstanding anything to the contrary herein, Galapagos shall not have the right to use such Independent Activities Data (other than safety data) for any purpose, except as set forth in Sections 7.4(c) or 7.5 or as otherwise agreed in writing by Gilead. Any Independent Activities Data generated by or on behalf of Galapagos shall be deemed to be Galapagos Foreground Know-How and, notwithstanding anything to the contrary herein, Gilead shall not have the right to use such Independent Activities Data (other than safety data) for any purpose, except as set forth in Sections 7.4(c) or 7.5 or as otherwise agreed in writing by Gilead. Any Independent Activities Data generated by or on behalf of Galapagos shall be deemed to be Galapagos Foreground Know-How and, notwithstanding anything to the contrary herein, Gilead shall not have the right to use such Independent Activities Data (other than safety data), except as set forth in Sections 7.4(c) or 7.5 or as otherwise agreed in writing by Galapagos.

(b) <u>Independent Activities Regulatory Documentation</u>. Notwithstanding Section 4.2(b), as between the Parties, the Independent Activities Party shall solely own all right, title and interest in and to all Regulatory Materials and Regulatory Approvals to the extent resulting solely from the applicable Independent Activities ("**Independent Activities Regulatory Documentation**"). Any Independent Activities Regulatory Documentation resulting solely from Independent Activities for which

Gilead is the Independent Activities Party shall be owned solely by Gilead and, notwithstanding anything to the contrary herein, Galapagos shall not have the right to use such Independent Activities Regulatory Documentation for any purpose, except as set forth in Sections 7.4(c) or 7.5 or as otherwise agreed in writing by Gilead. Any Independent Activities Regulatory Documentation resulting solely from Independent Activities for which Galapagos is the Independent Activities Party shall be owned solely by Galapagos and, notwithstanding anything to the contrary herein, Gilead shall not have the right to use such Independent Activities Regulatory Documentation for any purpose, except as set forth in Sections 7.4(c) or 7.5 or as otherwise agreed in writing by Galapagos.

(c) <u>Safety Data</u>. Notwithstanding anything to the contrary in this Section 7.4, any safety-related information (i) shall be covered by the PV Agreement, and (ii) may be used by the non-Independent Activities Party on the applicable Optioned Product label(s) and disclosed by the non-Independent Activities Party to any Regulatory Authority as required by Applicable Law or required by such Regulatory Authority in connection with the applicable Optioned Product or applicable Optioned Program, in each case, as being Developed and Commercialized by the non-Independent Activities Party, without payment of any fee or consideration to the Independent Activities Party and without any such additional agreement as contemplated in Section 7.5.

7.5 <u>Buy-In for Independent Activities Data or Independent Activities Regulatory Documentation</u>. The non-Independent Activities Party with respect to any Independent Activities Data or Independent Activities Regulatory Documentation may provide written notice to the Independent Activities Party that it desires to terminate the restrictions on use of such Independent Activities Data or Independent Activities Regulatory Documentation set forth in Section 7.4. Upon receipt of such notice, the Independent Activities Party shall provide the other Party with the total Research and Development Costs expended for such Independent Activities (calculated as if such activities were included in the applicable plan and budget) (the "Independent Activities Costs"). The non-Independent Activities Party may, in its discretion, elect to buy back the right to use such Independent Activities Data or Independent Activities Regulatory Documentation by paying to the Independent Activities Party an amount equal to [... ***...] of the portion of Independent Activities Costs the non-Independent Activities Party would have borne had such Independent Activities Party, and the restrictions on use of such Independent Activities Data or Independent Activities Party, (a) the restrictions on use of such Independent Activities Data or Independent Activities Party shall provide to the Independent Activities Data or Independent Activities Party shall have a license and right of reference for such Independent Activities Data or Independent Activities Party shall provide to the other Party are port covering such Independent Activities as set forth in Section 8.3 or 8.4, as applicable, and (b) the Independent Activities Party shall provide to the other Party a report covering such Independent Activities as set forth in clause (a) of Section 3.4.

ARTICLE VIII

GRANTS OF RIGHTS

8.1 <u>Grant of Research & Development License for Gilead Contributions</u>. In partial consideration for the Upfront Consideration, Galapagos, on behalf of itself and its Affiliates, hereby grants to Gilead as of the Effective Date an exclusive, sublicensable (solely in accordance with Section 8.6(a) (i)) research and development license under the Galapagos IP and Galapagos' interest in any Joint Collaboration IP, in each case, to conduct any Gilead Contributions (including Pre-Program Activities and the activities conducted by the Gilead' JDC representatives in meeting their responsibilities as members of the JDC, in each case, solely to the extent necessary to conduct such Gilead Contributions) pursuant to Section 2.2.

8.2 Gilead's Option Rights.

(a) <u>Option Grant</u>. For each Galapagos Program, subject to Section 8.2(d), Galapagos hereby grants to Gilead an exclusive option to obtain the exclusive and co-exclusive licenses described in Section 8.3 with respect to such Galapagos Program and all Galapagos Molecules and Galapagos Products included in such Galapagos Program (each, an "**Option**"). The Option with respect to each Galapagos Program shall be effective for the period beginning on (i) the Effective Date, for any Galapagos Program in existence as of the Effective Date, (ii) the date of acquisition by Galapagos or its applicable Affiliate, for any Acquired Galapagos Program that constitutes a Galapagos Program as of such date of acquisition, or (iii) the date on which a Galapagos Program comes into existence, for any Galapagos Program arising after the Effective Date from Pre-Program Activities (including, for clarity, with respect to any Acquired Galapagos Program that did not constitute a Galapagos Program upon acquisition), and, in each case ((i) through (iii)), ending on the earliest to occur with respect to such Option of (A) the applicable Option Exercise Closing with respect to each country in the Territory, (B) the end of the applicable Option Exercise Period prior to delivery by Gilead of an Option Exercise Notice for such Option, (C) the termination of such Option pursuant to Section 8.2(e), (D) the expiration or termination pursuant to ARTICLE XIV of this Agreement in its entirety or with respect to all Galapagos Molecules and all Galapagos Products in the applicable Galapagos Program, and (E) the third anniversary of the end of the Collaboration Term (such effectiveness period with respect to each such Option, the "**Galapagos Program Period**").

(b) Option Exercise Notice; Option Exercise Period.

(i) With respect to each Galapagos Program (and all Galapagos Molecules and Galapagos Products included in such Galapagos Program), Gilead may exercise the Option at any time during the applicable Option Exercise Period for such Galapagos Program by delivering a written notice of such exercise to Galapagos (each such notice for any Galapagos Program, an "**Option Exercise Notice**"). "**Option Exercise Period**" means, for each Galapagos Program, the period beginning on the Completion Date of the first Phase 2 Clinical Trial for a Galapagos Product in such Galapagos Program (whether or not such Clinical Trial is a Triggering Clinical Trial) and ending on the earlier to occur of:

(A) delivery of an Option Exercise Notice by Gilead for such Galapagos Program or Gilead' notice that it declines to exercise such Option, and

(B) 11:59 p.m. Pacific time on the later of (1) the [...***...] day after the date on which Galapagos has delivered to Gilead a Qualifying Data Package for such Galapagos Program and (2) [...***...] Business Days after a determination pursuant to [...***...] that a Qualifying Data Package has been delivered;

provided that for any Galapagos Program that is an Acquired Galapagos Program, clause (B)(1) above shall be the [...***...] day after the applicable delivery date; *provided further that* the Option Exercise Period for a given Galapagos Program shall [...***...].

(ii) During the Option Exercise Period for each Galapagos Program, Gilead may reasonably request additional information with respect to such Galapagos Program (including in connection with Gilead's review of the Galapagos Option Schedule of Exceptions for such Galapagos Program delivered pursuant to Section 8.2(c)(iii)), and to the extent that the information reasonably requested by Gilead is at such time reasonably available, Galapagos shall make full, accurate and timely responses to such requests. For clarity, (x) the foregoing sentence shall have no effect on the duration of the Option Exercise Period and (y) Galapagos shall not be required, as a result of such an information request by Gilead pursuant to this Section 8.2(b)(ii), to generate new information or to conduct any additional Development activities. If Gilead has not provided an Option Exercise Notice for such Galapagos Program to Galapagos in accordance with Section 8.2(b)(i) prior to the end of the Option Exercise Period for such Galapagos Program, then effective immediately after the end of such Option Exercise Period, such Galapagos Program shall be an Excluded Program.

(iii) If Gilead exercises the Option with respect to any Galapagos Program that is also an Acquired Galapagos Program, then the Parties shall adjust the terms of this Agreement to the extent necessary to account for any encumbrances or limitations on such Acquired Galapagos Program that restrict Gilead's activities relating to such Acquired Galapagos Program or rights to exploit the Acquired Galapagos Program pursuant to the terms of this Agreement, including Gilead's rights to Commercialize Optioned Products throughout the Gilead Territory in the Field. With respect to such Acquired Galapagos Program, [...***...] shall be responsible for any [...***...].

(iv) The Parties may agree to allow Gilead to exercise the Option for any Galapagos Program prior to the Option Exercise Period, and in such case, the allocation of Research and Development Costs set forth in Section 9.9(a) that would have applied following an Option Exercise Closing in the normal course shall [...***...].

(v) "Triggering Clinical Trial" means for each Galapagos Program either:

(A) the first Phase 2 Clinical Trial that reaches its Completion Date for a Galapagos Product in such Galapagos Program for a particular indication and that satisfies each of the following criteria: (1) [...***...]; (2) [...***...]; and (3) [...***...]; or

(B) the first Phase 3 Clinical Trial that reaches its Completion Date for a Galapagos Product in such Galapagos Program, if such Phase 3 Clinical Trial reaches its Completion Date prior to any Phase 2 Clinical Trial for such Galapagos Product in such Galapagos Program that satisfies the criteria in clause (A) above.

(vi) Without limitation of the foregoing, any Phase 2 Clinical Trial that satisfies [...***...] will be presumed to be a Triggering Clinical Trial if it is designed with [...***...] and [...***...]. For clarity, the foregoing sentence shall not be interpreted to mean that a Phase 2 Clinical Trial [...***...] is presumed to not be a Triggering Clinical Trial.

(vii) Notwithstanding anything to the contrary herein, in the event that Galapagos has Initiated one or more Phase 2 Clinical Trials for the same indication for a Galapagos Molecule or Galapagos Product in a Galapagos Program and proposes to Initiate a Phase 3 Clinical Trial prior to the completion of all such Phase 2 Clinical Trials, then Galapagos' obligations to deliver a Qualifying Data Package and Gilead's right to exercise its Option, in each case, with respect to such Galapagos Program shall be as set forth in this Section 8.2(b)(vii). Galapagos shall deliver a Qualifying Data Package for the first such Phase 2 Clinical Trial that constitutes a Triggering Clinical Trial and the JSC shall make its determination of whether such Phase 2 Clinical Trial constitutes a Triggering Clinical Trial, in accordance with Section 2.3(d). Thereafter, within the Option Exercise Period for such Galapagos Program, Gilead may elect one of the following: (A) deliver an Option Exercise Notice, (B) notify Galapagos that Gilead does not intend to exercise the Option for such Galapagos Program or (C) [...***...]. If Gilead delivers an Option Exercise Period shall end upon the earliest to occur of (w) [...***...], (x) [...***...], (y) [...***...] and (z) [...***...], Galapagos Shall deliver a Qualifying Data Package for each Phase 2 Clinical Trial that reaches its Completion Date no later than [...***...], Galapagos shall deliver a Qualifying Data Package for each Phase 2 Clinical Trial that reaches its Option, in each case, for such Galapagos Program. [...***...], Gilead may deliver an Option Exercise Notice or decline to exercise its Option, in each case, for such Galapagos Program. [...***...], Gilead shall be responsible for [...***...] percent [...***...], Gilead shall be responsible for [...***...] percent [...***...] and Galapagos shall be responsible for [...***...] of the costs of such Phase 3 Clinical Trial as if such costs were Research and Development Costs under Section 9.9(a) (without regard to Section 9.9(c)). If G

(viii) Notwithstanding anything to the contrary in this Agreement, with no further action or consent required on the part of either Party or any Committee, the ADAMTS-5 Existing Trial shall be deemed the Triggering Clinical Trial for the ADAMTS-5 Program, and the applicable Option Exercise Period shall commence, in accordance with this Section 8.2(b), upon the later of the Effective Date and the Completion Date of such ADAMTS-5 Existing Trial; *provided that* if another Clinical Trial meeting the criteria of a Triggering Clinical Trial reaches its Completion Date with respect to the ADAMTS-5 Program before the ADAMTS-5 Existing Trial reaches its Completion Date, then such other Clinical Trial shall be treated as the Triggering Clinical Trial with respect to the ADAMTS-5 Program.

(c) Option Exercise Closing.

(i) If Gilead delivers an Option Exercise Notice for a Galapagos Program during the Option Exercise Period for such Galapagos Program in accordance with Section 8.2(b)(i), subject to Section 8.2(c)(v), such Galapagos Program shall become an Optioned Program at 12:01 a.m. Pacific Time on the later of the [...***...] Business Day following (A) Galapagos' receipt of such Option Exercise Notice or (B) if any Antitrust Conditions apply to the Option for such Galapagos Program, the date Gilead notifies Galapagos pursuant to Section 8.2(d) that the Antitrust Conditions applicable to the Option for such Galapagos Program have been satisfied (with respect to each such Galapagos Program, an "**Option Exercise Closing**"). To the extent permitted by applicable Antitrust Laws, and subject to Section 8.2(c)(v), if the Antitrust Approvals have been obtained for [... ***...] with respect to a Galapagos Program, then (1) an Option Exercise Closing for such Galapagos Program shall occur for [... ***...] and each other jurisdiction in the Territory as to which Antitrust Approval is not required or has been obtained (such Option Exercise Closing, the "**Initial Option Closing**") and (2) subject to Section 8.2(e), the countries for which an Antitrust Approval is still required as of the Initial Option Closing (or any subsequent Option Exercise Closing for such Optioned Program) shall remain part of the Galapagos Territory unless and until the applicable required Antitrust Approval has been obtained. Upon each notice of receipt of any such subsequent Antitrust Approval for such Optioned Program, there shall occur an additional Option Exercise Closing for such Optioned Program with respect to the applicable jurisdictions.

(ii) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period, except as set forth in the initial Galapagos Option Schedule of Exceptions (which Galapagos shall deliver together with the Qualifying Data Package) for such Galapagos Program and subject to Section 8.2(c)(iii), the Galapagos Option Exercise Representations for such Galapagos Program shall be made (A) subject to the initial Galapagos Option Schedule of Exceptions, as of the date that the JSC determines, pursuant to Section 1.1(b)(ii) [...***...] that the data package delivered for such Galapagos Program constitutes a Qualifying Data Package and (B) subject to the updated Galapagos Option Schedule of Exceptions, as of the Initial Option Closing for such Galapagos Program (an "**Option Bringdown Date**").

(iii) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period, prior to each applicable Option Exercise Closing for such Galapagos Program, Galapagos shall promptly notify Gilead in writing if any of the Galapagos Option Exercise Representations for such Galapagos Program are no longer [...***...] and may update the disclosures in the Galapagos Schedule of Exceptions for such Galapagos Program with respect to the Galapagos Option Exercise Representations. [...***...]

(iv) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period, prior to each applicable Option Exercise Closing for such Galapagos Program, Gilead shall provide Galapagos with an updated list of any Gilead Programs with respect to such Galapagos Program.

(v) Notwithstanding anything to the contrary in this Agreement, if at any time after delivery of an Option Exercise Notice for a Galapagos Program and before the Initial Option Closing for such Galapagos Program, Galapagos notifies Gilead, or Gilead otherwise becomes aware, of facts or circumstances that would cause any of the Galapagos Option Exercise Representations to be [...***...] with respect to such Galapagos Program in a manner that reduces the full scope of rights that would have been granted to Gilead upon such Option Exercise Closing absent [...***...], then Gilead may [...***...]. If, after becoming aware of facts or circumstances that would cause any of the Galapagos Option Exercise Representations to be [...***...] with respect to such Galapagos Program, Gilead nevertheless decides not to [...***...] for such Galapagos Program, neither Galapagos nor its Affiliates shall have any liability to Gilead for damages arising from such [...***...] after such notification by Galapagos or awareness of Gilead; *provided that* the foregoing shall not restrict, limit or change in any manner Gilead's remedies arising from any breach of covenant by Galapagos or any of its Affiliates that gave rise to such [...***...] if Gilead [...***...].

(vi) Notwithstanding anything to the contrary in this Agreement, with no further action or consent required on the part of either Party or any Committee pursuant to ARTICLE I, Gilead's Option with respect to the Autotaxin Program shall be deemed exercised as of the Effective Date, and the Effective Date shall be deemed the Initial Option Closing for the Autotaxin Program.

(d) Antitrust Filing.

(i) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period (or prior to the applicable Option Exercise Period pursuant to Section 8.2(b)(iv)), upon either Party's request, the Parties shall work together in good faith to conduct an analysis of whether any Antitrust Filings are or may be required in connection with a proposed Option Exercise Closing with respect to such Galapagos Program.

(ii) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period (or prior to the applicable Option Exercise Period pursuant to Section 8.2(b)(iv)), following delivery of such Option Exercise Notice, both Parties shall file their respective Antitrust Filings as promptly as practicable with each applicable Antitrust Authority pursuant to any applicable Antitrust Laws, and in any event, with respect to notification and report forms filed with the Federal Trade Commission ("FTC") and the Department of Justice ("DOJ") pursuant to the HSR Act, if any, the Parties shall make such filings no later than [...***...] Business Days after such delivery of the Option Exercise Notice. Each Party will be responsible for its own costs and expenses associated with any Antitrust Filing, but Gilead shall be responsible for payment of all fees to the FTC and DOJ with respect to Antitrust Filings made pursuant to the HSR Act. The Parties shall provide each other promptly with information and assistance as may be reasonably necessary and use reasonable efforts, in each case, to obtain prompt clearance required under applicable Antitrust Laws for the consummation of the applicable Option Exercise Closing and the transactions contemplated thereby and shall keep each other apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC, DOJ and each other applicable Antitrust Authority and shall comply promptly with any such inquiry or request; *provided that* neither Party shall be required to consent to the divestiture or other disposition of any of its or its Affiliates' assets, consent to any other material structural or conduct remedy or otherwise restrict or limit its or its Affiliates' freedom of action. Each Party shall instruct its counsel to cooperate with the other Party's counsel and use reasonable efforts to facilitate and

expedite the identification and resolution of any such issues and, consequently, the expiration of the applicable HSR Act waiting period and any other applicable waiting period under Antitrust Laws, including if requested by Gilead, seeking early termination of any such waiting period. Such reasonable efforts and cooperation include counsels' undertaking: (A) to keep each other appropriately informed of communications from and to personnel of the reviewing Antitrust Authority; and (B) to confer with each other regarding appropriate contacts with and response to personnel of the FTC, DOJ or other applicable Antitrust Authority.

(iii) Subject to Section 8.2(d)(i), within [...***...] Business Days after the Parties have obtained the Antitrust Approval under the HSR Act with respect to any Galapagos Program, Gilead shall deliver to Galapagos a notice identifying the applicable Galapagos Program and attaching evidence of the Antitrust Approval under the HSR Act and each other applicable Antitrust Approval that has been obtained, if any, and the country(ies) to which each such Antitrust Approval relates.

(e) No Effective Option Exercise Closing.

(i) With respect to each Galapagos Program for which Gilead delivers an Option Exercise Notice during the applicable Option Exercise Period (or prior to the applicable Option Exercise Period pursuant to Section 8.2(b)(iv)) (during which period Galapagos can continue to conduct such Galapagos Program), notwithstanding anything to the contrary in Section 8.2(d), if within [...***...] days after the date on which both Parties have made the necessary initial Antitrust Filings in the applicable country(ies) with respect to such Galapagos Program, the Initial Option Closing has not occurred, then either Party may notify the other Party that it is terminating the Option relating to such Galapagos Program, and unless the non-notifying Party responds within [...***...] Business Days to the notifying Party, providing evidence that it is using reasonable efforts to secure the necessary Antitrust Approvals in the applicable jurisdictions (including in response to any additional request from Antitrust Authorities), such termination of the Option for such Galapagos Program. If the non-notifying Party responds within such [...***...] Business Day at the verifying Party responds within such [...***...] Business Day after the date on which both Parties have made such necessary initial Antitrust Filings, and such termination of the Option for such Galapagos Program shall occur at 12:01 a.m. Pacific time on [...***...]

(ii) With respect to each Optioned Program for which one (1) or more required Antitrust Approvals were not obtained pursuant to Section 8.2(e)(i), then each such country shall remain part of the Galapagos Territory for so long as such Optioned Program remains subject to this Agreement.

(f) [...***...].

8.3 <u>Grant of Expanded Development and Commercialization License for Optioned Programs</u>. For each Optioned Program, Galapagos, on behalf of itself and its Affiliates, hereby grants to Gilead, effective as of each Option Exercise Closing for such Optioned Program in the countries to which such Option Exercise Closing relates:

(a) a royalty-bearing, sublicensable (solely in accordance with Section 8.6(a)(i)) license under the Galapagos IP (subject to Section 7.4(a)) and Galapagos' interest in any Joint Collaboration IP, in each case, (A) to Exploit any Optioned Molecules or Optioned Products with respect to such Optioned Program in the Field in the Gilead Territory and (B) to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory for the purposes of Exploitation in the Gilead Territory and for purposes of performing activities assigned to it in this Agreement or in any Ancillary

Agreement, which license under the foregoing clause (A) is exclusive (including as to Galapagos but subject to Galapagos' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Gilead Territory for purposes of Exploitation in the Galapagos Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement) and under clause (B) is co-exclusive (with Galapagos);

(b) subject to Section 7.4(b), a royalty-bearing, sublicensable (solely in accordance with Section 8.6(a)(i)) right of reference under (or right of access to, if such right of reference is unavailable or insufficient) any Regulatory Materials or Regulatory Approvals Controlled by Galapagos with respect to any applicable Optioned Molecule or Optioned Product, in each case, (A) to Exploit such Optioned Molecules or Optioned Products with respect to such Optioned Program in the Field in the Gilead Territory and (B) to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory for the purposes of Exploitation in the Gilead Territory and for purposes of performing activities assigned to it in this Agreement or in any Ancillary Agreement, which right of reference (or right of access, if applicable) under the foregoing clause (A) is exclusive (including as to Galapagos but subject to Galapagos' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Gilead Territory for purposes of Exploitation in the Galapagos Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement) and under clause (B) is co-exclusive (with Galapagos). For clarity, with respect to any Regulatory Materials and Regulatory Approvals to be assigned and transferred to Gilead pursuant to Section 4.2(a)(i), this right of reference under (or right of access to) such Regulatory Materials and Regulatory Approvals applies until such time as they are effectively assigned and transferred to Gilead; and

(c) an exclusive (including as to Galapagos), royalty-free, sublicensable (solely in accordance with Section 8.6(a)(i)) license under the Global Promotional Materials for any applicable Optioned Product that are Controlled by Galapagos, to Commercialize such Optioned Product in the Field in the Gilead Territory.

For clarity, the licenses and other rights under this Section 8.3 shall be deemed to be granted as of the Effective Date with respect to the Autotaxin Program.

8.4 Grants to Galapagos.

(a) <u>Galapagos R&D Activities</u>. Gilead, on behalf of itself and its Affiliates, hereby grants to Galapagos a non-exclusive, royalty-free, sublicensable (solely in accordance with Section 8.6(a)(ii)) license under the Gilead Collaboration IP and Gilead's interest in any Joint Collaboration IP, in each case, to conduct any Galapagos R&D Activities in the Field in the Territory.

(b) <u>Optioned Programs</u>. For each Optioned Program, Gilead, on behalf of itself and its Affiliates, hereby grants to Galapagos, effective as of each Option Exercise Closing for such Optioned Program in the countries to which such Option Exercise Closing relates:

(i) a royalty-free, sublicensable (solely in accordance with Section 8.6(a)(ii)) license under the Gilead Collaboration IP (subject to Section 7.4(a)) and Gilead's interest in any Joint Collaboration IP, in each case, (A) to Exploit such Optioned Molecules or Optioned Products with respect to such Optioned Program in the Field in the Galapagos Territory and (B) to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Gilead Territory solely for purposes of Exploitation in the Galapagos Territory and for purposes of performing activities assigned to it in this Agreement or in any Ancillary Agreement, which license under the foregoing clause (A) is exclusive (even as to Gilead but subject to Gilead' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory and for purposes of Exploitation in the Gilead Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement, which license under the foregoing clause (A) is exclusive (even as to Gilead but subject to Gilead' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory for purposes of Exploitation in the Gilead Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement) and under clause (B) is co-exclusive (with Gilead);

(ii) subject to Section 7.4(b), a royalty-free, sublicensable (solely in accordance with Section 8.6(a)(ii)) right of reference under (or right of access to, if such right of reference is unavailable or insufficient) any Regulatory Materials or Regulatory Approvals Controlled by Gilead with respect to any applicable Optioned Molecule or Optioned Product, in each case, (A) to Exploit such Optioned Molecules or Optioned Products with respect to such Optioned Program in the Field in the Galapagos Territory and (B) to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Territory solely for purposes of Exploitation in the Galapagos Territory and for purposes of performing activities assigned to it in this Agreement or in any Ancillary Agreement, which right of reference (or right of access, if applicable) under the foregoing clause (A) is exclusive (even as to Gilead) but subject to Gilead' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement, which right of reference (or right of access, if applicable) under the foregoing clause (A) is exclusive (even as to Gilead) but subject to Gilead' retained rights to Develop, Manufacture and supply such Optioned Molecules and Optioned Products in the Galapagos Territory for purposes of Exploitation in the Gilead Territory and for purposes of performing activities assigned to it in this Agreement or any Ancillary Agreement and under clause (B) is co-exclusive (with Gilead); and

(iii) an exclusive (including as to Gilead), royalty-free, sublicensable (solely in accordance with Section 8.6(a)(ii)) license under the Global Promotional Materials for any applicable Optioned Product that are Controlled by Gilead, to Commercialize such Optioned Product in the Field in the Galapagos Territory.

(iv) For clarity, the licenses and other rights under this Section 8.4(b) shall be deemed to be granted as of the Effective Date with respect to the Autotaxin Program.

8.5 <u>Retained Rights</u>. Notwithstanding the exclusive licenses granted by Galapagos under Sections 8.1 and 8.3, Galapagos shall retain such rights under the Galapagos IP and Joint Collaboration IP as are necessary to perform (or have performed by permitted subcontractors hereunder) the activities assigned to Galapagos under this Agreement in accordance with the terms of this Agreement, including performing any Gilead Contribution that Gilead fails to perform. Notwithstanding the exclusive licenses granted by Gilead under Section 8.4(b), Gilead shall retain such rights under the Gilead Collaboration IP and Joint Collaboration IP as are necessary to perform (or have performed by permitted subcontractors hereunder) the activities assigned to Gilead under this Agreement in accordance with the terms of this Agreement.

8.6 Sublicensing.

(a) Scope of Permissible Sublicensing.

(i) <u>By Gilead</u>. Subject to Section 8.6(b), the licenses granted by Galapagos to Gilead hereunder may be sublicensed by Gilead, through multiple tiers, without any requirement of consent; *provided that* Gilead shall be liable for any act or omission of any Sublicensee that is a breach of any of Gilead's obligations under this Agreement as though the same were a breach by Gilead, and Galapagos shall have the right to proceed directly against Gilead with respect to such breach without any obligation to first proceed against such Sublicensee.

(ii) <u>By Galapagos</u>. Subject to Section 8.6(b), the licenses granted by Gilead to Galapagos hereunder may be sublicensed by Galapagos, through multiple tiers, without any requirement of consent; *provided that* Galapagos shall be liable for any act or omission of any Sublicensee that is a breach of any of Galapagos' obligations under this Agreement as though the same were a breach by Galapagos, and Gilead shall have the right to proceed directly against Galapagos with respect to such breach without any obligation to first proceed against such Sublicensee.

(b) <u>Sublicense Agreements</u>. Any Sublicense Agreement shall be consistent with and subject to the terms of this Agreement. In any Sublicense Agreement that either Party enters into with respect to any Optioned Product, such Party (the "**Sublicensing Party**") shall use [...***...] to require that the applicable Sublicense grant to such Sublicensing Party a royalty-free and sublicensable (through multiple tiers) (i) license under any (A) intellectual property rights (including any Information, Patents or Trademarks) conceived, discovered, developed, generated or otherwise made by or on behalf of such Sublicensee with respect to such Optioned Product in connection with such Sublicense Agreement and (B) Global Promotional Materials controlled by such Sublicensee with respect to such Optioned Product; and (ii) right of reference under (or right of access to) any Regulatory Materials or Regulatory Approvals controlled by such Sublicensee with respect to such Optioned Product; and entry in each case ((i) and (ii)), (1) as necessary to grant to the non-Sublicensing Party the rights that the Sublicensing Party would have granted under Sections 8.1, 8.3, 8.4 or 10.9(c) (as applicable) if such Sublicensing Party had Controlled such intellectual property rights, Global Promotional Materials, Regulatory Materials or Regulatory Approvals or (2) without limitation to clause (1), in the case of Gilead as the Sublicensing Party, as necessary to fulfill its obligations under Section 14.7(c).

8.7 Distributorships and Other Relationships.

(a) <u>Distributors; Contract Sales Forces and Co-Promotion Agreements</u>. Each Party shall have the right to appoint its Affiliates, and such Party and its Affiliates shall have the right, in their sole discretion, to appoint any commercially reasonable Distributor or contract sales force or to grant a Third Party co-promotion rights, in any country(ies) in the Territory, to distribute, market, and sell Optioned Products in its Respective Territory; *provided that* it shall provide the other Party [...***...] prior written notice before granting a Third party any such co-promotion rights. Any agreement between a Distributor or contract sales force and either Party or its Affiliates regarding an Optioned Product shall be on commercially reasonable and arm's length terms.

(b) Other Relationships.

(i) Except as provided in Section 8.7(a), if Gilead desires to license, sell or otherwise grant or transfer (including by option) rights to Commercialize (excluding granting co-promotion rights for) any Optioned Product with respect to [...***...], then Gilead shall notify Galapagos in writing and, at the Galapagos' request, the Parties shall enter into a separate written agreement on commercially reasonable terms pursuant to which Galapagos shall have such right to Commercialize such Optioned Product with respect to such one (1) or more countries. If the Parties do not reach agreement on the terms of any such agreement, such terms shall be determined pursuant to Section 15.1(b) and 15.2(d). If Galapagos does not make such request within [...***...] days of such notice, then Gilead shall have the right to proceed with such license, sale or other grant or transfer. For clarity, such right to license, sell or otherwise grant or transfer its rights or obligations under this Agreement shall not confer on Gilead any expanded or additional right to assign all or any portion of its rights or obligations under this Agreement pursuant to Section 17.6(a).

(ii) Except as provided in Section 8.7(a), if Galapagos desires to license, sell or otherwise grant or transfer (including by option), rights to Commercialize (excluding granting co-promotion rights for) any Optioned Product [...***...], then Galapagos shall notify Gilead in writing and, at Gilead's request, the Parties enter into a separate written agreement on commercially reasonable terms pursuant to which Gilead shall have such right to Commercialize such Optioned Product. If the Parties do not reach agreement on the terms of any such agreement, such terms shall be determined pursuant to Section

15.1(b) and 15.2(d). If Gilead does not make such request within [...***...] days of such notice, then Galapagos shall have the right to proceed with such license, sell or other grant or transfer. For clarity, such right to license, sell or otherwise grant or transfer its rights to Commercialize under this Agreement shall not confer on Galapagos any expanded or additional right to assign all or any portion of its rights or obligations under this Agreement pursuant to Section 17.6(a).

8.8 Reserved.

8.9 <u>No Implied Licenses</u>. Except as expressly provided in this Agreement, neither Party shall be deemed to have granted the other Party (by implication, estoppel or otherwise) any right, title, license or other interest in or with respect to any intellectual property rights, Information, Regulatory Materials or Regulatory Approvals Controlled by such Party.

8.10 Shared Program Process.

(a) [...***...].
(b) [...***...].
(c) [...***...].
(d) [...***...].

8.11 <u>Confirmatory Patent License</u>. Each Party shall, if requested to do so by the other Party and at the other Party's cost and expense, promptly enter into confirmatory license agreements in such form as may be reasonably requested by the other Party for purposes of recording the licenses granted under this Agreement with such patent offices in the such other Party's Respective Territory as other Party considers appropriate.

8.12 Excluded Programs. Notwithstanding anything to the contrary, (a) Gilead is granted no rights or licenses under this Agreement (including Section 8.3) with respect to any Excluded Programs and (b) neither Galapagos nor its Affiliates has any obligations with respect to Excluded Programs under this Agreement, in each case, at such times as any such program is an Excluded Program (*e.g.*, if a program is an Excluded Program as of the Effective Date, no rights are granted as of the Effective Date, and if a program becomes an Excluded Program after the Effective Date, any rights that were previously granted shall terminate at such time).

8.13 Existing Galapagos Third Party Obligations.

(a) For so long and to the extent applicable to this Agreement, (i) Gilead shall provide Galapagos such assistance and take such actions as necessary for Galapagos to be in compliance with the provisions set forth on the Existing Galapagos Third Party Obligation Schedule, as such provisions exist as of the Execution Date and have been fully disclosed to Gilead, and (ii) upon Galapagos' written request, Gilead shall use commercially reasonable efforts to provide Galapagos such assistance and take such actions as necessary for Galapagos to be in compliance with the provisions of the Existing Galapagos Third Party Agreements.

(b) For each Acquired Galapagos Program, Galapagos shall include in the Qualifying Data Package for such Acquired Galapagos Program a list of any agreements applicable to such Acquired Galapagos Program existing as of the date of the closing of the transaction pursuant to which Galapagos or its Affiliate acquired such Acquired Galapagos Program and [...***...]. If such Acquired Galapagos

Program becomes an Optioned Program, then for so long as it is an Optioned Program, (i) Gilead shall provide Galapagos such assistance and take such actions as necessary for Galapagos to be in compliance [...***...] (ii) upon Galapagos' written request, Gilead shall use commercially reasonable efforts to provide Galapagos such assistance and take such actions as necessary for Galapagos to be in compliance [...***...] (iii) upon Galapagos to be in compl

(c) For each Pre-Option In-License that relates to a Galapagos Program that was entered into prior to the Execution Date, Galapagos shall include in the Qualifying Data Package for such Galapagos Program [...***...]. If such Galapagos Program becomes an Optioned Program, then for so long as it is an Optioned Program, (i) Gilead shall provide Galapagos such assistance and take such actions as reasonably necessary for Galapagos to be in compliance with such provisions, as such provisions have been disclosed to Gilead and (ii) upon Galapagos' written request, Gilead shall use commercially reasonable efforts to provide Galapagos such assistance and take such actions as necessary for Galapagos to be in compliance with such Pre-Option In-License.

(d) In no event shall Gilead or any of its Affiliates have any financial obligations arising out of its or their obligations under this Section 8.13, except under Section 9.6(b), if applicable.

(e) Notwithstanding the obligations of Galapagos hereunder, with respect to reporting with regard to Pre-Program Activities, Galapagos Programs, Optioned Programs or Acquired Galapagos Programs, if Galapagos believes that it would be in breach of any Third Party obligations existing prior to the Execution Date (or in the case of an Acquired Galapagos Program or Pre-Option License Agreement, existing as of the date the closing of the transaction pursuant to which Galapagos or its Affiliates acquired such Acquired Galapagos Program or Pre-Option License Agreement), the Parties shall discuss such circumstance in good faith and reach a mutually acceptable resolution.

(f) Nothing in this Section 8.13 shall require Gilead to take, or omit to take, any action that would cause Gilead to be in breach of this Agreement.

ARTICLE IX

FINANCIALS

9.1 <u>Upfront Consideration</u>. In partial consideration of the licenses and the other rights granted to Gilead hereunder, Gilead shall pay to Galapagos a non-refundable, non-creditable upfront cash payment of Three Billion Nine-Hundred Fifty Million Dollars (\$3,950,000,000) (the "**Upfront Consideration**") no later than [...***...] Business Days after the Effective Date.

9.2 <u>Option Payment</u>. For each Optioned Program, excluding the Autotaxin Program and the ADAMTS-5 Program, Gilead shall pay to Galapagos an amount equal to One Hundred Fifty Million Dollars (\$150,000,000), and for the ADAMTS-5 Program, Gilead shall pay to Galapagos an amount equal to Two Hundred Fifty Million Dollars (\$250,000,000) (each such payment, an "**Option Payment**") within [...***...] days after the applicable Galapagos Program has become an Optioned Program pursuant to Section 8.2. For the Autotaxin Program, the payment for such Option shall be deemed made upon payment of the Upfront Consideration.

9.3 <u>Milestone Payments</u>. Each milestone payment in this Section 9.3 shall be payable only once. The payments under Sections 9.3(a), 9.3(b), and 9.3(c) total One Billion Seventy Five Million Dollars (\$1,075,000,000).

(a) <u>Development Milestones for ADAMTS-5</u>. Subject to Section 14.4(a)(ii), if the ADAMTS-5 Program becomes an Optioned Program and the applicable milestone event below occurs, Gilead shall pay Galapagos the corresponding milestone payment specified below. The maximum milestone payments due by Gilead under this Section 9.3(a) total [...***...] Dollars [...***...].

(i) If the ADAMTS-5 Program becomes an Optioned Program, Gilead shall pay Galapagos within [...***...] days after the Initial Option Closing for the ADAMTS-5 Program, one (but not both) of the following amounts:

(A) If the secondary endpoint of [...***...], as defined in the current protocol as of the Execution Date, is achieved for the ADAMTS-5 Existing Trial, Two Hundred Million Dollars (\$200,000,000); or

(B) If the secondary endpoint of [...***...], as defined in the current protocol as of the Execution Date, is not achieved, [... ***...] Dollars [...***...] for achievement of each of the following secondary endpoints: (1) [...***...], (2) [...***...], (3) [...***...], (4) [...***...], or (5) [...***...] for the ADAMTS-5 Existing Trial, up to a maximum total of [...***...] achievements for up to a maximum of [...***...] Dollars [... ***...].

If the ADAMTS-5 Existing Trial is not so completed, then the Parties shall negotiate an appropriate milestone for either a further Phase 2 Clinical Trial or Phase 3 Clinical Trial for the ADAMTS-5 Program. Any dispute regarding such an appropriate milestone shall be resolved pursuant to Section 15.2(d).

For purposes of this Section 9.3(a)(i), the endpoints are as they exist as of the Execution Date and were disclosed to Gilead unless the Parties agree in writing on modified endpoints.

(ii) If the FDA approves an NDA for an Optioned Product containing GLPG1972, Gilead shall pay to Galapagos a one-time payment of [...***...] within [...***...] days after the date of such FDA approval.

(b) <u>Sales Milestones for ADAMTS-5</u>. Gilead shall make each of the sales milestone payments indicated below to Galapagos if and when aggregate Net Sales in the Gilead Territory of all Optioned Products containing GLPG1972 across all indications in a given calendar year first meet the thresholds indicated below prior to the end of the expiration of the Royalty Term(s) for such Optioned Products in the Gilead Territory.

Aggregate Net Sales of Optioned Products Containing GLPG1972 in the Gilead Territory a

Given Calendar Year	Payment
[***]	[***]
[***]	$[^{***}]$
[***]	$[^{***}]$

Gilead shall notify and pay to Galapagos the amounts set forth in this Section 9.3(b) within [...***...] days after the end of the calendar quarter during which the applicable milestone event has been achieved.

(c) <u>Development Milestone for Autotaxin Program</u>. If the FDA approves an NDA for an Optioned Product containing GLPG1690, Gilead shall pay to Galapagos a one-time payment of Three Hundred Twenty-Five Million Dollars (\$325,000,000) within thirty (30) days after the date of such FDA approval.

9.4 Optioned Product Royalties.

(a) <u>Royalty Payments</u>. In partial consideration of the license granted by Galapagos under Section 8.3, during the applicable Royalty Term only, and on an Optioned Product-by-Optioned Product basis, Gilead shall pay to Galapagos non-refundable royalties in each calendar year on the amount of aggregate Net Sales of the applicable Optioned Product in the Gilead Territory (other than in Access Territory where such Optioned Product is available on an Affordable Basis), as calculated by multiplying the applicable royalty rates set forth below by the corresponding amount of incremental Net Sales of such Optioned Product in such calendar year.

Net Sales of such Optioned Product in the Gilead Territory	Royalty Rate
[***]	20%
[***]	$[^{***}]$
[***]	24%

For example, if the aggregate Net Sales of all Optioned Products in a given Optioned Program in a particular calendar year is (...***...], the amount of royalties payable under this Section 9.4(a) shall be as follows: (...***...]% of (...***...]%

From and after the expiration of the Royalty Term for an Optioned Product in a country, (i) Gilead shall no longer have any obligation to pay any royalty on Net Sales of such Optioned Product in such country and (ii) Net Sales of such Optioned Product in such country shall be excluded for purposes of determining the royalty tiers set forth above.

(b) <u>Royalty Term</u>. The "**Royalty Term**" shall commence, on an Optioned Product-by-Optioned Product and country-by-country basis, on the First Commercial Sale of such Optioned Product in such country and shall continue until [...***...].

(c) <u>Additional Royalty Provisions</u>. For each Optioned Product, the royalties payable under Sections 9.4(a) will be adjusted as follows with respect to each country in the Gilead Territory upon the first to occur of (i) or (ii) below with respect to such Optioned Product:

(i) Upon the [...***...], the royalty rates applicable to the Net Sales of such Optioned Product in such country shall be [...***...] of the rates set forth in Section 9.4(a). For clarity, [...***...], in each case, as of the First Commercial Sale of an Optioned Product in a country in the Gilead Territory, then as of such First Commercial Sale the royalty rates applicable to the Net Sales of such Optioned Product in such country shall be [...***...] of the rates set forth in Section 9.4(a); and

(ii) Upon [...***...], the royalty rates applicable to Net Sales of Optioned Products in such country shall be [...***...] of the rates set forth in Section 9.4(a).

For clarity, only one (1) reduction under either of the foregoing clauses (i) or (ii) will apply and the maximum reduction in the royalty rates set forth in Section 9.4(a) is [...***...].

(d) <u>Compulsory License</u>. If a Compulsory License is granted to a Third Party with respect to an Optioned Product in any country in the Gilead Territory, and such Third Party actually sells such Optioned Product in the country under such Compulsory License, then the Parties shall [... ***...], with Galapagos' [...***...] included in the royalty payments and reports made pursuant to Section 9.5.

9.5 <u>Royalty Payments and Reports</u>. For each Optioned Product, within [...***...] Business Days following the end of each calendar quarter, Gilead shall provide to Galapagos a statement setting forth on a country-by-country basis good faith estimates of the gross sales of the applicable Optioned Product in the Gilead Territory and an estimated calculation of Net Sales in the Gilead Territory with respect to such Optioned Product, in each case, reported in Dollars as determined in accordance with Section 9.13. All amounts payable to Galapagos pursuant to Section 9.4(a) with respect to Net Sales during a calendar quarter, shall be paid in Dollars within [...***...] days after the end of such calendar quarter. Each payment of royalties due to Galapagos shall be accompanied by a statement, on a country-by-country basis, of the amount of gross sales of each Optioned Product in the Gilead Territory, during the applicable calendar quarter, a calculation of Net Sales in the Gilead Territory with respect to each Optioned Product showing with reasonable specificity the aggregate deductions from gross sales provided for in the definition of Net Sales during such calendar quarter, and a calculation of the amount of royalty payment due on such sales for such calendar quarter, in all cases, reported in Dollars as determined in accordance with Section 9.13.

9.6 Payments to Third Parties.

(a) Negotiation of Post-Option In-Licenses.

(i) For each Optioned Program, if either Party desires to obtain a license under any Patent or any intellectual property right of a Third Party in any country to Exploit an applicable Optioned Product in such country, then, at such Party's request, the Parties shall meet and discuss whether to enter into a license agreement to obtain rights to such Patent or intellectual property right for the Exploitation of such Optioned Product in one (1) or more countries in the Territory and if so, the terms of such license agreement and which Party should obtain such license agreement. If the Parties agree to enter into such a license agreement, then the Party that the Parties agree should obtain such license agreement shall enter into such license agreement on the terms agreed by the Parties (such license agreement, a "Joint Post-Option In-License").

(ii) If the Parties do not agree to obtain a Joint Post-Option In-License with respect to any Patent or any intellectual property right of a Third Party in any country, then either Party may enter into a license agreement with such Third Party for rights under such Patent or intellectual property without the approval of the other Party (such license agreement, an "**Independent Post-Option In-License**"); *provided that* such Independent Post-Option In-License shall not impose any obligations on the other Party or preclude the other Party from negotiating such a license for its Respective Territory. Neither Party may enter into an Independent Post-Option In-License without first discussing such in-license agreement with the other Party.

(b) Responsibility for Payments.

(i) "**Third Party License Payments**" means, with respect to a license agreement with a Third Party for rights under such Third Party's Patent or other intellectual property, all license fees, milestones, royalties or other payments due to such Third Party under such license agreement.

(ii) <u>Pre-Option In-Licenses for Pre-Program Activities and Galapagos Programs</u>. [...***...] for any Third Party License Payments due under any Pre-Option In-License with respect to Pre-Program Activities and Galapagos Programs (other than Optioned Programs).

(iii) <u>Joint Post-Option In-License</u>. With respect to each Joint Post-Option In-License, the Third Party License Payments with respect thereto that are: (A) [...***...]; (B) [...***...]; (C) [...***...]; (D) [...***...] or (E) [...***...], shall be [...***...] by the Parties. In determining the reasonable allocation in foregoing clause (C) and clause (D), the Parties shall take into account [...***...].

(iv) With respect to each Pre-Option In-License for Optioned Programs, the Third Party License Payments with respect thereto that are: (A) [...***...]; (B) [...***...]; (C) [...***...]; or (D) [...***...], shall be allocated between the Parties [...***...]. In determining the reasonable allocation in foregoing clauses (C) and (D), the Parties shall take into account [...***...].

(v) <u>Independent Post-Option In-Licenses</u>. With respect to each Independent Post-Option In-License, subject to (A) Section 9.6(b)(vi) if Gilead or any of its Affiliates is a party to such Independent Post-Option In-License or (B) the terms of any Clinical Supply Agreement or Commercial Supply Agreement for which the Party that is a party to such Independent Post-Option In-License (or one of its Affiliates) is the supplying the Party, the Party that is a party to such Independent Post-Option In-License for the Third Party License Payments thereunder.

(vi) <u>Gilead Deduction</u>. With respect to any Third Party License Payments for an Optioned Product in the Gilead Territory (other than any Independent Post-Option In-License with respect to any improvement or enhancement to such Optioned Product in the Gilead Territory that is not being Developed or Commercialized in the Galapagos Territory) for which Gilead is solely responsible (*i.e.*, for which Galapagos is not also sharing costs pursuant to the other provisions of this Section 9.6), Gilead shall have the right to offset [...***...] percent [...***...] of such Third Party License Payments against any royalties or milestone payments owed to Galapagos under this Agreement with respect to such Optioned Product; *provided that* any such amount that cannot be applied to offset amounts due to Galapagos for any calendar quarter may be carried forward to be applied in the subsequent calendar quarter.

(vii) <u>Payment Process</u>. With respect to any Third Party License Payments not shared as Research and Development Costs (and therefore handled through the process in Section 9.9), on a quarterly basis, the Party initially bearing the Third Party License Payments shall invoice the other Party for such amount as is necessary to effectuate the responsibility for Third Party License Payments as set forth in this Section 9.6. The Party receiving such invoice shall pay it not later than [...***...] days following receipt thereof.

9.7 <u>Following Royalty Term</u>. Upon expiration of the Royalty Term with respect to an Optioned Product in a country, each Party's licenses from the other Party hereunder with respect to such Optioned Product, in such country, shall become fully paid-up, non-exclusive, perpetual, and irrevocable.

9.8 <u>Veterinary Products</u>. Neither Party, nor any of its Affiliates, shall Exploit or grant any (sub)licenses to any Third Party to Exploit any Galapagos Product or Optioned Product for Veterinary Use without the prior written consent of the other Party. With respect to Veterinary Use, unless set forth in a separate written agreement between the Parties, no research, development or commercialization costs and expenses are intended to be shared by the Parties and no milestones or royalties set forth herein shall apply to the Development or Commercialization of any Optioned Product for Veterinary Use.

9.9 Optioned Program Research and Development Costs.

(a) <u>Generally</u>. Subject to Section 9.9(b) and any other relevant terms in this Section 9.9, for each Optioned Program, the Parties shall share Research and Development Costs incurred by or on behalf of either Party or its Affiliates after the applicable Option Exercise Closing with respect to the countries relating to each Option Exercise Closing, as follows: Gilead shall be responsible for fifty percent

(50%) and Galapagos shall be responsible for fifty percent (50%). Within [...***...] Business Days after the end of each calendar quarter, each Party shall provide to the other Party a report in reasonable detail of any Research and Development Costs incurred by such Party in such calendar quarter for each Optioned Program, which may be based on such Party's good faith estimate. Such Research and Development Costs so reported shall be used for the calculation of the 50/50 split for the Research and Development Costs (and for clarity, any adjustments made to such reported amounts shall be taken in the calendar quarter in which such adjustments are recorded). Within [...***...] days following receipt of such report(s), for all Optioned Programs, the Party that incurred the higher aggregate Research and Development Costs for the applicable calendar quarter (across all Optioned Programs), shall invoice the other Party for one-half (1/2) of the difference between the Parties' respective aggregate Research and Development Costs for the applicable calendar quarter (across all Optioned Programs). The Party receiving such invoice shall pay it not later than [...***...] days following receipt thereof. The Parties will further work together and reasonably take into account the internal and external reporting requirements and timelines of the other Party (i) when preparing reports pursuant to this ARTICLE IX to the other Party, (ii) when aligning the respective R&D Plan and Budget, Global Commercialization Plan and Budget and Global Manufacturing Plan and Budget for each Optioned Program and Third Party License Payments and (iii) during any other planning-related exercise.

(b) Permitted R&D Cost Overrun. If in any calendar year the FTE Costs and out-of-pocket costs and expenses incurred by a Party or any of its Affiliates in connection with the activities allocated to such Party under the then-current R&D Plan and Budget exceeds the budget set forth in the then-current R&D Plan and Budget for such activities during such calendar year, then, if and to the extent that any such overspend (i) does not exceed [...***...] percent [...***...] of the budgeted costs and expenses set forth in the R&D Plan and Budget for such activities for such approval not to be unreasonably conditioned, withheld or delayed (a "**Permitted R&D Cost Overrun**"), then such Permitted R&D Cost Overrun shall be included in Research and Development Costs and expenses set forth in the R&D Plan and Budget for such activities approved by the Parties pursuant to Section 9.9(a). To the extent that any overspend exceeds [...***...] percent [...***...] of the budgeted costs and expenses set forth in the R&D Plan and Budget for such activities for such activities for such calendar year and was not otherwise approved by the Parties in writing, then such overspend shall not constitute Research and Development Costs in such calendar year and shall be borne by the Party incurring the same. The FTE Costs and out-of-pocket costs and expenses incurred by Galapagos or its Affiliates prior to the date the JSC approves the R&D Plan and Budget for the Autotaxin Program shall not count towards a Permitted R&D Cost Overrun.

(c) Post-Approval Commitments and Phase 4 Clinical Trials.

(i) <u>Mutual Post-Approval Commitments</u>. For each Optioned Program, any Research and Development Costs incurred by or on behalf of either Party or its Affiliates in fulfilling any Mutual Post-Approval Commitment (including in conducting any Requested or Required Phase 4 Clinical Trial to fulfill a Mutual Post-Approval Commitment) shall be included in the R&D Plan and Budget and the Research and Development Costs reported to the other Party and split by the Parties in accordance with Section 9.9(a). For clarity, any dispute regarding Mutual Post-Approval Commitments shall be subject to the same dispute resolution process generally applicable to the R&D Plan and Budget for such Optioned Program to which the Mutual Post-Approval Commitments relate. In the event of a Mutual Post-Approval Commitment arising from Regulatory Authorities in a non- Major Market, the Parties may discuss in good faith the impact of performing such Mutual Post-Approval Commitments on activities relating to the applicable Optioned Program in their Respective Territories.

(ii) <u>Other Post-Approval Commitments and Phase 4 Clinical Trials</u>. For each Optioned Program, any Research and Development Costs incurred by or on behalf of either Party or its Affiliates in (A) fulfilling any post-approval commitment other than a Mutual Post-Approval Commitment or (B) conducting any Phase 4 Clinical Trial that is not conducted to fulfill a Mutual Post-Approval Commitment, in each case ((A) and (B)), shall be excluded from the Research and Development Costs reported to the other Party pursuant to Section 9.9(a) and shall be borne by the Party conducting such activity, unless included in the R&D Plan and Budget or otherwise agreed by the Parties in writing.

(d) <u>Phase 3 Triggering Clinical Trial</u>. With respect to each Optioned Program for which the Triggering Clinical Trial is a Phase 3 Clinical Trial, if such Triggering Clinical Trial has reached its Completion Date, then Gilead shall pay to Galapagos an amount equal to the sum of (i) [...***...] percent [...***...] of the costs and expenses Gilead would have borne had such Triggering Clinical Trial been included in an R&D Plan and Budget and (ii) [...***...] Such payment shall be made to Galapagos within [...***...] days after receipt of an invoice with respect thereto, which invoice must be accompanied by a report in reasonable detail of such FTE Costs and out-of-pocket costs and expenses.

(e) <u>Delayed Initial Option Closing</u>. If the Initial Option Closing for a Galapagos Program has not occurred within [...***...] days after the date on which both Parties have made the necessary initial Antitrust Filings in the applicable country(ies) as set forth in Section 8.2(e), and such Initial Option Closing subsequently occurs, then Gilead shall be responsible for [...***...] percent [...***...] of the Research and Development Costs incurred for such Optioned Program beginning on the [...***...] day after the date on which both Parties have made the necessary initial Antitrust Filings in the applicable country(ies) in accordance with Section 9.9(a). Galapagos shall include any such costs in the invoices provided pursuant to Section 9.9(a).

9.10 Shared Commercialization Costs.

(a) For each Optioned Program, the Parties shall share the Shared Commercialization Costs incurred by or on behalf of either Party or its Affiliates after the applicable Option Exercise Closing with respect to the countries relating to each Option Exercise Closing, as follows: Gilead shall be responsible for fifty percent (50%) and Galapagos shall be responsible for fifty percent (50%). Within [...***...] Business Days after the end of each calendar quarter, each Party shall provide to the other Party a report in reasonable detail of any Shared Commercialization Costs incurred by such Party in such calendar quarter for each Optioned Program, which report may be based on such Party's good faith estimate. Such Shared Commercialization Costs so reported shall be used for the calculation of the 50/50 split for the Shared Commercialization Costs. Within [...***...] days following receipt of such report(s), for all Optioned Programs, the Party that incurred the higher aggregate Shared Commercialization Costs for the applicable period (across all Optioned Programs). The Party receiving such invoice shall pay it not later than [...***...] days following receipt thereof.

(b) If in any calendar year the Shared Commercialization Costs incurred by a Party exceed the budget set forth in the then-current Global Commercialization Plan and Budget for the activities allocated to it thereunder during such calendar year, then, if and to the extent that any such overspend (i) does not exceed [...***...] percent [...***...] of the budgeted costs and expenses set forth in the Global Commercialization Plan and Budget for such activities for such calendar year, or (ii) is otherwise approved by the Parties, such approval not to be unreasonably conditioned, withheld or delayed (a "**Permitted Commercialization Overrun**"), then such Permitted Commercialization Overrun shall be included in Shared Commercialization Costs and shared by the Parties pursuant to Section 9.10(a). To the extent that any overspend exceeds [...***...] percent [...***...] of the budgeted costs and expenses set forth in the Global Commercialization Plan and Budget for such activities in such calendar year and was not otherwise approved by the Parties, then such overspend shall not constitute Shared Commercialization Costs in such calendar year and shall be borne by the Party incurring the same.

9.11 Shared Patent Costs.

(a) For each Optioned Program, with respect to any Patent Costs to be [...***...] pursuant to ARTICLE X (the "**Shared Patent Costs**"), the Parties shall share such Shared Patent Costs as follows: Gilead shall be responsible for [...***...] percent [...***...] and Galapagos shall be responsible for [...***...] percent [...***...] within [...***...] Business Days after the end of each calendar quarter, each Party shall provide to the other Party a report in reasonable detail of any Shared Patent Costs incurred by such Party in such calendar quarter for each Optioned Program, which may be based on such Party's good faith estimate. Such Shared Patent Costs so reported shall be used for the calculation of the [...***...] the Shared Patent Costs. Within [...***...] days following receipt of such report(s), the Party that incurred the higher aggregate Shared Patent Costs for the applicable period shall invoice the other Party for [...***...] of the difference between the Parties' respective aggregate Shared Patent Costs for the applicable period. The Party receiving such invoice shall pay it not later than [...***...] days following receipt thereof.

(b) If Gilead requests that Galapagos incur Patent Costs with respect to a Galapagos Program and Galapagos agrees to incur such Patent Costs, then Section 9.11(a) shall apply to such Patent Costs as if such Patent Costs were Shared Patent Costs relating to an Optioned Program.

9.12 Taxes.

(a) <u>Taxes on Income</u>. Each Party shall be solely responsible for the payment of all Taxes imposed on its share of income arising directly or indirectly from the collaborative efforts of the Parties under this Agreement.

(b) <u>Tax Cooperation</u>. The Parties shall cooperate with one another and use reasonable efforts to mitigate or reduce Tax withholding or similar obligations in respect of royalties, milestone payments, and other payments made between the Parties under this Agreement. Without limiting the generality of the foregoing, each Party shall provide the other with any Tax forms and other information that may be reasonably necessary in order to eliminate or reduce withholding based on an applicable treaty. Each Party shall provide the other with reasonable assistance to enable the recovery, as permitted by Applicable Law, of withholding Taxes, VAT, or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding Tax or VAT.

(c) <u>VAT</u>. It is understood and agreed between the Parties that any payments made by Gilead under this Agreement are exclusive of any value added tax ("**VAT**") or similar Tax imposed upon such payments. Where VAT is properly added to a payment made under this Agreement, Gilead will pay the amount of VAT only on receipt of a valid Tax invoice issued in accordance with Applicable Law.

(d) <u>Payment of Tax</u>. To the extent Gilead is required by Applicable Law to deduct or withhold Taxes on any payment to Galapagos, Gilead shall pay the amounts of such Taxes to the proper Governmental Authority in a timely manner; *provided that* Gilead must provide Galapagos with at least [...***...] days' notice of the applicability of any withholding Tax with respect to such payment, and Gilead and Galapagos shall [...***...] work together during such [...***...] day period to eliminate or reduce such withholding. Any amount withheld or deducted by Gilead pursuant to this provision shall be treated as having been paid to Galapagos for purposes of this Agreement. Gilead must promptly (within [...***...] days of payment) transmit to Galapagos an official Tax certificate or other evidence of any withholding sufficient to enable Galapagos to claim available credits for withholding Tax deducted or withheld by Gilead.

(e) <u>No Partnership</u>. Galapagos and Gilead intend that this Agreement will not be treated as a partnership or joint venture for United States federal and state tax purposes, and each of Galapagos and Gilead will file all tax returns and will otherwise take all tax reporting positions in a manner consistent with such treatment.

9.13 Foreign Exchange. All payments shall be paid in Dollars. For purpose of computing such payments, the Net Sales of Optioned Products in countries other than the United States and other amounts reimbursable by the other Party hereunder shall be converted into Dollars in accordance with the standard practices used by the applicable Party or its Affiliate or Sublicensee receiving the Net Sales of the applicable Optioned Products or incurring the reimbursable expense in preparing its audited financial statements for the applicable calendar quarter. Gilead's standard worldwide currency conversion methodology on the Execution Date is [...***...]. Galapagos' standard worldwide currency conversion methodology prior to any such changes becoming effective.

9.14 <u>Late Payments</u>. If a Party does not receive payment of any sum due to it on or before the due date therefor, simple interest shall thereafter accrue on the unpaid sum due to such Party from the due date until the date of payment at a per-annum rate of [...***...] percent [...***...] above the prime rate as reported in The Wall Street Journal, Eastern Edition, or the maximum rate allowable by Applicable Law, whichever is less.

9.15 Financial Records; Audits. Each Party and its Affiliates shall use all reasonable efforts to maintain complete and accurate records in sufficient detail to permit the other Party to confirm the accuracy of the amounts to be reimbursed, pursuant to this ARTICLE IX, with respect to Research and Development Costs, Shared Commercialization Costs, Shared Patents Costs or other amounts to be reimbursed, credited, offset or shared hereunder incurred or generated (as applicable) by such Party's or its Affiliates' achievement of milestones, royalty payments and other compensation or reimbursement payable under this Agreement. Upon reasonable prior notice, such records shall be open during regular business hours for a period of [... ***...] years from the creation of individual records for examination not more often than once each calendar year, by an independent certified public accountant selected by the auditing Party and reasonably acceptable to the audited Party or its applicable Affiliate for the sole purpose of verifying for the auditing Party the accuracy of the financial statements or reports or sales milestone notices furnished by the audited Party or such Affiliate pursuant to this Agreement or of any payments made, or required to be made, by or to the audited Party or such Affiliate to the other pursuant to this Agreement. Any such auditor shall not disclose the audited Party's or its Affiliates' confidential information to the auditing Party, but shall, instead, report that there was or was not a discrepancy uncovered by the audit and if such a discrepancy was uncovered, the amount and direction of it. Any amounts shown to be owed but unpaid, or overpaid and in need of reimbursement, shall be paid or refunded (as the case may be) within [...***...] days after the auditor's report, plus interest (as set forth in Section 9.14) from the original due date (unless challenged in good faith by the audited Party, in which case any undisputed portion shall be paid in accordance with the foregoing timetable, any dispute with respect to such challenge shall be resolved in accordance with ARTICLE XV, any remaining disputed portion shall be paid within [...***...] days after resolution of the dispute, and interest shall not accrue with respect to the disputed portion during the period of time the dispute is being resolved). The auditing Party shall bear the full cost and expense of such audit unless such audit reveals an overpayment to, or an underpayment by, the audited Party or its Affiliates that resulted from a discrepancy in a report that the audited Party or its Affiliates provided to the other Party during the applicable audit period, which underpayment or overpayment was more than [...***...] percent [...***...] of the amount set forth in such report, in which case the audited Party or its applicable Affiliate shall bear the full cost and expense of such audit. Each Party, at the request of the other Party, shall make available to the other Party the results of any audit performed by the non-requesting Party on such non-requesting Party's Sublicensees hereunder. During the Term, each Party shall consider in good faith other reasonable procedures proposed by the other Party for sharing financial information in order to permit the other Party to close its books periodically in a timely manner.

9.16 <u>Manner and Place of Payment</u>. All payments owed under this Agreement shall be made by wire transfer in immediately available funds to a bank and account designated in writing by Galapagos or Gilead (as applicable), unless otherwise specified in writing by such Party.

9.17 No Double Counting. No cost or expense included in calculations pursuant to this ARTICLE IX shall be counted more than once.

ARTICLE X

INTELLECTUAL PROPERTY

10.1 <u>Ownership of Collaboration IP</u>. Subject to Section 7.4(a), Collaboration IP shall be owned as set forth in the remainder of this Section 10.1. Inventorship will be determined in accordance with U.S. Applicable Laws as to inventorship.

(a) <u>Galapagos Collaboration IP</u>. As between the Parties, Galapagos shall solely own all right, title and interest in and to all (i) Galapagos Foreground Know-How and Galapagos Program Period Know-How (collectively, the "**Galapagos Collaboration Know-How**") and (ii) Galapagos Foreground Patents and Galapagos Program Period Patents (collectively, the "**Galapagos Collaboration Know-How**") and (ii) Galapagos Foreground Patents and Galapagos Program Period Patents (collectively, the "**Galapagos Collaboration Row-How**") and all right, title and interest in and to the Galapagos. Gilead shall promptly disclose to Galapagos any inventions within the Galapagos Program Period Know-How conceived, discovered, developed, generated or otherwise made by or on behalf of Gilead or any of its Affiliates, and shall provide to Galapagos documentation regarding any such invention as Galapagos may reasonably request. Gilead, for itself and on behalf of its Affiliates and employees, hereby assigns (and to the extent such assignment can only be made in the future hereby agrees to assign) to Galapagos, Gilead's entire right, title and interest in and to the Galapagos Collaboration IP. Gilead shall, and shall cause its Affiliates to, cooperate with Galapagos to effectuate and perfect the foregoing ownership, including by promptly executing and recording assignments and other documents consistent with such ownership.

(b) <u>Gilead Collaboration IP</u>. As between the Parties, Gilead shall solely own all right, title and interest in and to all (i) Gilead Collaboration Know-How and (ii) Gilead Collaboration Patents, and all right, title and interest in and to the Gilead Collaboration Know-How and Gilead Collaboration Patents (collectively, the "**Gilead Collaboration IP**") shall automatically vest solely in Gilead.

(c) Joint Collaboration IP. As between the Parties, the Parties shall jointly own all right, title and interest in and to all Joint Collaboration Know-How and Joint Collaboration Patents (collectively "Joint Collaboration IP"). Subject to the terms of this Agreement (including, for clarity, Sections 8.1, 8.3, and 8.4, and the applicable Sections of this ARTICLE X), each Party shall have the right (without requiring the consent of, or accounting to, the other Party) to use, practice or otherwise exploit the Joint Collaboration IP in its Respective Territory and to use, practice or otherwise Exploit solely for internal research and development purposes the Joint Collaboration IP worldwide, and for no other purpose unless such Party obtains the prior written consent of the other Party (including through a license to the other Party's interest in such Joint Collaboration IP) to use, practice or otherwise Exploit any applicable Joint Collaboration IP for another purpose.

(d) <u>Joint Research Agreement</u>. As between the Parties, this Agreement shall be deemed to be a joint research agreement in accordance with 35 U.S.C. §103(c) or 35 U.S.C. §102(c), as applicable; *provided that* neither Party shall (i) unilaterally invoke the protections of or (ii) be required by this reference to have any Patent take advantage of or become subject to, 35 U.S.C. §103(c) or 35 U.S.C. §102(c), as applicable, except with the prior written consent of the other Party.

10.2 Prosecution of Patents

(a) <u>Patent Prosecution Committee</u>. Promptly (but no later than [...***...] days) after the Effective Date, the Parties will establish a patent prosecution committee (the "**Patent Prosecution Committee**"). The purpose of the Patent Prosecution Committee is to (i) discuss the general Prosecution strategies (including countries in which to file) regarding Patents for Pre-Program Activities and Galapagos Programs, (ii) review and agree on Prosecution strategies for the Optioned Programs (and Optioned Molecules and Optioned Products), (iii) discuss any election (or potential election) by a Party to cease Prosecution of any Patents for which it is responsible under this Section 10.2, (iv) discuss any election (or potential election) by a Party to cease sharing Patent Costs for any Patents for which it is [...***...], and (v) discuss the progress of any opposition proceeding, appeal or other action at the European Patent Office being defended pursuant to Section 10.5(b)(iii)(A). The Patent Prosecution Committee shall include members from both Parties who have appropriate experience with respect to Prosecution. Either Party may invite non-members to participate in the discussions and meetings of the Patent Prosecution Committee provided notice is given to the other Party, and such non-members shall be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII. The Patent Prosecution Committee will meet monthly or as otherwise agreed by the Parties. All final decisions related to the Prosecution of any Patent will be made by the Party with the right to control such Prosecution as set forth in this Section 10.2.

(b) Prosecution of Patents for Pre-Program Activities and Galapagos Programs.

(i) Galapagos Patents.

(A) Galapagos shall have the sole right and authority, but not the obligation, to prepare, file, prosecute and maintain (such activities, collectively, "**Prosecution**") the Galapagos Patents for any Pre-Program Activities and Galapagos Programs (other than Optioned Programs) in any jurisdiction using counsel of its choice, unless there is a conflict between said counsel and Gilead, in which case new counsel shall be selected that is reasonably acceptable to Gilead. Galapagos shall take into consideration the general strategies agreed by the Patent Prosecution Committee in connection with such Prosecution. If Galapagos intends to take any action that is inconsistent with such general strategies, it shall inform Gilead and reasonably consider in good faith any comments provided by Gilead with respect to such inconsistency.

(B) Upon Gilead's request (to be made no more than once per [...***...] period), Galapagos shall provide Gilead with highlevel summaries of the Galapagos Patents for any Pre-Program Activities and Galapagos Programs (other than Optioned Programs); *provided that* such summaries shall not include the [...***...].

(C) If Galapagos notifies Gilead of a Suspension or Termination of any Pre-Program Activities or Galapagos Program in accordance with Section 2.4 [...***...].

(D) Galapagos shall be solely responsible for all Patent Costs incurred in connection with the Prosecution of any Galapagos Patent for any Pre-Program Activities and Galapagos Programs (other than Optioned Programs).

(ii) <u>Gilead Collaboration Patents and Joint Collaboration Patents</u>. The Parties acknowledge that there will not be any Gilead Collaboration Patents or Joint Collaboration Patents claiming, covering, or arising from any Pre-Program Activities or Galapagos Programs (other than Optioned Programs) in any jurisdiction because all Collaboration Patents generated, solely or jointly, by or on behalf of Gilead or its Affiliates in connection with Pre-Program Activities or Galapagos Programs (other than Optioned Programs) would be through Gilead Contributions and, pursuant to Section 2.2, be deemed to be Galapagos Program Period Patents and therefore Galapagos Patents.

(c) Prosecution of Patents for Optioned Programs.

(i) Galapagos Patents and Joint Collaboration Patents.

(A) Galapagos shall have the first right and authority, but not the obligation, to Prosecute the Galapagos Patents for any Optioned Program and Joint Collaboration Patents in any jurisdiction using counsel of its choice, unless there is a conflict between said counsel and Gilead, in which case new counsel shall be selected that is reasonably acceptable to Gilead. Galapagos shall keep Gilead reasonably informed of all material matters relating to the Prosecution of such Galapagos Patents and Joint Collaboration Patents (including providing Gilead with copies of all material correspondence with Patent offices or other Governmental Authorities) and shall reasonably consider in good faith any comments provided by Gilead with respect to such submissions. With respect to any such Galapagos Patent and Joint Collaboration Patents filed after the closing of the Option for the applicable Optioned Program, Galapagos shall coordinate with Gilead through the Patent Prosecution Committee regarding the countries or territories in which such Galapagos Patent and Joint Collaboration Patents shall be filed. If Gilead desires that any such Galapagos, then upon Gilead's written request identifying such additional country(ies) or territory(ies), Galapagos shall cause such Galapagos Patent or Joint Collaboration Patents, as applicable, to be filed in such additional country(ies) or territory(ies).

(B) The Parties shall [...***...] all Patent Costs incurred in connection with the Prosecution of any Galapagos Patent for any Optioned Program and for any Joint Collaboration Patent.

(C) In the event that Gilead elects not to [...***...] Patent Costs for any Galapagos Patent for any Optioned Program or for any Joint Collaboration Patents, Gilead shall provide Galapagos with at least [...***...] days written notice thereof, and (1) if such Patent is a Galapagos Patent it shall, at the end of such notice period, be automatically deemed to be removed from the definition of "Galapagos Patents" under this Agreement, the licenses granted to Gilead and its Affiliates as to such Patent shall terminate, and Galapagos shall have no obligations or restrictions with respect to such Patent under this Agreement or (2) if such Patent is Joint Collaboration Patent it shall, at the end of such notice period, be automatically deemed to be removed from the definition of "Joint Collaboration Patents" under this Agreement, the licenses granted to Gilead and its Affiliates as to such Patent shall terminate, and Galapagos shall have no obligations or restrictions with respect to such Patent under this Agreement and shall be free to use, practice or otherwise Exploit such Joint Collaboration Patent for any purpose without the consent of, or accounting to, Gilead. In the event that Galapagos elects not to Prosecute any Galapagos Patent for any Optioned Program or any Joint Collaboration Patents, Galapagos shall notify Gilead in writing at least [...***...] days before any such Galapagos Patent or Joint Collaboration Patent would become abandoned or rights would otherwise be forfeited with respect thereto, and, unless Galapagos has a *bona fide* strategic reason for such election after considering, reasonably and in good faith, all input received from Gilead, Gilead shall have the right, but not the obligation, to assume Prosecution of such Galapagos Patent or Joint Collaboration Patent, in which case Gilead shall be solely responsible for all Patent Costs with respect to such Patent.

(D) On a [...***...] basis, Galapagos will provide summaries in reasonable detail of the status of Prosecution of the Galapagos Patents and Joint Collaboration Patents that Galapagos is Prosecuting pursuant to this Section 10.2(c)(i). Prior to September 1 of each year, the Parties will provide the Patent Prosecution Committee with an initial estimated budget of any Patent Costs [...***...] between the Parties under this Agreement. The Patent Prosecution Committee will evaluate such initial estimates and, prior to November 1 of each year, agree upon an estimated budget for any Patent Costs shared between the Parties for the upcoming year.

(ii) Gilead Collaboration Patents.

(A) Gilead shall have the first right and authority, but not the obligation, to Prosecute the Gilead Collaboration Patents in any jurisdiction using counsel of its choice, unless there is a conflict between said counsel and Galapagos, in which case new counsel shall be selected that is reasonably acceptable to Galapagos. Gilead shall be solely responsible for all Patent Costs incurred in connection with the Prosecution of the Gilead Collaboration Patents. Gilead shall keep Galapagos reasonably informed of all material matters relating to the Prosecution of the Gilead Collaboration Patents in the Major Markets (including providing Galapagos with copies of all material correspondence with Patent offices or other Governmental Authorities for the Major Markets) to the extent reasonably related to any Galapagos Molecule or Optioned Molecule, and shall reasonably consider in good faith any comments provided by Galapagos with respect to such correspondence. Galapagos shall bear any Patent Costs it may incur in connection with any review and consultation concerning any Gilead Collaboration Patent.

(B) In the event that Gilead elects not to Prosecute any Gilead Collaboration Patents, Gilead shall notify Galapagos in writing at least [...***...] days before any such Collaboration Patent would become abandoned or rights would otherwise be forfeited with respect thereto, and, unless Gilead has a *bona fide* strategic reason for such election after considering, reasonably and in good faith, all input received from Galapagos, Galapagos shall have the right, but not the obligation, to assume Prosecution of such Gilead Collaboration Patent, in which case Galapagos shall be solely responsible for all Patent Costs with respect to such Patent.

(d) <u>Cooperation in Prosecution</u>. Each Party shall provide the other Party all reasonable notice, assistance and cooperation in the Prosecution activities set forth in this Section 10.2, including providing any necessary powers of attorney and executing any other required documents or instruments for such prosecution.

10.3 <u>Patent Term Extensions</u>. Each Party shall have the sole right and authority, in consultation with the other Party, to apply for and obtain any Patent term extension or related extension of rights, including supplementary protection certificates and similar rights (collectively, "**Patent Term Extensions**"), for any Galapagos Patent, Gilead Collaboration Patent or Joint Collaboration Patent, in each case, in its Respective Territory. If the Parties disagree on the appropriate strategy with respect to any Patent Term Extension for any Galapagos Patent, Gilead Collaboration Patent or Joint Collaboration Patent, then such dispute shall be subject to resolution by the JSC; *provided that* if the JSC is unable to reach consensus on the strategy, then each Party shall have final decision-making authority with respect to such strategy in its Respective Territory and neither Party shall have the right to escalate such dispute beyond the JSC. In exercising such final decision-making authority after consideration by the JSC, each Party shall consider, reasonably and in good faith, all input received from the other Party. Each Party shall provide reasonable assistance to the other Party in connection with applying for and obtaining any Patent Term Extensions for any Galapagos Patent, Gilead Collaboration Patent. To the extent reasonably and legally required in order to obtain any Patent Term Extension in any country, each Party shall make available to the other a copy of the necessary documentation Controlled by such Party to enable such other Party to use the same for the purpose of obtaining such Patent Term Extension in such country. Any Patent Costs incurred in connection with any Patent Term Extension for any Galapagos Patent, Gilead Collaboration Patent, Gilead Collaboration Patent term Extension in such country. Any Patent Costs incurred in connection with any Patent Term Extension for any Galapagos Patent, Gilead Collaboration Patent term Extension.

(a) <u>Optioned Product Orange Book Listings</u>. With respect to any Galapagos Patent, Gilead Collaboration Patent or Joint Collaboration Patent, Gilead shall have the sole right and authority, in its discretion, to make any filing with respect to any Optioned Product in connection with the FDA's Orange Book or any preparation of a list of Patents under section 351 of the Public Health Service Act or any post-approval patent linkage and registration inside or outside the United States (any such filing, an "**Optioned Product Orange Book Listing**"). Gilead shall provide Galapagos a copy of all Optioned Product Orange Book Listings. Galapagos shall provide reasonable assistance to Gilead in connection with any Optioned Product Orange Book Listing. The Parties shall [...***...] all Patent Costs incurred in connection with any Optioned Product Orange Book Listing.

10.4 Infringement by Third Parties.

(a) <u>Notification of Program Infringement</u>. If either Party becomes aware of any infringement, threatened infringement, or alleged infringement (i) of any Galapagos Patent, Joint Collaboration Patent or Gilead Collaboration Patent by a Third Party that is manufacturing, using, importing, marketing or selling a Molecule or product that competes with (A) a Molecule or product that is the subject of such Pre-Program Activities, (B) a Galapagos Molecule or Galapagos Product, or (C) an Optioned Molecule or Optioned Product, as applicable, or (ii) as a result of a notification to such Party or any of its Affiliates pursuant to Sections 505(j)(2)(B) or 505(b)(3) of the FD&C Act (21 U.S.C. § 355(j)(2)(B) and 21 U.S.C. § 355(b)(3)) or Section 351(l) of the Public Health Service Act (42 U.S.C. 262(l)), or a foreign equivalent, of an application for approval of a Generic Product with respect to an applicable product, Molecule, Galapagos Molecule, Galapagos Product, Optioned Molecule or Optioned Product, as applicable (each of (i) and (ii), a "**Program Infringement**"), then such Party shall promptly notify the other Party in writing thereof and provide evidence in such Party's possession with respect thereto.

(b) Establishment of Patent Litigation Committee. If the notice provided under Section 10.4(a) is based on competition with, or a Generic Product of, an Optioned Molecule or Optioned Product then, promptly (but no later than [...***...] Business Days) after such notice (or after such Molecule or product becomes an Optioned Molecule or Optioned Product), the Parties will establish a patent litigation committee (the "**Patent Litigation Committee**"). The purpose of the Patent Litigation Committee is to facilitate the discussion and coordination of Patent enforcement and defense matters related to Optioned Programs in accordance with and subject to the terms of this Agreement. The Patent Litigation Committee shall include members from both Parties who have appropriate experience for the matters to be discussed. Either Party may invite non-members to participate in the discussions and meetings of the Patent Litigation Committee provided notice is given to the other Party, and such non-members shall be subject to confidentiality obligations at least as stringent as those set forth in ARTICLE XIII. The Patent Litigation Committee will meet monthly or as otherwise agreed by the Parties. All final decisions related to the enforcement or defense of any Patent will be made by the Party with the right to control such enforcement or defense, as applicable, as set forth in this ARTICLE X. If needed, the Parties shall put in place a common interest agreement to allow full sharing of information related to such enforcement or defense.

(c) Enforcement Based on Competition with Pre-Program Activities and Galapagos Programs

(i) <u>Galapagos Patents</u>. With respect to any Program Infringement of any Galapagos Patent based on competition with, or a Generic Product of, a Molecule or product that is the subject of Pre-Program Activities or a Galapagos Molecule or Galapagos Product, Galapagos shall have the sole right and authority, but not the obligation, to bring suit or other action to abate such infringement; *provided that*, Galapagos shall (1) provide at least [...***...] days' written notice to Gilead of its intent to bring such suit or other action; (2) consider, reasonably and in good faith, all input received from Gilead with respect thereto; and (3) conduct such suit or other action in a manner that Galapagos reasonably believes to be in the best interests of the Development and Commercialization of such Molecule, product, Galapagos Molecule or Galapagos Product.

(ii) <u>Gilead Collaboration Patents and Joint Collaboration Patents</u>. The Parties acknowledge that there will not be any Gilead Collaboration Patents or Joint Collaboration Patents claiming, covering, or arising from any Pre-Program Activities or Galapagos Programs (other than Optioned Programs) in any jurisdiction because all Collaboration Patents generated, solely or jointly, by or on behalf of Gilead or its Affiliates in connection with Pre-Program Activities or Galapagos Programs (other than Optioned Programs) would be through Gilead Contributions and, pursuant to Section 2.2, be deemed to be Galapagos Program Period Patents and therefore Galapagos Patents.

(d) Enforcement Based on Competition with Optioned Programs. With respect to any Program Infringement of any Galapagos Patent, Joint Collaboration Patent, or Gilead Collaboration Patent based on competition with, or a Generic Product of, an Optioned Molecule or Optioned Product, Gilead shall have the first right and authority, but not the obligation, to bring suit or other action to abate any Program Infringement of any Galapagos Patents, Joint Collaboration Patents and Gilead Collaboration Patent; *provided that*, Gilead shall (1) provide at least [...***...] days' written notice to Galapagos of its intent to bring such suit or other action; (2) consider, reasonably and in good faith, all input received from Galapagos and all discussions held by the Patent Litigation Committee with respect thereto and, with respect to the Galapagos Territory, incorporate any Galapagos comments in any filings and strategies; (3) conduct such suit or other action in a manner that Gilead reasonably believes (taking into account all input received from Galapagos and all discussions held by the Patent Litigation Committee) to be in the best interests of the Development and Commercialization of such Optioned Molecules and Optioned Products and (4) keep Galapagos and the Patent Litigation Committee fully informed and allow Galapagos to actively participate in all aspects of such suit or action. If Gilead does not inform Galapagos that it intends to bring such suit or other action to abate such Program Infringement within [...***...] days after notification of such Program Infringement pursuant to Section 10.4(a), then, unless, with respect to Gilead Territory only, Gilead has a *bona fide* strategic reason for not enforcing such Gilead Patent, Galapagos shall have the second right and authority, but not the obligation, to bring such suit or other action.

(e) Other Infringement.

(i) <u>Galapagos Patents and Joint Collaboration Patents</u>. With respect to any infringement, threatened infringement or alleged infringement of any Galapagos Patent or Joint Collaboration Patent that is not a Program Infringement, Galapagos shall have the sole right and authority, but not the obligation, to bring suit or other action to abate such infringement; *provided that* Galapagos shall (A) provide at least [...***...] days' written notice to Gilead of its intent to bring such suit or other action, (B) consider, reasonably and in good faith, all input received from Galapagos with respect thereto and (C) conduct such suit or other action in a manner that Galapagos reasonably believes to be in the best interests of the Development and Commercialization of Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products.

(ii) <u>Gilead Collaboration Patents</u>. With respect to any infringement, threatened infringement or alleged infringement of any Gilead Collaboration Patent that is not a Program Infringement, Gilead shall have the sole right and authority, but not the obligation, to bring suit or other action to abate such infringement; *provided that* Gilead shall (A) provide at least [...***...] days' written

notice to Galapagos of its intent to bring such suit or other action, (B) consider, reasonably and in good faith, all input received from Galapagos with respect thereto and, with respect to the Galapagos Territory, incorporate any Galapagos comments in any filings and strategies, (C) conduct such suit or other action in a manner that Gilead reasonably believes to be in the best interests of the Development and Commercialization of Molecules and Product that are the subject of Pre-Program Activities, Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products and (D) provide Galapagos with material submissions made in connection with such suit or action.

(f) <u>Cooperation and Information Sharing</u>. With respect to any suit or other action under this Section 10.4, the Party that is not bringing such suit or other action ("**Non-Enforcing Party**") shall cooperate fully as may be reasonably requested by the Party bringing such suit or other action ("**Enforcing Party**"), upon reasonable notice, to maintain such suit or other action, by executing and making available such documents as the Enforcing Party may reasonably request, and by performing all other acts which are or may become reasonably necessary to vest in the Enforcing Party the right to institute any such suit or other action, including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties. The Enforcing Party shall keep the Non-Enforcing Party regularly informed of the status and progress of such efforts, and shall reasonably consider the Non-Enforcing Party's comments on any such efforts.

(g) <u>Settlement</u>. Without the prior written consent of the other Party (such consent not to be unreasonably withheld, conditioned or delayed), neither Party shall settle any claim, suit or action that is brought under this Section 10.4 with respect to any Galapagos Patent, Gilead Collaboration Patent or Joint Collaboration Patent in any manner that would (i) reduce the scope of, or admit the invalidity or unenforceability of, such Patent, (ii) admit any liability by the other Party, or (iii) materially limit the rights of, or materially expand the obligations of, the other Party under this Agreement or any Ancillary Agreement.

(h) Costs and Expenses; Recoveries.

(i) <u>Enforcement Based on Competition with Pre-Program Activities and Galapagos Programs</u>. With respect to any suit or other action brought by either Party pursuant to Section 10.4(c), unless otherwise agreed by the Parties in writing, the Party filing such suit or taking such action shall be solely responsible for all Patent Costs incurred by such Party in connection with such suit or other action, and any Patent Costs incurred by the other Party or any of its Affiliates in connection with any cooperation requested by such Party with respect to such suit or other action. Any monetary damages recovered from a Third Party as a result of such suit or other action shall be [...***...].

(ii) <u>Enforcement Based on Competition with Optioned Programs</u>. With respect to any suit or other action brought by either Party pursuant to Section 10.4(d), unless otherwise agreed by the Parties in writing, the Parties shall [...***...] in (A) [...***...] and (B) [...***...].

(iii) <u>Other Infringement</u>. With respect to any suit or other action brought by either Party pursuant to Section 10.4(e), unless otherwise agreed by the Parties in writing, [...***...] shall be [...***...]. Any monetary damages recovered from a Third Party as a result of such suit or other action shall be retained by [...***...].

10.5 Defense of Patents.

(a) <u>Notification</u>. If either Party becomes aware of any Invalidity or Unenforceability Action with respect to any Galapagos Patent, Joint Collaboration Patent or Gilead Collaboration Patent, then such Party shall promptly notify the other Party in writing thereof and provide evidence in such Party's possession with respect thereto. If such Invalidity or Unenforceability Action is with respect to Program Claims that claim or cover an Optioned Molecule or Optioned Product or its exploitation, and a Patent Litigation Committee has not already been established, the Parties shall establish the Patent Litigation Committee as set forth in Section 10.4(b) within [...***...] Business Days of such notice.

(b) Patent Defense Rights.

(i) <u>Defense of Invalidity or Unenforceability Actions Brought as Defenses or Counterclaims</u>. Notwithstanding any other provision of this Section 10.5(b), if any Invalidity or Unenforceability Action with respect to any Galapagos Patent, Gilead Collaboration Patent or Joint Collaboration Patent is brought as a defense or counterclaim to a suit or other action enforcing such Patent under Section 10.4, then the Party enforcing such Patent under Section 10.4 shall have the sole right and authority to defend such Invalidity or Unenforceability Action; *provided that*, the applicable notice, consideration, conduct, information sharing and settlement provisions of the applicable subsection of Section 10.4 shall apply, *mutatis mutandis*.

(ii) Defense of Claims Not Covering an Optioned Molecule or Optioned Product.

(A) <u>Galapagos Patents and Joint Collaboration Patent</u>. Subject to Section 10.5(b)(i), Galapagos shall have the sole right and authority, but not the obligation, to defend any Invalidity or Unenforceability Action with respect to each claim in a Galapagos Patent and Joint Collaboration Patent that is not a Program Claim claiming or covering an Optioned Molecule or Optioned Product.

(B) <u>Gilead Collaboration Patents</u>. Subject to Section 10.5(b)(i), Gilead shall have the sole right and authority, but not the obligation, to defend any Invalidity or Unenforceability Action with respect to claim in a Gilead Collaboration Patent that is not a Program Claim claiming or covering an Optioned Molecule or Optioned Product; *provided that* Gilead shall (1) provide at least [...***...] days' written notice to Galapagos of its intent to defend such Invalidity or Unenforceability Action; (2) consider, reasonably and in good faith, all input received from Galapagos with respect thereto; (3) conduct such defense in a manner that Gilead reasonably believes to be in the best interests of the Development and Commercialization of Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products and (4) provide Galapagos with material submissions made in connection with such Invalidity and Unenforceability Action.

(iii) Defense of Program Claims Covering an Optioned Molecule or Optioned Product.

(A) Subject to Section 10.5(b)(i), if any Invalidity or Unenforceability Action is an opposition proceeding, appeal or other action at the European Patent Office with respect to any Program Claim in a Galapagos Patent or Joint Collaboration Patent that claims or covers an Optioned Molecule or Optioned Product or its exploitation, Galapagos shall have the first right and authority, but not the obligation, to defend such Invalidity or Unenforceability Action; *provided that* Galapagos shall (1) provide at least [...***...] days' written notice to Gilead of its intent to defend such Invalidity or Unenforceability Action; (2) consider, reasonably and in good faith, all input received from Gilead with respect thereto and, with respect to the Gilead Territory, incorporate any Gilead comments in any filings and strategies; and (3) keep Gilead reasonably informed on the progress of such proceeding, appeal or other action. If Galapagos does not elect to defend any such Invalidity or Unenforceability Action, then, unless Galapagos has a *bona fide* strategic reason for not defending such Galapagos Patent or Joint Collaboration Patent, subject to Section 10.5(b)(i), Gilead shall have the second right and authority, but not the obligation, to defend such Invalidity or Unenforceability Action with respect to any such Program Claim.

(B) Subject to Section 10.5(b)(i) and 10.5(b)(ii)(A), Gilead shall have the first right and authority, but not the obligation, to defend any Invalidity or Unenforceability Action with respect to each Program Claim in a Galapagos Patent, Joint Collaboration Patent, and Gilead Collaboration Patent that claims or covers an Optioned Molecule or Optioned Product, *provided that*, Gilead shall (1) provide at least [...***...] days' written notice to Galapagos of its intent to defend such action; (2) consider, reasonably and in good faith, all input received from Galapagos and all discussions held by the Patent Litigation Committee with respect thereto; (3) conduct such defense in a manner that Gilead reasonably believes (taking into account all input received from Galapagos and all discussions held by the Patent Litigation of such Optioned Molecules and Optioned Products, (4) with respect to the Galapagos Territory, incorporate any Galapagos comments in any filings and strategies and (5) keep Galapagos and the Patent Litigation Committee fully informed and allow Galapagos to actively participate in all aspects of such defense. If Gilead does not elect to defend any such Invalidity or Unenforceability Action, then it shall provide written notice to Galapagos at least [...***...] days' prior to the next deadline for taking any action with respect thereto, and, subject to Section 10.5(b) (i), unless, with respect to the Gilead Territory, Gilead has a *bona fide* strategic reason for not defending such Joint Collaboration Patent, Galapagos shall have the second right and authority, but not the obligation, to defend such Invalidity or Unenforceability Action with respect to any such Program Claim.

(c) <u>Cooperation</u>. Each Party shall provide to the Party defending any Invalidity or Unenforceability Action with respect to any Program Claim under this Section 10.5 all reasonable assistance in such defense, at such defending Party's request and expense.

(d) Costs and Expenses.

(i) <u>As Part of a Defense or Counterclaim</u>. If any Invalidity or Unenforceability Action with respect to any Galapagos Patent or Gilead Collaboration Patent is brought as a defense or counterclaim to a suit or other action enforcing such Patent under Section 10.5(b)(i), then all Patent Costs incurred by either Party in connection with defending such Invalidity or Unenforceability Action shall be allocated in accordance with Section 10.4(h).

(ii) <u>Defense of Claims Not Covering an Optioned Molecule or Optioned Product</u>. With respect to the defense of any Invalidity or Unenforceability Action controlled by a Party pursuant to Section 10.5(b)(ii), such Party shall be solely responsible for all Patent Costs incurred by such Party in connection with such defense, and any Patent Costs incurred by the other Party or any of its Affiliates in connection with any cooperation requested by such Party with respect to such defense.

(iii) <u>Defense of Program Claims Covering an Optioned Molecule or Optioned Product</u>. Patent Costs incurred by both Parties in connection with the defense of any Invalidity or Unenforceability Action shall be shared as follows: (x) with respect to the defense of any Invalidity or Unenforceability Action 10.5(b)(iii)(A), such Patent costs shall be [...***...] and shall b

10.6 Defense of Infringement or Misappropriation Actions.

(a) <u>Notification</u>. For each Galapagos Program, if either Party becomes aware of any potential claim, or claim, of infringement or misappropriation of Third Party intellectual property rights in connection with the Exploitation of any applicable Galapagos Molecule or Galapagos Product ("**Infringing Activity**"), then such Party shall promptly notify the other Party thereof. This Section 10.6(a) is not intended, and shall not be construed, as placing on either Party a duty of inquiry regarding Third Party intellectual property rights. If such notice is with respect to an Optioned Molecule or Optioned Product, and a Patent Litigation Committee has not already been established, the Parties shall establish the Patent Litigation Committee as set forth in Section 10.4(b) within [...***...] Business Days of such notice.

(b) <u>Pre-Program Activities; Galapagos Programs</u>. Galapagos shall have the sole right and authority, but not the obligation, to defend any action, suit, or other proceeding brought against either Party alleging Infringing Activity with respect to any Galapagos Program or any Pre-Program Activities, and Gilead shall reasonably cooperate with Galapagos (including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties), in connection with the defense of such action, suit or proceeding. Any costs and expenses incurred by Galapagos in connection with defending any action, suit or other proceeding under this Section 10.6(b), and any amounts payable to Third Parties for damages or other compensation in connection with any such action, suit or other proceeding, [...***...].

(c) Optioned Programs, Gilead shall have the first right and authority, but not the obligation, to defend any action, suit, or other proceeding brought against either Party alleging Infringing Activity with respect to any Optioned Program (or Optioned Molecule or Optioned Product) provided that, Gilead shall (1) provide at least [...***...] days' written notice to Galapagos of its intent to defend such action, suit or other proceeding; (2) consider, reasonably and in good faith, all input received from Galapagos and all discussions held by the Patent Litigation Committee with respect thereto; (3) conduct such defense in a manner that Gilead reasonably believes (taking into account all input received from Galapagos and all discussions held by the Patent Litigation Committee) to be in the best interests of the Development and Commercialization of the applicable Optioned Molecules and Optioned Products, (4) if such action, suit, or other proceeding relates to activities by or on behalf of Gilead or its Affiliates in the Gilead Territory, Gilead shall keep Galapagos reasonably informed and (5) if such action, suit or proceeding relates to the Galapagos Territory or to activities conducted by or on behalf of Galapagos or its Affiliates, keep Galapagos and the Patent Litigation Committee fully informed, allow Galapagos to actively participate in all aspects of such defense and incorporate all reasonable comments made by Galapagos. Galapagos shall reasonably cooperate with Gilead (including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties), in connection with the defense of such action, suit or proceeding. In the event that Gilead does not defend any such action, suit, or other proceeding, Galapagos shall have the right to do so, and Gilead shall reasonably cooperate with Galapagos (including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties), in connection with defense of such action, suit or proceeding. Any costs and expenses incurred by either Party in connection with defending any action, suit or other proceeding under this Section 10.6(c), and any amounts payable to Third Parties for damages or other compensation in connection with any such action, suit or proceeding, shall be (A) [...***...] or, (B) [...***...]. Notwithstanding the foregoing, a Party may not settle any action, suit or other proceeding under this Section 10.6(c), without the prior written consent of the Party against whom the Infringing Activity has been alleged.

10.7 <u>Patent Marking</u>. Each Party shall, and shall require its Affiliates and Sublicensees to, use Commercially Reasonable Efforts to mark Optioned Products sold by it hereunder (in a reasonable manner consistent with industry custom and practice) with appropriate Patent numbers or indicia to the extent permitted by Applicable Law, in those countries in which such markings or such notices impact recoveries of damages or equitable remedies available with respect to infringements of Patents.

10.8 <u>Personnel Obligations</u>. Prior to beginning work under this Agreement, each employee and contractor of Gilead or Galapagos or of either Party's respective Affiliates shall be bound by non-disclosure and invention assignment obligations which are consistent with the obligations of Gilead or Galapagos, as applicable, in this ARTICLE X and ARTICLE XIII, to the extent permitted by Applicable Law, including: (a) promptly reporting any invention, discovery, process or other intellectual property right; (b) assigning to Gilead or Galapagos, as appropriate, all of his or her right, title and interest in and to any invention, discovery, process or other intellectual property right; (c) in the case of employees working in the United States, taking actions reasonably necessary to secure Patent protection; (d) performing all acts and signing, executing, acknowledging and delivering any and all documents required for effecting the obligations and purposes of this Agreement; and (e) abiding by the obligations of confidentiality and non-use set forth in ARTICLE XIII. It is understood and agreed that such non-disclosure and invention assignment agreement need not reference or be specific to this Agreement.

10.9 Optioned Product Trademarks.

(a) <u>Global Trademarks</u>. For each Optioned Program, the JCC shall (a) oversee establishment of the Optioned Product Trademarks to be used on a worldwide basis with respect to the Exploitation of the applicable Optioned Products ("**Global Optioned Product Trademarks**"), (b) determine the allocation between the Parties of the ownership of and the control of obtaining, maintaining, enforcing and defending the Global Optioned Product Trademarks and (c) establish the strategy for use of the Global Optioned Product Trademarks in connection with the Commercialization of Optioned Products. Any costs and expense of either Party in connection therewith shall be deemed to be Shared Commercialization Costs and allocated in accordance with Section 9.10. If a Party wishes to use an Optioned Product Trademark for its Respective Territory, it must so notify the JCC and, if the other Party also wishes to use such Optioned Product Trademark, it shall be a Global Optioned Product Trademark hereunder (subject to any legal restrictions in the other Party's Respective Territory).

(b) <u>Regional Trademarks</u>. For each Optioned Program, if either Party desires to use any Optioned Product Trademark that is not a Global Optioned Product Trademark in the Commercialization of an applicable Optioned Product in its Respective Territory ("**Regional Optioned Product Trademarks**"), then such Party shall seek the consent of the JCC to use such Regional Optioned Product Trademark, which consent shall not be unreasonably withheld, conditioned or delayed. If the JCC approves the use of such Regional Optioned Product Trademark for such Optioned Product and Respective Territory, then such Party shall (i) have the right to use such Regional Optioned Product Trademark with respect to the Exploitation of such Optioned Product in its Respective Territory; (ii) control all matters with respect to obtaining, maintaining, enforcing and defending such Regional Optioned Product Trademark; and (iii) be solely responsible for any costs and expense of either Party in connection therewith.

(c) Trademark License.

(i) <u>License Grant</u>. With respect to each Optioned Program, subject to Section 10.9(c)(ii), the Party that owns any Global Optioned Product Trademarks in the other Party's Respective Territory for such Optioned Program (the "**Trademark Owner Party**"), on behalf of itself and its Affiliates, hereby grants to the other Party (the "**Trademark Licensee Party**"), effective as of each Option Exercise Closing for such Optioned Program, a royalty-free, exclusive, sublicensable (solely in accordance with Section 8.6(a)) license under the Global Optioned Product Trademarks to (A) if Gilead is the Trademark Owner Party and Galapagos is the Trademark Licensee Party, Commercialize any applicable Optioned Products in such Optioned Program in the countries in the Galapagos Territory (as it exists

immediately after the applicable Option Exercise Closing) and (B) if Galapagos is the Trademark Owner Party and Gilead is the Trademark Licensee Party, Commercialize any applicable Optioned Products in such Optioned Program in the countries in the Gilead Territory (as it exists immediately after the applicable Option Exercise Closing). At the request of the Trademark Licensee Party for any Global Optioned Product Trademarks, the Parties shall promptly enter into a separate trademark license agreement consistent with the licenses described in this Section 10.9(c)(i).

(ii) <u>Certain Conditions</u>. With respect to each Global Optioned Product Trademark, the Trademark Licensee Party shall not, and shall not permit its Affiliates to, (A) use in its or their respective businesses, any Trademark that is confusingly similar to, misleading or deceptive with respect to or that dilutes any (or any part) of such Global Optioned Product Trademark, or (B) do any act that endangers, destroys, or similarly affects, in any material respect, the value of the goodwill pertaining to such Global Optioned Product Trademark. With respect to each Global Optioned Product Trademark, each Party agrees, and shall cause its Affiliates, to (1) conform to the customary industry standards for the protection of Trademarks for pharmaceutical products and any guidelines of the applicable Trademark Owner Party with respect to manner of use (to the extent provided in writing by such Trademark Owner Party or included in the Global Commercialization Plan and Budget), and (2) to maintain the quality standards of the Trademark Owner Party with respect to the goods sold and services provided in connection with such Global Optioned Product Trademark. With respect to each Global Optioned Product Trademark Licensee Party shall not, and shall cause its Affiliates not to, (I) do any act that endangers, destroys, or similarly affects, in any material respect, the value of the goodwill pertaining to such Global Optioned Product Trademark or (II) attack, dispute, or contest the validity of or ownership of such Global Optioned Product Trademark anywhere in the Territory or any registrations issued or issuing with respect thereto.

(d) <u>Other Conditions</u>. Neither Party shall, without the other Party's prior written consent, use any Trademarks or house marks of the other Party (including the other Party's corporate name), or marks confusingly similar thereto, in connection with such Party's marketing or promotion of Optioned Products under this Agreement, except as may be expressly authorized in the Global Commercialization Plan and Budget (which, for clarity, shall include authorization related to the Global Optioned Product Trademark).

10.10 <u>Confirmatory Licenses</u>. Each Party shall, if so requested by the other Party, promptly enter into confirmatory license agreements, in a form consistent with the terms of this Agreement and reasonably acceptable to the Parties, for purposes of recording the licenses granted under this Agreement with any applicable Patent offices or other Governmental Authorities. Each Party shall bear its own filing costs and expenses and any costs and expenses of outside counsel or experts required with respect to such recordations.

10.11 <u>IP for Combination Product Activities</u>. The rights and obligations of both Parties with respect to any Patents that disclose the use of any active pharmaceutical ingredient that is owned or controlled by Gilead or any of its Affiliates with any Galapagos Molecule, Galapagos Product, Optioned Molecule or Optioned Product (including, for clarity, any Gilead Combination Product) shall be governed by the applicable Combination Product Activities Agreement.

10.12 <u>IP for Excluded Programs</u>. Gilead hereby covenants to Galapagos, on behalf of itself and its Affiliates, that, except as allowed under the Filgotinib Agreement or another written agreement between the Parties, it and its Affiliates shall not disclose in any Patent [...***...].

ARTICLE XI

REPRESENTATIONS AND WARRANTIES AND COVENANTS

11.1 <u>Mutual Representations, Warranties and Covenants</u>. Each Party hereby represents and warrants to the other Party as of the Execution Date and the Effective Date, and covenants (as applicable) as follows:

(a) <u>Corporate Existence and Power</u>. It is a company or corporation duly organized, validly existing, and in good standing under the laws of the jurisdiction in which it is incorporated, and has full corporate power and authority and the legal right to own and operate its property and assets and to carry on its business as it is now being conducted and as contemplated in this Agreement, including the right to grant the licenses granted by it hereunder.

(b) <u>Authority and Binding Agreement</u>. Except as set forth in this Section 11.1(b), (i) it has the corporate power and authority and the legal right to enter into this Agreement and perform its obligations hereunder; (ii) it has taken all necessary corporate action on its part required to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder; and (iii) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, valid, and binding obligation of such Party that is enforceable against it in accordance with its terms, except to the extent that enforcement of the rights and remedies created hereby is subject to (A) bankruptcy, insolvency, reorganization, moratorium and other similar laws of general application affecting the rights and remedies of creditors or (B) laws governing specific performance, injunctive relief and other equitable remedies.

(c) <u>No Conflict</u>. Neither it nor any of its Affiliates is a party to and will not enter into any agreement that would prevent it from granting the rights or exclusivity granted or intended to be granted to the other Party under this Agreement or performing its obligations under this Agreement except, in the case of Galapagos, as set forth in the Galapagos Schedule of Exceptions.

(d) <u>No Debarment</u>. Neither it nor any of its Affiliates is debarred, has been convicted, or is subject to debarment or conviction pursuant to Section 306 of the FD&C Act. Such Party has not used in connection with any activity in its business, any employee, agent or independent contractor who has been debarred by any Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority or has been convicted pursuant to Section 306 of the FD&C Act. In the course of conducting its activities under this Agreement, each Party shall not, and shall cause its Affiliates not to, use any employee, agent or independent contractor who has been debarred by any Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority or has been convicted pursuant to Section 306 of the FD&C Act. Each Party shall promptly notify the other Party of any debarment or debarment proceeding that could have an impact on the use of the results of any Clinical Trials relating to any Galapagos Program or any Pre-Program Activities, including Nonclinical Studies.

(e) <u>Anti-Corruption</u>. Each Party, its Affiliates and their respective directors, officers, employees, agents or other persons or entities acting on its behalf (all the foregoing collectively "**Representatives**") have conducted and will conduct their respective activities under this Agreement (which, for clarity, with respect to Galapagos includes all Pre-Program Activities and all activities relating to any Galapagos Program) in compliance with the US Foreign Corrupt Practices Act of 1977, as amended (such act, including the rules and regulations thereunder, the "FCPA"), the UK Bribery Act 2010 ("**Bribery Act**") and any other applicable anti-corruption laws, rules or regulations (collectively, "**Anti-Corruption Laws**"). Without limiting the foregoing, each Party shall ensure that neither it, nor any of its Representatives, shall offer, pay, promise, solicit or receive, directly or indirectly, any remuneration, benefit

or advantage to or from any physician or other health care practitioner, governmental or political official, political party, candidate for public office, hospital, medical insurance company or similar provider organization, customer or other person in order to induce or encourage approval, referrals, purchase, or reimbursement or to obtain any other improper business advantage in violation of any Anti-Corruption Laws. Without limiting the generality of the foregoing, Gilead represents and warrants that it has and will have, and Galapagos represents and warrants that it will have, a compliance program as set forth in Section 5.2(e).

11.2 <u>Representations and Warranties of Galapagos</u>. Galapagos hereby represents and warrants and, with respect to Sections 11.2(c) and 11.2(g), covenants to Gilead as of the Execution Date except as set forth in **Schedule 11.2** (the "**Galapagos Schedule of Exceptions**"), and as of the Effective Date except as set forth in the Galapagos Schedule of Exceptions (as it may be amended prior to the Effective Date), as follows:

(a) <u>Title; Encumbrances</u>. Galapagos or one of its Affiliates solely owns or exclusively licenses and Controls the Existing Galapagos Patents relating to any Galapagos Program, *provided*, *however*, that the foregoing shall not constitute a representation or warranty of non-infringement of a Third Party's intellectual property rights. Galapagos or one of its Affiliates has the right to grant the licenses to Gilead as purported to be granted pursuant to this Agreement. Neither Galapagos nor any of its Affiliates has entered into any agreement (other than agreements with subcontractors) granting any right, interest or claim in or to, any Existing Galapagos Patents or Galapagos Know-How to any Third Party that would conflict with the rights and licenses to Gilead as purported to be granted pursuant to this Agreement.

(b) <u>Assignment of Rights</u>. Each Person who has or has had any rights in or to any Existing Galapagos Patents or any material Galapagos Know-How has assigned and has executed an agreement assigning its entire right, title and interest in and to such Existing Galapagos Patents and Galapagos Know-How to Galapagos or one of its Affiliates. To Galapagos' Knowledge, no current officer, employee, agent, or consultant of Galapagos or any of its Affiliates is in violation of any term of any assignment or other agreement regarding the protection of Patents or other intellectual property or proprietary information of Galapagos or such Affiliate or of any employment contract relating to the relationship of any such Person with Galapagos.

(c) <u>Patents</u>. All Existing Galapagos Patents are identified in Section 11.2(c) of the Galapagos Schedule of Exceptions. All Existing Galapagos Patents are subsisting and are being diligently prosecuted in the patent offices indicated in Section 11.2(c) of the Galapagos Schedule of Exceptions in accordance with Applicable Law and to Galapagos' Knowledge are not invalid or unenforceable in whole or in part. All applicable fees with respect thereto have been timely paid or will be timely paid (taking account of any permitted extensions). The Existing Galapagos Patents in Section 11.2(c) of the Galapagos Schedule of Exceptions represent all Patents within Galapagos' or its Affiliates' ownership or control (by license or otherwise) that Galapagos reasonably believes include claims covering the making, using or composition of matter of the Galapagos Molecules or Galapagos Products, or the Exploitation of any such Galapagos Molecule or Galapagos Product. To the extent required, Galapagos or one of its Affiliates has properly recorded in the relevant U.S. and foreign patent offices the assignments, or other necessary documents, supporting its legal title to the Existing Galapagos Patents in Section 11.2(c) of the Galapagos Schedule of Exceptions. Galapagos and its Affiliates have presented, or will present prior to the pertinent patent office, in connection with the prosecution of which they and the inventors are aware to the relevant patent examiner at the pertinent patent office, in connection with the prosecution of the pending patent applications included in the Existing Galapagos Patents.

(d) <u>No Infringement</u>. To Galapagos' Knowledge, no Person is infringing or threatening to infringe or misappropriating or threatening to misappropriate any Existing Galapagos Patents or any Galapagos Know-How or Regulatory Documentation. To Galapagos' Knowledge, there are no activities by Third Parties that would constitute infringement or misappropriation of the Galapagos IP (in the case of pending claims, evaluating them as if issued).

(e) <u>No Conflicts</u>. The execution and delivery of this Agreement by Galapagos does not, and the consummation of the transactions contemplated by this Agreement will not, (i) except for the rights granted to Gilead in this Agreement, result in the creation of any encumbrance on any of the material properties or assets relating to any Galapagos Program, or (ii) conflict with, or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, or require any consent, approval or waiver from any Person pursuant to (A) any provision of the organizational or governing documents of Galapagos, in each case as amended to date, or (B) any material agreement applicable to Galapagos' or any of its Affiliates' material properties or assets relating to the Autotaxin Program or the ADAMTS-5 Program or any other Galapagos Program.

(f) In-Licenses and Restrictions on Business Activities. None of the Existing Galapagos Patents licensed hereunder by Galapagos to Gilead are owned or controlled in whole or in part by any Third Party. There is no agreement, judgment, injunction, order or decree of a Governmental Authority binding upon Galapagos or any of its Affiliates with respect to any Galapagos Program that has or would reasonably be expected to have, whether before or after the Effective Date, the effect of prohibiting or impairing any current or presently proposed business practice of Galapagos or any of its Affiliates or the conduct of business by Galapagos or any of its Affiliates as currently conducted or as presently proposed to be conducted by Galapagos or any of its Affiliates for such Galapagos Program.

(g) <u>Copyrightable IP</u>. To Galapagos' Knowledge, all works of authorship and all other materials subject to copyright protection included in Information owned or otherwise controlled by Galapagos or any of its Affiliates that is reasonably necessary or useful to Exploit any Galapagos Molecule or Galapagos Product are original and were either created by employees of Galapagos or its Affiliates within the scope of their employment or are otherwise works made for hire, or right, title and interest in and to such materials have been legally assigned or licensed to Galapagos or such Affiliate to the extent necessary to provide Gilead with the rights granted to it hereunder, and all rights in all inventions and discoveries made, developed, or conceived by any employee or independent contractor of Galapagos or any of its Affiliates during the course of their employment (or other retention) by Galapagos or such Affiliate, and relating to or included in the Galapagos Know-How or that are the subject of one or more Existing Galapagos Patents, have been or will be assigned in writing to Galapagos or such Affiliate to the extent necessary to provide Gilead with the rights granted to it hereunder.

(h) <u>Transfer of Rights</u>. Galapagos or one of its Affiliates has obtained the right (including under any Patents and other intellectual property rights) to use all Information and all other materials (including any formulations and manufacturing processes and procedures) developed or delivered by any Third Party under any agreements between Galapagos or one of its Affiliates and any such Third Party with respect to any Galapagos Program to the extent necessary to provide Gilead with the rights granted to it hereunder, and Galapagos or one of its Affiliates has the rights under each such agreement to transfer such Information or other materials to Gilead and its designees and to grant Gilead the right to use such Information or other materials in the Development or Commercialization of the Galapagos Molecules or Galapagos Products in any Galapagos Program as required to enable Gilead to Exploit the Galapagos Molecules and the Galapagos Products in the Gilead Territory.

(i) <u>Confidentiality of Know-How</u>. With respect to those portions of the Galapagos Know-How the confidentiality of which is material to the Exploitation of any Significant Pre-Program Activities, Galapagos Molecule or Galapagos Product, such portions of the Galapagos Know-How have been kept confidential or have been disclosed to Third Parties only under terms of confidentiality or if published or otherwise publicly disclosed, were published or publicly disclosed in a manner that would not reasonably be expected to adversely impact the patentability of such Galapagos Know-How. To Galapagos' and its Affiliates' Knowledge, no breach of such confidentiality has been committed by any Third Party.

(j) No Proceedings.

(i) No claim or litigation has been brought or threatened in writing by any Person against Galapagos or any of its Affiliates alleging, and Galapagos has no Knowledge of any reasonable basis for any such claim or allegation, whether or not asserted, that (A) any Existing Galapagos Patents are invalid or unenforceable, or (B) that the use or practice of any Existing Galapagos Patents or any Regulatory Materials or Galapagos Know-How, or the disclosing, copying, making, assigning or licensing of any Existing Galapagos Patents or any such Regulatory Materials or Galapagos Know-How, or the Development, Commercialization or other Exploitation of the Galapagos Molecules or Galapagos Products included in any Galapagos Program as contemplated herein or Significant Pre-Program Activities, does or will violate, infringe, misappropriate or otherwise conflict or interfere with, any Patent or other intellectual property or proprietary right of any Third Party.

(ii) There is no private or Governmental Authority action, suit, proceeding, claim, mediation, arbitration or investigation pending before any Governmental Authority, or, to Galapagos' Knowledge, threatened against Galapagos or any of its Affiliates or any of its or their respective assets or properties relating to any of its or their Pre-Program Activities or Galapagos Programs or, to Galapagos' Knowledge, any of their respective directors, managers, officers or employees (in their capacities as such or relating to their employment, services or relationship with Galapagos or any of its Affiliates), nor, to Galapagos' Knowledge, is there any reasonable basis for any such action, suit, proceeding, claim, mediation, arbitration or investigation. There is no judgment, decree, injunction or order against Galapagos or any of its Affiliates, or, to Galapagos' Knowledge, any of its or their respective assets or properties, relating to any Pre-Program Activities or Galapagos Programs or, to Galapagos' Knowledge, any of their respective directors, managers, officers or employees (in their capacities as such or relating to their employment, services or relationship with Galapagos or any of its Affiliates). Neither Galapagos nor any of its Affiliates has any action, suit, proceeding, claim, mediation, arbitration pending against any other Person relating to any of its Pre-Program Activities or Galapagos Programs.

(k) <u>No Misappropriation</u>. The conception and reduction to practice of any inventions and the use or development of any other Information within the Galapagos IP (i) owned by Galapagos have not constituted or involved the misappropriation of trade secrets or other rights or property of any Third Party and (ii) in-licensed by Galapagos, to Galapagos' Knowledge, have not constituted or involved the misappropriation of trade secrets or other rights or property of any Third Party; *provided, however*, that the foregoing clauses (i) and (ii) shall not constitute a representation or warranty of non-infringement of a Third Party's intellectual property rights.

(1) <u>Full Disclosure</u>. Galapagos has provided or made available to Gilead all material adverse information with respect to the safety and efficacy of the Galapagos Molecules or Galapagos Products relating to the ADAMTS-5 Program and the Autotaxin Program. Galapagos has provided a true, complete and correct list of each Galapagos Molecule and Galapagos Product that has received IND approval from the FDA (i) for the ADAMTS-5 Program and (ii) for the Autotaxin Program.

(m) Compliance and Qualifications.

(i) To Galapagos' Knowledge, Galapagos and its Affiliates have conducted, and their respective contractors, licensees and consultants, have conducted in all material respects all Development under the Galapagos Programs in accordance with all Applicable Laws, including current Good Clinical Practices, Good Laboratory Practices, Good Manufacturing Practices and the Declaration of Helsinki.

(ii) Galapagos and its Affiliates and licensees have generated, prepared, maintained and retained all Regulatory Documentation that is required to be maintained or retained with respect to the Autotaxin Program and the ADAMTS-5 Program pursuant to and in accordance with Applicable Law, and all such Regulatory Documentation is true, complete and correct and what it purports to be.

(iii) In connection with the collection, storage, transfer (including any transfer across national borders) or use of any information relating to identifiable natural persons (collectively "**Personal Information**") by or on behalf of Galapagos or any of its Affiliates relating to any Pre-Program Activities or Galapagos Program, Galapagos and its Affiliates are in compliance in all material respects with all Applicable Laws in all relevant jurisdictions, internal privacy policies and the requirements of any contract or codes of conduct to which Galapagos or any of its Affiliates is a party. Galapagos and its Affiliates have commercially reasonable technical and organizational measures in place to ensure the security of all Personal Information it controls or processes. Galapagos and its Affiliates are in compliance in all material respects with all Applicable Laws relating to breaches of security affecting Personal Information and associated notification obligations. Neither Galapagos nor any of its Affiliates has received a complaint regarding its collection, storage, transfer or use of Personal Information. Galapagos agrees to execute or cause to be executed any additional clauses or agreements necessary to comply with data protection laws prior to the transfer of Personal Information.

(n) <u>No Misrepresentation</u>. To Galapagos' Knowledge, neither Galapagos nor any of its Affiliates or licensees, nor any of its or their respective officers, employees or agents, has made an untrue statement of material fact or fraudulent statement to the FDA or any other Regulatory Authority with respect to the Development of the Galapagos Molecules or Galapagos Products, failed to disclose a material fact required to be disclosed to the FDA or any other Regulatory Authority with respect to the Development of the Galapagos Molecules or Galapagos Products, or committed an act, made a statement, or failed to make a statement with respect to the Development of the Galapagos Molecules or Galapagos Products that could reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities," set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto or any analogous Applicable Laws in the Territory.

(o) <u>Royalties and other Payments</u>. There are no milestone payments, profit share obligations, royalty payments or other amounts in each case that are based on the Manufacture, Development or Commercialization of the Galapagos Molecules or Galapagos Products required to be paid to a Third Party as a result of the Manufacture, Development or Commercialization of the Galapagos Molecules or Galapagos Products under any agreement to which Galapagos or any of its Affiliates is a party.

(p) <u>No Government Funding</u>. To Galapagos' Knowledge, the inventions claimed in the Existing Galapagos Patents (i) were not conceived, discovered, developed, generated or otherwise made in connection with any research activities funded, in whole or in part, by the federal government of the United States or any agency thereof, (ii) are not a "subject invention" as that term is described in 35 U.S.C. Section 201(f), and (iii) are not otherwise subject to the provisions of the Bayh-Dole Act.

(q) <u>No Governmental Consents</u>. Neither the execution, delivery and performance by Galapagos of this Agreement, the Subscription Agreement and the Filgotinib Amendment, nor the consummation by Galapagos of the transactions contemplated hereby or thereby, will, other than with respect to the HSR Act or any other Applicable Antitrust Law, the Belgian Act of May 2, 2007 on the disclosure of major shareholdings in issuers the shares of which are admitted to trading on a regulated market, the Belgian Royal Decree of February 2, 2008 on the disclosure of major shareholdings, EU Regulation No. 596/2014 of 16 April 2014 on market abuse (the "**Market Abuse Regulation**"), the consents to be obtained and filings to be made in relation to or for the purpose of the admission to trading of Galapagos' shares on Euronext Brussels and Euronext Amsterdam as contemplated by the Subscription Agreement, and the other filings contemplated or permitted by this Agreement (including Section 13.3(c)) and the Subscription Agreement, require Galapagos or any of its Affiliates to (i) obtain any consent or authorization of, or (ii) give any notice to, or make any filing or registration with, any Governmental Authority or other Person.

(r) <u>Material Adverse Effect</u>. Since the date of the 2018 Annual Report of Galapagos, no Material Adverse Event has occurred with respect to Galapagos or any of its Affiliates, taken as a whole, except as previously publicly disclosed. "**Material Adverse Effect**" means any change, event, violation, circumstance or effect that, individually or collectively, and regardless of whether or not such change, event, violation, circumstance or effect constitutes a breach of the representations or warranties made by Galapagos hereunder, is, or is reasonably likely to, (i) have a material adverse effect on the condition (financial or otherwise), properties, assets (including intangible assets), liabilities, business, operations or results of operations of Galapagos and its Affiliates, taken as a whole or (ii) materially impede or delay Galapagos' ability to consummate the transactions contemplated by this Agreement in accordance with its terms and Applicable Law; *provided that* none of the following, either alone or in combination, will constitute, or be considered in determining whether there has been, a Material Adverse Effect: any change, event, circumstance, effect or other matter resulting from or related to (A) any outbreak or escalation of war or major hostilities or any act of terrorism, (B) changes in financial markets, general economic conditions or political conditions, (E) any action required to be taken or prohibited by this Agreement or action taken or failed to be taken at the request of, or consented to by, Gilead, or (F) the public announcement of the transactions contemplated by this Agreement (except, with respect to clauses (A), (B), (C) and (D), to the extent that any such change, event, violation, circumstance or effect, alone or in combination, disproportionately has a greater adverse impact on Galapagos and its Affiliates, taken as a whole, as compared to the other companies operating in the same industries and markets in which Galapagos and its Affiliates, taken as a whole, as compared to the oth

(s) [...***...].

(t) [...***...].

11.3 <u>Representations and Warranties of Gilead</u>. Gilead hereby represents and warrants to Galapagos, as of the Execution Date and the Effective Date, as follows:

(a) <u>No Conflicts</u>. The execution and delivery of this Agreement by Gilead does not, and the consummation of the transactions contemplated by this Agreement will not, conflict with, or result in any violation of or default under (with or without notice, lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, or require any consent, approval or waiver from any Person pursuant to any provision of the organizational or governing documents of Gilead, in each case as amended to date.

(b) <u>No Governmental Consents</u>. Neither the execution, delivery and performance by Gilead of this Agreement, the Subscription Agreement and the Filgotinib Amendment, nor the consummation by Gilead of the transactions contemplated hereby or thereby (for clarity, excluding the possible option exercise transactions), will, other than filings required by the HSR Act or any other applicable Antitrust Law, require Gilead to (i) obtain any consent or authorization of, or (ii) give any notice to, or make any filing or registration with, any Governmental Authority or other Person.

(c) <u>Tax Matters</u>. Gilead is, for United States Tax purposes, resident in the United States and subject to the United States ordinary corporate income Tax regime on its worldwide income (subject to an allocation of taxing rights to other jurisdictions based on Double Tax Treaties). Hence, Gilead is, for United States Tax purposes, not considered to have only a branch in the United States. Gilead is not considered as Tax resident in any other jurisdiction.

(d) [...***...].

11.4 <u>No Other Representations or Warranties</u>. EXCEPT AS EXPRESSLY STATED IN THIS ARTICLE XI OR IN ANY ANCILLARY AGREEMENT, NO REPRESENTATIONS OR WARRANTIES WHATSOEVER, WHETHER EXPRESS OR IMPLIED, INCLUDING WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT, OR NON-MISAPPROPRIATION OF THIRD PARTY INTELLECTUAL PROPERTY RIGHTS, IS MADE OR GIVEN BY OR ON BEHALF OF A PARTY. EXCEPT AS EXPRESSLY STATED IN THIS AGREEMENT OR IN ANY ANCILLARY AGREEMENT, ALL REPRESENTATIONS AND WARRANTIES, WHETHER ARISING BY OPERATION OF LAW OR OTHERWISE, ARE HEREBY EXPRESSLY EXCLUDED.

11.5 Post-Effective Date Covenants.

(a) <u>Use of Proceeds</u>. Without limiting Galapagos' obligations under Section 2.1, over the course of the Term, Galapagos shall in good faith dedicate and apply the Upfront Consideration, plus all net proceeds paid to Galapagos by Gilead pursuant to the Subscription Agreement to the Galapagos R&D Activities, Pre-Program Activities, research and development activities across all current and future programs, Molecules and products and other similar activities, including corporate development activities, intended to, directly or indirectly, support the foregoing activities. The foregoing funds shall not be used at any time during the Term to acquire, license or otherwise obtain rights to any assets, business or company if such acquisition, license or rights to assets do not include [...***...]. For purposes of this Section 11.5(a), [...***...].

(b) Conduct of Business.

(i) Galapagos shall use good faith efforts to conduct the Galapagos R&D Activities during the Term. Galapagos shall, and shall cause each of its Affiliates to, except as otherwise contemplated by the terms of this Agreement, (A) conduct its business consistent with its mission to develop first-in-class medicines based on the discovery of novel targets, as such mission may be updated by Galapagos or its Affiliates, as applicable, from time to time, (B) preserve intact its material assets to the extent necessary to preserve the rights granted to Gilead hereunder, and (C) maintain business relationships with licensors, licensees, Governmental Authorities and others having material business dealings with Galapagos to the extent necessary to preserve the rights granted to Gilead hereunder.

(ii) The Upfront Consideration, plus all net proceeds paid to Galapagos by Gilead pursuant to the Subscription Agreement, shall not be used to pay dividends or make any other distributions on capital stock of the Company.

(c) Conduct of Program-Related Activities.

(i) Each Party covenants that with respect to all intellectual property that it licenses to the other Party under this Agreement that is, may be or becomes subject to the Bayh-Dole Act, such licensing Party shall, and shall cause its Affiliates and the relevant research partners to, continue to comply with the applicable provisions of the Bayh-Dole Act, in a manner that protects and preserves such licensing Party's right, title and interest in the subject intellectual property to the maximum extent permitted by Applicable Law.

(ii) Each Party shall, and shall cause its Affiliates and its and their respective contractors, licensees and consultants to, conduct all Pre-Program Activities and Development of the Galapagos Molecules, Galapagos Products and all other activities undertaken pursuant to this Agreement in accordance with Applicable Law. Without limitation of the foregoing sentence, each Party shall, and shall cause its Affiliates and its and their respective licensees, to employ Persons with appropriate education, knowledge and experience to conduct and to oversee the conduct of Nonclinical Studies and Clinical Trials with respect to the Galapagos Molecules or Galapagos Products.

(iii) With respect to all Pre-Program Activities, each Galapagos Program until the end of the Galapagos Program Period for such Galapagos Program and each Optioned Program until the end of the Term for such Optioned Program, neither Galapagos nor its Affiliates shall enter into any agreement with any Third Party, whether written or oral, with respect to, or otherwise assign, transfer, license, convey, dispose of or encumber (other than as required by the terms of the Existing Galapagos Third Party Agreements) its right, title or interest in or to, the Galapagos IP relating to such Pre-Program Activities, Galapagos Program or Optioned Program, as applicable, that creates a conflict with the rights granted by (or purported to be granted by) Galapagos to Gilead under this Agreement or that prevents Galapagos from performing its obligations under this Agreement or that prevents Gilead from exercising its rights hereunder.

(iv) Except in a manner that would not create a material conflict with the rights granted or purported to be granted by Galapagos to Gilead under this Agreement or as expressly permitted under this Agreement, with respect to (A) all Pre-Program Activities, (B) each Galapagos Program until the end of the Galapagos Program Period for such Galapagos Program, and (C) each Optioned Program until the end of the Term for such Optioned Program, neither Galapagos nor its Affiliates shall, sell, out-license or otherwise dispose of any material assets or other material rights relating to any portion of such Pre-Program Activities, such Galapagos Program or such Optioned Program, as applicable (other than non-exclusive, non-commercial licenses granted in the ordinary course, including through clinical trial agreements, research agreements with academic institutions and non-profit organizations, service agreements, material transfer agreements and other similar agreements); grant any security interest or otherwise encumber any of the foregoing (including Galapagos IP and all other intellectual property relating to such Galapagos Program or Optioned Program, as applicable) relating to any portion of such Pre-Program Activities, such Galapagos Program or such Optioned Program, as applicable, other than pursuant to the Security Agreement referenced in Section 16.5(f).

(d) [<u>...***...</u>].

(i) No later than [...***...] Business Days after the Completion Date for the Triggering Clinical Trial for [...***...], Galapagos shall provide notice to [...***...] that it is considering granting a license for commercial rights to the Optioned Products in [...***...] to Gilead for the Gilead Territory with respect to [...***...].

(ii) Galapagos shall promptly, and in no event later than the [...***...] notify Gilead of any [...***...] necessary or useful to Exploit any Galapagos Molecules from [...***...] that is disclosed by [...***...] to Galapagos or of which Galapagos or any of its Affiliates otherwise becomes aware. Galapagos shall use commercially reasonable efforts to obtain consent from [...***...] to grant Gilead the license set forth in Section 8.3 with respect to the [...***...] necessary or useful to Exploit any Galapagos Molecules from [...***...].

(e) [...***...].

11.6 Compliance Violations. Without limiting the indemnification, termination, dispute resolution and other rights of either Party hereunder, in the event that either Party has a good faith belief that the other Party has violated or is violating Applicable Law in connection with any Pre-Program Activities, a Galapagos Program or an Optioned Program, or if it believes in good faith that the other Party has breached any of its compliance-related representations and covenants in this Agreement or in any Ancillary Agreement and desires to have a discussion regarding same, then upon such Party's request, the Parties shall promptly convene a meeting of appropriate representatives from each Party within [...***...] Business Days after such request, which may, at the request of the requesting Party, be required to include either or both of each Party's general counsel or chief compliance officer. At such meeting, the Parties' representatives shall agree in writing upon a plan to rectify the situation and the notified Party shall take such action as required under the plan. If the Parties' representatives are not able to agree upon a plan within [...***...] Business Days after such meeting begins, then the requesting Party may designate an independent Third Party with experience in compliance counseling in the area of concern to conduct an audit of the conduct about which the requesting Party is concerned. Such Third Party shall be chosen within [...***...] Business Days after the end of the preceding [...***...] Business Day period, shall complete its work as soon as practicable as determined by such Third Party and shall provide its report to both Parties within [...***...] Business Days of completing its audit. The Parties shall then, to the extent necessary, negotiate an implementation plan within [...***...] Business Days after the receipt of such Third Party's report. If the Parties are unable to agree upon such implementation plan, the Third Party shall select one of the Parties' proposed plans as the implementation plan. The Party that is the subject of the implementation plan shall use its best efforts to conduct the activities set forth in the implementation plan as promptly as practicable. If the Party that is subject to the implementation plan is unable to implement the plan within [...***...] days, the other Party shall be entitled to require the implementing Party to have all matters subject to remediation submitted to the other Party for approval until such implementation is appropriately completed, and the implementing Party shall reasonably compensate the approving Party for such efforts. Unless and until the implementation plan has been established and the requesting Party is reasonably satisfied that the implementation plan has been reasonably implemented, the requesting Party may suspend the performance of any or all of its obligations hereunder that are related to such violation, other than its payment obligations.

ARTICLE XII

INDEMNIFICATION

12.1 <u>Indemnification by Galapagos</u>. Galapagos shall defend, indemnify and hold Gilead, its Affiliates and its and their respective officers, directors, employees and agents (the "**Gilead Indemnitees**") harmless from and against any and all damages or other amounts payable to a Third Party claimant, as well as any reasonably incurred attorneys' fees and costs and expenses of litigation (collectively, "**Losses**") incurred by such Gilead Indemnitees, to the extent resulting from claims, suits, proceedings or causes of action brought by or on behalf of such Third Party (collectively, "**Third Party Claims**") that arise from or are based on:

(a) the breach of any of Galapagos' representations, warranties or obligations under this Agreement;

(b) the willful misconduct or gross negligence of any of the Galapagos Indemnitees in connection with this Agreement;

(c) the violation of Applicable Law by any of the Galapagos Indemnitees in connection with this Agreement;

(d) except as otherwise provided in Section 10.6, the Exploitation of any Optioned Molecule or Optioned Product by or on behalf of Galapagos or any of its Affiliates, excluding any Shared Development Losses (which shall be allocated as set forth in Section 12.4);

(e) except as otherwise provided in Section 10.6 or in a separate written agreement, (i) the conduct of the Pre-Program Activities or Exploitation of a molecule or product that is the subject of any Pre-Program Activities or any Galapagos Molecule or Galapagos Product or (ii) the Exploitation of any molecule or product under a Galapagos Program that becomes an Excluded Program after the date on which such Galapagos Program became an Excluded Program, in each case ((i) and (ii)), by or on behalf of Galapagos or any of its Affiliates, or any of its or their licensees, Sublicensees or Distributors;

(f) the Exploitation, in accordance with the provisions of ARTICLE XIV, after the effective date of such termination, of any Galapagos Molecule or Galapagos Product with respect to which this Agreement has been terminated by or on behalf of Galapagos or any of its Affiliates, or any of its or their licensees, Sublicensees or Distributors;

(g) the (i) exercise of any rights under any license or right of reference by or on behalf of Galapagos or any of its Affiliates, or any of its or their licensees, Sublicensees or Distributors or (ii) use of any Regulatory Materials, Regulatory Approvals, Trademarks or Information by or on behalf of Galapagos or any of its Affiliates, in each case ((i) and (ii)), granted, transferred or made available to Galapagos or any of its Affiliates by or on behalf of Gilead or any of its Affiliates in accordance with the provisions of ARTICLE XIV following or in connection with termination of this Agreement with respect to any Galapagos Product or Optioned Product; or

(h) except as otherwise provided in a separate written agreement, any Independent Activities conducted by or on behalf of Galapagos or any of its Affiliates;

except, in the case of clauses (a) through (h), for those Losses for which Gilead, in whole or in part, has an obligation to indemnify any Galapagos Indemnitee pursuant to Section 12.2 or under any Ancillary Agreement, as to which Losses each Party shall indemnify the other to the extent of their respective responsibility for the Losses. Galapagos acknowledges and agrees that the terms of any separate written agreement contemplated to be entered into in accordance with this Agreement shall include indemnity obligations generally consistent with the allocation of liability between the Parties in this Agreement; *provided, however, that* any Supply Agreement shall include customary limitations on the liability of the supplying Party.

12.2 <u>Indemnification by</u> Gilead. Gilead shall defend, indemnify and hold Galapagos, its Affiliates and its and their respective officers, directors, employees and agents (the "**Galapagos Indemnitees**") harmless from and against any and all Losses incurred by such Galapagos Indemnitees, to the extent resulting from Third Party Claims that arise from or are based on:

(a) the breach of any of Gilead's representations, warranties or obligations under this Agreement;

(b) the willful misconduct or gross negligence of any of the Gilead Indemnitees in connection with this Agreement;

(c) the violation of Applicable Law by any of the Gilead Indemnitees in connection with this Agreement;

(d) except as otherwise provided in Section 10.6, the Exploitation of any Optioned Molecule or Optioned Product by or on behalf of Gilead or any of its Affiliates, excluding any Shared Development Losses (which shall be allocated as set forth in Section 12.4);

(e) except as otherwise provided in Section 10.6 or in a separate written agreement, the conduct of any Gilead Contributions;

(f) except as otherwise provided in a separate written agreement, any Independent Activities conducted by or on behalf of Gilead or any of its Affiliates; or

(g) any activity for which Galapagos or its Affiliates has no liability pursuant to the last sentence of Sections 3.7, the last sentence of Section 4.2(a)(i)(B), or Section 8.2(c)(v) except to the extent resulting from the negligence of any of the Galapagos Indemnitees.

except, in the case of clauses (a) through (g), for those Losses for which Galapagos, in whole or in part, has an obligation to indemnify any Gilead Indemnitee pursuant to Section 12.1 or under any Ancillary Agreement, as to which Losses each Party shall indemnify the other to the extent of their respective responsibility for the Losses. Gilead acknowledges and agrees that the terms of any separate written agreement contemplated to be entered into in accordance with this Agreement shall include indemnity obligations generally consistent with the allocation of liability between the Parties in this Agreement; *provided, however, that* any Supply Agreement shall include customary limitations on the liability of the supplying Party.

12.3 Indemnification Procedures. All indemnification claims with respect to a Gilead Indemnitee or Galapagos Indemnitee shall be made solely by Gilead or Galapagos, as applicable. The Party claiming indemnity under this ARTICLE XII (the "Indemnified Party") shall give written notice to the Party from which indemnity is being sought (the "Indemnifying Party") promptly after learning of the Third Party Claim for which indemnity is being sought. The Indemnifying Party's obligations to defend, indemnify, and hold harmless pursuant to Section 12.1 or 12.2, as applicable, shall be reduced to the extent the Indemnified Party's delay in providing notification pursuant to the previous sentence results in prejudice to the Indemnifying Party. At its option, the Indemnifying Party may assume the defense of any Third Party Claim for which indemnity is being sought by giving written notice to the Indemnified Party within [...***...] days after receipt of the notice of the Third Party Claim. The assumption of defense of a Third Party Claim shall not be construed as an acknowledgment that the Indemnifying Party is liable to indemnify any Indemnified Party in respect of the Third Party Claim, nor shall it constitute a waiver by the Indemnifying Party of any defenses it may assert against the Indemnified Party's claim for indemnification. The Indemnified Party shall provide the Indemnifying Party with reasonable assistance, at the Indemnifying Party's expense, in connection with the defense of any Third Party Claim. The Indemnified Party may participate in and monitor such defense with counsel of its own choosing at its sole expense. subject to the Indemnifying Party's right to assume and conduct the defense of the Third Party Claim with counsel of its choice. If the Indemnifying Party does not assume and conduct the defense of a Third Party Claim as provided above, then (a) the Indemnified Party may defend against such Third Party Claim (and the Indemnified Party need not consult with the Indemnifying Party in connection therewith), and (b) the Indemnified Party reserves any rights it may have under this ARTICLE XII to obtain indemnification from the Indemnifying Party with respect to such Third Party Claim. The Indemnifying Party shall not settle any

Third Party Claim without the prior written consent of the Indemnified Party, not to be unreasonably withheld, conditioned or delayed, unless the settlement involves only the payment of money for which the Indemnifying Party is responsible. The Indemnified Party shall not settle any Third Party Claim for which it has or will exercise its right under this ARTICLE XII to obtain indemnification from the Indemnifying Party without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld, conditioned or delayed.

12.4 Shared Development Losses.

(a) The Parties shall share equally any Losses incurred by any Gilead Indemnitee or Galapagos Indemnitee (collectively, "**Party Indemnitees**") to the extent resulting from Third Party Claims against any such Party Indemnitee that arise from or are based on the performance of any activities under an R&D Plan and Budget with respect to which Research and Development Costs are shared equally by the Parties, to the extent (i) not arising from or based on the events described in clauses (a), (b), (c), (e), (f), (g), and (h) of Section 12.1 or clauses (a), (b), (c), (e), (f), and (g) of Section 12.2 or (ii) not otherwise provided in Section 10.6 (any such Losses, "**Shared Development Losses**" and any such Third Party Claims, "**Shared Development Claims**"). If either Party receives notice of any Shared Development Claim, such Party shall inform the other Party in writing as soon as reasonably practicable, and the Parties shall discuss a strategy for defending such Shared Development Claim. For clarity, this Section 12.4 is not intended, and shall not be construed, to alter either Party's liability under Section 12.1(d) or Section 12.2(d) (as applicable) for any Losses that arise from or are based on any commercially sold Optioned Product.

(b) At its option, Gilead may assume the defense of any Shared Development Claim by giving written notice to Galapagos within [... ***...] days after the notice of the Shared Development Claim. If Gilead does not assume and conduct the defense of a Shared Development Claim as provided in the preceding sentence, then Galapagos may defend against such Shared Development Claim. The non-defending Party shall provide the defending Party with reasonable assistance in connection with the defense of any Shared Development Claim. The non-defending Party may participate in and monitor such defense with counsel of its own choosing. subject to the defending Party's right to assume and conduct the defense of the Shared Development Claim with counsel of its choice. Neither Party shall settle any Shared Development Claim without the prior written consent of the other Party, not to be unreasonably withheld, conditioned or delayed.

12.5 Limitation of Liability. NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, PUNITIVE, OR INDIRECT DAMAGES, INCLUDING FOR LOSS OF PROFITS SUFFERED BY THE OTHER PARTY, ARISING FROM OR RELATING TO ANY BREACH OF THIS AGREEMENT OR ANY TORT CLAIMS ARISING HEREUNDER, REGARDLESS OF ANY NOTICE OF THE POSSIBILITY OF SUCH DAMAGES. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS SECTION 12.5 IS INTENDED, OR SHALL BE CONSTRUED, TO LIMIT OR RESTRICT (A) THE RIGHTS OR OBLIGATIONS OF EITHER PARTY UNDER SECTIONS 12.1, 12.2 OR 12.4; (B) LIABILITY FOR A PARTY'S BREACH OF ITS CONFIDENTIALITY OBLIGATIONS UNDER ARTICLE XIII; OR (C) LIABILITY IN THE CASE OF A PARTY'S FRAUD, GROSS NEGLIGENCE OR INTENTIONAL MISCONDUCT.

12.6 <u>Insurance</u>. Each Party shall maintain in full force and effect during the Term insurance required by Applicable Law in each country where such Party performs any activities under this Agreement. Without limiting the foregoing, each Party shall maintain in full force and effect during the Term either reasonable self-insurance with the ability to cover the liabilities of such Party that could reasonably occur in view of the activities of such Party under this Agreement and the Ancillary Agreements, or insurance policies with the following insurance coverages, with limits of liability not less than those specified below:

(a) Commercial general liability with minimum limits of \$[...***...] each occurrence, \$[...***...] general aggregate and \$[...***...] products/completed operations aggregate, including coverage for premises liability, personal and advertising injury, products and completed operations liability, clinical trial liability, contractual liability and broad form property damage. Each policy shall name the other Party as an additional insured with respect to liability arising from premises rented or owned and liability arising from all ongoing operations. Such insurance may be provided on a claims-made basis; *provided, however, that* such insurance shall have a retroactive date prior to the date that any activities will be performed pursuant to this Agreement, and shall be maintained (or shall have an extended reporting period) of at least [...***...] years after the effective date of termination of this Agreement. The use of primary and excess limits to achieve the total required limits is acceptable.

(b) Workers' compensation insurance in compliance with Applicable Law of the state or other jurisdiction in which activities are performed under this Agreement and employer's liability insurance in amounts not less than [...***...] bodily injury by accident-each accident, [...***...] bodily injury by disease-policy limit and [...***...] bodily injury by disease-each employee. Where permitted by Applicable Law, such policies shall contain a waiver of the insurer's subrogation rights against the other Party.

(c) All insurance programs required to be maintained hereunder shall be from insurers having an A.M. Best rating of [...***...] or better, or its equivalent.

(d) Automobile liability insurance for bodily injury, property damage and automobile contractual liability covering all owned, hired and non-owned autos with a combined single limit of liability for each accident of not less than \$[...***...].

(e) To the extent requested by the other Party, each Party shall provide the other with an original certificate of insurance evidencing that (i) all such insurance coverages are in effect, and (ii) none of the required policies of insurance shall be terminated or canceled by insures except upon at least [...***...] days' written notice to the other Party. Nothing contained in this Section 12.6 is intended, or shall be construed, to limit either Party's indemnity obligations.

ARTICLE XIII

CONFIDENTIALITY

13.1 Confidentiality Obligations.

(a) <u>Confidential Information</u>. "**Confidential Information**" means, with respect to a Party or any of its Affiliates, and subject to Section 13.1(a)(i) through Section 13.1(a)(v), all information that is disclosed by or on behalf of such Party or any of its Affiliates to the other Party or any of its Affiliates under this Agreement, any Ancillary Agreement, the Subscription Agreement, the Filgotinib Amendment or the Security Agreement, except to the extent any such other agreement expressly provides otherwise. Except to the extent expressly authorized by this Agreement or otherwise agreed in writing by the Parties, each Party agrees that, during the Term and for [...***...] years thereafter, it shall, and shall cause its Affiliates to, keep confidential and not publish or otherwise disclose to any Third Party, and not use for any purpose other than as provided for in this Agreement or any Ancillary Agreement, any Confidential Information of the other Party or any of its Affiliates. Galapagos Collaboration Know-How shall be deemed the Confidential Information of Galapagos (and Galapagos shall be deemed to be the disclosing Party and Gilead the receiving Party with respect thereto). Gilead Collaboration Know-How shall be deemed the Confidential Information of Gilead (and Gilead shall be deemed to be the disclosing Party and Galapagos the receiving Party with respect thereto). Joint Collaboration Know-How shall be deemed the Confidential

Information of both Parties (and both Parties shall be deemed to be the disclosing Party and the receiving Party with respect thereto). Notwithstanding the foregoing, Confidential Information of a Party or its Affiliates shall exclude that portion of such information that the receiving Party (or the receiving Party's applicable Affiliate) can demonstrate by competent written proof:

(i) was already known to the receiving Party or its Affiliate, other than under an obligation of confidentiality, at the time of disclosure to the receiving Party or its Affiliate;

Affiliate;

(ii) was generally available to the public or part of the public domain at the time of its disclosure to the receiving Party or its e;

(iii) became generally available to the public or part of the public domain after its disclosure and other than through any wrongful act, fault, or negligence of the receiving Party or its Affiliate;

(iv) was subsequently disclosed to the receiving Party or its Affiliate by a Third Party without obligations of confidentiality with respect thereto; or

(v) was independently discovered or developed by the receiving Party or its Affiliate without the aid, application, or use of the other Party's Confidential Information;

provided that specific disclosures made under this Agreement shall not be deemed to be subject to any of the foregoing exceptions merely because they are embraced by general disclosures in the public knowledge or literature or in the possession of the receiving Party or its Affiliates, and any combination of features disclosed under this Agreement shall not be deemed subject to the above exceptions merely because individual features are in the public knowledge or literature or in the possession of the receiving Party or its Affiliates are in the public knowledge or literature or in the possession of the receiving Party or its Affiliates. The Parties acknowledge that Confidential Information has been provided by the Parties (or their Affiliates) to each other prior to the Effective Date pursuant to the Existing Confidentiality Agreement, including the terms and conditions thereof. The Parties agree that as of the Effective Date, all such Confidential Information shall be protected by the terms and conditions of this Agreement, which shall replace those of such Existing Confidentiality Agreement.

(b) Restrictions on Galapagos Know-How.

(i) With respect to any Galapagos Know-How that is Confidential Information and material to the Exploitation of any Pre-Program Activities, Galapagos Molecule, Galapagos Product, Optioned Molecule or Optioned Product, (A) subject to subsection (B), Galapagos shall not, and shall cause its Affiliates not to, disclose any such material Galapagos Know-How to any Third Party unless such Third Party is bound by obligations of confidentiality and non-use at least as protective as those set forth in Section 13.1 and Section 13.2, except to the extent such disclosure would be permitted under Section 13.2 if such Galapagos Know-How were Confidential Information of Gilead and (B) Galapagos may publish or otherwise publicly disclose such Galapagos Know-How if such publication or other public disclosure complies with the obligations under Section 13.4 and either (1) Galapagos reasonably determines that there is no need to secure Patent protection on such Galapagos Know-How or (2) it would not reasonably be expected to adversely impact the patentability of such Galapagos Know-How.

(ii) With respect to any Gilead Collaboration Know-How that is Confidential Information and material to the Exploitation of any Optioned Molecule or Optioned Product, (A) subject to subsection (B), Gilead shall not, and shall cause its Affiliates not to, disclose any such material Gilead Collaboration Know-How to any Third Party unless such Third Party is bound by obligations of

confidentiality and non-use at least as protective as those set forth in Section 13.1 and Section 13.2, except to the extent such disclosure would be permitted under Section 13.2 if such Gilead Collaboration Know-How were Confidential Information of Gilead and (B) Gilead may publish or otherwise publicly disclose such Gilead Collaboration Know-How if such publication or other public disclosure complies with the obligations under Section 13.4 and either (1) Gilead reasonably determines that there is no need to secure Patent protection on such Gilead Collaboration Know-How or (2) it would not reasonably be expected to adversely impact the patentability of such Gilead Collaboration Know-How.

13.2 Authorized Disclosure or Use of Confidential Information.

(a) <u>Authorized Disclosure</u>. Notwithstanding Section 13.1, each Party may disclose Confidential Information of the other Party to the extent such disclosure is reasonably necessary in the following situations:

(i) filing or prosecuting Galapagos Patents, Gilead Collaboration Patents or Joint Collaboration Patents in accordance with ARTICLE X with the consent of the other Party, such consent not to be unreasonably withheld, delayed or conditioned;

(ii) regulatory filings and other filings with Governmental Authorities (including Regulatory Authorities), including filings with the SEC or FDA, with respect to an Optioned Product as permitted hereunder, *provided that* any such disclosure in a filing with the SEC is, in the opinion of outside counsel, required;

(iii) responding to a valid order of a court of competent jurisdiction or other competent authority, or in the opinion of the receiving Party's legal counsel, making such disclosure as required by Applicable Law or the rules of a stock exchange on which the securities of the disclosing Party (or its parent entity) are listed (or to which an application for listing has been submitted); *provided that* the receiving Party shall, to the extent reasonably practicable under the circumstances, first have given to the disclosing Party notice and a reasonable opportunity to quash the order or obtain a protective order requiring that the Confidential Information be held in confidence or used only for the purpose for which the order was issued or such disclosure was required by Applicable Law or such rules; and *provided further that* if such order is not quashed or a protective order is not obtained, the Confidential Information disclosed shall be limited to the information that is legally required to be disclosed;

(iv) disclosure to its Affiliates and its and its Affiliates' officers, directors, employees, agents and advisors, and any other Third Parties, in each case, only on a need-to-know basis and solely in connection with the performance by the disclosing Party of its obligations or the exercise of its rights under this Agreement (including with respect to the Development, Manufacturing and Commercialization of Optioned Products), *provided that*, prior to any such disclosure, each disclosee other than an advisor must be bound by obligations of confidentiality and non-use at least equivalent in scope as those set forth in Section 13.1 and Section 13.2 and each advisor must be bound by obligations of confidentiality and non-use that are commercially reasonable;

(v) with prior notice to the other Party as permitted by Applicable Law, disclosure of the material terms of this Agreement or any Ancillary Agreement to any *bona fide* potential or actual investor, investment banker, acquirer, merger partner or other potential or actual financial partner; *provided that* each disclose must be bound by obligations of confidentiality and non-use at least as restrictive as those set forth in Section 13.1 and Section 13.2 prior to any such disclosure, except that, where the disclose is an investor, investment banker or financial partner, such disclose shall only need to be bound by commercially reasonable confidential terms; and

(vi) disclosure of any Collaboration Know-How or status reports (including data from any Clinical Trials) by either Party (i) with the consent of the other Party, such consent not to be unreasonably withheld, delayed or conditioned; *provided that* each disclosee must be bound by obligations of confidentiality and non-use at least equivalent in scope as those set forth in Section 13.1 and Section 13.2 prior to any such disclosure, except that, where the disclosee is an investor, investment banker or financial partner, such disclosee shall only need to be bound by commercially reasonable confidential terms or (ii) pursuant to Section 13.4.

Notwithstanding the foregoing, in the event that a Party is required to make a disclosure of the other Party's Confidential Information pursuant to Sections 13.2(a)(i), 13.2(a)(ii) or 13.2(a)(iii), it will, except where impracticable, give reasonable advance notice to the other Party of such disclosure and use reasonable efforts to secure confidential treatment of such information. In any event, the Parties agree to take all reasonable action to avoid disclosure of Confidential Information hereunder.

(b) <u>Authorized Use</u>. [...***...].

13.3 <u>Terms of Agreements</u>.

(a) The Parties agree that the terms of this Agreement, any Ancillary Agreements, the Subscription Agreement and the Security Agreement are the Confidential Information of both Parties, subject to the special authorized disclosure provisions set forth in Section 13.2 and this Section 13.3. The Parties have agreed to make a joint public announcement of the execution of this Agreement substantially in the form of the press release attached as **Schedule 13.3(a)** on the Execution Date.

(b) After release of such press release, if either Party or any of its Affiliates desires to make a press release or other similar public announcement concerning the terms of this Agreement or any Ancillary Agreement, such Party shall give reasonable prior advance notice of the proposed text of such press release or announcement to the other Party for its prior review and approval (except as otherwise provided herein), such approval not to be unreasonably withheld, conditioned or delayed, except that, subject to Section 13.3(c), in the case of a press release or filings with a Governmental Authority required by Applicable Law, such Party shall provide the other Party with such advance notice as it reasonably can and shall not be required to obtain approval therefor. The other Party shall provide its comments, if any, within [...***...] Business Days after receiving the press release for review. Neither Party shall be required to seek the permission of the other Party to repeat any information regarding the terms of this Agreement or any Ancillary Agreement that have already been publicly disclosed by such Party or such Party's Affiliates, or by the other Party or any of its Affiliates, in accordance with this Section 13.3; *provided that* such information remains accurate as of such repeat.

(c) The Parties acknowledge that either or both Parties (or their respective parent entities) may be obligated to make a filing (including to file a copy of this Agreement and the Subscription Agreement) with the SEC or other Governmental Authorities. Each Party shall be entitled to make such a required filing, *provided that* it shall (i) agree (such agreement not to be unreasonably withheld, conditioned or delayed) with the other Party in advance regarding the form of redacted copy of this Agreement and the Subscription Agreement to be so filed (the "**Redacted Agreements**"), (ii) request, and use commercially reasonable efforts consistent with Applicable Laws to obtain, confidential treatment of all terms redacted from this Agreement and the Subscription Agreement, as reflected in the Redacted Agreement, for a period of at least [...***...] years, (iii) promptly deliver to the other Party any written correspondence received by it or its representatives from such Governmental Authority with respect to such confidential treatment request, (iv) upon the written request of the other Party, if legally justifiable, request an appropriate extension of the

term of the confidential treatment period, and (v) if such Governmental Authority requests any changes to the redactions set forth in the Redacted Agreement, use commercially reasonable efforts consistent with Applicable Laws to support the redactions in the Redacted Agreement as originally filed and not agree to any changes to the Redacted Agreement without, to the extent practical, first discussing such changes with the other Party and taking the other Party's comments into consideration when deciding whether to agree to such changes. Each Party shall be responsible for its own legal and other external costs and expenses in connection with any such filing, registration or notification.

13.4 Public Disclosures of Data.

(a) Publications.

(i) **"Publication**" means, with respect to any information, data or results, the public disclosure of such information, data and results, including any oral presentation or abstract of such data and results at scientific and medical conferences or publications of, or investor communications regarding, such information, data and results in peer-reviewed journals.

(ii) <u>Pre-Program Activities</u>. Galapagos may make Publications regarding the Pre-Program Activities in sole its discretion and no JCRC or Gilead review or approval shall be required; *provided*, *however*, Galapagos shall use reasonable efforts to provide Gilead and the JCRC advanced written notice of such proposed Publication.

(iii) <u>Independent Activities</u>. Each Party shall maintain a Publication plan regarding Publications that contain any information, data or results with respect to any Independent Activities (including any data or results of Clinical Trials or Nonclinical Studies) for which such Party is the Independent Activities Party. Each Party shall provide its Publication plan to the JCRC for its review. Each Party shall disclose such Publications consistent with the applicable plan.

(iv) <u>Optioned Programs</u>. Each Party shall maintain a Publication plan regarding Publications that contain any information, data or results with respect to any Optioned Molecule or Optioned Product under an Optioned Program (including any data or results of Clinical Trials or Nonclinical Studies). Each Party shall provide its Publication plan to the JCRC for its review and approval. Each Party shall disclose such Publications consistent with such approved plan; *provided that* the Party proposing a Publication shall provide the other Party or the JCRC the opportunity to review the proposed Publication at least [...***...] Business Days prior to its intended submission for publication, oral presentation or abstract of any data and results at scientific and medical conferences. If the other Party or the JCRC offers any comments on the Publication, the submitting Party shall consider such comments in good faith.

(b) <u>Press Releases</u>. Except to the extent required by Applicable Law or the rules of a stock exchange on which the securities of the disclosing Party (or its parent entity) are listed (or to which an application for listing has been submitted), at least [...***...] Business Days prior to either Party or its Affiliates issuing a press release with respect to any Optioned Program, including in connection with any Independent Activities for which it is the Independent Activities Party, such Party shall provide a copy of such press release to (i) the other Party and (ii) the JCRC for its initial review and subsequent submission to the JSC for approval. Except to the extent required by Applicable Law or the rules of a stock exchange on which the securities of the disclosing Party (or its parent entity) are listed (or to which an application for listing has been submitted), neither Party nor its Affiliates shall issue any such press release unless and until such press release is approved by the JSC.

13.5 Destruction of Galapagos Confidential Information. Upon the earlier of (a) the expiration of the Galapagos Program Period for each Galapagos Program (with respect to Confidential Information of Galapagos related to such Galapagos Program and that is not related to a Galapagos Program or Optioned Program for which this Agreement remains in effect), (b) the expiration of the Collaboration Period (with respect to any Confidential Information of this Agreement remains in effect), (c) the effective date of termination of this Agreement for a given Pre-Program Activity, Galapagos Program or Optioned Program, as applicable, and not related to any Galapagos Program or Optioned Program for which this Agreement in its entirety (with respect to all Confidential Information of Galapagos), Gilead shall, as soon as reasonably practicable, destroy all copies of such Confidential Information in its and its Affiliates possession and confirm such destruction in writing to Galapagos; *provided that* Gilead may retain one copy of such Confidential Information for the sole purpose of performing any continuing obligations hereunder, as required by Applicable Law, or for archival purposes.

13.6 <u>Market Abuse Regulation Considerations</u>. Each Party notes and acknowledges, and shall ensure that its relevant Affiliates and its and its Affiliates' officers, directors, employees, agents and advisors receiving information in relation to or involving the other Party in connection with the performance by such receiving Party of its obligations or the exercise of its rights under this Agreement (including with respect to Development, Manufacturing and Commercialization of Optioned Products) could constitute, in relation to the other Party or the financial instruments of the other Party, "inside information" as defined in Article 7 of the Market Abuse Regulation, and that in that case and pending the disclosure thereof by the other Party, such receiving Party and its relevant Affiliates and its and its Affiliates' officers, directors, employees, agents and advisors are (a) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information relates; (b) prohibited from using that information, or attempting to use that information; and (c) obliged to keep that information confidential. Each Party shall not, and shall instruct its relevant Affiliates' officers, directors, employees, agents and advisors not to, act or use such information in relation to or involving the other Party in any way that contravenes Article 8 of MAR (insider dealing), Article 10 of MAR (unlawful disclosure of inside information), Article 12 of MAR (market manipulation), Article 14 of MAR (

ARTICLE XIV

TERM AND TERMINATION

14.1 Term. This Agreement shall become effective on the Effective Date and, unless terminated earlier pursuant to this ARTICLE XIV, shall expire in its entirety as of the expiration of the Collaboration Term; *provided that* (a) with respect to each Galapagos Program in existence as of the expiration of the Collaboration Term (including any Galapagos Program that is Suspended or Terminated), the Post-IND Term Extension shall apply and this Agreement shall continue in effect solely for such Galapagos Program until the expiration of the Post-IND Term Extension (and, for clarity, this Agreement shall expire with respect to all Pre-Program Activities), and (b) with respect to each Optioned Program in existence as of the expiration of the Collaboration Term, this Agreement shall continue in effect until the end of each applicable Royalty Term in effect for any Optioned Product included in such Optioned Program anywhere

in the Gilead Territory; *provided further* that, no later than [...***...] prior to the end of the Post-IND Term Extension, solely with respect to each Galapagos Program (i) for which the Post-IND Term Extension applied pursuant to foregoing clause (a) and (ii) that is in existence at such time (including any Galapagos Program that is Suspended or Terminated), Galapagos shall, in its sole discretion, notify Gilead that Galapagos elects to either (A) apply the Final Term Extension to such Galapagos Program or (B) allow Gilead to exercise the Option for such Galapagos Program prior to the end of the Post-IND Term Extension pursuant Section 8.2(b)(iv), but subject to [...***...] (the period from the Effective Date through expiration (or earlier termination) of this Agreement as described in this Section 14.1, with respect to each Galapagos Program and Optioned Program, the "**Term**"). If, with respect to (1) a Galapagos Program for which Galapagos allows Gilead to exercise its Option pursuant to foregoing clause (B), Gilead fails to exercise such Option prior to the expiration of the Post-IND Term Extension, then this Agreement shall terminate with respect to such Galapagos Program and (2) a Galapagos Program for which Galapagos applies the Final Term Extension pursuant to foregoing clause (A), Gilead fails to exercise such Option prior to the expiration of the Final Term Extension, then this Agreement shall terminate with respect to such Galapagos Program.

14.2 <u>Outside Date</u>. The provisions of this Agreement, the Subscription Agreement and the Filgotinib Amendment that are binding on the Parties as of the Execution Date shall terminate automatically, without any further action required by either Party, if the Antitrust Clearance Date has not occurred on or before the date that is [...***...] days following the Execution Date of this Agreement (the "**Outside Date**"); *provided, however, that* the Parties may mutually agree in writing to postpone the Outside Date by sequential [...***...] day increments if the Antitrust Clearance Date has not occurred by the initial Outside Date.

14.3 Termination by Gilead at Will or by Galapagos Under Section 5.1.

(a) <u>Termination by Gilead at Will</u>. Gilead shall have the right, at any time after the Effective Date and in its sole discretion, to terminate this Agreement with respect to any Pre-Program Activities (which termination shall be with respect to all molecules or products that are the subject of such Pre-Program Activities), any Galapagos Program (which termination shall be with respect to all Galapagos Molecules and Galapagos Products included in such Galapagos Program) or any Optioned Program (which termination shall be with respect to all Optioned Molecules and Optioned Products included in such Optioned Program) with respect to any country(ies) in the Gilead Territory upon [...***...] days' prior written notice to Galapagos. If Gilead exercises such termination right for any country (but not all countries) in the Gilead Territory, then this Agreement shall remain in effect with respect to all other countries in the Gilead Territory and such country with respect to which Gilead exercised such termination right shall become a country in the Galapagos Territory with respect to the applicable Galapagos Program or Optioned Program, as the case may be.

(b) <u>Termination by Galapagos Under Section 5.1</u>. Galapagos shall have the right to terminate this Agreement pursuant to Section 5.1.

14.4 Termination by Either Party for Material Breach.

(a) Termination and Other Rights.

(i) <u>Rights of Galapagos for Gilead's Breach</u>. Subject to Section 14.4(b) and Section 14.4(c), if Gilead materially breaches its obligations under this Agreement with respect to any Galapagos Program or Optioned Program, then Galapagos shall have the right to terminate this Agreement with respect to such Galapagos Program (which termination shall be with respect to all Galapagos Molecules and Galapagos Products included in such Galapagos Program) or Optioned Program (which

termination shall be with respect to all Optioned Molecules and Optioned Products included in such Optioned Program) by providing (A) in the case of a failure by Gilead to make any payment due hereunder, [...***...] days' or (B) in the case of any other material breach by Gilead, [...***...] days', in each case ((A) and (B)), prior written notice to Gilead identifying such material breach in reasonable detail and the Galapagos Program(s) or Optioned Program(s) to which such material breach relates; *provided that* (1) the materiality of any breach by Gilead shall be measured in relation to the entire Gilead Territory for the applicable Galapagos Program or Optioned Program (and any material breach with respect to the United States shall be deemed a material breach with respect to the entire territory) and (2) any such proposed termination shall not become effective if Gilead cures the breach specified in such notice during the applicable notice period set forth in clause (A) or clause (B) (or in the case of clause (B), if not curable within such [...***...]-day period, then for up to a maximum additional [...***...]-day period but only so long as, during such additional [...***...]-day period, Gilead is using reasonable efforts to do so). Notwithstanding the foregoing, in the event that the termination for material breach is solely with respect to one or more countries other than the United States but not all countries, then for such Galapagos Program or Optioned Program, as the case may be, the Gilead Territory shall no longer include such country or countries and in lieu of termination of this Agreement under this Section 14.4(a)(i), such country or countries shall become part of the Galapagos Territory and a transition analogous to the transition described in Section 14.4(a)(i), then Galapagos shall have the right to terminate this Agreement in its entirety with respect to such Optioned Program pursuant to this Section 14.4(a)(i).

(ii) Rights of Gilead for Galapagos' Breach. Subject to Section 14.4(b) and Section 14.4(c), if Galapagos materially breaches its obligations under this Agreement with respect to any Galapagos Program or Optioned Program, then Gilead shall have the right to terminate this Agreement with respect to such Galapagos Program (which termination shall be with respect to all Galapagos Molecules and Galapagos Products included in such Galapagos Program) or Optioned Program (which termination shall be with respect to all Optioned Molecules and Optioned Products included in such Optioned Program) by providing (A) in the case of a failure by Galapagos to make any payment due hereunder, [...***...] days' or (B) in the case of any other material breach by Galapagos, [...***...] days', in each case ((A) and (B)), prior written notice to Galapagos identifying such material breach in reasonable detail and the Galapagos Program(s) or Optioned Program(s) to which such material breach relates; provided that any such proposed termination shall not become effective if Galapagos cures the breach specified in such notice during the applicable notice period set forth in clause (A) or clause (B) (or in the case of clause (B), if not curable within such [...***...]-day period, then for up to a maximum additional [...***...]-day period but only so long as, during such additional [...***...]-day period, Galapagos is using reasonable efforts to do so). Without limitation to the foregoing, in lieu of Gilead's right to terminate this Agreement pursuant to this Section 14.4(a)(ii) with respect to any Galapagos Program or Optioned Program, Gilead shall have the right to elect (as its sole remedy for such breach), in its written notice to Galapagos identifying the applicable material breach, to (1) maintain the effectiveness of this Agreement with respect to such Galapagos Program or Optioned Program and (2) reduce by [...***...] percent [...***...] any amounts payable to Galapagos under ARTICLE IX with respect to such Galapagos Program (if and when it becomes an Optioned Program) or Optioned Program (in each case, including with respect to all Galapagos Molecules, Galapagos Products, Optioned Molecules and Optioned Products included in such Galapagos Program or Optioned Program, as the case may be); provided that (x) any such proposed reduction shall not become effective if Galapagos cures the breach specified in such notice during the applicable notice period or during an additional [...***...]-day period thereafter and (y) such reduction shall terminate and Gilead shall again be responsible for the full amounts payable to Galapagos under ARTICLE IX at such time as [...***...].

(b) <u>Disputed Breach</u>. If, during any applicable notice period described in Section 14.4(a), the allegedly breaching Party provides the other Party with (i) written notice disputing in good faith (A) the existence or materiality of a breach specified in a notice provided by such other Party in accordance with Section 14.4(a) or (B) whether a material breach has been cured within such notice period and (ii) invokes and continues to pursue in good faith the dispute resolution procedures set forth in Section 15.1(b), then such other Party shall not have the right to terminate this Agreement for such breach unless and until (1) it has been determined by arbitration in accordance with Section 15.2(a) that such allegedly breaching Party has materially breached this Agreement or that such Party has failed to cure a material breach (as applicable) and (2) such allegedly breaching Party fails to cure such material breach within (I) [...***...] days following such determination, in the case of any failure to make payment or (II) [...***...] days following such determination, in the case of any such dispute, all of the terms and conditions of this Agreement shall remain in effect.

(c) [...***...].

14.5 Termination by Either Party for Insolvency. If, at any time during the Term (a) a case is commenced by or against either Party under Title 11, United States Code, as amended, or analogous provisions of Applicable Law outside the United States (the "**Bankruptcy Code**") and, in the event of an involuntary case under the Bankruptcy Code, such case is not dismissed within [...***...] days after the commencement thereof, (b) either Party files for or is subject to the institution of bankruptcy, liquidation or receivership proceedings (other than a case under the Bankruptcy Code), (c) either Party assigns all or a substantial portion of its assets for the benefit of creditors, (d) a receiver or custodian is appointed for either Party's business, or (e) a substantial portion of either Party's business is subject to attachment or similar process and such attachment or similar process is not dismissed or withdrawn within one hundred twenty (120) days after the commencement thereof, then, in any such case ((a), (b), (c), (d) or (e)), the other Party may terminate this Agreement upon written notice to such Party to the extent permitted under Applicable Law.

14.6 [...***...].

14.7 Effects of Termination of this Agreement.

(a) Conduct During Termination Notice Period.

(i) Following any notice of termination permitted under this ARTICLE XIV, during any applicable termination notice period with respect to any Galapagos Program, Galapagos Molecule, Galapagos Product, Optioned Program, Optioned Molecule or Optioned Product (as applicable, each, a "**Termination Notice Period**"), each Party shall continue to perform all of its obligations under this Agreement with respect to such Galapagos Program, Galapagos Product, Optioned Program, Optioned Molecule or Optioned Product, including performing all activities allocated to it with respect thereto pursuant to any R&D Plan and Budget, Global Manufacturing Plan and Budget or Global Commercialization Plan and Budget then in effect, in each case, in accordance with the terms and conditions of this Agreement. Each Party shall also continue to bear its share of all Research and Development Costs, Shared Commercialization Costs and Shared Patent Costs incurred during the Termination Notice Period with respect to any such Galapagos Program, Galapagos Molecule, Galapagos Product, Optioned Program, Optioned Molecule or Optioned Product (as applicable).

(ii) [...***...].

(b) <u>Galapagos Programs</u>. In the case of a termination of this Agreement with respect to any Galapagos Program, Galapagos Molecule or Galapagos Product, all rights and licenses granted to Gilead under this Agreement with respect to such Galapagos Program, Galapagos Molecule or Galapagos Product, as applicable, shall terminate.

(c) <u>Optioned Programs</u>. In the case of a termination of this Agreement with respect to any Optioned Program ("**Terminated Program**"), (including all Optioned Molecules ("**Terminated Molecules**") and Optioned Products ("**Terminated Products**") that are the subject of such Optioned Program), in each case ((i) through (iii)), the following shall apply with respect to such Terminated Program (including all Terminated Molecules and Terminated Products that are the subject of such Terminated Program) (in addition to any other rights and obligations under this ARTICLE XIV or otherwise under this Agreement with respect to such termination) in the applicable Terminated Regions:

(i) Licenses. The licenses and other rights granted to Gilead under this Agreement shall terminate with respect to any such Terminated Program (including all Terminated Molecules and Terminated Products that are the subject of such Terminated Program) for the Terminated Region(s) (subject to the sell-off period set forth in Section 14.7(c)(vi)). Effective as of the effective date of termination of this Agreement with respect to any applicable Reversion Product(s), Gilead shall, subject to Section 14.7(c)(v), grant to Galapagos an exclusive, perpetual, irrevocable license, with the right to grant multiple tiers of sublicenses, under (A) any Gilead Collaboration Patents and Gilead's interest in any Joint Collaboration Patents and (B) any other Patents Controlled by Gilead as of the effective date of such termination that claim or cover such Reversion Product or that are being used in the Development, Manufacture or Commercialization or other Exploitation of the Reversion Product as of the effective date of termination ((A) and (B), the "**Gilead Reversion Patents**"), to Develop, Manufacture and Commercialize and otherwise Exploit such Reversion Product(s) in the Field for the applicable Terminated Region(s); *provided, however, that* [...***...]. For clarity, the foregoing license under Gilead Reversion Patents extends solely to those elements of such a Reversion Product that were incorporated into such Reversion Product as of the effective date of termination and shall not be construed as a right to modify such elements or to incorporate additional elements or technology that would infringe a Gilead Reversion Patent.

(ii) <u>Optioned Product Trademarks</u>. Gilead shall assign to Galapagos all of its right, title and interest in and to any Optioned Product Trademarks Controlled by Gilead and used exclusively with any applicable Reversion Product(s) (excluding any such Optioned Product Trademarks to the extent that it includes, in whole or part, any corporate name or logo of Gilead or any of its Affiliates or Sublicensees) in the applicable Terminated Region(s).

(iii) <u>Regulatory Materials</u>. Gilead shall grant to Galapagos a right of reference under all Regulatory Materials and Regulatory Approvals Controlled by Gilead, in each case, that relate to any applicable Reversion Product(s) in or for the benefit of the applicable Terminated Region(s), unless and until such Regulatory Materials or Regulatory Approvals are assigned to Galapagos pursuant to any Transition Agreement.

(iv) On-Going Clinical Trials.

(A) With respect to each Clinical Trial for any Terminated Product in or for the benefit of the applicable Terminated Region(s) being conducted by or on behalf of Gilead that was Initiated prior to the effective date of termination (each such Clinical Trial, a "**Gilead Clinical Trial**"), Gilead shall, at Galapagos' election, (1) continue to timely perform all activities necessary to continue such Gilead Clinical Trial through its database lock in accordance with the protocol in effect for such Gilead Clinical Trial as of the effective date of the applicable termination, (2) work with Galapagos to promptly transfer such Gilead Clinical Trial to Galapagos in an orderly manner, or (3) promptly wind-down such Gilead Clinical Trial in an orderly manner consistent with ethical and clinical obligations and Applicable Law. For each Gilead Clinical Trial that Galapagos elects to have Gilead promptly wind-down, Gilead shall be solely responsible for all costs for such Gilead Clinical Trial.

(B) With respect to each Gilead Clinical Trial for which Galapagos elects to have Gilead either perform all activities necessary to continue such Clinical Trial through its database lock or promptly transfer such Clinical Trial to Galapagos in an orderly manner, the costs and expenses to conduct such Gilead Clinical Trial or to transfer such Gilead Clinical Trial to Galapagos, as applicable, shall be allocated as follows:

(i) If Galapagos terminated the Terminated Product for such Gilead Clinical Trial pursuant to Section 14.4(a)(i), 14.5 or 14.6, (1) with respect to each Gilead Clinical Trial set forth in the R&D Plan and Budget for the applicable Terminated Program, such costs and expenses shall [...***...] in accordance with the terms of this Agreement and (2) with respect to each Gilead Clinical Trial set forth in an Independent Activities Plan, such costs shall be treated [...***...], and, in each case, [...***...] in accordance with the terms of this Agreement.

(ii) If Gilead terminated the Terminated Product for such Gilead Clinical Trial pursuant to Section 14.4(a)(ii) or 14.5, Galapagos shall reimburse Gilead for [...***...] percent [...***...] of such costs and expenses. Such payment shall be made to Gilead within [...***...] days after receipt of an invoice with respect thereto, which invoice must be accompanied by a report in reasonable detail of such costs and expenses.

(iii) If either Party terminated the Terminated Product for such Gilead Clinical Trial pursuant to Section 14.3, (1) with respect to each Gilead Clinical Trial set forth in the R&D Plan and Budget, such costs and expenses shall continue to be treated as [...***...] in accordance with the terms of this Agreement and (2) with respect to each Gilead Clinical Trial set forth in an Independent Activities Plan, Galapagos shall reimburse Gilead for [...***...] percent [...***...] of such costs and expenses. Such payment shall be made to Gilead within [...***...] days after receipt of an invoice with respect thereto, which invoice must be accompanied by a report in reasonable detail of such costs and expenses.

(iv) <u>Transition Agreement</u>. For any Terminated Molecule or Terminated Product, in each case, with respect to the applicable Terminated Region(s), the Parties shall enter into a written agreement (a "**Transition Agreement**") that would effectuate the terms and conditions of this Section 14.7(c)(iv) and would include other reasonable terms and conditions, including terms allocating costs and expenses, describing the Parties' indemnification obligations, setting forth the Parties' obligations with respect to unauthorized sales, and setting forth other coordination obligations. If, despite such efforts, the Parties are unable to agree upon the terms and conditions of any Transition Agreement within [...***...] days after the effective date of termination of this Agreement with respect to the applicable Terminated Molecule or Terminated Product, then either Party may refer the dispute for resolution by arbitration in accordance with Section 15.2(d).

(A) <u>Transition Assistance</u>. The Transition Agreement shall require Gilead, [...***...], to disclose and provide to Galapagos (1) [...***...] and that is necessary for Galapagos to Develop, Manufacture or Commercialize the applicable Reversion Product(s) with respect to the applicable Terminated Region(s) and (2) at Galapagos' request, all then-existing commercial arrangements of Gilead or any of its Affiliates to the extent relating solely and specifically to the applicable Reversion Product(s) and the applicable Terminated Region(s) (which may be redacted to the extent necessary to comply with Gilead' or its Affiliate's obligation to the Third Party to any such commercial arrangement), and to provide reasonable consultation and assistance with respect thereto for a period of no more than [...***...] days following the completion of such disclosure and provision. The foregoing obligation

described in clause (2) shall include assigning to Galapagos, upon request of Galapagos, any agreements between Gilead or any of its Affiliates and Third Party suppliers or vendors to the extent such agreements solely and specifically relate to the supply or sale of the applicable Reversion Product(s) with respect to the applicable Terminated Region(s). If any such agreement is not assignable to Galapagos (whether by such agreement's terms or because such agreement does not solely and specifically relate to the supply or sale of the applicable Reversion Product(s) with respect to the applicable Terminated Region(s)) but is otherwise reasonably necessary or useful for Galapagos to Develop, Manufacture or Commercialize any such Reversion Product(s) with respect to the applicable Terminated Region(s), then Gilead shall reasonably cooperate with Galapagos in Galapagos' efforts to obtain from such Third Party the assignment of such agreement or of that portion of such agreement that solely and specifically relates to the supply or sale of such Reversion Product(s) with respect to the applicable Terminated Region(s). With respect to any applicable Reversion Product(s) and the applicable Terminated Region(s), unless and until any such agreements are assigned to Galapagos pursuant to the preceding sentences, or if Gilead Manufactures such Reversion Product(s) itself (and thus there is no agreement to assign), the Transition Agreement shall require Gilead to supply bulk finished quantities of such Reversion Product(s) to Galapagos with respect to the applicable Terminated Region(s) for a reasonable period following the effective date of termination (not to exceed [...***...] months) to enable Galapagos to establish an alternate, validated source of supply for such Reversion Product(s). The cost to Galapagos for supply of any such Reversion Product shall be [...***...].

(B) <u>Regulatory Materials</u>. The Transition Agreement shall require Gilead to transfer and assign to Galapagos [... ***...], that relate to any applicable Reversion Product(s) in or for the benefit of the applicable Terminated Region(s). The Transition Agreement shall contain terms governing the coordination of any ongoing regulatory responsibilities with respect to such Reversion Product(s) that are required under Applicable Law to be conducted by Gilead with respect to the applicable Terminated Region(s).

(v) <u>Third Party Agreements</u>. To the extent that any payments would be owed by Gilead or any of its Affiliates to any Third Party (including royalties, milestones and other amounts) under any Third Party agreements that are applicable to the grant to Galapagos of any (sub)license, right of reference or other right provided in this Section 14.7 or any Transition Agreement, or that are applicable to the exercise by Galapagos or any of its Affiliates or Sublicensees of any sublicense or other right with respect thereto, Gilead shall notify Galapagos of the existence and anticipated amounts of such payments and Galapagos shall have the right either to decline such (sub)license, right of reference or other right provided in this Section 14.7 or such Transition Agreement or to accept the same, in which case Galapagos shall (A) comply with any obligation under any such Third Party agreement that apply to Galapagos (*provided that* Gilead has notified Galapagos of such obligation in writing) and (B) be responsible for any such payments.

(vi) Remaining Inventories.

(A) Gilead shall be entitled, during the [...***...] days following termination of this Agreement with respect to any Terminated Product(s) and Terminated Region(s), to finish any work-in-progress and to sell in the applicable Terminated Region(s) any inventory of Terminated Product(s) that remains on hand as of the effective date of such termination. Gilead shall pay Galapagos the royalties applicable to such sales in accordance with the terms and conditions of this Agreement.

(B) At any time within [...***...] days after the effective date of termination of this Agreement with respect to any Terminated Product(s) and Terminated Region(s), Galapagos shall have the right, upon written notice to Gilead, to purchase from Gilead any or all of the inventory of Reversion Product(s) held by Gilead with respect to any such Terminated Region(s) as of the date of such notice solely for distribution in the applicable Terminated Region(s) and other countries in the Galapagos Territory and not for distribution in the remainder of the Gilead Territory (which inventory of Reversion Product(s) is not committed to be supplied to any Third Party as of such date) at a price equal to [...***...].

14.8 <u>Other Remedies</u>. Termination or expiration of this Agreement for any reason shall not release either Party from any liability or obligation that already has accrued prior to such expiration or termination, nor affect the survival of any provision hereof to the extent it is expressly stated to survive such termination. Termination or expiration of this Agreement for any reason shall not constitute a waiver or release of, or otherwise be deemed to prejudice or adversely affect, any rights, remedies or claims, whether for damages or otherwise, that a Party may have hereunder or that may arise out of or in connection with such termination or expiration.

14.9 <u>Rights in Bankruptcy</u>. All rights and licenses granted under or pursuant to this Agreement by Galapagos and Gilead are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the U.S. Bankruptcy Code, licenses of right to "intellectual property" as defined under Section 101 of the U.S. Bankruptcy Code. The Parties agree that each Party, as licensee of certain rights under this Agreement, shall retain and may fully exercise all of its rights and elections under the U.S. Bankruptcy Code. The Parties further agree that, in the event of the commencement of a bankruptcy proceeding by or against a Party (such Party, the "**Bankrupt Party**") under the U.S. Bankruptcy Code, the other Party shall be entitled to a complete duplicate of (or complete access to, as appropriate) any intellectual property licensed to such other Party pursuant to the terms of this Agreement and all embodiments of such intellectual property, which, if not already in such other Party's possession, shall be promptly delivered to it (a) upon any such commencement of a bankrupt proceeding upon such other Party's written request therefor, unless the Bankrupt Party elects to continue to perform all of its obligations under this Agreement or (b) if not delivered under clause (a), following the rejection of this Agreement by the Bankrupt Party upon written request therefor by the other Party. In the case of an Insolvency that is governed by non-U.S. bankruptcy law, the Parties agree that, to the extent not prohibited by the applicable Insolvency law, the non-Bankrupt Party will be entitled to the same rights and protections afforded by the U.S. Bankruptcy Code, including survival of the licenses granted hereunder even if the Bankrupt Party revokes or terminates this Agreement and a copy of the embodiments of such intellectual property, without conditions other than payment of any royalties due hereunder.

14.10 <u>Survival</u>. Termination or expiration of this Agreement shall not affect rights or obligations of the Parties under this Agreement that have accrued prior to the date of termination or expiration of this Agreement. Notwithstanding anything to the contrary, the following provisions shall survive and apply after expiration or termination of this Agreement in its entirety: ARTICLE XII, ARTICLE XII, ARTICLE XV, ARTICLE XVII, and Sections 2.6(b), 2.6(d) (other than the last sentence), 4.2(b) (subject to Section 14.7(c) and the terms of any Transition Agreement, if applicable), 9.5, 9.6(b) (expiration only), 9.7 (expiration only), 9.12, 9.13, 9.14, 9.15, 9.16, 9.17, 10.1 (subject to Section 14.7(c) and the terms of any Transition Agreement, if applicable), 4.3, 7.4 (subject to Section 14.2, only the provisions referenced in Section 16.5 and any applicable definitions of any capitalized terms, including those set forth in Appendix A; shall survive and apply. In addition, except in case of termination pursuant to Section 14.2, the other applicable provisions of ARTICLE IX shall survive and apply to the extent required to make final reimbursements, reconciliations or other payments incurred or accrued prior to the date of termination or expiration. For any surviving provisions requiring action or decision by a Committee or an Executive Officer, each Party will appoint representatives to act as its Committee members or Executive Officer, as applicable. All provisions not surviving in accordance with the foregoing shall terminate upon expiration or termination of this Agreement and be of no further

force and effect. If this Agreement is terminated with respect to one or more Terminated Regions but not in its entirety, then following such termination the foregoing provisions of this Agreement shall remain in effect with respect to the Terminated Regions (to the extent they would survive and apply in the event the Agreement expires or is terminated in its entirety), and all provisions not surviving in accordance with the foregoing shall terminate upon termination of this Agreement with respect to the applicable Terminated Region(s) and be of no further force and effect (and for the avoidance of doubt all provisions of this Agreement shall remain in effect with respect to any countries that are not terminated).

ARTICLE XV

DISPUTE RESOLUTION

15.1 Disputes.

(a) <u>Generally</u>. It is the objective of the Parties to establish procedures to facilitate the resolution of disputes arising under this Agreement in an expedient manner by mutual cooperation and without resort to litigation.

(b) Non-Committee Disputes. In the event of any dispute that may arise between the Parties out of or in relation to or in connection with this Agreement ("Dispute") that does not arise from a matter within the decision-making jurisdiction of a Committee (a "Non-Committee Dispute"), including any alleged failure to perform, or breach, of this Agreement, or any issue relating to the interpretation or application of this Agreement, either Party may refer such Non-Committee Dispute to (i) with respect to any Non-Committee Dispute regarding Development or Commercialization, the Parties' applicable Subject Matter Experts and (ii) with respect to any other Non-Committee Dispute, the Executive Officers of the Parties. Following any such referral to Subject Matter Experts, such Subject Matter Experts shall attempt to reach consensus on such Non-Committee Dispute during a period of [...***...] Business Days, and any final decision agreed to in writing by the Subject Matter Experts with respect to such Non-Committee Dispute to the Executive Officers of the Parties for resolution. Following referral to the Executive Officers (either after referral to Subject Matter Experts or as an initial matter, as applicable), the Executive Officers shall attempt to reach consensus on such Non-Committee Dispute of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers shall attempt to reach consensus on such Non-Committee Officers (either after referral to Subject Matter Experts or as an initial matter, as applicable), the Executive Officers shall attempt to reach consensus on such Non-Committee Dispute within such period of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers shall attempt to reach consensus on such Non-Committee Dispute during a period of [...***...] Business Days thereafter, and any final decision agreed to in writing by the Executive Officers shall attempt to reach consensus on such Non-Committee Dispute durin

(c) <u>Committee Disputes</u>. In the event of a Dispute that arises from a Committee and is within the decision-making jurisdiction of such Committee (a "**Committee Dispute**"), the Parties shall first attempt to resolve such Committee Dispute pursuant to ARTICLE I, including Section 1.5. If the Committee Dispute is not resolved pursuant to ARTICLE I, then such Committee Dispute shall be resolved in accordance with Section 1.5 [... ***...].

15.2 Arbitration.

(a) <u>Arbitration of Non-Committee Disputes</u>. With respect to any Non-Committee Dispute other than a Non-Committee Dispute covered by Section 15.2(d) or Section 15.7, either Party may, following the end of the [...***...] Business Day period referenced in Section 15.1(b), refer such Non-Committee Dispute to binding arbitration in accordance with Section 15.2(c) by submitting a written notice of such request to the other Party.

(b) <u>Arbitration of Committee Disputes [...***...]</u>. Following the end of the period set forth in Section 1.5(b)(i)(C) for the Executive Officers to discuss any Dispute regarding whether [...***...], either Party may refer such Dispute to arbitration in accordance with Section 15.2(c) by submitting a written notice of such request to the other Party. With respect to the arbitration of any Dispute under this Section 15.2(b), the following shall apply: (i) the Parties shall each choose their respective arbitrator pursuant to Section 15.2(c) within [...***...] Business Days after referring such Dispute for arbitration, and such arbitrators shall mutually agree upon the third arbitrator pursuant to Section 15.2(c) with [...***...] Business Days after their collective appointment, which arbitrators shall have expertise with respect to development in the pharmaceutical and biotechnology industries; (ii) the arbitrators must determine, as the only method to resolve such a Dispute, whether [...***...], (iii) within [...***...] Business Days after appointment of such third arbitrator, each Party shall deliver to the arbitrators and to the other Party a memorandum in support of its position; (iv) within [...***...] Business Days after the delivery of such memorandum, the Parties and arbitrators shall meet to discuss the Dispute; and (v) within [...***...] Business Days after such meeting, the arbitrators shall issue their determination with respect to such Dispute in accordance with Section 15.2(c). If the arbitrators determine that [...***...], then the arbitrators shall set forth in writing [...***...].

(c) Arbitration Procedures. For disputes referred to arbitration in accordance with Section 15.2(b), Section 15.2(b) shall control in the event of any conflict between this Section 15.2(b) and Section 15.2(c). Any arbitration shall be conducted in accordance with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators. Each Party will select one arbitrator, and the two (2) arbitrators so selected by the Parties shall select the third arbitrator. The arbitrators shall have significant experience and shall have expertise in licensing and partnering agreements in the pharmaceutical and biotechnology industries, including expertise in the applicable subject matter of the Dispute. The place of arbitration shall be New York City and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither Party shall be required to give general discovery of documents, but may be required only to produce specific, identified documents that are relevant to the Dispute. The Parties shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Parties, and shall be governed by the terms and conditions hereof, including the limitation on damages set forth in Section 12.5. The Parties agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of the State of New York applicable to the commencement of a lawsuit shall apply to the commencement of arbitration under this ARTICLE XV. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each Party. All proceedings and decisions of the arbitrators shall be deemed Confidential Information of both of the Parties, and shall be subject to ARTICLE XIII. The arbitrators shall issue appropriate protective orders to safeguard each Party's Confidential Information. Except as required by Applicable Law, neither Party shall make (or instruct the arbitrators to make) any public announcement with respect to the proceedings or decision of the arbitrators without prior written consent of the other Party. The existence of any Dispute submitted to arbitration, and the award, shall be kept in confidence by the Parties and the arbitrators, except as required in connection with the enforcement of such award or as otherwise required by Applicable Law.

(d) <u>Baseball Arbitration</u>. With respect to any Baseball Matter, either Party may, following the end of the period referenced in Section [... ***...] or Section [...***...], as applicable, refer such Baseball Matter to arbitration in accordance with Section 15.2(c) by submitting a written notice of such request to the other Party. In addition to the procedures set forth in Section 15.2(c), the following shall apply with respect to any such Baseball Matter:

(i) Within [...***...] Business Days after appointment of the third arbitrator pursuant to Section 15.2(c), each Party shall deliver to the arbitrators and to the other Party its proposal regarding such Baseball Matter (each, a "**Proposal**"), as applicable, and a memorandum in support thereof. Within [...***...] Business Days after receipt of the other Party's Proposal and memorandum, each Party may submit to the arbitrators (with a copy to the other Party) a response to the other Party's Proposal. Except as directed by the arbitrators and in any event with both Parties present or participating, neither Party may have any other communications (either written or oral) with the arbitrators other than for the sole purpose of engaging the arbitrators.

(ii) Within [...***...] days after the receipt of the Proposals and memoranda from both Parties, the arbitrators shall select the Proposal provided by one Party (without modification) that the arbitrators believe is [...***...]. The arbitrators must select, as the only method to resolve such a Dispute, the Proposal of one Party and, absent agreement by the Parties, may not combine elements of both of the Parties' Proposals or terms or award any other relief or take any other action. The selection by the arbitrators of the applicable Proposal shall be binding and conclusive on the Parties.

(iii) With respect to any Dispute regarding [...***...], the arbitrators shall [...***...].

(e) <u>Survivability</u>. Any obligation to arbitrate under this Agreement shall remain in effect and be enforceable after termination of this Agreement for any reason.

15.3 <u>Governing Law</u>. This Agreement shall be governed by and construed under the substantive laws of the State of New York, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of this Agreement to the substantive law of another jurisdiction.

15.4 <u>Award</u>. Any award to be paid by one Party to the other Party as determined by the arbitrators pursuant to Section 15.2 shall be promptly paid in Dollars free of any Tax, deduction or offset; and any costs, expenses, fees or Taxes incident to enforcing the award shall, to the maximum extent permitted by Applicable Law, be charged against the Party resisting enforcement. Each Party agrees to abide by the award rendered in any arbitration conducted pursuant to this ARTICLE XV, and agrees that, subject to the U.S. Federal Arbitration Act, 9 U.S.C. §§ 1-16, judgment may be entered upon the final award in the U.S. Federal District Court for the Southern District of New York and that other courts may award full faith and credit to such judgment in order to enforce such award.

15.5 <u>Injunctive Relief</u>. Nothing in this ARTICLE XV will preclude either Party from seeking equitable relief or interim or provisional relief from a court of competent jurisdiction, including a temporary restraining order, preliminary injunction or other interim equitable relief, concerning a Dispute either prior to or during any arbitration if necessary to protect the interests of such Party or to preserve the status quo pending the arbitration proceeding. Therefore, in addition to its rights and remedies otherwise available at law, including the recovery of damages for breach of this Agreement, upon an adequate showing of material breach, and without further proof of irreparable harm other than this acknowledgement, such non-breaching Party shall be entitled to seek (a) immediate equitable relief, specifically including both interim and permanent restraining orders and injunctions, and (b) such other and further equitable relief as the court may deem proper under the circumstances. For the avoidance of doubt, nothing in this Section 15.5 shall otherwise limit a breaching Party's opportunity to cure a material breach as permitted in accordance with Section 14.4.

15.6 Jurisdiction. For the purposes of this ARTICLE XV, the Parties acknowledge their diversity (Gilead having its principal place of business in the State of California and Galapagos having its principal place of business in Belgium, each, as of the Execution Date), and except as provided in Section 15.7, agree to accept the jurisdiction of any United States District Court located in New York for the purposes of enforcing or appealing any awards entered pursuant to this ARTICLE XV and for enforcing the agreements reflected in this ARTICLE XV.

15.7 <u>Patent and Trademark Disputes</u>. Notwithstanding Section 15.2, any Dispute between the Parties or their respective Affiliates relating to the scope, validity, enforceability or infringement of any Patents or Trademarks covering the manufacture, use, importation, offer for sale or sale of Galapagos Molecules, Galapagos Products, Optioned Molecules or Optioned Products shall be submitted to a court of competent jurisdiction in the country in which such patent or trademark rights were granted or arose.

ARTICLE XVI

CLOSING CONDITIONS; EFFECTIVENESS

16.1 Antitrust Requirement.

(a) Following the Execution Date, both Parties shall file their respective Antitrust Filings required by the Antitrust Laws in respect of the transactions contemplated by this Agreement, the Subscription Agreement and the Filgotinib Amendment required by the Antitrust Laws in respect of the transactions contemplated by this Agreement, the Subscription Agreement and the Filgotinib Amendment as promptly as practicable with each applicable Antitrust Authority pursuant to any applicable Antitrust Laws, and in any event, with respect to notification and report forms filed with the FTC and DOJ pursuant to the HSR Act, if any, the Parties shall make such filings no later than [...***...] Business Days after the Execution Date. Each Party will be responsible for its own costs and expenses associated with any Antitrust Filing, but Gilead shall be responsible for payment of all fees to the FTC and DOJ with respect to Antitrust Filings made pursuant to the HSR Act. The Parties shall provide each other promptly with information and assistance as may be reasonably necessary and use reasonable efforts, in each case, to obtain prompt clearance required under applicable Antitrust Laws for the consummation of this Agreement and the transactions contemplated hereby and shall keep each other apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC, DOJ and each other applicable Antitrust Authority and shall comply promptly with any such inquiry or request; provided that neither Party shall be required to consent to the divestiture or other disposition of any of its or its Affiliates' assets, consent to any other material structural or conduct remedy or otherwise restrict or limit its or its Affiliates' freedom of action. Each Party shall instruct its counsel to cooperate with the other Party's counsel and use reasonable efforts to facilitate and expedite the identification and resolution of any such issues and, consequently, the expiration of the applicable HSR Act waiting period and any other applicable waiting period under Antitrust Laws, including, if requested by Gilead, seeking early termination of any such waiting period. Such reasonable efforts and cooperation include counsels' undertaking: (A) to keep each other appropriately informed of communications from and to personnel of the reviewing Antitrust Authority; and (B) to confer with each other regarding appropriate contacts with and response to personnel of the FTC, DOJ or other applicable Antitrust Authority.

(b) "Antitrust Clearance Date" means the first date on which each of the following criteria is met with respect to the Subscription Agreement, the Filgotinib Amendment and this Agreement: (i) the Parties shall have complied with all applicable requirements of the HSR Act (if any) and each other applicable Antitrust Law (if any); (ii) the waiting period under the HSR Act (if any) and each other applicable Antitrust Law (if any); shall have expired or earlier been terminated; (iii) no judicial or administrative proceeding opposing consummation of all or any part of this Agreement, the Subscription

Agreement or the Filgotinib Amendment shall be pending; (iv) no injunction (whether temporary, preliminary or permanent) prohibiting consummation of the transactions contemplated by this Agreement, the Subscription Agreement or the Filgotinib Amendment or any material portion of such agreement or amendment, as applicable, shall be in effect; and (v) no requirements or conditions shall have been formally requested or imposed by the DOJ, FTC, or any other Antitrust Authority in connection therewith that are not reasonably and mutually satisfactory to the Parties.

16.2 <u>Pre-Closing Negative Covenants</u>. During the period beginning on the Execution Date and ending on the Effective Date, Galapagos shall not, and shall cause its Affiliates not to, without the prior written consent of Gilead (such consent not to be unreasonably withheld, conditioned or delayed):

(a) enter into any agreement with any Third Party, whether written or oral, with respect to, or otherwise assign, transfer, license or convey its right, title or interest in or to, the Galapagos IP relating to any Galapagos Program or any Pre-Program Activities, in each case, in a manner that creates a material conflict with the rights granted or purported to be granted by Galapagos to Gilead under this Agreement;

(b) (i) sell, out-license or otherwise dispose of any assets or rights relating to any Galapagos Program or any Pre-Program Activities, in each case, in a manner that creates a material conflict with the rights granted or purported to be granted by Galapagos to Gilead under this Agreement, (ii) amend any agreements, licenses or other rights of Galapagos or any of its Affiliates relating to any Galapagos Program or any Pre-Program Activities, in each case, in a manner that creates a material conflict with the rights granted or purported to be granted by Galapagos to Gilead under this Agreement, or (iii) grant any security interest or otherwise encumber material assets and properties (including Galapagos IP, other than pursuant to the Security Agreement referenced in Section 16.5(f)), relating to any Galapagos Program or any Pre-Program Activities;

(c) (i) compromise, settle or agree to settle any litigation, dispute, action or other proceeding or institute any such litigation, dispute, action or other proceeding, in each case, concerning any Existing Galapagos Patents or any other Galapagos IP that is material to any Galapagos Program or any Pre-Program Activities, or (ii) fail to take any action necessary or advisable to protect or maintain any Galapagos IP that is material to any Galapagos or any of its Affiliates to commence any such litigation, dispute, action or other proceeding; or

(d) (i) enter into any material agreement relating to any Galapagos Program or any Pre-Program Activities or (ii) enter into any agreement pertaining to a merger, sale, acquisition, licensing, development, manufacturing, distribution, co-development, marketing or co-marketing arrangement, or any contract containing exclusivity provisions or restrictive covenants relating to any Galapagos Program or any Pre-Program Activities, in each case ((i) and (ii)), that creates a conflict with the rights granted or purported to be granted by Galapagos under this Agreement.

16.3 <u>Affirmative Covenants</u>. During the period beginning on the Execution Date and ending on the Effective Date, (a) Galapagos shall, and shall cause its Affiliates to comply with the covenants set forth in Section 11.5(c)(i) and Section 11.5(c)(ii) and (b) Galapagos shall promptly notify Gilead of the occurrence of a Key Product Event, and in no event more than [...***...] Business Days after such occurrence.

16.4 Key Product Event. If (a) a Key Product Event occurs for the Autotaxin Program or ADAMTS-5 Program after the Execution Date, [... ***...].

16.5 <u>Effectiveness</u>. Notwithstanding anything to the contrary in this Agreement, this ARTICLE XVI (Closing Conditions; Effectiveness), Sections 11.1 (Mutual Representations and Warranties) and 11.2 (Representations and Warranties of Galapagos), Section 14.2 (Outside Date) and Section 13.3 (Terms of Agreements) shall be binding upon the Parties as of the Execution Date. The remainder of this Agreement shall not take effect, and commencement of the Term shall not occur, until the last of the following conditions is met or waived by mutual agreement of the Parties with respect to Section 16.5(a) and Section 16.5(b), or by Gilead with respect to Section 16.5(c),and Section 16.5(f), or by Galapagos with respect to Section 16.5(d) in accordance with Applicable Law; *provided that* neither Party shall be permitted to assert the failure of any such condition that such Party has caused in order to prevent the occurrence of the Effective Date:

(a) The Antitrust Clearance Date shall have occurred.

(b) The Closing of the Subscription Agreement (as "Closing" is defined in the Subscription Agreement) shall have occurred or shall occur concurrently upon the Effective Date.

(c) Each of the representations and warranties of Galapagos set forth in Section 11.1 and Section 11.2 shall be [...***...] as of the Execution Date and the Effective Date in each case as qualified by the Schedule of Exceptions as of such date, and Galapagos shall have delivered a complete updated Galapagos Schedule of Exceptions to Gilead at least [...***...] Business Days prior to the Effective Date, with any updates to such Galapagos Schedule of Exceptions not having a material adverse effect on the rights granted to Gilead under this Agreement.

(d) Each of the representations and warranties of Gilead set forth in Section 11.1 and Section 11.3 shall be [...***...] as of the Execution Date and the Effective Date.

(e) Galapagos shall have delivered a certificate signed on its behalf by its Chief Executive Officer certifying whether or not a Key Product Event shall have occurred as of immediately prior to the Effective Date.

(f) Each Party shall have delivered or caused to be delivered on or before the Effective Date a duly executed copy of the Security Agreement.

ARTICLE XVII

MISCELLANEOUS

17.1 Entire Agreement; Amendment. This Agreement, including the Schedules hereto, and the Ancillary Agreements set forth the complete, final and exclusive agreement and all the covenants, promises, agreements, warranties, representations, conditions and understandings between the Parties with respect to the subject matter hereof and supersedes all prior agreements and understandings between the Parties existing as of the Effective Date with respect to the subject matter hereof, including the Existing Confidentiality Agreement. In the event of any inconsistency between any plan hereunder (including any R&D Plan and Budget, Global Commercialization Plan and Budget or Global Manufacturing Plan and Budget) and this Agreement, the terms of this Agreement shall prevail. There are no covenants, promises, agreements, warranties, representations, conditions or understandings, either oral or written, between the Parties other than as are set forth herein and therein. No subsequent alteration, amendment, change or addition to this Agreement shall be binding upon the Parties unless reduced to writing and signed by an authorized officer of each Party.

17.2 <u>No Third Party Beneficiary Rights</u>. Except as set forth in ARTICLE XII, this Agreement is not intended to and shall not be construed to give any Third Party any interest or rights (including any Third Party beneficiary rights) with respect to or in connection with any agreement or provision contained herein or contemplated hereby.

17.3 Force Majeure. Each Party shall be excused from the performance of its obligations under this Agreement to the extent that such performance is prevented or delayed by force majeure and the nonperforming Party promptly provides notice of the prevention to the other Party. Such excuse shall be continued so long as the condition constituting force majeure continues and the nonperforming Party takes reasonable efforts to remove such condition. For purposes of this Agreement, force majeure shall mean conditions beyond the control of the Parties, including an act of God, war, civil commotion, terrorist act, labor strike or lock-out, epidemic, failure or default of public utilities or common carriers, destruction of production facilities or materials by fire, earthquake, storm or like catastrophe, and failure of plant or machinery (*provided that* such failure could not have been prevented by the exercise of skill, diligence, and prudence that would be reasonably and ordinarily expected from a skilled and experienced person engaged in the same type of undertaking under the same or similar circumstances). Notwithstanding the foregoing, a Party shall not be excused from making payments owed hereunder because of a force majeure affecting such Party.

17.4 <u>Notices</u>. Any notice or other communication required or permitted to be given under this Agreement shall be in writing (whether or not specifically stated), shall specifically refer to this Agreement, and shall be addressed to the appropriate Party at the address specified below or such other address as may be specified by such Party in writing in accordance with this Section 17.4, and shall be deemed to have been given for all purposes (a) when received, if hand-delivered, sent by a reputable international expedited delivery service (with receipt confirmed) or facsimile (with transmission confirmed), or (b) [...***...] Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. This Section 17.4 is not intended to govern the day-to-day business communications necessary between the Parties in performing their obligations under the terms of this Agreement (for which e-mail or other methods of communications shall suffice).

If to Galapagos:

With a copy to (which shall not constitute notice):

Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: Chief Executive Officer Fax: [...***...]

Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: Legal Department Fax: [...***...]

With a copy to (which shall not constitute notice):

If to Gilead:

With a copy to (which shall not constitute notice):

Baker & McKenzie LLP 452 Fifth Avenue New York, New York 10018 United States Attention: Olivia Tyrrell Oren Livne Fax: [...***...]

Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 Attention: Alliance Management

Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 Attention: General Counsel

Covington & Burling LLP Salesforce Tower 415 Mission Street, Suite 5400 San Francisco, CA 94105-2533 Attention: Amy L. Toro, Esq. Facsimile number: [...***...]

17.5 <u>No Strict Construction; Headings</u>. This Agreement has been prepared jointly and shall not be strictly construed against either Party. Ambiguities, if any, in this Agreement shall not be construed against either Party, irrespective of which Party may be deemed to have authored the ambiguous provision. The headings of each Article and Section in this Agreement have been inserted for convenience of reference only and are not intended to limit or expand on the meaning of the language contained in the particular Article or Section.

17.6 Assignment, Change of Control.

(a) Neither Party may assign or transfer this Agreement or any rights or obligations hereunder without the prior written consent of the other Party, except that [...***...]. Any permitted successor or assignee of rights or obligations hereunder shall, in a writing to the other Party, expressly assume performance of such rights or obligations (and in any event, any Party assigning this Agreement to an Affiliate shall remain bound by the terms and conditions hereof). Any permitted assignment shall be binding on the successors of the assigning Party. Any assignment or attempted assignment by either Party in violation of the terms of this Section 17.6 shall be null, void and of no legal effect.

(b) If Galapagos undergoes a Change of Control, then (i) Patents, Information, Molecules and products existing as of the effective date of such Change of Control that are owned or controlled, by Persons that are Affiliates of Galapagos as a result of such Change of Control (collectively, the "Galapagos Acquirer") and (ii) Patents, Information, Molecules, and products that are conceived, discovered, developed, generated or otherwise made by or on behalf of the Galapagos Acquirer or that become owned or controlled by the Galapagos Acquirer after such Change of Control, in each case ((i) and (ii)), [...***...].

(c) <u>Performance by Affiliates</u>. Subject to the limitations of Section 8.6, each Party may discharge any obligations and exercise any right hereunder through any of its Affiliates. Each Party hereby guarantees the performance by its Affiliates of such Party's obligations under this Agreement, and shall cause its Affiliates to comply with the provisions of this Agreement in connection with such performance. Any breach by a Party's Affiliate of any of such Party's obligations under this Agreement shall be deemed a breach by such Party, and the other Party may proceed directly against such Party without any obligation to first proceed against such Party's Affiliate.

17.7 Subcontractors. Without the prior written approval of the other Party, and subject to the terms of any Ancillary Agreement, each Party may itself or any of its Affiliates, Sublicensees or subcontractors may exercise such Party's rights or perform such Party's obligations under this Agreement through one or more (sub)contractors or consultants; *provided that* (a) such Party remains responsible for the work allocated to, and payment to, such (sub)contractors and consultants to the same extent it would if it had done such work itself; (b) such Party conducts appropriate risk-based due diligence to assess the capabilities, compliance and reputation of such subcontractors or consultants; (c) the (sub)contractor or consultant undertakes in writing obligations of confidential Information pursuant to ARTICLE XIII hereof; and (d) (i) with respect to any subcontracting of any Development or Manufacturing activities other than Independent Activities, such Party requires that the subcontractor or consultant undertakes in writing to assign or exclusively license back (with the right to sublicense) to such Party all intellectual property is conceived, discovered, developed, generated or otherwise made by or on behalf of such Subcontractor or consultant in the course of performing any such work and (ii) with respect to (sub)contracting of any other activities, such Party uses reasonable efforts to provide that the (sub)contractor or consultant undertakes in writing to assign or exclusively license) to such Party all intellectual property sic noceived, discovered, developed, generated or otherwise made by or on behalf of such Party uses reasonable efforts to provide that the (sub)contractor or consultant undertakes in writing to assign or exclusively license back (with the right to sublicense) to such Party with respect to Galapagos Products or Optioned Molecules or Optioned Molecules or Optioned Products which intellectual property is conceived, discovered, developed, generated or otherwise made by or on behalf of such

17.8 <u>Compliance with Applicable Law</u>. Each Party shall comply with Applicable Law in the course of performing its obligations or exercising its rights pursuant to this Agreement.

17.9 <u>Further Actions</u>. Each Party agrees to execute, acknowledge and deliver such further instruments, and to do all such other acts, as may be necessary or appropriate in order to carry out the purposes and intent of this Agreement.

17.10 <u>Severability</u>. If any one or more of the provisions of this Agreement is held to be invalid or unenforceable by an arbitrator or by any court of competent jurisdiction from which no appeal can be or is taken, the provision shall be considered severed from this Agreement and shall not serve to invalidate any remaining provisions hereof. The Parties shall make a good faith effort to replace any invalid or unenforceable provision with a valid and enforceable one such that the objectives contemplated by the Parties when entering into this Agreement may be realized.

17.11 <u>No Waiver</u>. Any delay in enforcing a Party's rights under this Agreement or any waiver as to a particular default or other matter shall not constitute a waiver of such Party's rights to the future enforcement of its rights under this Agreement, except with respect to an express written and signed waiver relating to a particular matter for a particular period of time.

17.12 <u>Independent Contractors</u>. Each Party shall act solely as an independent contractor, and nothing in this Agreement shall be construed to give either Party the power or authority to act for, bind, or commit the other Party in any way. Nothing herein shall be construed to create the relationship of partners, principal and agent, or joint-venture partners between the Parties.

17.13 <u>Counterparts</u>. This Agreement may be executed in one (1) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Signature pages of this Agreement may be exchanged by email or in .pdf or other electronic means without affecting the validity thereof.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have executed this Agreement in duplicate originals by their duly authorized officers as of the Execution Date.

	GALAPAGOS NV		GILEAD SCIENCES, INC.
By:		By:	
Name:		Name:	
Title:		Title:	
By:		By:	
Name:		Name:	
Title:		Title:	

[SIGNATURE PAGE TO OPTION, LICENSE AND COLLABORATION AGREEMENT]

APPENDIX A

DEFINITIONS

As used in this Agreement, the following initially capitalized terms, whether used in the singular or plural form, shall have the meanings set forth in this <u>Appendix A</u>. In addition, the terms "includes," "including," "include" and derivative forms of them shall be deemed followed by the phrase "without limitation" (regardless of whether it is actually written (and drawing no implication from the actual inclusion of such phrase in some instances after such terms but not others)) and the term "or" has the inclusive meaning represented by the phrase "and/or" (regardless of whether it is actually written (and drawing no implication from the actual use of the phrase "and/or" in some instances but not in others)). Unless specified to the contrary, references to Articles, Sections or Schedules shall refer to the particular Articles, Sections or Schedules of or to this Agreement and references to this Agreement include all Schedules hereto. The word "day," "quarter" or "year" (and derivatives thereof, *e.g.*, "quarterly") shall mean a calendar day, calendar quarter or calendar year unless otherwise specified. The word "hereof," "herein," "hereby" and derivative or similar word refers to this Agreement (including any Schedules). The words "will" and "shall" shall have the same obligatory meaning. Provisions that require that a Party or Parties hereunder "agree," "consent" or "approve" or the like shall require that such agreement, consent or approval be specific and in writing, whether by written agreement, letter or otherwise. Words of any gender include the other gender. Words using the singular or plural number also include the plural or singular number, respectively. References to any specific law or article, section or other division thereof, shall be deemed to include the thencurrent amendments or any replacement law thereto, and any rules and regulations promulgated thereunder.

"Access Territory" means, with respect to an Optioned Product, any and all countries and territories where Gilead (itself or through its Affiliates) has publicly announced a policy to generally sell or otherwise make available such Optioned Product and one or more other Gilead products at a significantly discounted price to patients in such countries or territories. The list of countries and territories included in the Access Territory as of the Execution Date is set forth on Schedule A, which list shall be updated by Gilead on at least an annual basis.

"Acquired Galapagos Program" means a Galapagos Program with respect to which Galapagos or any of its Affiliates acquires rights (whether such acquisition occurs prior to the date on which such acquired program becomes a Galapagos Program or thereafter) as the result of, other than a Change of Control of Galapagos, any license, merger, acquisition, reorganization, consolidation or combination or any other transaction after the Execution Date.

"Active Gilead Program" has the meaning set forth in Schedule 8.10-1.

"ADAMTS-5 Existing Trial" means the Clinical Trial titled "A Study to Assess Efficacy and Safety of GLPG1972/S201086 in Patients With Knee Osteoarthritis (Roccella)" (ClinicalTrial.gov Identifier NCT03595618) that is ongoing as of the Execution Date.

"ADAMTS-5 Program" means the Galapagos Program ongoing as of the Execution Date, as it is continued thereafter, including as an Optioned Program, which program includes GLPG1972, an ADAMTS-5 inhibitor, as the Lead Molecule.

"Affiliate" means, with respect to a particular Person, a person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such Person, for so long as such control exists, regardless of whether such Person is or becomes an Affiliate on or after the Effective Date. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms

"controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, by contract or otherwise. Notwithstanding the foregoing, Fidelta is deemed not to be an Affiliate of Galapagos and no material, Information, databases, Patents, Trademarks, promotional materials, regulatory materials or regulatory approvals owned or controlled by Fidelta shall be considered Controlled by Galapagos or otherwise subject to this Agreement; *provided* that, the foregoing exclusion of Fidelta shall no longer apply on or after such time that Fidelta (a) [...***...] (b) [...***...].

"Affordable Basis" means, with respect to an Optioned Product in the Access Territory, selling or otherwise making such Optioned Product available to patients at a price where the revenue per unit shall not exceed [...***...].

"Ancillary Agreement" means (a) any agreement entered into by the Parties or their designated Affiliates pursuant to this Agreement and (b) any other agreement in effect between the Parties or their designated Affiliates which specifies that it is an "Ancillary Agreement" as defined under this Agreement, but excluding the Subscription Agreement, the Filgotinib Agreement and the Security Agreement.

"Antitrust Approval" means, as the context requires, any consent, approval or other authorization required under the applicable Antitrust Laws from the applicable Antitrust Authority to effect either, (a) Gilead's exercise of an Option with respect to a Galapagos Program or (b) the transactions contemplated by the Subscription Agreement, the Filgotinib Amendment or this Agreement (including any prospective exercise by Gilead of an Option under this Agreement).

"Antitrust Authority" means any applicable Governmental Authority exercising authority with respect to any Antitrust Laws.

"Antitrust Condition" means with respect to any subsequent Option Exercise Closing, as applicable, that (a) all waiting periods (and any extension thereof) applicable to Gilead's exercise of such Option pursuant to Section 8.2(b) and Section 8.2(d) under any and all applicable Antitrust Laws shall have expired or been terminated, and (b) if applicable, any applicable Antitrust Approvals necessary for the exercise of such Option under such Antitrust Laws shall have been received.

"Antitrust Filing" means, as the context requires a filing or notification, together with all required documentary attachments thereto, by the Parties with or to the applicable Antitrust Authority as required by the Antitrust Laws with respect to (a) Gilead's exercise of an Option with respect to a Galapagos Program pursuant to Section 8.2(b)(viii) and Section 8.2(d), or (b) the transactions contemplated by the Subscription Agreement or this Agreement.

"Antitrust Laws" means any Applicable Law governing merger control, competition, monopolies or restrictive trade practices, including the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"**Applicable Law**" means the applicable laws, rules and regulations, including any rules, regulations, guidelines or other requirements of Governmental Authorities, including Regulatory Authorities, that may be in effect from time to time, and including, with respect to Pre-Program Activities, Galapagos Programs and Optioned Programs, GCP, GLP and GMP to the extent applicable to any such activities or programs.

"**Autotaxin Program**" means the Galapagos Program ongoing as of the Execution Date, as it is continued thereafter, including as an Optioned Program, which program includes GLPG1690, an autotaxin inhibitor, as the Lead Molecule. For clarity, the Autotaxin Program shall be a Galapagos Program as of the Execution Date and an Optioned Program as of the Effective Date, except that for purposes of Section 11.1 and Section 11.2, the Autotaxin Program shall be treated as a Galapagos Program as of the Effective Date.

"Backup" [...***...].

"Baseball Matters" [...***...].

"Belgian Companies and Associations Code" means the Belgian Companies and Associations Code of 23 March 2019, as amended from time to time, and the rules and regulations promulgated thereunder.

"Belgian Companies Code" means the Belgian Companies Code of 7 May 1999, as amended from time to time, and the rules and regulations promulgated thereunder.

"**BLA**" means a biologics license application for Regulatory Approval of a Galapagos Product or Optioned Product that is filed with FDA under Section 351 of the Public Health Service Act, including all amendments and supplements to any such application, and any equivalent application, amendment or supplement to the equivalent Regulatory Authority in any other regulatory jurisdiction.

"**Business Day**" means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States, (c) a bank or other public holiday in Brussels, Belgium, (d) a bank or other public holiday in Dublin, Ireland or (e) the period commencing on December 25th and ending on January 1st (inclusive).

"Change of Control" means, with respect to Galapagos, a change of control of Galapagos as defined and determined in accordance with the provisions of the Belgian Companies Code, and (as of from the application of the relevant provisions of such code to Galapagos) the Belgian Companies and Associations Code.

"Clinical Data" means all data and results with respect to any Galapagos Molecule or Galapagos Product or Optioned Molecule or Optioned Product made, collected or otherwise generated under or in connection with the conduct of Clinical Trials.

"Clinical Trial" means any human clinical trial of a Galapagos Product or Optioned Product.

"**CMC Activities**" means those Manufacturing activities and regulatory activities designed to support preparation of the Chemistry, Manufacturing and Controls sections of any Regulatory Materials or Regulatory Approval.

"Collaboration IP" means Collaboration Know-How and Collaboration Patents.

"**Collaboration Know-How**" means Information that is conceived, discovered, developed, generated or otherwise made by or on behalf of either Party or its Affiliates, solely or jointly, during the Term, in performing activities under this Agreement (including, for clarity, any Pre-Program Activities or any activities with respect to any Galapagos Program or Optioned Program, including any Independent Activities and, for clarity, excluding the performance of activities under a separate written agreement).

"Collaboration Patent" means any Patent claiming or covering Collaboration Know-How.

"Collaboration Term" means the period beginning on the Effective Date and ending on the tenth anniversary of the Effective Date.

"**Combination Product Activities**" means any activities by either Party or its Affiliates relating to any of the molecules that are the subject of the Pre-Program Activities, Galapagos Molecules or Optioned Molecules, which activities involve use of filgotinib or any active pharmaceutical ingredient that is owned or controlled by Gilead or any of its Affiliates (including, for clarity, any Gilead Combination Product).

"**Commercialization**" means (a) any and all activities directed to marketing, promoting, distributing, importing, exporting, using, offering to sell, selling or having sold a product, including activities related to the commercial manufacture, marketing, promotion, sale or distribution of a product in the Territory, and (b) Medical Affairs Activities. Commercialization shall include commercial activities conducted in preparation for a product launch. "**Commercialize**" has a correlative meaning.

"**Commercially Reasonable Efforts**" means, with respect to the Development, Manufacture or Commercialization of an Optioned Product, that level of efforts and resources commonly dedicated in the research-based pharmaceutical industry by a company to the analogous development, manufacture or commercialization activities of a product of similar commercial potential at a similar stage in its lifecycle, [...***...].

"**Committee**" means (a) the Joint Steering Committee, the Joint Development Committee, the Joint Commercialization Committee, or the Joint Communications Review Committee or (b) any other committee established by the Parties pursuant to Section 1.8. For clarity, the Patent Prosecution Committee and Patent Litigation Committee are not intended, and shall not be construed, to be a "Committee".

"**Completion Date**" means, with respect to a Clinical Trial, the earlier of (a) the date of completion of the final study report for such Clinical Trial or (b) the [...***...] day after final database lock for such Clinical Trial.

"**Compulsory License**" means, with respect to an Optioned Product and a country or territory, a license or rights granted to a Third Party by a Governmental Authority for such country or territory to sell or offer for sale such Optioned Product in such country or territory under any patent rights owned or controlled by Gilead or its Affiliates, without direct or indirect authorization from Gilead or its Affiliates, for example a right granted pursuant to requests under the 30 August 2003 WTO decision.

"**Control**" means, with respect to any material, Information, databases, Patent, Trademark, Global Promotional Materials, Regulatory Materials or Regulatory Approvals, the possession (whether by ownership or license (other than by operation of the licenses and other rights granted in Sections 8.1, 8.3, 8.4 or 10.9(c))) by a Party or its Affiliates of the ability to grant to the other Party a license, right of reference or other right as provided herein to such item, without violating the terms of any agreement or other arrangement with any Third Party, or being obligated to pay any royalties or other consideration therefor (other than royalties or other consideration shared by the Parties pursuant to Section 9.6), in existence as of the time such Party or its Affiliates would first be required hereunder to grant the other Party such license, right of reference or other right. For clarity, [...***...].

"**Cover**," "**Covering**" or "**Covered**" means, with respect to a Patent, in the absence of a license to a Valid Claim included in such Patent, the applicable activity, or, to the extent the applicable activity is not specified, the Exploitation of the applicable invention, discovery, process or product, would infringe such Valid Claim (or, in the case of a Valid Claim that has not yet issued, would infringe such Valid Claim if it were to issue).

"CPI" means the Consumer Price Index for the U.S. City Average (all times).

"Data Room" means an electronic data room hosted by a Third Party vendor reasonably acceptable to Gilead.

"**Development**" means (a) conducting research or development for a Galapagos Program or Optioned Program, including with respect to any Galapagos Molecule, Galapagos Product or Galapagos Target or Optioned Molecule, Optioned Product or Target that is the subject of an Optioned Program; (b) obtaining or maintaining Regulatory Approval of a Galapagos Product or Optioned Product for one or more indications; or (c) developing the process for the Manufacture of clinical and commercial quantities of a Galapagos Product or Optioned Product. "Development" includes (i) the conduct of Nonclinical Studies and Clinical Trials (including Phase 4 Clinical Trials) and (ii) the preparation, submission, review and development of data or information in support of a submission to a Regulatory Authority to obtain or maintain Regulatory Approval of a Galapagos Product or Optioned Product, but excluding Commercialization (including the Manufacture and accumulation of commercial inventory of a Galapagos Product or Optioned Product). "**Develop**" has a correlative meaning.

"**Development Manufacturing Costs**" means, with respect to an Optioned Product (or placebo or comparator if required for the applicable Clinical Trial pursuant to this Agreement), FTE Costs incurred by either Party or any of its Affiliates and all out-of-pocket costs and expenses incurred by or on behalf of either Party or any of its Affiliates, in each case, in Manufacturing such Optioned Product (or placebo or comparator) for Development activities, including costs and expenses incurred in connection with (a); (b) [...***...]; (c) [...***...] (e) [...***...]; (f) [...***...]; (g) [...***...]; (h) [...***...] and (i) [...***...]. "Development Manufacturing Costs" shall further include:

(i) [...***...]; and

(ii) [...***...].

"Directed To" means, with respect to any Molecule and a Target, that such Molecule (a) [...***...] and (b) [...***...].

"**Distributor**" means a Third Party appointed by a Party or any of its Affiliates to distribute, market and sell any Optioned Product in a specified country or region, which Third Party (a) purchases all of its requirements of such Optioned Product from such Party or its Affiliates or its or their Sublicensees and (b) is not a sublicensee of such Party or any of its Affiliates or its or their Sublicensees under the rights granted to such Party by the other Party pursuant to Section 8.1, 8.3, 8.4 or 10.9(c), as applicable (except to the extent licensed or sublicensed to finish or package such Optioned Product).

"Dollars" or "\$" means the lawful currency of the United States.

"Effective Date" means the date on which the last of the conditions set forth in Section 16.5 is satisfied.

"EMA" means the European Medicines Agency or its successor.

"European Union" means all of the European Union member states as its membership may be constituted from time to time,.

"Excluded Programs" means (a) the Filgotinib Collaboration Program; (b) any Galapagos Program that does not become an Optioned Program before the end of the Galapagos Program Period with respect thereto, but solely after such Galapagos Program Period has ended; (c) any program or activities with respect to a Galapagos Molecule or Galapagos Product that become an Excluded Program pursuant to any of Sections 2.2, 2.4, 2.5, 2.6, or 8.2(e), or as provided on Schedule 8.10 or [...***...] or otherwise under this Agreement; (d) any Pre-Program Activity, Galapagos Program or Optioned Program with respect to which this Agreement is terminated pursuant to ARTICLE XIV; (e) any programs or activities of Fidelta for so long as [...***...], and (f) the programs set forth on Schedule B; provided, that, with respect to any of the programs set forth on Schedule B, if any time after the Execution Date, Galapagos is able to grant to Gilead the rights contemplated hereunder without violating any agreement with a Third Party to which Galapagos or any of its Affiliates is a party to as of the Execution Date, then such program shall cease to be an Excluded Program and shall become an Acquired Galapagos Program (or, if not the subject of an IND filed with a Regulatory Authority at such time, will become an Acquired Galapagos Program if and when an IND for such program is filed with a Regulatory Authority).

"Executive Officer" means, with respect to Galapagos, its Chief Executive Officer, and with respect to Gilead, the Chief Executive Officer of Gilead Sciences, Inc.

"Existing Confidentiality Agreement" means that certain Mutual Confidential Disclosure Agreement entered into by Gilead and Galapagos, dated [...***...] and amended as of [...***...] and [...***...].

"Existing Galapagos Patents" means, as of the Effective Date or Execution Date, as applicable, the Galapagos Patents existing as of such date.

"Existing Galapagos Third Party Agreement" means the agreements set forth on Schedule C.

"Existing Galapagos Third Party Obligations Schedule" means the obligations set forth on the Existing Galapagos Third Party Obligations Schedule.

"Existing Galapagos Third Party Obligations Schedule" means the schedule attached hereto as Schedule C, as such schedule may be amended from time to time pursuant to Section 8.13.

"**Exploit**" means, collectively, research, develop, use, manufacture, have manufactured, sell, have sold, offer for sale, commercialize, import, have imported, distribute, have distributed, export, have exported and otherwise exploit (including, for clarity, to Develop or Commercialize, including to Manufacture therefor). "**Exploitation**" has a correlative meaning.

"FD&C Act" means the United States Federal Food, Drug and Cosmetic Act, as amended.

"FDA" means the United States Food and Drug Administration or its successor.

"Fidelta" means Fidelta d.o.o., a Croatian limited liability company.

"Field" means all uses.

"Filgotinib Agreement" means that certain License and Collaboration Agreement by and between Gilead Biopharmaceutics Ireland UC and Galapagos dated as of December 16, 2015, as amended concurrent herewith (such amendment to take effect on the Effective Date) ("Filgotinib Amendment").

"Filgotinib Collaboration Program" means the activities governed by the Filgotinib Agreement, including such activities with respect to the Licensed Compound and Licensed Product(s) (each, as defined in the Filgotinib Agreement).

"Final Term Extension" [...***...].

"**First Commercial Sale**" means, with respect to an applicable product in a country, the first sale in an arm's length transaction to a Third Party by or on behalf of a Party or any of its Affiliates or Sublicensees in the Field, other than for Veterinary Uses, in such country following Regulatory Approval of such product in such country. For the avoidance of doubt, a first sale for compassionate use or named patient program sales shall not constitute a First Commercial Sale for purposes of this Agreement.

"FTE" means the equivalent of the work of one (1) employee full time for one (1) calendar year (consisting of at least a total of [...***...] hours per calendar year) of work directly related to the Development of an Optioned Product. No additional payment shall be made with respect to any person who works more than [...***...] hours per calendar year and any person who devotes less than [...***...] hours per calendar year (or such other number as may be agreed by the JDC in the case of the R&D Plan and Budget) shall be treated as an FTE on a pro rata basis based upon the actual number of hours worked divided by [...***...].

"**FTE Costs**" means, with respect to a Party for any period, the applicable FTE Rate multiplied by the applicable number of FTEs of such Party or its Affiliates performing the applicable activities during such period in accordance with the applicable R&D Plan and Budget or the Global Manufacturing Plan and Budget.

"FTE Rate" means [...***...] Dollars [...***...], increased or decreased on January 1 of each calendar year beginning January 1, 2020, [... ***...].

"GAAP" means, in the case of Gilead, United States Generally Accepted Accounting Principles and in the case of Galapagos, International Financial Reporting Standards, in each case as consistently applied by a Party in its accounting practices.

"GLPG1690" means Galapagos' autotaxin inhibitor that, as of the Execution Date, is in Phase 3 Clinical Trials for idiopathic pulmonary fibrosis, the structure of which inhibitor is set forth on Schedule D.

"GLPG1972" means Galapagos' ADAMTS-5 inhibitor that, as of the Execution Date, is in Phase 2 Clinical Trials for osteoarthritis, the structure of which inhibitor is set forth on Schedule E.

"Galapagos Background IP" means the Galapagos Background Know-How and the Galapagos Background Patents.

"Galapagos Background Know-How" means, with respect to each Galapagos Program or Optioned Program, or any Pre-Program Activities, Information Controlled by Galapagos as of the Effective Date or during the Term that is reasonably necessary or useful to Exploit any applicable Galapagos Molecule or Galapagos Product or Optioned Molecule or Optioned Product or Molecule or product that is the subject of such Pre-Program Activities, excluding any Collaboration Know-How.

"Galapagos Background Patents" means, with respect to each Galapagos Program or Optioned Program, or any Pre-Program Activities, Patents Controlled by Galapagos as of the Effective Date or during the Term that (a) claim or cover Galapagos Background Know-How for such Galapagos Program or Optioned Program or Pre-Program Activities or (b) are otherwise reasonably necessary or useful to Exploit any applicable Galapagos Molecule or Galapagos Product or Optioned Molecule or Optioned Product or a Molecule or product that is the subject of such Pre-Program Activities, excluding, in each case ((a) and (b)), any Collaboration Patents. For clarity, the Galapagos Background Patents include the Existing Galapagos Patents. "Galapagos Foreground Know-How" means Collaboration Know-How conceived, discovered, developed, generated or otherwise made solely by or on behalf of Galapagos or its Affiliates, excluding any Galapagos Program Period Know-How.

"Galapagos Foreground Patents" means Collaboration Patents (a) claiming or covering Galapagos Foreground Know-How and (b) not claiming or covering Galapagos Program Period Know-How, Gilead Collaboration Know-How or Joint Collaboration Know-How.

"Galapagos IP" means the Galapagos Know-How and Galapagos Patents.

"Galapagos Know-How" means the Galapagos Background Know-How and Galapagos Collaboration Know-How, in each case to the extent Controlled by Galapagos or its Affiliates.

"Galapagos Molecules" means, for each Galapagos Program, (a) [...***...], (b) [...***...] and (c) [...***...].

"Galapagos Option Exercise Representations" means the representations and warranties set forth in Schedule F that Galapagos shall make as of each Option Bringdown Date with respect to each Optioned Program.

"Galapagos Option Schedule of Exceptions" means the schedule of exceptions that Galapagos shall deliver with respect to the Galapagos Option Exercise Representations in connection with each Option Exercise Closing. Any information disclosed in one schedule to the Galapagos Option Schedule of Exceptions shall be deemed to be disclosed with respect to, and shall be deemed to apply to qualify, all other representations and warranties of Galapagos to the extent the relevance of such item to such other representations and warranties is reasonably apparent.

"Galapagos Patents" means the Galapagos Background Patents and Galapagos Collaboration Patents, in each case to the extent Controlled by Galapagos or its Affiliates.

"Galapagos Product" means any product [...***...].

"Galapagos Program" means any program Controlled by Galapagos or any of its Affiliates (including any research or development program and any commercial program) that exists as of the Execution Date or thereafter comes into existence at any time prior to the end of the Collaboration Term (including any Acquired Galapagos Program upon the date of its acquisition), other than an Optioned Program or Excluded Program, [...***...] (such Molecule, the "Lead Molecule"). Any Galapagos Program shall be deemed to include any Related Molecule(s) and Backups with respect to the applicable Lead Molecule. The Galapagos Programs existing as of the Execution Date are set forth on Schedule G.

"Galapagos Program Period Know-How" means, (a) for each Galapagos Program, Collaboration Know-How to the extent (i) related to a Galapagos Molecule or Galapagos Product within such Galapagos Program and (ii) conceived, discovered, developed, generated or otherwise made during the Galapagos Program Period for such Galapagos Program and (b) any Information deemed to be Galapagos Program Period Know-How (including pursuant to Section 2.2).

"Galapagos Program Period Patents" means, for each Galapagos Program, Collaboration Patents claiming or covering Galapagos Program Period Know-How for such Galapagos Program and not claiming or covering Gilead Collaboration Know-How or Joint Collaboration Know-How.

"Galapagos Target" means, for any Galapagos Program, a Target that is the subject of such Galapagos Program.

"Galapagos Territory" means, with respect to each Optioned Program at any time, subject to Section 3.7(c)(iii), Section 8.2(e) and Section 14.3, (a) the countries set forth on Schedule H and (b) each country for which a required Antitrust Approval has not been obtained with respect to such Optioned Program as of the most recent Option Exercise Closing for such Optioned Program; *provided that*, with respect to the [...***...], the Galapagos Territory includes (x) the countries in subclause (a) and (y) any country not in subclause (a) for which, at the applicable time, [...***...]. For clarity, as of the Effective Date, the Galapagos Territory with respect to the [...***...] Program including all countries in the world other than the United States.

"Generic Product" means, with respect to a product, a generic version of a product containing the same active Molecule as such product that is approved for marketing by a Third Party in a given country either: (a) pursuant to Section 505(j) of the FD&C Act (21 U.S.C. 355(j)), 505(b)(2) of the FD&C Act (21 U.S.C. 355(b)(2)), 42 USC 262(k), or a foreign equivalent of any of the foregoing, by reference to a Marketing Approval of such product, or (b) pursuant to any other Applicable Law where such approval is based on a demonstration of bioequivalence or biosimilarity to such product. Under no circumstances shall a product be considered a Generic Product if Gilead or its Affiliates or its Sublicensees (i) transferred a Marketing Approval of such product to such Third Party, (ii) transferred an application for Marketing Approval of such product to such Third Party, or (iii) provided a right of reference to such Third Party in order to enable such Third Party to commercialize such product, except that this exclusion shall not apply if the right of reference is limited to qualifying the generic version for a Governmental Authority's program for providing medicines at no or low cost to countries in the Access Territory.

"Gilead Collaboration IP" means the Gilead Collaboration Know-How and Gilead Collaboration Patents.

"Gilead Collaboration Know-How" means Collaboration Know-How conceived, discovered, developed, generated or otherwise made solely by or on behalf of Gilead or its Affiliates, excluding any Galapagos Program Period Know-How.

"Gilead Collaboration Patents" means Collaboration Patents (a) claiming or covering Gilead Collaboration Know-How and (b) not claiming or covering Galapagos Foreground Know-How, Galapagos Program Period Know-How or Joint Collaboration Know-How.

"Gilead Combination Product" means a pharmaceutical product containing a Galapagos Molecule or Optioned Molecule In Combination with one or more other active pharmaceutical ingredients that are owned or controlled by Gilead or any of its Affiliates, in any and all finished forms, presentations, delivery systems, strength, dosages, and formulations.

"Gilead Territory" means, for the applicable Optioned Program at any time, all countries in the world other than the countries included in the Galapagos Territory with respect to such Optioned Program at such time.

"Good Clinical Practice" or "GCP" means the then-current standards for Clinical Trials for pharmaceuticals or biologics set forth in the ICH Guideline for Good Clinical Practices, as amended from time to time, FDA regulations set forth under Title 21 of the C.F.R. Parts 50, 54, 56 and 312 (as amended from time to time) together with related FDA guidance, and such standards of good clinical practice as are required by the European Union and other organizations and Governmental Authorities in countries in which any Clinical Trial is conducted, to the extent such standards are not less stringent than the ICH guidelines.

"Good Laboratory Practice" or "GLP" means the then current standards for laboratory activities for pharmaceuticals or biologics, as set forth in the FDA's GLP regulations as set forth under Title 21 of the C.F.R. Part 58, or the GLP principles of the Organization for Economic Co-Operation and Development (OECD), as amended from time to time, and such standards of good laboratory practice as are required by the European Union and other organizations and Governmental Authorities in countries in which any Clinical Trial is conducted, to the extent such standards are not less stringent than the FDA GLP regulations.

"Good Manufacturing Practice" or "GMP" means all current regulatory requirements that apply to the manufacture of active ingredients and pharmaceutical or biologic products, including the regulations set forth under Title 21 of the C.F.R., Parts 210, 211 and 600, as may be amended from time to time, as well as applicable guidance published by the FDA from time to time, and such standards of good manufacturing practice as are required in the European Union, and foreign equivalents, in each case, as applicable to any Clinical Trial.

"Governmental Authority" means any multi-national, federal, state, local, municipal or other government authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, court or other tribunal, as well as any securities exchange or securities exchange authority).

"HSR Act" means the U.S. Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"**In Combination**" means, with respect to any Galapagos Molecule or Optioned Molecule and any other active pharmaceutical ingredient(s), that such Galapagos Molecule or Optioned Molecule and other active pharmaceutical ingredient(s) are included in a product that is sold either as a fixed dose combination or with separate doses in a single package.

"**IND**" means (a) an investigational new drug application as described in the FD&C Act and applicable regulations promulgated thereunder by the FDA, including all amendments and supplements to any such application or (b) the equivalent application to the equivalent Regulatory Authority in any other regulatory jurisdiction, including all amendments and supplements to any such application, the filing of which is necessary to initiate a Clinical Trial of a pharmaceutical product in humans in such jurisdiction.

"Information" means any data, results, and information of a scientific or technical nature, in any tangible or intangible form, including know-how, trade secrets, practices, techniques, methods, processes (including manufacturing processes (including for active pharmaceutical ingredients and drug products)), inventions, developments, specifications, formulations, formulae, materials or compositions of matter of any type or kind (patentable or otherwise), software, algorithms, Clinical Trial and Nonclinical Study reports, technology, test data including pharmacological, biological, chemical, biochemical, toxicological, and clinical test data, analytical and quality control data, stability data, studies and procedures, excluding any Regulatory Materials and Regulatory Approvals (but, for clarity, not excluding any such data, results or information of a scientific or technical nature contained in any Regulatory Materials or Regulatory Approvals).

"Initiation" means, with respect to a Clinical Trial, the date of [...***...] in such Clinical Trial. "Initiate" shall have a corresponding meaning.

"**Invalidity or Unenforceability Action**" means, with respect to any Patent claim, any written allegation of invalidity or unenforceability of such Patent claim by a Third Party, including (a) in a declaratory judgment action; (b) as a defense or counterclaim to a suit or other action enforcing such Patent; or (c) in any proceeding originating in a patent office, including any opposition proceeding, *inter partes* review proceeding, post grant review proceeding, interference proceeding, reissue proceeding, reexamination proceeding or other post-grant proceeding originating in a patent office. "Joint Collaboration Know-How" means Collaboration Know-How conceived, discovered, developed, generated or otherwise made jointly by or on behalf of both Parties or their respective Affiliates, excluding Galapagos Program Period Know How.

"Joint Collaboration Patents" means Collaboration Patents claiming or covering (a) Joint Collaboration Know-How or (b) both (i) Galapagos Foreground Know-How or Galapagos Program Period Know-How, on the one hand and (ii) Gilead Collaboration Know-How, on the other hand.

"Key Product Event" means any event with respect to [...***...] that: (a) [...***...]; and (b) [...***...].

"Knowledge" [...***...].

"**Major Market Material Regulatory Communication**" means any material communications from the European Medicines Agency or European Commission or any Regulatory Authority in a Major Market, including with respect to: Clinical Trial protocols and amendments thereto, meeting requests and materials, requests for information and responses thereto, clinical hold notices, investigator's brochures, and Marketing Authorization Application submissions.

"Major Markets" means the following countries: [...***...].

"Manufacture" means, with respect to a Galapagos Molecule or Galapagos Product or Optioned Molecule or Optioned Product, the synthesis, manufacturing, processing, formulating, packaging, labeling, holding, quality control testing and release (as applicable) of such Galapagos Molecule or Galapagos Product or Optioned Molecule or Optioned Product and such other manufacturing-related activities that support the Development (including the seeking and obtaining of Regulatory Approvals) and Commercialization of such Galapagos Molecule, Galapagos Product, Optioned Molecule or Optioned Product, including manufacturing process development and scale-up, validation, qualification and audit of clinical and commercial manufacturing facilities, bulk production and fill/finish work, related quality assurance technical support activities and CMC Activities. "Manufacturing" has a correlative meaning.

"**Marketing Approval**" means, with respect to a Marketing Authorization Application and a particular country or jurisdiction, the approval by a Regulatory Authority of such Marketing Authorization Application for such country or jurisdiction.

"Marketing Authorization Application" or "MAA" means an application for Regulatory Approval in a country, territory or possession, including an NDA or BLA.

"**Medical Affairs Activities**" means activities designed to ensure or improve appropriate medical use of, conduct medical education of, or further research regarding, an Optioned Product sold in the Territory, including by way of example: (a) activities of medical scientific liaisons who, among their other functions, may (i) conduct service-based medical activities including providing input and assistance with consultancy meetings, recommend investigators for clinical trials and provide input in the design of such trials and other research related activities, and (ii) deliver non-promotional communications and conduct non-promotional activities including presenting new clinical trial and other scientific information; (b) grants to support continuing medical education, symposia, or Third Party research related to an Optioned Product in the Territory; (c) development, publication and dissemination of publications relating to an Optioned Product in the Territory; (d) medical information services provided in response to inquiries communicated via sales representatives or received by letter, phone call or email; (e) conducting advisory board meetings or other consultant programs; (f) the support of investigator-initiated trials; and (g) establishment and implementation of risk, evaluation and mitigation and strategies (REMS).

"**Molecule**" means both small molecules and large molecules (such as biologics), including any such molecules that are Directed To multiple Targets (*e.g.*, a bi-specific or multi-specific antibody).

"**Mutual Post-Approval Commitment**" means any post-Regulatory Approval commitment with respect to a given Optioned Product that is applicable to at least the United States and the European Union.

"NDA" means a new drug application, as defined in the FD&C Act and applicable regulations promulgated thereunder by FDA, including all amendments and supplements to any such application, and any equivalent application, amendment or supplement to the equivalent Regulatory Authority in any other regulatory jurisdiction.

"Net Receipts" [...***...].

"Net Sales" [...***...].

"Nonclinical Studies" means all non-human animal studies, including preclinical studies and toxicology studies, of Galapagos Molecules and Optioned Molecules.

"**Optioned Molecule**" means, for any Optioned Program, each Galapagos Molecule included in the applicable Galapagos Program as of immediately prior to the applicable Option Exercise Closing for such Optioned Program.

"**Optioned Product**" means any product containing an Optioned Molecule, other than a Gilead Combination Product. Except as set forth in the preceding sentence, an "Optioned Product" includes all products containing the same Optioned Molecule, alone or In Combination with one or more other active pharmaceutical ingredients (other than (a) any other Galapagos Molecule or Galapagos Product that is not another Optioned Molecule or Optioned Product, (b) any Molecule or product that is the subject of any Pre-Program Activities or (c) any Molecule or product that is the subject of any Excluded Programs), in any and all finished forms, presentations, delivery systems, strength, dosages, and formulations.

"**Optioned Product Trademarks**" means the Trademark(s) to be used by either Party or its respective Affiliates, or its or their respective Sublicensees, for the Commercialization of Optioned Products in the Territory and any registrations thereof or any pending applications relating thereto in the Territory (excluding, in any event, any trademarks, service marks, names or logos that include any corporate name or logo of the Parties or their Affiliates).

"Optioned Program" means any Galapagos Program with respect to which the Option has been exercised and the Initial Option Closing has occurred.

"**Patent**" means (a) any national, regional or international patent or patent application, including any provisional patent application; (b) any patent application claiming priority from such a patent, patent application or provisional application or from an application claiming priority from any of these, including any divisional, continuation, continuation-in-part, converted provisional or continued prosecution application; (c) any patent that has issued or in the future issues from any of the foregoing patent applications ((a) and (b)), including any utility model, petty patent, design patent or certificate of invention; (d) any extension or restoration by existing or future extension or restoration mechanisms, including any revalidation, reissue, re-examination, review and extension (including any supplementary protection certificate and the like) of any of the foregoing patent applications ((a), (b) and (c)); and (e) any similar rights, including so-called pipeline protection, or any importation, revalidation, confirmation or introduction patent or registration patent or patent of additions to any such foregoing patent application or patent.

"**Patent Costs**" means the out-of-pocket costs and expenses paid by a Party or its Affiliates to outside legal counsel, patent offices or other Governmental Authorities, or other Third Parties, including filing and maintenance expenses, in each case, with respect to (a) the Prosecution of any Patent; (b) the enforcement of any Patent; (c) the defense of any Invalidity or Unenforceability Action with respect to any claim of a Patent; (d) Patent Term Extension of any Patent or (e) Orange Book Listing for any Patent.

"**Person**" means an individual, sole proprietorship, partnership, limited partnership, limited liability partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture or other similar entity or organization, including a government or political subdivision, department or agency of a government.

"**Phase 1 Clinical Trial**" means a Clinical Trial of a Galapagos Product or Optioned Product to determine initial tolerance, safety, pharmacokinetic or pharmacodynamic information in single dose, single ascending dose, multiple dose or multiple ascending dose regimens, which is prospectively designed to generate sufficient data (if successful) to permit design of a Phase 2 Clinical Trial of such Galapagos Product or Optioned Product.

"**Phase 2 Clinical Trial**" means a Clinical Trial of a Galapagos Product or Optioned Product regarding the safety, dose ranging and efficacy of a pharmaceutical product that is prospectively designed to generate sufficient data (if successful) to commence pivotal clinical trials, as described in 21 C.F.R. §312.21(b) (as amended or any replacement thereof), or a similar Clinical Trial prescribed by the Regulatory Authorities in a foreign country, including the phase 2 portion of a Clinical Trial that is both a Phase I Clinical Trial and a Phase 2 Clinical.

"**Phase 3 Clinical Trial**" means a Clinical Trial of a Galapagos Product or Optioned Product on a sufficient number of subjects that is designed to establish that such product is safe and efficacious for its intended use, and to determine warnings, precautions, and adverse reactions that are associated with such product in the dosage range to be prescribed, which Clinical Trial is intended to support Regulatory Approval of such Galapagos Product or Optioned Product, as described in 21 C.F.R. 312.21(c) (as amended or any replacement thereof), or a similar Clinical Trial prescribed by the Regulatory Authorities in a foreign country.

"**Phase 4 Clinical Trial**" means (a) a Clinical Trial of an Optioned Product, conducted following commencement of a pivotal Clinical Trial for such Optioned Product, that is not required for receipt of approval of Marketing Authorization Application, but that may be useful in support of the post-approval Exploitation of such Optioned Product; or (b) a Clinical Trial of an Optioned Product conducted after Marketing Approval of such Optioned Product has been obtained from an appropriate Regulatory Authority due to a request or requirement of such Regulatory Authority.

"PhRMA Code" means the PhRMA Code on Interactions with Health Care Professionals.

"**Post-IND Term Extension**" means, for any Galapagos Program, an extension of the applicable Galapagos Program Period until the earlier of (a) delivery of an Option Exercise Notice for such Galapagos Program, (b) the end of the applicable Option Exercise Period prior to delivery by Gilead of an Option Exercise Notice for such Option, and (c) the third (3rd) anniversary of the end of the Collaboration Term.

"**Pre-Option In-License**" means, with respect to each Optioned Program, any agreement between Galapagos or any of its Affiliates (including by way of assignment or other transfer in connection with an Acquired Galapagos Program) and any Third Party that governs a license of Galapagos IP related to such Optioned Program, and that was entered into by Galapagos or any of its Affiliates prior to the Initial Option Closing for such Optioned Program.

"**Pre-Program Activities**" means any discovery, target screening, development, research, pre-clinical, non-clinical and manufacturing activities of Galapagos or any its Affiliates during the Collaboration Term and any Post-IND Term Extension, other than such activities conducted under a Galapagos Program, Optioned Program or Excluded Program.

"**Preliminary Target Information**" means any information disclosed by or on behalf of one Party to the other Party with respect to Target identification, Target or other screening (including resulting hits) results, selection or prioritization. For clarity, Preliminary Target Information shall not include [...***...].

"**Program Claim**" means, with respect to (a) any Patent and (b) any Pre-Program Activities, Galapagos Program or Optioned Program, a claim of such Patent that claims or covers (i) a molecule or product that is the subject of such Pre-Program Activities or (ii) an applicable Galapagos Molecule, Galapagos Product, Optioned Molecule or Optioned Product, or, in each case ((i) or (ii)), the Exploitation thereof.

"Qualifying Data Package" means, with respect to each Galapagos Program, a downloadable copy of each item set forth on Schedule I, delivered via a Data Room.

"R&D Plan and Budget" means, with respect to each Optioned Program, a plan and budget containing [...***...].

"**Regulatory Approval**" means all approvals (including licenses, registrations or authorizations) from any applicable Regulatory Authority in a given country or countries (and, if applicable, the European Union) necessary for the Manufacture, marketing, commercial distribution, importation and sale of a Galapagos Product or Optioned Product for one or more indications in the Field other than for Veterinary Uses and in such country or regulatory jurisdiction, which may include satisfaction of all applicable regulatory and notification requirements and, where applicable, labeling approval, but which, shall exclude any pricing and reimbursement approvals. Regulatory Approvals include Marketing Approvals.

"**Regulatory Authority**" means, in a particular country or regulatory jurisdiction, any applicable Governmental Authority involved in granting any Regulatory Approval for the applicable product in such country or regulatory jurisdiction or otherwise exercising authority with respect to the Development, Manufacture or Commercialization of a Galapagos Product or Optioned Product in such country or jurisdiction, including (a) the FDA, (b) the EMA and (c) the European Commission or the successor of any such Governmental Authority.

"**Regulatory Exclusivity**" means, with respect to any country in the Territory, an additional market protection, other than Patent protection, granted by a Regulatory Authority in such country which confers an exclusive Commercialization period during which a Party or its Affiliates or Sublicensees have the exclusive right to market and sell an Optioned Product in such country through a regulatory exclusivity right (*e.g.*, new chemical entity exclusivity, new use or indication exclusivity, new formulation exclusivity, orphan drug exclusivity, pediatric exclusivity, or any applicable data exclusivity).

"**Regulatory Materials**" means (a) regulatory applications, submissions, notifications, registrations, or other filings made to or with a Regulatory Authority, and (b) correspondence and reports submitted to or received from a Regulatory Authority (including minutes and official contact reports relating to any communication with any Regulatory Authority) and all supporting documents with respect thereto, including all regulatory drug lists, advertising and promotion documents, adverse event files, and complaint files, in each case ((a) and (b)), that are necessary or reasonably desirable in order to Develop, Manufacture, market, sell or otherwise Commercialize a Galapagos Product or Optioned Product in a particular country or regulatory jurisdiction. Regulatory Materials include INDs and Marketing Authorization Applications (for clarity, as applications, but not the approvals with respect thereto).

"Related Molecule(s)" means, with respect to any Lead Molecule, any and all other Molecule(s) [...***...].

"**Requested or Required Phase 4 Clinical Trial**" means a Phase 4 Clinical Trial that is conducted due to a request or requirement of a Regulatory Authority, including any post-approval commitment, postmarketing requirement under Section 505(o)(3), Section 506(c)(2)(A), or Section 505B of the FD&C Act or FDA regulations, including subpart H or I of 21 C.F.R. part 314, subpart E or H of 21 C.F.R. part 601 or any Clinical Trial that is imposed by the applicable Regulatory Authority as a condition of the Marketing Approval in the European Union.

"Research and Development Costs" means, with respect to an Optioned Program: [...***...].

"**Respective Territory**" means at a given time, with respect to Gilead, the Gilead Territory and with respect to Galapagos, the Galapagos Territory, in each case, at such time.

"**Reversion Product(s)**" means, with respect to a given termination of this Agreement, the Terminated Molecules and Terminated Product(s) with respect thereto, in the forms that are as of the effective date of such termination the subject of clinical Development or Commercialization hereunder.

"SEC" means the U.S. Securities and Exchange Commission.

"Security Agreement" means that certain security agreement between the Parties to be agreed prior to the Closing consistent with the key terms set forth on Schedule J.

"Shared Commercialization Costs" means, with respect to an Optioned Program, [...***...].

"Standard Cost of Manufacturing" means, with respect to a given Optioned Product, [...***...].

"**Subject Matter Expert**" means (a) for Development matters, including Manufacturing in support thereof, with respect to Galapagos, Chief Scientific Officer (or his or her designee) and, with respect to Gilead, its Chief Scientific Officer (or his or her designee) and (b) for Commercialization matters, including Manufacturing in support thereof, with respect to Galapagos, Chief Operating Officer (or his or her designee) and, with respect to Gilead (or his or her designee), its Chief Commercial Officer. Each Party may change its respective Subject Matter Expert upon written notice to the other Party with another representative of equivalent seniority, knowledge and expertise.

"Sublicense Agreement" means any agreement pursuant to which a Party grants a sublicense to a Third Party under any license granted to it in Sections 8.1, 8.3, 8.4 or 10.9(c), as applicable.

"Sublicensee" means any Third Party that is granted a sublicense under the rights licensed to a Party hereunder.

"Subscription Agreement" has the meaning set forth in the Recitals.

"Suspension or Termination" means a Party's decision to suspend [...***...], other than due to any delay due [...***...]) or terminate any Pre-Program Activities, Galapagos Program or any Optioned Program, as applicable; *provided that* any inactivity with respect to a Pre-Program Activity, Galapagos Program or Optioned Program that lasts for [...***...] or longer (not including any delay in activity due to [...***...] will be deemed a Suspension or Termination. (any such determination to suspend or terminate, a "Suspend or Terminate" has a correlative meaning.

"Target" means one or more genes, proteins, nucleic acids, receptors, ligands, antigens or other Molecules or targets. [...***...].

"**Tax or Taxes**" means any taxes of any kind including, but not limited to those measured on, measured by or referred to as, income, alternative or add-on minimum, gross receipts, escheat, capital, capital gains, sales, use, ad valorem, franchise, profits, license, privilege, transfer, withholding, payroll, employment, social security, excise, severance, stamp, occupation, premium, value added, property, environmental or windfall profits taxes, customs duties or similar fees, assessments or charges of any kind whatsoever, including any contractual obligation to indemnify another Person for Taxes, together with any interest and any penalties, additions to tax or additional amounts imposed by any Governmental Authority.

"**Terminated Region**" means, with respect to a Pre-Program Activity, Galapagos Program or Optioned Program, as applicable, the countries in the Gilead Territory as to which the applicable termination is effective or, if all countries in the Gilead Territory are or have been terminated, then the Gilead Territory.

"Territory" means all countries in the world.

"Third Party" means any entity other than Galapagos or Gilead or an Affiliate of either of them.

"**Trademark**" means any trademark, trade name, service mark, service name, brand, domain name, trade dress, logo, slogan, or other indicia of origin or ownership, including registrations and applications therefor and the goodwill and activities associated with each of the foregoing.

"U.S." or "United States" means the United States of America (including all possessions and territories thereof).

"Valid Claim" means, with respect to a particular country, (a) any claim of an issued and unexpired Patent in such country that (i) has not been held permanently revoked, unenforceable or invalid by a decision of a court or Governmental Authority of competent jurisdiction, which decision is unappealable or unappealed within the time allowed for appeal and (ii) has not been abandoned, disclaimed, denied or admitted to be invalid or unenforceable through reissue or disclaimer or otherwise in such country, or (b) a claim of a pending Patent application that has been pending without issuance for a period not longer than [...***...] years from the earliest priority date of such application, which claim is being diligently prosecuted and has not been abandoned or finally disallowed without the possibility of appeal or re-filing of the application.

"Veterinary Use" means Exploitation of an Optioned Molecule or Optioned Product for the prevention or treatment of veterinary medical conditions in animals. For clarity, humans are not animals for purposes of this definition.

"Voluntary Phase 4 Clinical Trial" means a Phase 4 Clinical Trial that is not a Requested or Required Phase 4 Clinical Trial.

Each of the following terms is defined in the Section set forth opposite such term:

Term	Section
Active Galapagos Target	Schedule 8.10
Agreement	Preamble
Alliance Manager	1.9
Anti-Corruption Laws	11.1(e)
Antitrust Clearance Date	16.1(b)
[***]	2.3(c)(iii)
Bankrupt Party	14.9
Bankruptcy Code	14.5
Bribery Act	11.1(e)
Clinical Supply Agreement	6.2(e)
Combination Product Activities Agreement	2.6(c)(iii)
Commercial Supply Agreement	6.2(f)
Committee Dispute	15.1(c)
Confidential Information	13.1(a)
Dispute	15.1(b)
DOJ	8.2(d)(ii)
FCPA	11.1(e)
FTC	8.2(d)(ii)
Galapagos	Preamble
Galapagos Acquirer	17.6(b)
Galapagos Collaboration IP	10.1(a)
Galapagos Collaboration Know-How	10.1(a)
Galapagos Collaboration Patents	10.1(a)
Galapagos Indemnitees	12.2
Galapagos Program Period	8.2(a)
Galapagos R&D Activities	2.1
Galapagos Schedule of Exceptions	11.2
Enforcing Party	10.4(f)
Execution Date	Preamble
Gilead	Preamble
Gilead Collaboration IP	10.1(b)
Gilead Contributions	2.2
Gilead Indemnitees	12.1
Gilead Reversion Patents	14.7(c)(i)
Global Commercialization Plan and Budget	5.2(a)
Global Manufacturing Plan and Budget	6.2(a)

Term	Section
Global Optioned Product Trademarks	10.9(a)
Global Promotional Materials	5.2(c)
ICC Rules	15.2(c)
Indemnified Party	12.3
Indemnifying Party	12.3
Independent Activities	7.1
Independent Activities Costs	7.5
Independent Activities Data	7.4(a)
Independent Activities Party	7.2(a)
Independent Activities Plan	7.2(a)
Independent Activities Regulatory Documentation	7.4(b)
Independent Post-Option In-License	9.6(a)(ii)
Infringing Activity	10.6(a)
Initial Option Closing	8.2(c)(i)
Joint Collaboration IP	10.1(c)
Joint Commercialization Committee or JCC	1.3(a)
Joint Communication Review Committee" or "JCRC	1.4(a)
Joint Development Committee or JDC	1.2(a)
Joint Post-Option In-License	9.6(a)(i)
Joint Steering Committee or JSC	1.1(a)
	"Galapagos
	Program"
Lead Molecule	definition
	"Related
	Molecules"
Lead Patent	definition
Losses	12.1
Market Abuse Regulation	11.2(q)
Material Adverse Effect	11.2(r)
Non-Committee Dispute	15.1(b)
Non-Enforcing Party	10.4(f)
Option	8.2(a)
Option Bringdown Date	8.2(c)(ii)
Option Exercise Closing	8.2(c)(i)
Option Exercise Notice	8.2(b)(i)
Option Exercise Period	8.2(b)(i)
Option Payment	9.2
Optioned Product Orange Book Listing	10.3(a)
Optioned Program R&D Activities	3.2
Opt-Out Notice	3.7(a)
Outside Date	14.2
Party Indemnitees	12.4(a)

<u>Term</u>	Section
Party or Parties	Preamble
Patent Litigation Committee	10.4(b)
Patent Prosecution Committee	10.2(a)
Patent Term Extensions	10.3
Permitted Commercialization Overrun	9.10(b)
Permitted R&D Cost Overrun	9.9(b)
Personal Information	11.2(m)(iii)
Program Infringement	10.4(a)
Proposal	15.2(d)(i)
	10.2(b)(i)
Prosecution	(A)
Publication	13.4(a)(i)
PV Agreement	4.2(g)(i)
Redacted Agreements	13.3(c)
Regional Optioned Product Trademarks	10.9(b)
Regulatory Transition Date	4.2(a)(i)
Representatives	11.1(e)
Royalty Term	9.4(b)
	"Galapagos
f and l	Territory"
[***]	definition
	"Galapagos
[444]	Territory"
	definition
Shared Development Claims	12.4(a)
Shared Development Losses	12.4(a)
Shared Patent Costs	9.11(a)
[***]	8.10(a)
	8.10(a)
Significant Pre-Program Activities	2.4
Special Option Exercise Representations	8.2(c)(iii)
Sublicensing Party	8.6(b)
Subscription Agreement	Preamble
Suspension or Termination	2.4
TCT Determination Package	2.3(c)(i)
Technology Transfer	6.3(b)
Technology Transfer Plan	6.3(b)
Term	14.1
Terminated Molecule	14.7(c)
Terminated Product	14.7(c)
Terminated Program	14.7(c)
Termination Notice Period	14.7(a)(i)
Third Party Claims	12.1
Third Party License Payments	9.6(b)(i)

<u>Term</u> Trademark Licensee Party Trademark Owner Party Transition Agreement Triggering Clinical Trial Upfront Consideration VAT Working Groups
 Section

 10.9(c)(i)

 10.9(c)(i)

 14.7(c)(iv)

 8.2(b)(v)

 9.1

 9.12(c)

 1.8(b)

* * *

CERTAIN CONFIDENTIAL INFORMATION IN THIS EXHIBIT HAS BEEN OMITTED AND REPLACED WITH "[...***...]" BECAUSE IT IS NOT MATERIAL AND WOULD BE COMPETITIVELY HARMFUL IF PUBLICLY DISCLOSED.

SCHEDULE A

Access Territory

SCHEDULE B

Excluded Programs

	Molecule	Molecule Target(s)	
1.	GLPG0187	Integrin inhibitor	ThromboGenics
2.	GLPG0634	JAK-1 inhibitor	Gilead
3.	GLPG3535	ASK-1 inhibitor	Calchan
4.	MOR106	Anti-IL-17C	Novartis

SCHEDULE C

Existing Galapagos Third Party Obligations

SCHEDULE D

Structure of GLPG1690

SCHEDULE E

Structure of GLPG1972

SCHEDULE F

Galapagos Option Exercise Representations

The Galapagos Program that is the subject of the Option Exercise Notice to which these Option Exercise Representations and the corresponding Galapagos Option Schedule of Exceptions relate is referred to as the "Subject Program." Except as set forth in such Galapagos Option Schedule of Exceptions, Galapagos hereby represents and warrants and, with respect to Sections 4 and 8 below, covenants to Gilead as of the applicable Option Exercise Closing, except as set forth in the corresponding Galapagos Option Schedule of Exceptions, as follows with respect to the Subject Program:

- 1. **No Debarment**. Neither Galapagos nor any of its Affiliates is debarred, has been convicted, or is subject to debarment or conviction pursuant to Section 306 of the FD&C Act. Galapagos has not used in connection with any activity in its business, any employee, agent or independent contractor who has been debarred by any Regulatory Authority, or, to the best of Galapagos' knowledge, is the subject of debarment proceedings by a Regulatory Authority or has been convicted pursuant to Section 306 of the FD&C Act. Galapagos shall promptly notify Gilead of any debarment or debarment proceeding that could have an impact on the use of the results of any Clinical Trials relating to the Subject Program.
- 2. **Title; Encumbrances**. Galapagos or one of its Affiliates solely owns or exclusively licenses and Controls the Existing Galapagos Patents and other Galapagos IP relating to the Subject Program, *provided*, *however*, that the foregoing shall not constitute a representation or warranty of non-infringement of a Third Party's intellectual property rights. Galapagos or one of its Affiliates has the right to grant the licenses for the Subject Program to Gilead as purported to be granted pursuant to this Option Exercise Closing. Neither Galapagos nor any of its Affiliates has entered into any agreement (other than agreements with subcontractors) granting any right, interest or claim in or to, any Existing Galapagos Patents or Galapagos Know-How relating to the Subject Program to any Third Party that would conflict with the rights and licenses to Gilead as purported to be granted pursuant to this Option Exercise.
- 3. Assignment of Rights. Each Person who has or has had any rights in or to any Existing Galapagos Patents or any material Galapagos Know-How relating to the Subject Program has assigned and has executed an agreement assigning its entire right, title and interest in and to such Existing Galapagos Patents and Galapagos Know-How relating to the Subject Program to Galapagos or one of its Affiliates. To Galapagos' Knowledge, no current officer, employee, agent, or consultant of Galapagos or any of its Affiliates is in violation of any term of any assignment or other agreement regarding the protection of Patents or other intellectual property or proprietary information of Galapagos or such Affiliate relating to the Subject Program or of any employment contract relating to the relationship of any such Person with Galapagos.

- 4. Patents. All Existing Galapagos Patents for the Subject Program are identified on Schedule 4. All Existing Galapagos Patents for the Subject Program are subsisting and are being diligently prosecuted in the patent offices indicated on Schedule 4 in accordance with Applicable Law and to Galapagos' Knowledge are not invalid or unenforceable in whole or in part. All applicable fees with respect thereto have been timely paid or will be timely paid (taking account of any permitted extensions). The Existing Galapagos Patents on Schedule 4 represent all Patents within Galapagos' or its Affiliates' ownership or control (by license or otherwise) that Galapagos reasonably believes include claims covering the making, using or composition of matter of the Galapagos Molecules or Galapagos Products, or the Exploitation of any such Galapagos Molecule or Galapagos Product, in each case, in the Subject Program. To the extent required, Galapagos or one of its Affiliates has properly recorded in the relevant U.S. and foreign patent offices the assignments, or other necessary documents, supporting its legal title to the Existing Galapagos Patents for the Subject Program on Schedule 4. Galapagos and its Affiliates have presented, or will present prior to the pertinent patent office deadlines, all relevant references, documents or information of which they and the inventors are aware to the relevant patent examiner at the pertinent patent office, in connection with the prosecution of the pending patent applications included in the Existing Galapagos Patents for the Subject Program.
- 5. **No Infringement**. To Galapagos' Knowledge, no Person is infringing or threatening to infringe or misappropriating or threatening to misappropriate any Existing Galapagos Patents or any Galapagos Know-How or Regulatory Documentation relating to the Subject Program. To Galapagos' Knowledge, there are no activities by Third Parties that would constitute infringement or misappropriation of the Galapagos IP relating to the Subject Program (in the case of pending claims, evaluating them as if issued).
- 6. **No Conflicts.** The consummation of this Option Exercise Closing does not, and the consummation of the transactions contemplated by this Option Exercise Closing will not, (i) except for the rights granted to Gilead in this Option Exercise, result in the creation of any encumbrance on any of the material properties or assets relating to the Subject Program, or (ii) conflict with, or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, or require any consent, approval or waiver from any Person pursuant to (A) any provision of the organizational or governing documents of Galapagos, in each case as amended to date, or (B) any material agreement applicable to Galapagos' or any of its Affiliates' material properties or assets relating to the Subject Program.
- 7. **In-Licenses and Restrictions on Business Activities.** None of the Existing Galapagos Patents for the Subject Program licensed hereunder by Galapagos to Gilead are owned or controlled in whole or in part by any Third Party. There is no agreement, judgment, injunction, order or decree of a Governmental Authority with respect to the Subject Program binding upon Galapagos or any of its Affiliates that has or would reasonably be expected to have, whether before or after the Option Exercise Closing, the effect of prohibiting or impairing any current or presently proposed business practice of Galapagos or any of its Affiliates or the conduct of business by Galapagos or any of its Affiliates as currently conducted or as presently proposed to be conducted by Galapagos or any of its Affiliates for such Subject Program.

- 8. **Copyrightable IP**. To Galapagos' Knowledge, all works of authorship and all other materials subject to copyright protection included in Information owned or otherwise controlled by Galapagos or any of its Affiliates that is reasonably necessary or useful to Exploit any Galapagos Molecule or Galapagos Product in the Subject Program are original and were either created by employees of Galapagos or its Affiliates within the scope of their employment or are otherwise works made for hire, or right, title and interest in and to such materials have been legally assigned or licensed to Galapagos or such Affiliate to the extent necessary to provide Gilead with the rights granted to it hereunder, and all rights in all inventions and discoveries made, developed, or conceived by any employee or independent contractor of Galapagos or any of its Affiliates during the course of their employment (or other retention) by Galapagos or such Affiliate, and relating to or included in the Galapagos Know-How or that are the subject of one or more Existing Galapagos Patents, in each case relating to the Subject Program, have been or will be assigned in writing to Galapagos or such Affiliate to the extent necessary to provide Gilead with the rights granted to it hereunder.
- 9. **Transfer of Rights**. Galapagos or one of its Affiliates has obtained the right (including under any Patents and other intellectual property rights) to use all Information and all other materials (including any formulations and manufacturing processes and procedures) developed or delivered by any Third Party under any agreements between Galapagos or one of its Affiliates and any such Third Party with respect to the Subject Program to the extent necessary to provide Gilead with the rights granted to it hereunder, and Galapagos or one of its Affiliates has the rights under each such agreement to transfer such Information or other materials to Gilead and its designees and to grant Gilead the right to use such Information or other materials in the Development or Commercialization of the Galapagos Molecules or Galapagos Products in the Subject Program as required to enable Gilead to Exploit the Galapagos Molecules and the Galapagos Products in the Subject Program in the Gilead Territory.
- 10. **Confidentiality of Know-How**. With respect to those portions of the Galapagos Know-How the confidentiality of which is material to the Exploitation of any Galapagos Molecule or Galapagos Product in the Subject Program, such portions of the Galapagos Know-How have been kept confidential or have been disclosed to Third Parties only under terms of confidentiality or if published or otherwise publicly disclosed, were published or publicly disclosed in a manner that would not reasonably be expected to adversely impact the patentability of such Galapagos Know-How relating to the Subject Program. To Galapagos' and its Affiliates' Knowledge, no breach of such confidentiality relating to the Subject Program has been committed by any Third Party.

11. No Proceedings.

a. No claim or litigation has been brought or threatened in writing by any Person against Galapagos or any of its Affiliates alleging, and Galapagos has no Knowledge of any reasonable basis for any such claim or allegation, whether or not asserted, that (A) any Existing Galapagos Patents for the Subject Program are invalid or unenforceable, or (B) that the use or practice of any Existing Galapagos Patents or any Regulatory Materials or Galapagos Know-How, or the disclosing,

copying, making, assigning or licensing of any Existing Galapagos Patents or any such Regulatory Materials or Galapagos Know-How, in each case relating to the Subject Program, or the Development, Commercialization or other Exploitation of the Galapagos Molecules or Galapagos Products in the Subject Program as contemplated herein, does or will violate, infringe, misappropriate or otherwise conflict or interfere with, any Patent or other intellectual property or proprietary right of any Third Party.

- b. There is no private or Governmental Authority action, suit, proceeding, claim, mediation, arbitration or investigation pending before any Governmental Authority, or, to Galapagos' Knowledge, threatened against Galapagos or any of its Affiliates or any of its or their respective assets or properties relating to the Subject Program or, to Galapagos' Knowledge, any of their respective directors, managers, officers or employees (in their capacities as such or relating to their employment, services or relationship with Galapagos or any of its Affiliates), nor, to Galapagos' Knowledge, is there any reasonable basis for any such action, suit, proceeding, claim, mediation, arbitration or investigation. There is no judgment, decree, injunction or order against Galapagos or any of its Affiliates, or, to Galapagos' Knowledge, any of its or their respective assets or properties, relating to the Subject Program or, to Galapagos' Knowledge, any of its or their respective assets or properties, relating to the Subject Program or, to Galapagos' Knowledge, any of its or their respective assets or properties, relating to the Subject Program or, to Galapagos' Knowledge, any of its or their respective assets or properties, relating to the Subject Program or, to Galapagos' Knowledge, any of their respective directors, managers, officers or employees (in their capacities as such or relating to their employment, services or relationship with Galapagos or any of its Affiliates). Neither Galapagos nor any of its Affiliates has any action, suit, proceeding, claim, mediation, arbitration or investigation pending against any other Person or relating to the Subject Program.
- 12. **No Misappropriation**. The conception and reduction to practice of any inventions and the use or development of any other Information relating to the Subject Program (i) owned by Galapagos have not constituted or involved the misappropriation of trade secrets or other rights or property of any Third Party and (ii) in-licensed by Galapagos, to Galapagos' Knowledge, have not constituted or involved the misappropriation of trade secrets or other rights or property of any Third Party; *provided, however*, that the foregoing clauses (i) and (ii) shall not constitute a representation or warranty of non-infringement of a Third Party's intellectual property rights.
- 13. **Full Disclosure**. Galapagos has provided or made available to Gilead all material adverse information with respect to the safety and efficacy of the Galapagos Molecules or Galapagos Products in the Subject Program. Galapagos has provided a true, complete and correct list of each Galapagos Molecule and Galapagos Product in the Subject Program that has received IND approval from the FDA.

14. Compliance and Qualifications.

a. To Galapagos' Knowledge, Galapagos and its Affiliates have conducted, and their respective contractors, licensees and consultants, have conducted in all material respects all Development under the Subject Program in accordance with all Applicable Laws, including current Good Clinical Practices, Good Laboratory Practices, Good Manufacturing Practices and the Declaration of Helsinki.

- b. Galapagos and its Affiliates and licensees have generated, prepared, maintained and retained all Regulatory Documentation that is required to be maintained or retained with respect to the Subject Program pursuant to and in accordance with Applicable Law, and all such Regulatory Documentation is true, complete and correct and what it purports to be.
- c. In connection with the collection, storage, transfer (including any transfer across national borders) or use of any information relating to identified or identifiable natural persons (collectively "**Personal Information**") by or on behalf of Galapagos or any of its Affiliates relating to the Subject Program, Galapagos and its Affiliates are in compliance in all material respects with all Applicable Laws in all relevant jurisdictions, internal privacy policies and the requirements of any contract or codes of conduct to which Galapagos or any of its Affiliates is a party. Galapagos and its Affiliates have commercially reasonable technical and organizational measures in place to ensure the security of all Personal Information relating to the Subject Program it controls or processes. Galapagos and its Affiliates are in compliance in all material respects with all Applicable Laws relating to breaches of security affecting Personal Information relating to the Subject Program and associated notification obligations. Neither Galapagos nor any of its Affiliates has received a complaint regarding its collection, storage, transfer or use of Personal Information relating to the Subject Program. Galapagos agrees to execute or cause to be executed any additional clauses or agreements necessary to comply with data protection laws prior to the transfer of Personal Information.
- 15. **No Misrepresentation**. To Galapagos' Knowledge, neither Galapagos nor any of its Affiliates or licensees, nor any of its or their respective officers, employees or agents, has made an untrue statement of material fact or fraudulent statement to the FDA or any other Regulatory Authority with respect to the Development of the Galapagos Molecules or Galapagos Products in the Subject Program, failed to disclose a material fact required to be disclosed to the FDA or any other Regulatory Authority with respect to the Development of the Galapagos Products in the Subject Program, or committed an act, made a statement, or failed to make a statement with respect to the Development of the Galapagos Molecules or Galapagos Products in the Subject Program, or committed an act, made a statement, or failed to make a statement with respect to the Development of the Galapagos Molecules or Galapagos Products in the Subject Program that could reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities," set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto or any analogous Applicable Laws in the Territory.
- 16. **No Royalties**. There are no milestone payments, profit share obligations, royalty payments or other amounts in each case that are based on the Manufacture, Development or Commercialization of the Galapagos Molecules or Galapagos Products in the Subject Program required to be paid to a Third Party as a result of the Manufacture, Development or Commercialization of such Galapagos Molecules or Galapagos Products under any agreement to which Galapagos or any of its Affiliates is a party.

- 17. **No Government Funding**. To Galapagos' Knowledge, the inventions claimed in the Existing Galapagos Patents for the Subject Program (i) were not conceived, discovered, developed, generated or otherwise made in connection with any research activities funded, in whole or in part, by the federal government of the United States or any agency thereof, (ii) are not a "subject invention" as that term is described in 35 U.S.C. Section 201(f), and (iii) are not otherwise subject to the provisions of the Bayh-Dole Act.
- 18. **No Governmental Consents**. Neither the performance by Galapagos of this Option Exercise Closing for the Subject Program, nor the consummation by Galapagos of the transactions contemplated hereby, will, other than with respect to the HSR Act or any other Applicable Antitrust Law or the EU Regulation No. 596/2014 of 16 April 2014 on market abuse (the "Market Abuse Regulation"), require Galapagos or any of its Affiliates to (i) obtain any consent or authorization of, or (ii) give any notice to, or make any filing or registration with, any Governmental Authority or other Person.
- 19. To be included among the Option Exercise Representations with respect to an Option Exercise Closing for [...***...] only: [...***...]

SCHEDULE G

Galapagos Programs Existing as of the Execution Date

	Lead Molecule	Target(s)	Indication(s) and Stage of Clinical Development
1.	GLPG0555	JAK1	[***]
		inhibitor	
2.	GLPG0778	JAK1	Inflammation; program on hold
		inhibitor	
3.	GLPG1205	GPR84 inhibitor	 Idiopathic Pulmonary Fibrosis (Phase 2 Clinical Trial Ongoing)
			Ulcerative Colitis – Failed Phase 2
4.	GLPG1690	Autotaxin	 Idiopathic Pulmonary Fibrosis (Phase 3 Clinical Trial Ongoing)
		inhibitor	Scleroderma (Phase 2a Clinical Trial Ongoing)
5.	GLPG1837	CFTR	[***]
6.	GLPG1972	ADAMTS-5	Osteoarthritis (Phase 2 Clinical Trial Ongoing)
		inhibitor	
7.	GLPG3121	JAK1/TYK2	Inflammation (Phase 1 Clinical Trial Ongoing)
8.	GLPG3312	[***]	Inflammation (Phase 1 Clinical Trial Ongoing)
9.	[***]	[***]	[***]

Acquired Galapagos Program:

1. [***] [***] [***]

SCHEDULE H

Galapagos Territory

- The member states of the European Union as constituted as of the Execution Date (Austria, Belgium Bulgaria, Croatia, Cyprus, Czechia, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Ireland, Italy, Latvia, Lithuania, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia, Spain, Sweden, and the United Kingdom), and all other territories and possessions of the foregoing and all other member states of the European Union as constituted as of the Execution Date.
- Iceland
- Norway
- Lichtenstein
- Switzerland

SCHEDULE I

Qualifying Data Package

A Qualifying Data Package with respect to a given Galapagos Program shall include copies of the following items, [...***...], in each case relating to any Galapagos Molecule or Galapagos Product under such Galapagos Program [...***...]:

SCHEDULE J

Form of Security Agreement

SCHEDULE 3.1(B)

Initial R&D Plan and Budget for Autotaxin Program

SCHEDULE 8.10-1 Shared Program Process Definitions

SCHEDULE 8.10-2 Shared Program Process

SCHEDULE 8.10-3 Baseball Arbitration Bookends

SCHEDULE 11.2 Schedule of Exceptions

Schedule 11.2(a)

Title; Encumbrances

Schedule 11.1(c)

No Conflict

Schedule 11.2(c)

Patents

Schedule 11.2(d)

<u>No Infringement</u>

Schedule 11.2(e)

No Conflicts

Schedule 11.2(f)

In-Licenses and Restrictions on Business Activities

Schedule 11.2(h)

Transfer of Rights

Schedule 11.2(i)

Confidentiality of Know-How

Schedule 11.2(j)

No Proceedings

Schedule 11.2(0)

<u>Royalties</u>

Schedule 11.2(s) [...***...] IP Matters

Form of Press Release

See attached.

CERTAIN CONFIDENTIAL INFORMATION IN THIS EXHIBIT HAS BEEN OMITTED AND REPLACED WITH "[...***...]" BECAUSE IT IS NOT MATERIAL AND WOULD BE COMPETITIVELY HARMFUL IF PUBLICLY DISCLOSED.

AMENDED AND RESTATED LICENSE AND COLLABORATION AGREEMENT

BY AND BETWEEN

GALAPAGOS NV

AND

GILEAD BIOPHARMACEUTICS IRELAND UC

DATED AS OF AUGUST 23, 2019

ARTICLE 1 DEFINITIONS	Page 2
ARTICLE 2 PROGRAM; GOVERNANCE	17
ARTICLE 3 DEVELOPMENT	26
ARTICLE 4 REGULATORY MATTERS	29
ARTICLE 5 COMMERCIALIZATION; MEDICAL AFFAIRS	31
ARTICLE 6 MANUFACTURE AND SUPPLY	39
ARTICLE 7 LICENSES AND EXCLUSIVITY	40
ARTICLE 8 FINANCIALS	44
ARTICLE 9 INTELLECTUAL PROPERTY	50
ARTICLE 10 REPRESENTATIONS AND WARRANTIES AND COVENANTS	56
ARTICLE 11 INDEMNIFICATION	61
ARTICLE 12 CONFIDENTIALITY	64
ARTICLE 13 TERM AND TERMINATION	67
ARTICLE 14 DISPUTE RESOLUTION	72
ARTICLE 15 MISCELLANEOUS	74
i	

AMENDED AND RESTATED LICENSE AND COLLABORATION AGREEMENT

THIS AMENDED AND RESTATED LICENSE AND COLLABORATION AGREEMENT (the "Agreement") is entered into and effective as of the Amendment Effective Date (as defined below) by and between GALAPAGOS NV, a corporation organized under the laws of Belgium and having its principal place of business at Generaal de Wittelaan L11 A3, 2800 Mechelen, Belgium ("Galapagos"), and GILEAD BIOPHARMACEUTICS IRELAND UC, an unlimited liability company formed under the laws of Ireland with its registered address at 70 Sir John Rogerson's Quay Dublin 2, Ireland ("Gilead"). Galapagos and Gilead are sometimes referred to herein individually as a "Party" and collectively as the "Parties."

BACKGROUND

WHEREAS, Galapagos and Gilead entered into that certain License and Collaboration Agreement (the "License and Collaboration Agreement"), dated as of December 16, 2015 (the "Execution Date") pursuant to which Galapagos granted a license under certain intellectual property rights with respect to the Licensed Compound (as defined therein) and Licensed Products (as defined therein) to develop and commercialize Licensed Products in the Territory (as defined therein), in each case in accordance with the terms and conditions set forth therein;

WHEREAS, Gilead Therapeutics A1 Unlimited Company (as successor-in-interest to Gilead) and Galapagos are parties to the Subscription Agreement attached as <u>Exhibit A</u> hereto ("**Subscription Agreement**");

WHEREAS, Galapagos delivered to Gilead that certain Notice of Exercise of Co-Commercialization Option, dated as of December 14, 2017 pursuant to the License and Collaboration Agreement and thereafter the Parties negotiated the allocation of co-commercialization activities in France, Germany, Italy, Spain and the United Kingdom ("**EU5 Countries**") as well as the Benelux Countries (as defined below);

WHEREAS, the Parties entered into that certain option, license and collaboration agreement dated as of July 14, 2019 (the "**Option, License and Collaboration Agreement**") and, in connection therewith, executed that certain First Amendment (the "**First Amendment**") to the License and Collaboration Agreement as of July 14, 2019 (the "**Amendment Execution Date**"), which First Amendment provided that, on and after the Amendment Effective Date, subject to the oversight of the Committees established thereunder and those existing under the License and Collaboration Agreement, Galapagos will be responsible for Commercialization activities for the Benelux Countries and the Parties will jointly Commercialize the Licensed Product and the Gilead Combination Products in the EU5 Countries;

WHEREAS, the Parties desire and intend to work together leveraging each Party's expertise to collaborate with respect to the Development, Manufacture and Commercialization of the Licensed Compound and Licensed Products in the Field in the Territory, as and to the extent set forth in this Agreement (the "**Collaboration**"); and

WHEREAS, the Parties now desire to amend and restate the License and Collaboration Agreement to incorporate the terms of the First Amendment as set forth herein.

NOW THEREFORE, in consideration of the foregoing premises and the mutual promises, covenants and conditions contained in this Agreement, the Parties agree as follows:

ARTICLE 1

DEFINITIONS

As used in this Agreement, the following initially capitalized terms, whether used in the singular or plural form, shall have the meanings set forth in this Article 1. In addition, the terms "includes," "including," "include" and derivative forms of them shall be deemed followed by the phrase "without limitation" (regardless of whether it is actually written there (and drawing no implication from the actual inclusion of such phrase in some instances after such terms but not others)) and the term "or" has the inclusive meaning represented by the phrase "and/or" (regardless of whether it is actually written (and drawing no implication from the actual use of the phrase "and/or" in some instances but not in others)). Unless otherwise stated, dollar amounts set forth herein are U.S. dollars. Unless specified to the contrary, references to Articles, Sections or Exhibits shall refer to the particular Articles, Sections or Exhibits of or to this Agreement and references to this Agreement include all Exhibits hereto. The word "day," "quarter" or "year" (and derivatives thereof, e.g., "quarterly") shall mean a calendar day, calendar quarter or calendar year unless otherwise specified. The word "notice" shall mean notice in writing (whether or not specifically stated) and shall include notices, consents, approvals and other written communications contemplated under this Agreement. The word "hereof," "herein," "hereby" and derivative or similar word refers to this Agreement (including any Exhibits). The words "will" and "shall have the same obligatory meaning. Provisions that require that a Party, the Parties or a Committee hereunder "agree," "consent" or "approve" or the like shall require that such agreement, consent or approval be specific and in writing, whether by written agreement, letter, approved minutes or otherwise. Words of any gender include the other gender. Words using the singular or plural number also include the plural or singular number, respectively. References to any specific law or article, section or other division thereof, shall be deemed to include the then-current amendments or any replacement law thereto. Each reference to "the Co-Commercialization Agreement" shall be deemed to mean and be interpreted as "any Co-Commercialization Agreement" or "all Co-Commercialization Agreements" (as the context so requires).

1.1 [...***...]

1.2 "Access Territory" means, with respect to a Licensed Product or Gilead Combination Product, any and all countries and territories where Gilead (itself or through its Affiliates) has publicly announced a policy to generally sell or otherwise make available such Licensed Product or Gilead Combination Product and one or more other Gilead products at a significantly discounted price to patients in such countries or territories. The list of countries and territories included in the Access Territory as of the Execution Date is set forth on Exhibit F, which list shall be updated by Gilead on at least an annual basis.

1.3 "Affiliate" means, with respect to a particular Person, a person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such Person. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, by contract or otherwise.

1.4 "Affordable Basis" means, with respect to a Licensed Product or a Gilead Combination Product in the Access Territory, selling or otherwise making such Licensed Product or Gilead Combination Product available to patients at a price where the revenue per unit shall not exceed [...***...].

1.5 "Agreement" has the meaning set forth in the preamble hereto.

1.6 "Alliance Manager" has the meaning set forth in Section 2.1(b).

1.7 "**Amendment Effective Date**" means the Effective Date (as defined in the Option, License, and Collaboration Agreement).

1.8 "Amendment Execution Date" has the meaning set forth in the Recitals.

1.9 "**Ancillary Agreement**" means the SDEA, the Subscription Agreement, the Clinical Supply Agreement, the Co-Commercialization Agreement or the Transition Agreement.

1.10 "**Applicable Law**" means the applicable laws, rules and regulations, including any rules, regulations, guidelines or other requirements of Governmental Authorities, including Regulatory Authorities, that may be in effect from time to time, including the Foreign Corrupt Practices Act of 1977, as amended.

1.11 "Bankrupt Party" has the meaning set forth in Section 13.7.

1.12 "Bankruptcy Code" has the meaning set forth in Section 13.4(d).

1.13 "Benelux Co-Commercialization Agreement" means that certain Co-Commercialization Agreement (Benelux Countries) contemplated to be entered into by and between Galapagos and Gilead or their respective Affiliates pursuant to this Agreement to cover Co-Commercialization in the Benelux Countries.

1.14 "Benelux Countries" means the following countries: Belgium, The Netherlands, and Luxembourg.

1.15 "Business Conduct Policies" means Gilead's business conduct policies provided by Gilead to Galapagos from time to time.

1.16 "**Business Day**" means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States or (c) a bank or other public holiday in Brussels, Belgium.

1.17 "Claim" has the meaning set forth in Section 11.3.

1.18 "Clinical Supply Agreement" has the meaning set forth in Section 6.3.

1.19 "Clinical Trial" means any human clinical trial of a Licensed Product.

1.20 **"CMC Activities**" means those Manufacturing activities and regulatory activities designed to support preparation of the Chemistry, Manufacturing and Controls sections of any Regulatory Materials or Regulatory Approval.

1.21 "**Co-Commercialization**" means Commercialization activities in or for the Shared Territory, including sales and marketing activities, Pricing Matters, the local adaptation of global or regional Promotional Materials, strategic marketing, and Medical Affairs Activities as set forth in, and subject to, the applicable Co-Commercialization Agreement, including Detailing and promotional activities (including performing sales calls) related to a Licensed Product or, as applicable, a Gilead Combination Product undertaken by personnel of either Party with respect to the Licensed Product or, as applicable, a Gilead Combination Product, in each case in the Shared Territory. For clarity, [...***...]. "Co-Commercialize," "Co-Commercializing" and "Co-Commercialized" shall have corollary meanings therefor.

1.22 "**Co-Commercialization Agreement**" means the co-commercialization agreement setting forth the terms and conditions of Galapagos' Co-Commercialization of Licensed Products or applicable Gilead Combination Products in the respective country, which agreement shall be consistent with the terms of this Agreement, and additional reasonable and customary terms and conditions. For clarity, "**Co-Commercialization Agreements**" include without limitation, Benelux Co-Commercialization Agreement and EU5 Countries Co-Commercialization Agreement.

1.23 "**Co-Commercialization Costs**" means all costs incurred by or on behalf of either Party that are reasonably and directly attributable to the Co-Commercialization of Licensed Products or, if included in a Co-Commercialization Agreement with respect to one or more countries in the Shared Territory, Gilead Combination Products, including Distribution Costs, Sales and Marketing Costs, Medical Affairs Costs, Market Access Costs and all other costs (including in all cases, FTE Costs) related to Co-Commercialization of Licensed Products or, if included in a Co-Commercialization Agreement, a Gilead Combination Product, in each case with respect to one or more countries in the Shared Territory, including regulatory costs for the Shared Territory other than for the Regulatory Approvals, including those associated with approval of promotional materials in the Shared Territory.

1.24 "Co-Commercialization Non-Selling Party" means with respect to a country, the Party that is not the Co-Commercialization Selling Party.

1.25 **"Co-Commercialization Product**" means the Initial Co-Commercialization Product, and any and all other Licensed Products and Gilead Combination Products that the Parties Co-Commercialize in accordance with the terms hereof and the Co-Commercialization Agreements.

1.26 "**Co-Commercialization Selling Party**" means, with respect to a country in the Shared Territory, the Party that is in charge of the physical distribution of the products and will be (or one of its Affiliates will be) the legal entity selling the applicable Licensed Product or Gilead Combination Product in such country. As of the Amendment Effective Date, for Licensed Products, Galapagos is the Co- Commercialization Selling Party for the Benelux Countries, France, Spain and Italy, and Gilead is the Co-Commercialization Selling Party for the United Kingdom and Germany. For any Gilead Combination Product, for the Shared Territory, the Co- Commercialization Selling Party will be determined by the JSC in anticipation of launch of any Gilead Combination Product.

1.27 **"Co-Commercialization Term**" means, with respect to a country in the Shared Territory, the period from the Amendment Effective Date through the end of the term of the applicable Co-Commercialization Agreement with respect to the applicable country.

1.28 "Collaboration" has the meaning set forth in the Recitals.

1.29 "**Collaboration Know-How**" means, to the extent not Gilead Combination Know-How, all Information related exclusively to a Licensed Product or the Licensed Compound and that is conceived, discovered, developed or otherwise made jointly by the Parties, in each case optionally with their Affiliates or any Third Parties or any employees, consultants or agents of any of the foregoing, in performing the activities under this Agreement (including performance of the Development Plan or Shared Territory Commercialization Plan and Budget).

1.30 "Collaboration Patents" has the meaning set forth in Section 9.1(a).

1.31 "**Commercialization**" means any and all activities directed to marketing, promoting, distributing, importing, exporting, using, offering to sell, selling or having sold a product, including activities related to the commercial manufacture, marketing, promotion, sale or distribution of a product in the Territory, and, for purposes of setting forth the rights and obligations of the Parties under this Agreement, shall be deemed to include conducting Medical Affairs Activities. Commercialization shall include commercial activities conducted in preparation for a product launch and Phase 4 Clinical Trials other than those expressly set forth in the definition of Development. "**Commercialize**" has a correlative meaning.

1.32 "**Commercially Reasonable Efforts**" means, with respect to the Development or Commercialization of a Licensed Product, that level of efforts and resources commonly dedicated in the research-based pharmaceutical industry by a company to the Development or Commercialization, as the case may be, of a product of similar commercial potential at a similar stage in its lifecycle, [...***...].

1.33 "**Committee**" means the Joint Steering Committee, Joint Development Committee, or Joint Commercialization Committee, or any other subcommittee established under <u>Article 2</u>, as applicable.

1.34 "Competing Program" has the meaning set forth in Section 15.6(d).

1.35 "Competition Law Guidelines" means any global, regional or local guidelines relating to information exchange and antitrust and competition law compliance included in the Co-Commercialization Agreements or otherwise mutually agreed in writing by the Parties.

1.36 **"Compulsory License"** means, with respect to a Licensed Product or a Gilead Combination Product in a country or territory, a license or rights granted to a Third Party by a governmental agency within such country or territory to sell or offer for sale such Licensed Product or a Gilead Combination Product in such country or territory under any patent rights owned or controlled by Gilead or its Affiliates, without direct or indirect authorization from Gilead or its Affiliates, for example a right granted pursuant to requests under 30 August 2003 WTO decision.

1.37 "**Confidential Information**" means, with respect to a Party or any of its Affiliates, and subject to <u>Section 12.2</u>, all Information of such Party or such Affiliate that is disclosed to the other Party or any of its Affiliates under this Agreement, any Co-Commercialization Agreement, any Supply Agreement (or related quality agreement) or the Subscription Agreement.

1.38 "**Control**" means, with respect to any material, Information, Patent, Regulatory Materials or Regulatory Approvals, the possession (whether by ownership or license) by a Party or its Affiliates of the ability to grant to the other Party a license as provided herein to such item, without violating the terms of any agreement or other arrangement with any Third Party, or being obligated to pay any royalties or other consideration therefor that a Party has not expressly agreed to pay in relation to the activities under this Agreement, in existence as of the time such Party or its Affiliates would first be required hereunder to grant the other Party such license.

1.39 "**Cost of Goods Sold**" means, with respect to the supply of a Licensed Product, the product of the Standard Cost of Manufacturing such Licensed Product and the number of units of the applicable Licensed Product.

1.40 "**Cover**," "**Covering**" or "Covered" means, with respect to a Patent, in the absence of a license to a Valid Claim included in such Patent, the applicable activity, or, to the extent the applicable activity is not specified, the Exploitation of the applicable invention, discovery, process or product, would infringe such Valid Claim (or, in the case of a Valid Claim that has not yet issued, would infringe such Valid Claim if it were to issue).

1.41 "CPI" means the Consumer Price Index for the US City Average (all times).

1.42 [...***...].

1.43 "**Detail**" or "**Detailing**" means, with respect to a Licensed Product or Gilead Combination Product in the Shared Territory, the communication by a Sales Representative to a physician or healthcare practitioner permitted by law to prescribe Licensed Products or Gilead Combination Products during a sales call, which in the case of any counting of details under the KPIs set forth in the Shared Territory Commercialization Plan and Budget, such details shall meet the standards set forth therein or in the applicable Co-Commercialization Agreement. For the avoidance of doubt, discussions at conventions or other scientific meetings shall not constitute "**Details**" or "**Detailing**."

1.44 "**Development**" means all activities that relate to (a) obtaining or maintaining Regulatory Approval of a Licensed Product or Gilead Combination Product for one or more indications, (b) developing the process for the Manufacture of clinical and commercial quantities of a Licensed Product or a Gilead Combination Product, or (c) the conduct of Nonclinical Studies and Clinical Trials (including Required Phase 4 Clinical Trials and other Phase 4 Clinical Trials that are either (i) [...***...] or (ii) [...***...], including the preparation, submission, review and development of data or information in support of a submission to a Regulatory Authority to obtain or maintain Regulatory Approval of a Licensed Compound, Licensed Product, or Gilead Combination Product, as applicable, including the services of outside advisors in connection therewith, including its legal counsel and regulatory consultants, but excluding (A) Commercialization and (B) the Manufacture and accumulation of commercial inventory of a Licensed Product or Gilead Combination Product, as applicable. "**Develop**" has a correlative meaning. For clarity, the Clinical Trials set forth on Exhibit M are deemed included in "Development."

1.45 "Development Budget" means the budget included in the Development Plan setting forth the anticipated Development Costs.

1.46 "**Development Costs**" means (a) Development Manufacturing Costs and (b) all out-of-pocket costs incurred by or on behalf of either Party or any of its Affiliates that are reasonably and directly attributable to the Development of Licensed Products in the Territory, excluding any such amounts to the extent included as Development Manufacturing Costs. [...***...].

1.47 "Development Manufacturing Costs" means [...***...].

1.48 "Development Plan" has the meaning set forth in Section 3.2(a).

1.49 "Distribution Costs" means [...***...].

1.50 "**Distribution Matters**" means all issues and decisions regarding the distribution of Licensed Products or Gilead Combination Products in the Shared Territory, including decisions as to whether and with which wholesalers and distributors to contract, and the terms of contracts with such wholesalers and distributors.

1.51 "Distributor" means a Person appointed by a Party to distribute, market, and sell Licensed Products or Gilead Combination Products in a specified country or region.

1.52 "Drug Company" has the meaning set forth in Section 15.6(b).

1.53 "Earliest Termination Date" has the meaning set forth in Section 13.3(a)(i).

1.54 "Effective Date" means January 19, 2016.

1.55 [...***...] has the meaning set forth in Exhibit J.

1.56 "EMA" means the European Medicines Agency or its successor.

1.57 "EU" means all of the European Union member states as of the applicable time during the Term.

1.58 "EU5 Countries" has the meaning set forth in the Recitals.

1.59 "EU5 Countries Co-Commercialization Agreement" means that certain Co-Commercialization Agreement (EU5 Countries) contemplated to be entered into by and between Galapagos and Gilead or their respective Affiliates pursuant to this Agreement to cover Co-Commercialization in the EU5 Countries.

1.60 "Execution Date" has the meaning set forth in the Recitals.

1.61 "Executive Officer" means, with respect to Galapagos, its chief executive officer, and with respect to Gilead, its Chief Operating Officer.

1.62 "Executive Officer Referral Notice" has the meaning given to it in Section 2.4(b).

1.63 **"Existing Confidentiality Agreement**" means the mutual confidential disclosure agreement entered into by Gilead and Galapagos, dated September 28, 2015.

1.64 "Existing Galapagos Patents" means those Patents set forth on Exhibit B.

1.65 **"Existing Regulatory Documentation**" means Regulatory Materials filed with or prepared to be filed with any Regulatory Authority as of the Effective Date.

1.66 **"Exploit**" means, collectively, research, develop, use, manufacture, have manufactured, sell, have sold, offer for sale, commercialize, import, have imported, distribute, have distributed, export, have exported and otherwise exploit. **"Exploitation**" has a correlative meaning.

1.67 "FD&C Act" means the United States Federal Food, Drug and Cosmetic Act, as amended.

1.68 "FDA" means the United States Food and Drug Administration or its successor.

1.69 "Field" means all uses.

1.70 "Finance Officers" has the meaning set forth in Section 8.9(a)(ii)(B).

1.71 "First Amendment" has the meaning set forth in the Recitals.

1.72 "**First Commercial Sale**" means, with respect to the applicable product in a country, the first sale in an arm's length transaction to a Third Party by a Party or any of its Affiliates or Sublicensees in the Field other than for Veterinary Uses in such country following Regulatory Approval of such product in such country. For the avoidance of doubt, a first sale for compassionate use or named patient program sales shall not constitute a First Commercial Sale for purposes of this Agreement.

1.73 "**FTE**" means the equivalent of the work of one (1) individual (whether an employee or individual contractor) engaged in the activities under the Co-Commercialization Agreement full time for one (1) calendar year (consisting of at least a total of [...***...] hours per calendar year) of work directly related to activities under any Co-Commercialization Agreement. No additional payment shall be made with respect to any person who works more than [...***...] hours per calendar year and any person who devotes less than [...***...] hours per calendar year shall be treated as an FTE on a pro rata basis based upon the actual number of hours worked divided by [...***...] hours per calendar year (or such other number as may be so agreed).

1.74 "**FTE Cost**" means the cost of an FTE based on the FTE Rate applicable to such FTE. Notwithstanding the foregoing, in the event a Party elects to provide a greater number of FTEs than are contemplated by the Shared Territory Commercialization Plan and Budget, the costs attributable to such additional FTEs shall not be included in FTE Cost.

1.75 **"FTE Rates**" means the rates (determined and adjusted in accordance with the applicable Co-Commercialization Agreement) to be used by both Parties in determining the cost of an FTE in the applicable functional area and region.

1.76 "GAAP" means, in the case of Gilead, United States Generally Accepted Accounting Principles and in the case of Galapagos, International Financial Reporting Standards, in each case as consistently applied by a Party in its accounting practices.

1.77 [...***...] has the meaning set forth in <u>Section 9.2(a)</u>.

1.78 "Galapagos" has the meaning set forth in the preamble to this Agreement.

1.79 "Galapagos Claims" has the meaning set forth in Section 11.2.

1.80 **"Galapagos Combination Product**" means a pharmaceutical product containing the Licensed Compound in combination with at least one active pharmaceutical ingredient other than the Licensed Compound, which active pharmaceutical ingredient is either a Galapagos Molecule or an Optioned Molecule under the Option, License and Collaboration Agreement.

1.81 "Galapagos Damages" has the meaning set forth in Section 11.2.

1.82 "Galapagos Foreground Know-How" means (a) all Information conceived, discovered, developed or otherwise made solely by Galapagos, optionally with its Affiliates or any Third Parties or any employees, consultants or agents of any of the foregoing, in performing the activities under this Agreement (including performance of the Development Plan or Shared Territory Commercialization Plan and Budget). For clarity, Galapagos Foreground Know-How shall exclude rights under any Galapagos Patents.

1.83 "Galapagos Indemnitees" has the meaning set forth in Section 11.2.

1.84 "**Galapagos Know-How**" means (a) all Information Controlled as of the Effective Date by Galapagos or its Affiliate(s) and reasonably necessary or useful to Exploit Licensed Compound or Licensed Products in the Field, and which is inherently linked to the Galapagos Patents, and (b) all Collaboration Know-How, and (c) all Galapagos Foreground Know-How. For clarity, Galapagos Know-How shall exclude rights under any Galapagos Patents.

1.85 "Galapagos Patents" means (a) the Existing Galapagos Patents (including, for clarity, [...***...]), (b) the Collaboration Patents, (c) any Patent that is not a Collaboration Patent and that is filed on or after the Effective Date, (d) those Patents that claim priority, directly or indirectly, in whole or in part, to any of the Patents in (a) and (b), that in each case (a), (b), (c) and (d) is Controlled by Galapagos or its Affiliate(s) during the Term and that is related to the Licensed Compound, the Licensed Product or a Gilead Combination Product, and that in each case (a), (b), (c), and (d) is reasonably necessary or useful to make, use, sell, offer for sale, import or export Licensed Compound or Licensed Products or Gilead Combination Products in the Field.

1.86 "Galapagos Technology" means the Galapagos Patents, Galapagos Know-How, Galapagos's interest in Joint Patents and Joint Foreground Know-How.

1.87 [...***...].

1.88 "Generic Product" shall mean a generic version of a product containing an active component (including the Licensed Compound) of a Gilead Combination Product or Licensed Product that is approved for marketing by a Third Party in the Field either: (i) pursuant to Section 505(j) of the FD&C Act (21 U.S.C. 355(j)), 505(b)(2) of the FD&C Act, (21 U.S.C. 355(b)(2)), or a foreign equivalent of either, by reference to a Regulatory Approval of such product, or (ii) pursuant to any other law or regulation where such approval is based on a demonstration of bio-equivalence or biosimilarity to such product. Under no circumstances shall a product be considered a Generic Product if Gilead or its Affiliates or its Sublicensees (a) transferred a Regulatory Approval of such product to such Third Party, (b) transferred an application for Regulatory Approval of such product to such Third Party in order to enable such Third Party to commercialize such product, except for a right of reference limited to qualifying the generic version for a government and/or nonprofit entity's program for providing medicines at no or low cost to countries in the Access Territory.

1.89 "Gilead" has the meaning set forth in the preamble to this Agreement.

1.90 "Gilead Background Patents" has the meaning set forth in Section 13.5(a).

1.91 "Gilead Claims" has the meaning set forth in Section 11.1.

1.92 "Gilead Combination Know-How" has the meaning set forth in Section 9.1(b).

1.93 "Gilead Combination Patents" has the meaning set forth in Section 9.1(b).

1.94 "**Gilead Combination Product**" means a pharmaceutical product containing the Licensed Compound in combination with at least one active pharmaceutical ingredient other than the Licensed Compound, which active pharmaceutical ingredient is Controlled by Gilead or its Affiliates. The term "in combination," covers instances where the Licensed Compound and at least one active pharmaceutical ingredient Controlled by Gilead or its Affiliates are administered in a single formulation and where the Licensed Compound and at least one active pharmaceutical ingredient Controlled by Gilead or its Affiliates are sold either as a fixed dose combination or with separate doses in a single package.

1.95 "Gilead Combination Technology" has the meaning set forth in Section 9.1(b).

1.96 "Gilead Damages" has the meaning set forth in <u>Section 11.1</u>.

1.97 "**Gilead Foreground Know-How**" means all Information conceived, discovered, developed or otherwise made solely by Gilead, optionally with its Affiliates or any Third Parties or any employees, consultants or agents of any of the foregoing, in performing the activities under this Agreement (including performance of the Development Plan or Shared Territory Commercialization Plan and Budget), and necessary for the Exploitation of the Licensed Compound or Licensed Products. For clarity, Gilead Foreground Know-How shall exclude rights under any Gilead Foreground Patents and Gilead Combination Know-How.

1.98 "Gilead Foreground Patents" has the meaning set forth in Section 9.1(b).

1.99 "Gilead Indemnitees" has the meaning set forth in <u>Section 11.1</u>.

1.100 "Gilead Technology" means Gilead Combination Technology, Gilead Foreground Know-How and any Patents arising from Gilead Combination Technology or Gilead Foreground Know-How and Gilead's interest in the Joint Patents and Joint Foreground Know-How.

1.101 "Global Commercialization Strategy" has the meaning set forth in Section 5.2(a)(i).

1.102 "Global Pricing Strategy" shall have the meaning set forth under Section 5.2(a)(ii).

1.103 "**Governmental Authority**" means any multi-national, federal, state, local, municipal or other government authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, court or other tribunal, as well as any securities exchange or securities exchange authority).

1.104 [...***...].

1.105 "HSR Act" means the U.S. Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

1.106 "**IND**" means (a) an Investigational New Drug Application as defined in the FD&C Act and applicable regulations promulgated thereunder by the FDA, or (b) the equivalent application to the equivalent Regulatory Authority in any other regulatory jurisdiction, the filing of which is necessary to initiate or conduct clinical testing of a pharmaceutical product in humans in such jurisdiction.

1.107 "Indemnified Party" has the meaning set forth in Section 11.3.

1.108 "Indemnified Person" means, in the case of Gilead, any Gilead Indemnitee, and in the case of Galapagos, any Galapagos Indemnitee.

1.109 "**Indemnifying Party**" has the meaning set forth in <u>Section 11.3</u>.

1.110 "Industry Transaction" has the meaning set forth on <u>Section 15.6(b)</u>.

1.111 "**Information**" means any data, results, and information of any type whatsoever, in any tangible or intangible form, including know-how, trade secrets, practices, techniques, methods, processes, inventions, developments, specifications, formulations, formulae, materials or compositions of matter of any type or kind (patentable or otherwise), software, algorithms, marketing reports, clinical and non-clinical study reports, regulatory submission documents and summaries, expertise, stability, technology, test data including pharmacological, biological, chemical, biochemical, toxicological, and clinical test data, analytical and quality control data, stability data, studies and procedures.

1.112 "Initial Co-Commercialization Indications" means [...***...].

1.113 "**Initial Co-Commercialization Product**" means that certain Licensed Product containing or comprising filgotinib as the sole active pharmaceutical ingredient, which, as of the Amendment Effective Date, is being Developed by or on behalf of Gilead under this Agreement.

1.114 [...***...].

1.115 [...***...].

1.116 [...***...].

1.117 "Joint Commercialization Committee" or "JCC" means the committee formed by the Parties as described in <u>Section 2.3(a)</u>.

1.118 "Joint Commercialization Costs" means [...***...].

1.119 "Joint Development Committee" or "JDC" means the committee formed by the Parties as described in Section 2.2(a).

1.120 "Joint Foreground Know-How" means, to the extent not Collaboration Know-How or Gilead Combination Know-How, all Information conceived, discovered, developed or otherwise made jointly by the Parties, optionally with their respective Affiliates, in performing the activities under this Agreement (including performance of the Development Plan or Shared Territory Commercialization Plan and Budget), and necessary for the Exploitation of the Licensed Products. Joint Foreground Know-How shall exclude Collaboration Know-How, rights under any Joint Patents and Gilead Combination Know-How.

1.121 "Joint Patents" has the meaning set forth in <u>Section 9.1(c)</u>.

1.122 "Joint Steering Committee" or "JSC" means the committee formed by the Parties as described in Section 2.1(a).

1.123 "Knowledge" means [...***...].

1.124 "License and Collaboration Agreement" has the meaning set forth in the Recitals.

1.125 "Licensed Compound" means the compound known as GLPG0634, or filgotinib, and any metabolite, salt, ester, hydrate, solvate, isomer, enantiomer, free acid form, free base form, crystalline form, co-crystalline form, amorphous form, pro-drug, racemate, polymorph, chelate, stereoisomer, tautomer, or optically active form of the foregoing.

1.126 "Licensed Product(s)" means any product, other than a Gilead Combination Product, which product contains a Licensed Compound. Licensed Product includes all such products containing the same Licensed Compound, alone or in combination with one or more active pharmaceutical ingredients, in any and all finished forms, presentations, delivery systems, strength, dosages, and formulations. By the term in combination or intended to include where the Licensed Compound and the one or more active pharmaceutical ingredients are sold either as a fixed dose combination or with separate doses in a single package.

1.127 "Licensed Territory" means all countries of the world other than the Shared Territory.

1.128 "Major Markets" means the following countries: [...***...].

1.129 "**Manufacture**" means, with respect to a Licensed Product, those manufacturing-related activities that support the Development (including the seeking and obtaining of Regulatory Approvals) and Commercialization of such Licensed Product, including manufacturing process development and scale-up, validation, qualification and audit of clinical and commercial manufacturing facilities, bulk production and fill/finish work, related quality assurance technical support activities and CMC Activities, and including, in the case of a clinical and commercial supply of such Licensed Product, the synthesis, manufacturing, processing, formulating, packaging, labeling, holding, quality control testing and release of such Licensed Product. "**Manufacturing**" has a correlative meaning.

1.130 "**Market Access Activities**" means pricing and reimbursement approvals as well as supporting activities, including payor advisory boards, health economic modelling, real world evidence generation, pricing research, pricing, reimbursement and value dossier preparation, negotiation, national and sub-national payor engagement and negotiations and other market access activities that are typical and customary in the pharmaceutical industry.

1.131 "**Market Access Costs**" means all [...***...] incurred by or on behalf of either Party that are reasonably and directly attributable to Market Access Activities. Notwithstanding the foregoing, Market Access Costs shall exclude [...***...].

1.132 "Marketing Authorization Application" or "MAA" means an application for Regulatory Approval in a country, territory or possession.

1.133 "Marks" has the meaning set forth in <u>Section 9.9</u>.

1.134 "**Material Communications**" means any material communications with a Regulatory Authority, including clinical study protocols and amendments thereto, meeting requests and materials, request for information and responses thereto, clinical hold notices, investigator's brochures, and supplemental NDA submissions.

1.135 "**Materials Transfer Agreement**" means that certain Materials Transfer Agreement by and between the Parties dated November 18, 2015, as amended.

1.136 "Medical Affairs Activities" means activities designed to ensure or improve appropriate medical use of, conduct medical education of, or further research regarding, a Licensed Product or Gilead Combination Product, as applicable, sold in the Territory, including by way of example: (a) activities of medical scientific liaisons who, among their other functions may (i) conduct service-based medical activities including providing input and assistance with consultancy meetings, recommend investigators for clinical trials and provide input in the design of such trials and other research related activities, and (ii) deliver non-promotional communications and conduct non-promotional activities including presenting new clinical trial and other scientific information; (b) grants to support continuing medical education, symposia, or Third Party research related to a Licensed Product or Gilead Combination Product, as applicable, in the Territory; (c) development, publication and dissemination of publications relating to a Licensed Product or Gilead Combination Product, as applicable, in the Territory; (d) medical information services provided in response to inquiries communicated via Sales Representatives or received by letter, phone call or email; (e) conducting advisory board meetings or other consultant programs; (f) the support of investigator-initiated trials; (g) establishment and implementation of risk, evaluation and mitigation and strategies (REMS); and (h) Voluntary Phase 4 Clinical Trials.

1.137 "**Medical Affairs Costs**" means all [...***...] incurred by or on behalf of either Party that are reasonably and directly attributable to Medical Affairs Activities, whether prior to or after receipt of Regulatory Approvals. Notwithstanding the foregoing, Medical Affairs Costs shall exclude [...***...].

1.138 "NDA" means a New Drug Application, as defined in the FD&C Act and applicable regulations promulgated thereunder by the FDA.

1.139 [...***...] has the meaning set forth in <u>Section 8.4</u>.

1.140 "Net Receipts" means [...***...].

1.141 "**Net Sales**" means [...***...].

1.142 "Nonclinical Studies" means all non-human animal studies, including preclinical studies and toxicology studies, of Licensed Products.

1.143 "**Operating Profit (or Loss)**" means, for a given period of time, [...***...]. For clarity, Operating Profit (or Loss) shall be determined prior to application of any income taxes, and if such terms are used individually, "**Operating Profit**" shall mean a positive Operating Profit (or Loss), and "**Operating Loss**" shall mean a negative Operating Profit (or Loss).

1.144 "Option, License and Collaboration Agreement" has the meaning set forth in the Recitals.

1.145 "Other Co-Commercialization Indications" means indications other than the Initial Co-Commercialization Indications and excluding [... ***...].

1.146 "**Other Indication**" means a discrete clinically recognized form of a disease or any precursor condition thereof for which at least [...***...] patients [...***...] would be eligible for treatment using the respective product. By way of example, the following diseases shall be considered Other Indications: [...***...] and [...***...].

1.147 "Party" or "Parties" has the meaning set forth in the preamble to this Agreement.

1.148 "**Patent**" means (a) any national, regional or international patent or patent application, including any provisional patent application, (b) any patent application filed either from such a patent, patent application or provisional application or from an application claiming priority from any of these, including any divisional, continuation, continuation-in-part, provisional, converted provisional, and continued prosecution application, (c) any patent that has issued or in the future issues from any of the foregoing patent applications ((a) and (b)), including any utility model, petty patent, design patent and certificate of invention, (d) any extension or restoration by existing or future extension or restoration mechanisms, including any revalidation, reissue, re-examination and extension (including any supplementary protection certificate and the like) of any of the foregoing patent applications ((a), (b) and (c)), and (e) any similar rights, including so-called pipeline protection, or any importation, revalidation, confirmation or introduction patent or registration patent of additions to any such foregoing patent application or patent.

1.149 "Patent Challenge" has the meaning set forth in Section 13.3(b)(i).

1.150 "Patent Committee" has the meaning set forth in Section 9.11.

1.151 **"Patent Costs**" means the out-of-pocket costs and expenses paid to outside legal counsel, patent offices and other governmental departments and other Third Parties, and filing and maintenance expenses, incurred in the preparation, filing, prosecution and maintenance of Patents, as well as re-examinations, reissues and the like with respect to any Patent, together with the conduct of interferences, the defense of oppositions and other similar proceedings with respect to any Patent.

1.152 "**Person**" means an individual, sole proprietorship, partnership, limited partnership, limited liability partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture or other similar entity or organization, including a government or political subdivision, department or agency of a government.

1.153 **"Phase 1 Clinical Trial**" means a human Clinical Trial of a product with the endpoint of determining initial tolerance, safety, pharmacokinetic or pharmacodynamic information in single dose, single ascending dose, multiple dose or multiple ascending dose regimens, which is prospectively designed to generate sufficient data (if successful) to commence a Phase 2 (as defined in 21 C.F.R. 312.21(b) as amended from time to time, or the corresponding foreign regulations) of such product, as further defined in 21 C.F.R. 312.21(a), as amended from time to time, or the corresponding foreign regulations.

1.154 "**Phase 2 Clinical Trial**" means a Clinical Trial of a Licensed Product or Gilead Combination Product, as applicable, regarding the safety, dose ranging and efficacy of a pharmaceutical product, which Clinical Trial is prospectively designed to generate sufficient data (if successful) to commence pivotal clinical trials, as described in 21 C.F.R. §312.21(b) (as amended or any replacement thereof), or a similar clinical study prescribed by the Regulatory Authorities in a foreign country.

1.155 "**Phase 3 Clinical Trial**" means a Clinical Trial of a Licensed Product on a sufficient number of subjects that is designed to establish that a pharmaceutical product is safe and efficacious for its intended use, and to determine warnings, precautions, and adverse reactions that are associated with such pharmaceutical product in the dosage range to be prescribed, which Clinical Trial is intended to support Regulatory Approval of such product, as described in 21 C.F.R. 312.21(c) (as amended or any replacement thereof), or a similar clinical study prescribed by the Regulatory Authorities in a foreign country.

1.156 "**Phase 4 Clinical Trial**" means (a) a Clinical Trial of a Licensed Product, conducted following commencement of a pivotal Clinical Trial for such Licensed Product whether or not required for receipt of approval of the NDA or MAA (whether such Clinical Trial is conducted prior to or after receipt of such approval), but that may be useful in support of the post-approval Exploitation of a Licensed Product or a Gilead Combination Product; or (b) a Clinical Trial of a Licensed Product or a Gilead Combination Product or such Gilead Combination Product, respectively, has been obtained from an appropriate Regulatory Authority whether or not due to a request or requirement of such Regulatory Authority. Investigator sponsored trials are part of Medical Affairs activities and do not constitute Phase 4 Clinical Trials.

1.157 "PhRMA Code" means the PhRMA Code on Interactions with Health Care Professionals.

1.158 "Pricing Matters" means all issues and decisions regarding (a) price, price terms and other contract terms with respect to Licensed Product or Gilead Combination Product sales in the Shared Territory, including discounts, rebates, other price concessions and service fees to payors and purchasers and (b) reimbursement programs applicable to a Licensed Product or Gilead Combination Product in the Shared Territory. For clarity, "Pricing Matters" includes all financial issues and financial decisions with respect to contracting with managed care entities, hospitals, pharmacies, group purchasing organizations, pharmacy benefit managers, and government, and specifically includes issues and decisions about the offer of discounts or rebates for formulary placement for Licensed Products or Gilead Combination Products, in all cases, as disclosed or discussed or agreed by or between the Parties only as provided in the Competition Law Guidelines.

1.159 "Product Infringement" has the meaning set forth in Section 9.3(a).

1.160 "**Promotional Materials**" means all sales representative training materials and all written, printed, graphic, electronic, audio or video matter, including journal advertisements, sales visual aids, leave-behind items, reprints, direct mail, internet postings and sites and broadcast advertisements intended for use or used by or on behalf of either Party or their respective Affiliates in connection with any promotion of a Licensed Product or Gilead Combination Product.

1.161 "**Publication**" has the meaning set forth in <u>Section 12.4(b)</u>.

1.162 "**Redacted Agreements**" has the meaning set forth in <u>Section 12.3(c)</u>.

1.163 "**Regulatory Approval**" means all approvals (including licenses, registrations or authorizations) from any applicable Regulatory Authority in a given country or countries (and, if applicable, the EU) necessary for the Manufacture as applicable, marketing, commercial distribution, importation and sale of a Licensed Product or a Gilead Combination Product for one or more indications in the Field other

than for Veterinary Uses and in such country or regulatory jurisdiction, which may include satisfaction of all applicable regulatory and notification requirements and, where applicable, labeling approval, but which, shall exclude any pricing and reimbursement approvals. Regulatory Approvals include approvals by Regulatory Authorities of MAAs or NDAs.

1.164 "**Regulatory Authority**" means, in a particular country or regulatory jurisdiction, any applicable Governmental Authority involved in granting Regulatory Approval, including (a) the FDA, (b) the EMA and (c) the European Commission or the successor of any such Governmental Authority.

1.165 "**Regulatory Exclusivity**" means any exclusive marketing rights or data exclusivity rights conferred by any Regulatory Authority with respect to a Licensed Product other than Patents, including, without limitation, rights conferred in the U.S. to an NDA holder under the Hatch-Waxman Act or the FDA Modernization Act of 1997 (including pediatric exclusivity), or rights similar thereto outside the U.S.

1.166 "**Regulatory Materials**" means regulatory applications, submissions, notifications, registrations, or other filings made to or with a Regulatory Authority that are necessary or reasonably desirable in order to Develop, manufacture, market, sell or otherwise Commercialize a Licensed Product in a particular country or regulatory jurisdiction. Regulatory Materials include INDs, MAAs and NDAs (as applications, but not the approvals with respect thereto).

1.167 "**Required Phase 4 Clinical Trial**" means a Phase 4 Clinical Trial that is conducted due to a request or requirement of a Regulatory Authority.

1.168 **"Regulatory Transition Date**" has the meaning set forth in <u>Section 4.1</u>

1.169 [...***...].

1.170 [...***...].

1.171 "**Reversion Product**" means, with respect to any Terminated Region(s), any Licensed Product that is or has been the subject of clinical Development or Commercialization hereunder.

1.172 "**Royalty Term**" has the meaning set forth in <u>Section 8.3(c)</u>.

1.173 [...***...] has the meaning set forth in <u>Section 13.3(a)(ii)</u>.

1.174 "Sales and Marketing Costs" means [...***...].

1.175 "**Sales Representative**" means a pharmaceutical sales representative engaged or employed by either Party to conduct Detailing and other promotional efforts with respect to the Licensed Products or Gilead Combination Products in the Shared Territory in accordance with the terms of this Agreement.

1.176 "SDEA" has the meaning set forth in <u>Section 4.7(a)</u>.

1.177 "SEC" means the U.S. Securities and Exchange Commission.

1.178 "Shared Program Activities" means any activities with respect to a Licensed Product or, to the extent included in a Co-Commercialization Agreement, Gilead Combination Product in the framework of the Co-Commercialization pursuant to the terms of the Co-Commercialization Agreement and conducted by either Party or any of its Affiliates, Sublicensees or subcontractors at any time on or after the Effective Date during the Term consisting of (a) the development (including Development) for the purpose of, or in support of, (i) obtaining or maintaining Regulatory Approval in the Shared Territory or (ii) Commercialization of any such Licensed Product or Gilead Combination Product in the Shared

Territory, in each case ((i) and (ii)) pursuant to any Development Plan, (b) Commercialization of any such Licensed Product or Gilead Combination Product in the Shared Territory, (c) Medical Affairs Activities with respect to any such Licensed Product or Gilead Combination Product in the Shared Territory or (d) the Manufacture of any such Licensed Product or Gilead Combination Product (including any intermediate thereof or any Licensed Compound or other material contained therein) for use in any activities under clause (a), (b) or (c). For clarity, the foregoing definition of Shared Program Activities shall not be construed as granting Galapagos any rights to perform activities in the Shared Territory other than those expressly set forth in this Agreement and the applicable Co-Commercialization Agreement.

1.179 "**Shared Program Damages**" means damages or other amounts payable by either Party (or any of its Indemnified Persons) to any Third Party claimant, as well as any reasonable attorneys' fees and costs of litigation incurred by either Party (or any of its Indemnified Persons) from Third Party claims that arise from or are based on Shared Program Activities, including such damages and other amounts (and attorneys' fees) from claims of infringement or a Third Party's Patent and other intellectual property rights; *provided*, *however*, *that* "Shared Program Damages" shall exclude any and all damages and other amounts (including attorneys' fees) for which a Party has an obligation to indemnify pursuant to <u>Section 11.1(a)</u> or <u>11.1(b)</u>, <u>11.2(c)</u> or <u>11.2(d)</u>.

1.180 "**Shared Territory**" means the EU5 Countries and the Benelux Countries, in each case other than countries that have been removed from the Shared Territory pursuant to this Agreement or the applicable Co-Commercialization Agreement.

1.181 **"Shared Territory Commercialization Plan and Budget**" means that certain plan and budget for Co-Commercialization Costs, which plan and budget is approved by the JSC as contemplated hereunder.

1.182 "Shared Territory JCC" has the meaning set forth in Section 2.6(a).

1.183 "**Specific Disclosures**" means the disclosures listed in <u>Exhibit K</u>.

1.184 "Standard Cost of Manufacturing" shall mean [...***...].

1.185 "Sublicense Agreement" has the meaning set forth in Section 7.2(b).

1.186 "**Sublicensee**" means any Third Party other than a Distributor, which Third Party is granted a sublicense by a Party under the rights licensed to such Party hereunder.

1.187 "Subscription Agreement" has the meaning set forth in the Recitals.

1.188 "Target Indications" means [...***...].

1.189 "Term" has the meaning set forth in <u>Section 13.1</u>.

1.190 "Terminated Region" has the meaning set forth in <u>Section 13.5</u>.

1.191 "Termination at Will" has the meaning set forth in Section 13.3(a)(i).

1.192 "Termination Notice Period" has the meaning set forth in Section 13.5(d)(i).

1.193 "**Territory**" means all countries in the world.

1.194 "Third Party" means any entity other than Galapagos or Gilead or an Affiliate of either of them.

1.195 "Transition Agreement" has the meaning set forth in <u>Section 13.5(e)</u>.

1.196 [...***...].

1.197 "U.S." means the United States of America (including all possessions and territories thereof).

1.198 "Valid Claim" means, with respect to a particular country, (a) any claim of an issued and unexpired Patent in such country that (i) has not been held permanently revoked, unenforceable or invalid by a decision of a court or governmental agency of competent jurisdiction, which decision is unappealable or unappealed within the time allowed for appeal and (ii) has not been abandoned, disclaimed, denied or admitted to be invalid or unenforceable through reissue or disclaimer or otherwise in such country, or (b) a claim of a pending Patent application that has been pending without issuance for a period not longer than [...***...] years from the earliest priority date of such application, which claim is being diligently prosecuted and has not been abandoned or finally disallowed without the possibility of appeal or re-filing of the application.

1.199 "Veterinary Use" means Exploitation of a Licensed Compound and any product containing a Licensed Compound for the prevention or treatment of veterinary medical conditions in animals. For clarity, humans are not animals for purposes of this definition.

1.200 "Voluntary Phase 4 Clinical Trial" means a Phase 4 Clinical Trial that is not a Required Phase 4 Clinical Trial.

ARTICLE 2

PROGRAM; GOVERNANCE

2.1 Joint Steering Committee.

(a) <u>Purpose; Formation</u>. The Parties hereby establish a joint steering committee (the "**JSC**") which will monitor and provide strategic oversight of the activities under this Agreement and facilitate communications between the Parties with respect to the Development, Manufacture and Commercialization of Licensed Compound, Licensed Products, all in accordance with this <u>Section 2.1</u>.

(b) <u>Composition</u>. Each Party shall initially appoint three (3) representatives to the JSC, all of whom will have sufficient seniority within the applicable Party to make decisions arising within the scope of the JSC's responsibilities. Each Party will also appoint an employee who is not a representative on the JSC to have alliance management responsibility (such employee, an "**Alliance Manager**") who will attend JSC meetings as observers. The Parties' initial representatives to the JSC are set forth on <u>Exhibit C</u>. The JSC may change its size from time to time if agreed by consensus among its members, *provided that* the JSC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JSC representatives at any time upon written notice to the other Party *provided, however, that* neither Party may replace a representative on the JSC (except for the Party's Alliance Manager) with an individual with lower seniority without the approval of the other Party, which approval shall not be unreasonably withheld, conditioned or delayed. The JSC may invite non-members to participate in the discussions and meetings of the JSC, *provided that* such participants shall have no voting authority at the JSC. The JSC shall have a chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos and Gilead. The initial chairperson shall be selected by [... ***...]. The role of the chairperson shall be to convene and preside at meetings of the JSC. The Alliance Managers shall work with the chairperson to prepare and circulate agendas and ensure the preparation and execution of meeting minutes. The chairperson shall have no additional powers or rights beyond those held by the other JSC representatives.

(c) <u>Specific Responsibilities</u>. In addition to its overall responsibility for monitoring and providing strategic oversight with respect to the Parties' activities under this Agreement, the JSC shall in particular:

(i) discuss the activities of the Parties under this Agreement;

(ii) review and approve each (A) Development Plan prepared by Gilead and (B) Shared Territory Commercialization Plan and Budget prepared by the Parties and reviewed by the Shared Territory JCC and, in the case of all plans, amendments thereto;

(iii) approve all Development Budgets and amendments thereto;

(iv) approve the Global Commercialization Strategy, and amendments thereto;

(v) review any proposal from the Shared Territory JCC regarding changes to [...***...] for applicable units of products;

(vi) discuss strategies regarding intellectual property, new indications and Gilead Combination Products;

(vii) discuss progress of Gilead Combination Products, as applicable;

(viii) attempt to resolve issues presented to it by, and disputes within, the JDC, JCC, Shared Territory JCC or any other subcommittee (but for clarity, certain subcommittee disputes shall be first escalated to the JCC as specified herein or in a Co-Commercialization Agreement); and

(ix) establish such additional joint subcommittees as it deems necessary to achieve the foregoing objectives.

(d) <u>Meetings</u>. The JSC shall meet at least once per calendar quarter during the Term unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JSC, the chairperson of the JSC shall prepare and circulate an agenda for such meeting; <u>provided</u>, <u>however</u>, <u>that</u> either Party may propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JSC (by videoconference, teleconference, or in person) by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JSC and Alliance Managers of both Parties to provide the members of the JSC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision on the matters to be considered. The Parties may mutually agree to additional meetings on shorter or longer notice in the event that matters arise requiring JSC consideration. The JSC may meet in person, by videoconference or by teleconference. Notwithstanding the foregoing, at least one (1) meeting per calendar year shall be in person unless the Parties mutually agree in writing to waive such requirement. In-person JSC meetings will be held at locations alternately selected by Galapagos and by Gilead. Each Party will bear the expense of its respective JSC members' participation in JSC meetings. Meetings of the JSC shall be effective only if at least one (1) representative of each Party (which representative is not such Party's Alliance Manager) is present or participating in such meeting. The Gilead Alliance Manager will be responsible for preparing reasonably detailed written minutes of all JSC meetings

that reflect material decisions made and action items identified at such meetings. The Gilead Alliance Manager shall send draft meeting minutes to each member of the JSC for review and approval within [...***...] Business Days after each JSC meeting. Such minutes will be deemed approved unless one or more members of the JSC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case the Gilead Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JSC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines as set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. Minutes will be officially endorsed by the JSC at the next JSC meeting, and will be signed by the Alliance Managers.

(e) <u>Decision-Making</u>. Subject to <u>Section 2.4</u>, in addition to resolving issues specifically delegated to it, the JSC shall have the authority to resolve disputes within the jurisdiction of the JDC, JCC and any other committees that the Parties may subsequently create to assist in governance of the Collaboration, but otherwise shall have no authority except where expressly specified in this Agreement or mutually agreed by the Parties in writing. The representatives from each Party will have, collectively, one (1) vote on behalf of that Party, and all decision making shall be by consensus. Disputes at the JSC shall be handled in accordance with <u>Section 2.4</u>.

(f) <u>Disbandment</u>. The JSC may be dissolved (i) upon the mutual agreement of the Parties or (ii) in the event of an Industry Transaction of Galapagos, where Gilead shall have the right at any time and for any reason, effective upon written notice, to disband the JSC pursuant to, but only to the extent provided in, <u>Section 15.6(c)</u>.

2.2 Joint Development Committee.

(a) <u>Formation</u>; <u>Composition</u>. Within [...***...] days after the Effective Date, the Parties shall establish a committee to oversee Development of Licensed Product(s) in the Territory in accordance with the Development Plan(s) for the same and to coordinate the Development activities of the Parties, and review and discuss the Development and Manufacture of Licensed Compound and Licensed Products (the "**JDC**"). Each Party shall initially appoint three (3) representatives to the JDC, with each representative having knowledge and expertise in the development of compounds and products similar to the Licensed Products and having sufficient seniority within the applicable Party to make decisions arising within the scope of the JDC's responsibilities. The JDC may change its size from time to time if agreed by consensus among its members, *provided that* the JDC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JDC representatives at any time upon written notice to the other Party. The JDC may invite non-members to participate in the discussions and meetings of the JDC, *provided that* such participants shall have no voting authority at the JDC. The JDC shall have a chairperson, who shall serve for a term of one (1) year, and who shall be selected alternately, on an annual basis, by Galapagos or Gilead. The initial chairperson shall be selected by [...***...]. The role of the chairperson shall be to convene and preside at meetings of the JDC and to ensure the preparation of minutes, but the chairperson shall have no additional powers or rights beyond those held by the other JDC representatives.

(b) <u>Specific Responsibilities of the JDC</u>. The JDC shall have the following responsibilities:

(i) Discuss all Development Plans prepared by Gilead (including Development Budgets), and all annual and interim amendments to Development Plans (including Development Budgets) for, Licensed Compound in the Territory;

(ii) review the conduct of the Development Plans;

(iii) discuss Development of new indications for Licensed Products in the Territory:

Gilead;

(iv) implement and review the overall strategy created for global Development and the design of all Clinical Trials and Nonclinical Studies conducted under each Development Plan;

(v) discuss whether and when to initiate or discontinue any Clinical Trial and any Nonclinical Study under each Development Plan, *provided that* nothing is intended to limit a Party's ability to comply with Applicable Law or manage subject safety;

(vi) review the conduct of all Clinical Trials and Nonclinical Studies under each Development Plan, including Required Phase 4 Clinical Trials or any other Phase 4 Clinical Trial included in Development;

(vii) facilitate the flow of information between the Parties with respect to the Development of Licensed Compound;

(viii) implement and review the overall strategy regarding Regulatory Approval of Licensed Products in the Territory created by

(ix) discuss Manufacturing, including progress with formulation, validation scaleup, and other activities to maintain supply;

(x) review the regulatory strategy with respect to discussions with and commitments to or agreements with Regulatory Authorities (including post-approval commitments) with respect to risk management or Required Phase 4 Clinical Trials or any other Phase 4 Clinical Trial included in Development;

(xi) review any material submission to, or any material agreement with or material commitment made to, a Regulatory Authority by Gilead with respect to a Licensed Product, such as any NDA or MAA, or any submission, agreement or commitment with respect to Licensed Product labeling, any risk management plans, any Required Phase 4 Clinical Trial or any other Phase 4 Clinical Trial included in Development or other post-approval commitment for such Licensed Product, in each case with respect to the Major Markets;

(xii) facilitate the flow of information between the Parties with respect obtaining Regulatory Approval for Licensed Products; and

(xiii) review, discuss and coordinate the Parties' scientific presentation and publication strategy relating to the Licensed Products in the Territory.

(c) <u>Meetings</u>. The JDC shall meet at least once per calendar quarter, unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JDC, the chairperson of the JDC shall prepare and circulate an agenda for such meeting; <u>provided</u>, <u>however</u>, <u>that</u> either Party shall be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JDC (by videoconference, teleconference, or in person) by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JDC to provide the members of the JDC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision. The JDC

may meet in person, or at the request of either Party, by videoconference, or by teleconference. In-person JDC meetings will be held at locations alternately selected by Galapagos and by Gilead or at any other location mutually agreed by the members of the JDC. Each Party shall report to the JDC on all material issues relating to the Development of Licensed Products for and in the Territory promptly after such issues arise. Each Party will bear the expense of its respective JDC members' participation in JDC meetings. The chairperson will be responsible for preparing reasonably detailed written minutes of JDC meetings that reflect all decisions made and action items identified at such meetings. The JDC chairperson shall send meeting minutes to each member of the JDC for review and approval and to the Alliance Managers for informational purposes within [...***...] Business Days after each JDC meeting. Minutes will be deemed approved unless one or more members of the JDC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case the Gilead Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JDC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. Minutes will be officially endorsed by the JDC at the next JDC meeting, and will be signed by the Alliance Managers.

(d) <u>Decision-Making</u>. Subject to the remainder of this <u>Section 2.2(d)</u> and <u>Section 2.4</u>, the JDC shall act by consensus. The representatives from each Party will have, collectively, one (1) vote on behalf of that Party. If the JDC cannot reach consensus on an issue that comes before the JDC within [...***...] days of the meeting such issue was raised and over which the JDC has oversight, then either Party may refer such matter to the JSC for resolution in accordance with <u>Section 2.1(e)</u> and 2.4.

(e) <u>Disbandment</u>. Upon Regulatory Approval of the last Licensed Product developed pursuant to the Development Plan, unless otherwise mutually agreed in writing, the JDC shall have no further responsibilities or authority under this Agreement and will be considered dissolved by the Parties. Additionally, the JDC may be dissolved upon the mutual agreement of the Parties or in the event of an Industry Transaction of Galapagos, where Gilead shall have the right at any time and for any reason, effective upon written notice, to disband the JDC pursuant to, but only to the extent provided in, <u>Section 15.6(c)</u>.

2.3 Joint Commercialization Committee.

(a) <u>General</u>. With respect to Licensed Products and Gilead Combination Products, as applicable in the Shared Territory, within [...***...] days of the Effective Date, the Parties shall establish a committee to oversee Commercialization of Licensed Products in the Territory, and review and discuss the Commercialization of the Licensed Compound, Licensed Products and Gilead Combination Products, as applicable in the Shared Territory (the "JCC").

(b) <u>Formation; Composition</u>. Each Party shall initially appoint three (3) representatives to the JCC, with each representative having knowledge and expertise in the commercialization of products similar to the Licensed Products and Gilead Combination Products, as applicable in the Shared Territory, and having sufficient seniority within the applicable Party to make decisions arising within the scope of the JCC's responsibilities. The JCC may change its size from time to time if agreed by consensus among its members, *provided that* the JCC shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its JCC representatives at any time upon written notice to the other Party. The JCC may invite non-members to participate in the discussions and meetings of the JCC, *provided that* such participants shall have no voting authority at the JCC. The JCC shall have a chairperson, who shall be selected by Gilead. The role of the chairperson shall be to convene and preside at meetings of the JCC and to ensure the preparation of minutes, but the chairperson shall have no additional powers or rights beyond those held by the other JCC representatives.

(c) <u>Specific Responsibilities of the JCC</u>. Subject to <u>Section 2.3(f)</u>, the JCC shall have the following responsibilities:

(i) review the Global Commercialization Strategy and Global Pricing Strategy and updates thereto, provide feedback to the Parties, and submit the final Global Commercialization Strategy and Global Pricing Strategy for approval by the JSC;

(ii) review the plan prepared by Gilead for the Licensed Product brand in the United States and progress against plan after launch;

(iii) review and discuss uses of the Marks with respect to Licensed Products in the Territory;

(iv) attempt to resolve issues presented to it by, and disputes within, the Shared Territory JCC; and

(v) manage any activities related to adherence by the Parties to the corporate compliance programs required to be maintained pursuant to <u>Section 5.2(b)(ix)</u>.

(d) <u>Meetings</u>. The JCC shall meet at least once per calendar quarter, unless the Parties mutually agree in writing to a different frequency. No later than [...***...] Business Days prior to any meeting of the JCC, the chairperson of the JCC shall prepare and circulate an agenda for such meeting; provided, however, that either Party shall be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Either Party may also call a special meeting of the JCC (by videoconference, teleconference, or in person) by providing at least [...***...] Business Days prior written notice to the other Party if such Party reasonably believes that a significant matter must be addressed prior to the next scheduled meeting, in which event such Party shall work with the chairperson of the JCC to provide the members of the JCC no later than [...***...] Business Days prior to the special meeting with an agenda for the meeting and materials reasonably adequate to enable an informed decision. The JCC may meet in person, by videoconference, or by teleconference. In-person JCC meetings will be held at locations alternately selected by Galapagos and by Gilead or at any other location mutually agreed by the members of the JCC. Meetings of the JCC shall be effective only if at least one (1) representative of each Party is present or participating in such meeting. Each Party shall report to the JCC on all material issues relating to the Commercialization of Licensed Products and Gilead Combination Products, as applicable, promptly after such issues arise. Each Party will bear the expense of its respective JCC members' participation in JCC meetings. The chairperson will be responsible for preparing reasonably detailed written minutes of JCC meetings that reflect all decisions made and action items identified at such meetings. The JCC chairperson shall send meeting minutes to each member of the JCC for review and approval and the Alliance Managers for informational purposes within [...***...] Business Days after each JCC meeting. Minutes will be deemed approved unless one or more members of the JCC objects (whereby an objection by e-mail shall suffice) to the accuracy of such minutes within [...***...] Business Days of receipt, in which case the Gilead Alliance Manager shall amend the draft meeting minutes accordingly and send the revised draft meeting minutes to each member of the JCC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines set out in the immediately preceding sentence shall apply to such revised draft meeting minutes. Minutes will be officially endorsed by the JCC at the next JCC meeting, and will be signed by the Alliance Managers.

(e) <u>Decision-Making</u>. Subject to the remainder of this <u>Section 2.3(e)</u> and <u>Section 2.4</u>, the JCC shall act by consensus. The representatives from each Party will have, collectively, one (1) vote on behalf of that Party. If the JCC cannot reach consensus on an issue that comes before the JCC within [...***...] days of the meeting such issue was raised and over which the JCC has oversight, then either Party may refer such matter to the JSC for resolution in accordance with <u>Section 2.1(e)</u> and 2.4.

(f) <u>Disbandment</u>. The JCC may be dissolved (i) upon the mutual agreement of the Parties, or (ii) in the event of an Industry Transaction of Galapagos, Gilead shall have the right at any time and for any reason, effective upon written notice, to disband the JCC pursuant to, but only to the extent provided in, <u>Section 15.6(c)</u>.

2.4 Resolution of Committee Disputes.

(a) <u>Within Operating Committees</u>. If a dispute arises which cannot be resolved within the Committees, then if such dispute relates to a matter within the jurisdiction of the applicable Committee, the representatives of either Party may cause such matter to be referred to the JSC for resolution as provided in <u>Section 2.1(e)</u>.

(b) <u>Within the JSC</u>. Subject to the exceptions specified below in this <u>Section 2.4(b)</u>, all decisions within the JSC (whether originating there, or referred to it by an operating Committee) shall be made by consensus. If a matter is referred by an operating Committee to the JSC, the JSC shall use good faith efforts, in compliance with <u>Section 2.4(d)</u>, to resolve promptly such matter. If the JSC is unable to reach consensus on any issue for which it is responsible, other than those addressed in the last sentence of this <u>Section 2.4(b)</u>, within [...***...] Business Days after a Party affirmatively states that a decision needs to be made, if the matter relates to [...***...] either Party may elect, by written notice to the other Party (the "**Executive Officer Referral Notice**") to submit such issue to the Parties' Executive Officers in accordance with Section 2.4(c). [...***...]

(c) <u>Referral To Executive Officers</u>. If a Party makes an election under <u>Section 2.4(b)</u> to refer a matter to the Executive Officers, the JSC shall submit, in writing and within [...***...] days of the Executive Officers Referral Notice, the respective positions of the Parties to their respective Executive Officers. Such Executive Officers shall use good faith efforts, in compliance with <u>Section 2.4(d)</u>, to resolve promptly such matter, which good faith efforts shall include at least one meeting between such Executive Officers within [...***...] days after the JSC's submission of such matter to them. [...***...]

(d) <u>Good Faith</u>. In conducting themselves on Committees, and in exercising their rights under this <u>Section 2.4</u>, all representatives of both Parties shall consider reasonably and in good faith all input received from the other Party. [...***...].

2.5 <u>Appointment of Alliance Managers</u>. Each Party shall appoint an appropriately qualified individual to serve as Alliance Manager under this Agreement. Such persons shall endeavor to assure clear and responsive communication between the Parties and the effective exchange of information, and may serve as a single point of contact for any matters arising under this Agreement. The Alliance Managers may attend meetings of all Committees and subcommittees under this Agreement. The Alliance Managers shall not have any authority under this Agreement.

2.6 Shared Territory Governance.

(a) <u>Shared Territory JCC; Formation; Composition</u>. Within [...***...] days after the Amendment Effective Date, the Parties shall establish a joint commercialization committee for the Shared Territory (the "**Shared Territory JCC**") whose role shall be to coordinate and integrate the activities of, and to facilitate the communication and exchange of information between, the Parties with respect to Co-Commercialization of Licensed Products and if included in a Co-Commercialization Agreement, Gilead Combination Products for the Shared Territory, including (i) defining and overseeing the execution of a strategic plan for each country in the Shared Territory, (ii) establishing an operational framework and overseeing launch readiness in each country in the Shared Territory JCC, with each Party's representatives having knowledge and expertise in

Commercialization (including marketing, Medical Affairs Activities, and Market Access Activities) with respect to compounds and products similar to the Licensed Products or applicable Gilead Combination Products and having sufficient seniority to make decisions within the scope of the Shared Territory JCC's responsibilities. The Shared Territory JCC may invite non-members to participate in the discussions and meetings of the Shared Territory JCC, *provided that* such participants shall have no voting authority at the Shared Territory JCC. For any Pricing Matters before the Shared Territory JCC, each Party may substitute one or more of its representatives to the Shared Territory JCC with a representative having knowledge and expertise with respect to pricing, *provided* that the foregoing shall not change the total number of representatives of each Party on the Shared Territory JCC or effect the Party's voting rights under <u>Section 2.6(d)</u> with respect to Pricing Matters. The Shared Territory JCC may change its size from time to time if agreed by consensus among its members; *provided that* the Shared Territory JCC representatives at any time upon written notice to the other Party. The Shared Territory JCC shall have a chairperson, who shall be selected alternately for each Shared Territory JCC meeting by Galapagos and Gilead. The initial chairperson shall be selected by [...***...]. The role of the chairperson shall be solely to convene and preside at meetings of the Shared Territory JCC meeting by Gother Shared Territory JCC and to ensure the preparation of minutes, and the chairperson shall have no authority, power or rights beyond those of other Shared Territory JCC meetings.

(b) <u>Specific Responsibilities of the Shared Territory JCC</u>. The Shared Territory JCC shall have the following responsibilities, which may be assigned to expert sub-committees with equal expert representation by both Parties for joint decision, including a Shared Territory pricing committee:

(i) adapt the Global Commercialization Strategy for execution in the Shared Territory;

(ii) review the draft Shared Territory Commercialization Plan and Budget prepared by the Parties, provide feedback to the Parties, and submit the final draft Shared Territory Commercialization Plan and Budget for approval by the JSC;

(iii) oversee Co-Commercialization in the Shared Territory, including implementation of the Shared Territory Commercialization Plan and Budget, including implementation of any hiring plans contained therein and the activities of the Joint Teams;

(iv) with respect to the EU5 Countries, (A) formulating a regional pricing strategy based on the Global Pricing Strategy, including establishing the public list price and net pricing parameters for the applicable Licensed Product or Gilead Combination Product in each applicable country, and (B) considering requests from the Joint Teams for approval to deviate from such EU5 Countries' pricing strategy;

(v) with respect to the Benelux Countries, making non-binding recommendations with respect to the public list price parameters (such public list prices to be determined solely by Gilead) and net pricing parameters (such net prices to be determined solely by Galapagos) for the applicable Licensed Product or Gilead Combination Product in each applicable country;

(vi) provide a forum for discussion of any compliance-related concerns in the Shared Territory raised by the Joint Teams or either Party and establish appropriate plans to address any compliance concerns relating to the Shared Territory;

(vii) review the costs associated with Distribution Costs on an annual basis and propose changes to the JSC, if any, to [...***...];

(viii) review the KPIs set forth in the applicable Shared Territory Commercialization Plan and Budget, including [...***...], and [...

***...]; and

(ix) perform such other functions as are specified in this Agreement or the applicable Co-Commercialization Agreement.

(c) <u>Meetings</u>. The Shared Territory JCC shall meet (i) in person for at least one hour once per month or as otherwise agreed by the Parties and (ii) by videoconference or by teleconference for at least one hour every second week, or, in each case, at such other frequency as is agreed by the Parties. No later than [...***...] Business Days prior to any meeting of the Shared Territory JCC, the chairperson shall prepare and circulate an agenda for such meeting; *provided, however, that* either Party shall be free to propose additional topics to be included on such agenda, either prior to or in the course of such meeting. Meetings of the Shared Territory JCC shall be effective only if at least one representative of each Party is present or participating in such meeting. Each Party will bear the expense of its respective Shared Territory JCC members' participation in Shared Territory JCC will be responsible for preparing reasonably detailed written minutes of such meetings that summarize the discussions had and action items identified at such meetings. The Shared Territory JCC chairperson shall send meeting minutes to each member of the Shared Territory JCC for review and approval within [...***...] Business Days of receipt, in which case the Shared Territory JCC for review and approval within [...***...] Business Days of receipt of the objection(s). The same review procedures and timelines set out in the immediately preceding sentence shall apply to such revised draft meeting minutes.

(d) <u>Decision-Making</u>. Subject to the remainder of this <u>Section 2.6(d)</u> and <u>Section 2.4</u>, the Shared Territory JCC shall act by consensus. The representatives from each Party will have, collectively, one (1) vote on behalf of that Party. If the Shared Territory JCC cannot reach consensus on an issue that comes before the Shared Territory JCC within [...***...] days of the meeting at which such issue was raised and over which the Shared Territory JCC has oversight, then either Party may refer such matter to the JCC for resolution in accordance with <u>Sections 2.1(e)</u> and <u>2.4</u>; *provided*, *however*, that in the case of any matters relating to pricing, the decision shall be expedited as needed.

(e) Joint Teams. For each country in the EU5 Countries, no later than [...***...] months prior to the anticipated launch of the Initial Licensed Product in such country, the Parties shall establish a joint project team (each, a "Joint Team") whose role shall be to coordinate the Parties' respective Co-Commercialization activities in such country, and to facilitate the communication and exchange of information between the Parties, with respect to the Commercialization of Licensed Products and Gilead Combination Products in such country. For each country in the EU5 Countries, each Party shall initially appoint at least one member to the Joint Team for such country. It is expected that the members of the Joint Teams will increase over time as reasonably necessary to ensure that both Parties are involved in and coordinate regarding the Commercialization of the Licensed Product in such country. To achieve such involvement and coordination, the Shared Territory JCC may change the size of the Joint Teams if agreed by consensus among the Shared Territory JCC's members; *provided that* each Joint Team shall consist at all times of an equal number of representatives of each of Galapagos and Gilead. Each Party may replace its Joint Team representatives in a country at any time upon written notice to the other Party with other representatives in such country, *provided that* such Party continues to have adequate representation to remain involved in and to coordinate regarding the Commercialization of the Licensed Products and Gilead Combination Products in such country. Each Joint Team shall have a chairperson, who shall be selected by [...***...]. The role of the chairperson shall be solely to convene and preside at meetings of the Joint Team and to ensure the preparation of minutes, and the chairperson shall have no authority, power or rights beyond those of other Joint Team members.

2.7 <u>General Committee Authority</u>. Each Committee shall have solely the powers expressly assigned to it in this <u>Article 2</u> and elsewhere in this Agreement. No Committee shall have any power to amend, modify, or waive compliance with this Agreement. It is expressly understood and agreed that the [...***...], so as to resolve a disagreement or deadlock on a Committee for any matter will not authorize either Party to perform any function or exercise any decision-making right not delegated to a Committee or such Party, and that neither Galapagos nor Gilead shall have any right to unilaterally modify or amend, or waive its own compliance with, the terms of this Agreement.

2.8 <u>Joint Asset Committee</u>. The Parties shall consider from time to time if it would be mutually beneficial to establish a single global joint committee to oversee both the Development and Commercialization of the Licensed Product and to merge the plans under this Agreement into a consolidated global asset strategy plan to be reviewed and approved by such joint committee.

2.9 Local Pricing Committee. For each EU5 Country, the applicable Co-Commercialization Agreement will establish a local pricing committee established with at least one representative from each Party to allow the Parties' local representatives to establish pricing and discounting for such country. The local pricing committee must seek approval from the Shared Territory JCC for any deviations from pricing parameters established by the Shared Territory JCC (or, if applicable, Shared Territory pricing committee). The local pricing committees will make decisions by consensus and any disputes will be escalated to the Shared Territory JCC.

ARTICLE 3

DEVELOPMENT

3.1 <u>Overview of Development</u>. The Parties' respective responsibilities for the Development of the Licensed Products and Gilead Combination Products are set forth in this <u>Article 3</u> and in the Development Plan. Gilead shall be primarily responsible for Development and seeking Regulatory Approval of the Licensed Product and Gilead Combination Products in the Territory and shall use Commercially Reasonable Efforts with respect thereto for the first Licensed Product in each of the Major Markets in each of the Target Indications. Notwithstanding anything to the contrary in this Agreement, the immediately foregoing obligation with respect to Commercially Reasonable Efforts in the Major Markets and the Target Indications shall be Gilead's sole diligence obligation with respect to Development of Licensed Products and Gilead Combination Products, and Gilead shall have the sole right, but not any obligation, to Develop other Licensed Products and Gilead Combination Products in other countries or for other indications and no diligence obligation of Gilead shall arise under this Agreement with respect to Development of Gilead Combination Products. Galapagos shall use Commercially Reasonable Efforts as requested by Gilead to assist Gilead with Development activities with respect to [...***...].

3.2 Development Plans.

(a) <u>General</u>. All Development of the Licensed Product for any indication, as applicable, pursuant to this Agreement shall be conducted in accordance with the terms of this Agreement and a development plan and budget (each such plan and budget, a "**Development Plan**") created by Gilead substantially in a form and substance typically employed by Gilead for development plans for products at a similar stage of development as the applicable Licensed Product. By way of example, a Development Plan may describe (A) the proposed overall program of Development for the Licensed Compound and

Licensed Product, including Clinical Trials and Nonclinical Studies, if any, and, regulatory plans and other elements of obtaining Regulatory Approval(s) in the Territory; (B) with respect to Clinical Trials, the number of arms, number of subjects per arm, comparator treatment and proposed dosage of the Licensed Product; (C) the anticipated start dates and data availability dates of such Clinical Trials and Nonclinical Studies, and anticipated timelines for filing of applications for Regulatory Approvals in the Territory; (D) the respective roles and responsibilities of each Party in connection with such activities; and (E) a detailed budget for all such activities in the Territory, including cost sharing between the Parties. In the event of any inconsistency between a Development Plan and this Agreement, the terms of this Agreement shall prevail. The initial Development Plan is attached hereto as Exhibit D.

(b) <u>Amendments to the Development Plan</u>. On an annual basis, or more often as Gilead deems appropriate, Gilead, in accordance with its usual practices with respect to product development plans, shall prepare any needed amendments to the then-current Development Plan for review and approval by the JSC.

3.3 <u>Operational Responsibilities for Development</u>. Unless the Parties agree in writing upon an alternate allocation of responsibility or as set forth in the Development Plan, (a) Gilead shall hold primary responsibility for completing all Development activities relating to Licensed Compound and Licensed Products in the Territory and shall use Commercially Reasonable Efforts with respect thereto for the first Licensed Product in each of the Major Markets in the Target Indications, and (b) Galapagos shall use Commercially Reasonable Efforts to complete any Development activities assigned to it in the Development Plan relating to the Licensed Compound and Licensed Products in the Territory. Without limiting the foregoing, [...***...].

3.4 <u>Development Costs</u>. The Parties shall share Development Costs incurred by or on behalf of either Party or its Affiliates after the Effective Date solely to the extent related to Development of the Licensed Compounds or Licensed Products in the Territory, such sharing to be as follows: (1) for Development Costs incurred prior to the Amendment Effective Date, Gilead shall be responsible for eighty percent (80%) and Galapagos shall be responsible for twenty percent (20%) and (2) for Development Costs incurred on or after the Amendment Effective Date, each Party shall be responsible for fifty percent (50%) (regardless of the date of invoice or payment). Within [...***...] Business Days after the end of each calendar quarter, each Party shall provide to the other Party a report in reasonable detail of any Development Costs incurred by such Party in such calendar quarter for each Licensed Product. The Development Costs so reported will be used for the calculation of the 80%/20% split or the 50%/50% split for the Development Costs, as applicable. Within [...***...] Business Days after the end of each calendar quarter for each Licensed detail regarding such Development Costs incurred by each Party for such calendar quarter. Within [...***...] days following receipt of such report, the Party whose Development Cost expenditures exceed the portion of the total such expenditures by both Parties for such calendar quarter allocated to such Party in this <u>Section 3.4</u> shall invoice the other Party for the amount of funds necessary to account for such excess. The Party receiving such invoice shall pay it not later than [...***...] days following receipt thereof. For clarity, Gilead shall be solely responsible for all costs and expenses incurred by or on behalf of Gilead in the Development of Gilead Combination Products in the Territory. [...***...].

3.5 <u>Development Reports</u>. Each Party shall keep the JDC reasonably informed regarding the progress and results of Development activities for Licensed Compound, Gilead Combination Products and Licensed Products in the Territory, including by providing an annual report in reasonable detail of results versus goals (as such goals are set forth in the Development Plan(s)) as, in the case of Gilead, is typically generated by Gilead regarding its product development efforts.

3.6 <u>Clinical Trial Reporting</u>. Each Party agrees that (a) each Clinical Trial conducted pursuant to a Development Plan, as applicable, that is required to be posted pursuant to Applicable Law or applicable industry codes, including the PhRMA Code, on clinicaltrials.gov or any other similar registry shall be so posted, and (b) all results of such Clinical Trials that are necessary for obtaining a Regulatory Approval for a Licensed Product or Gilead Combination Product in the Territory shall be posted on clinicalstudyresults.org and on any other registry with requirements consistent with the registration and publication guidelines of the International Committee of Medical Journal Editors, to the extent required. All data and Information generated under such Clinical Trial posted on clinicalstudyresults.org or any other registry pursuant to this <u>Section 3.6</u> shall be subject to <u>Section 12.4(b)</u> as if such posting were a Publication.

3.7 Development Records. As promptly as reasonably practicable following the Effective Date, and as promptly as reasonably practicable on an ongoing basis with respect to newly-created development records, Galapagos shall use Commercially Reasonable Efforts to transfer to Gilead copies of all development records Controlled by Galapagos as of the Effective Date that are necessary or useful for Gilead to Exploit the Licensed Compound or Licensed Product under this Agreement. Each Party shall maintain complete and accurate records (in the form of technical notebooks or electronic files where appropriate) of all work conducted by it under the Development Plans and all Information resulting from such work. Such records shall fully and properly reflect all work done and results achieved in the performance of the Development Plans in sufficient detail and in good scientific manner appropriate for patent and regulatory purposes. Each Party shall have the right to access such records maintained by the other Party to the extent reasonably necessary to perform obligations (or in Gilead's case, to exercise rights) under this Agreement. The JDC shall determine the means by which such access will be provided.

3.8 <u>Subcontracts</u>. Gilead may perform any of its Development, Manufacturing, regulatory, and Commercialization obligations under this Agreement, and Galapagos my perform any of its Development obligations under this Agreement, in each case through one or more subcontractors or consultants, *provided that* (a) such Party remains responsible for the work allocated to, and payment to, such subcontractors and consultants to the same extent it would if it had done such work itself; (b) the subcontractor or consultant undertakes in writing obligations of confidentiality and non-use regarding Confidential Information that are substantially the same as those undertaken by the Parties with respect to Confidential Information pursuant to <u>Article 12</u> hereof; and (c) such Party undertakes all reasonable efforts to provide that the subcontractor or consultant undertakes in writing to assign or exclusively license back (with the right to sublicense) all intellectual property with respect to Licensed Compound or Licensed Products developed in the course of performing any such work to such Party. Each Party may also subcontract work on terms other than those set forth in this <u>Section 3.8</u> with the prior approval of the other Party.

3.9 <u>Combination of the Licensed Compound with Molecules under the Option, License and Collaboration Agreement</u>. Notwithstanding anything to the contrary in this Agreement or in the Option, License and Collaboration Agreement:

(a) Neither Party will [...***...].

(b) In the event the Parties [...***...], the Parties shall [...***...].

ARTICLE 4

REGULATORY MATTERS

4.1 <u>Ownership of Regulatory Filings</u>. Gilead will own all INDs, applications for Regulatory Approval and related regulatory documentation in the Territory with respect to any Licensed Product or Gilead Combination Product. As soon as reasonably practicable but not later than [...***...] months after the Effective Date, Galapagos will assign and transfer to Gilead all INDs, applications for Regulatory Approval, related regulatory documentation submitted to any Regulatory Authority in the Territory, as well as any safety and clinical databases, with respect to such Licensed Product and any Regulatory Approvals and related documentation, in each case, that is in the possession or Control of Galapagos. Each Party will submit all filings, letters and other documentation necessary to effect such assignments and transfers to the applicable Regulatory Authority. Until the date that the transfer of all INDs and related regulatory documents filed with or submitted to any Regulatory Authority in the Territory documents filed with or submitted to any Regulatory Authority in the Territory that related to such Licensed Product becomes effective (the "**Regulatory Transition Date**"), Galapagos shall, under the direction of Gilead, handle all matters related to each Licensed Product involving Regulatory Authorities, to the extent not yet assigned and transferred to Gilead, and shall keep Gilead fully informed of all regulatory matters relating to any Licensed Product in the Territory including providing Gilead with reasonable advance notice of all formal meetings and teleconferences with Regulatory Authorities in the Territory pertaining to any Licensed Product. Galapagos shall permit Gilead to have, at Gilead's expense, representatives of Gilead to lead such formal meetings and teleconferences with Regulatory Authorities in the Territory pertaining to such Licensed Product.

4.2 <u>Responsibility for Regulatory Matters</u>. Following the Regulatory Transition Date, Gilead will be solely responsible for all regulatory matters relating to any Licensed Product or Gilead Combination Product in the Territory, and shall use Commercially Reasonable Efforts with respect to such regulatory matters for the first Licensed Product in each Major Market in each Target Indication, and Galapagos shall use Commercially Reasonable Efforts as requested by Gilead to assist Gilead with such regulatory matters. Notwithstanding anything to the contrary in this Agreement, the immediately foregoing obligation with respect to Commercially Reasonable Efforts in the Major Markets and the Target Indications shall be Gilead's sole diligence obligation with respect to regulatory matters for ther Licensed Products and Gilead Combination Products, and Gilead shall have the sole right, but not any obligation, with respect to regulatory matters for other Licensed Products and Gilead Combination Products in other countries or for other indications and no diligence obligation of Gilead shall arise under this Agreement with respect to seeking or obtaining Regulatory Approvals for Gilead Combination Products. Gilead's sole responsibility shall include (i) overseeing, monitoring and coordinating regulatory actions, communications and filings with, and submissions to, Regulatory Authority in the Territory with respect to Licensed Products and Gilead Combination Products; (ii) interfacing, corresponding and meeting with Regulatory Authorities in the Territory with respect to Licensed Products and Gilead Combination Products; and (iv) maintaining regulatory filings in the Territory with respect to Licensed Products and Gilead Combination Products; and (iv) maintaining and submitting records required to be maintained or required to be submitted to any Regulatory Authority in the Territory with respect to Licensed Products and Gilead Combination Products.

4.3 <u>Communications with Regulatory Authorities</u>. Following the Regulatory Transition Date, within [...***...] days after receipt of any Material Communication from a Regulatory Authority in a Major Market with respect to any Licensed Product, Gilead will provide Galapagos, through its Alliance Manager, with a brief written description of the principal issues raised in such Material Communication. Upon Galapagos' reasonable request after receiving a notice from Gilead in accordance with the immediately preceding sentence, Gilead will provide to Galapagos complete copies of such correspondence within a reasonable period of time following such request. Gilead will allow Galapagos [...***...] Business Days to review and comment on Gilead's proposed response to any such Material Communications with any Regulatory Authority in the Major Markets with respect to any Licensed Product in advance of the transmission of such response, and Gilead will reasonably consider all comments timely provided by Galapagos in connection therewith.

4.4 <u>Meetings with Regulatory Authorities</u>. Following the Regulatory Transition Date, Gilead shall provide Galapagos with reasonable advance notice of all formal meetings and teleconferences with the FDA and EMA pertaining to any Licensed Product, or with as much advance notice as practicable under the circumstances. Gilead shall permit Galapagos to have, at Galapagos' expense, two representatives of Galapagos attend as observer, such formal meetings and teleconferences with the FDA and EMA pertaining to such Licensed Product.

4.5 <u>Submissions</u>. With respect to each Licensed Product, Gilead will allow Galapagos [...***...] Business Days to review and comment on all filings and other submissions to Regulatory Authorities in the Major Markets related to such Licensed Product in advance of submission of any such filings, and Gilead will reasonably consider all comments provided by Galapagos in connection therewith. With respect to each Licensed Product, Gilead shall provide Galapagos with prompt written notice of each of the following events (but in any event within [...***...] days) after the occurrence of such event in the Major Markets: (i) the filing of any IND for such Licensed Product; (ii) the submission of any filings or applications for Regulatory Approval (including orphan drug applications and designations) of such Licensed Product to any Regulatory Authority; and (iii) receipt or denial of Regulatory Approval for such Licensed Product; *provided*, *however*, *that* Gilead shall inform Galapagos of such event prior to public disclosure of such event.

4.6 Costs of Regulatory Affairs. With respect to costs and expenses incurred following the Effective Date in connection with applying for Regulatory Approval with respect to Licensed Products in the Territory, and related regulatory affairs activities (excluding any costs expressly set out hereunder as being at one Party's cost), such costs and expenses shall be treated as Development Costs and Gilead shall be responsible for eighty percent (80%) and Galapagos shall be responsible for twenty percent (20%). Gilead shall be solely responsible for all costs and expenses incurred for regulatory affairs activities following Regulatory Approval with respect to the Licensed Product or Gilead Combination Product in a particular indication, provided that, any costs and expenses incurred in connection with regulatory activities with respect to any Phase 4 Clinical Trial and any other fees paid to Regulatory Authorities following Regulatory Approval, in each case with respect to Licensed Products and not Gilead Combination Products, shall be shared by the Parties as Development Costs in accordance with Section 3.4. Within [...***...] Business Days after the end of the second (2nd) month of each calendar quarter, Galapagos shall provide to Gilead a report in reasonable detail of any such costs, expenses and fees incurred by Galapagos in the last calendar month of the preceding guarter and the first two calendar months of the current guarter for each Licensed Product. The costs, expenses and fees so reported will be used for the calculation of the 80/20 split for Development Costs and will be included in the reconciliation, invoicing and payment made pursuant to Section 3.4. The percentages in effect under Section 4.6 are hereby modified so that the Parties shall share costs and expenses in connection with applying for Regulatory Approval with respect to Licensed Products in the Territory, and related regulatory affairs activities (excluding any costs expressly set out hereunder as being at one Party's cost) equally (50%/50%) commencing with any such costs incurred on or after the Amendment Effective Date, regardless of when invoiced or paid. Such costs, expenses and fees reported by Galapagos to Gilead pursuant to Section 4.6 will be used for the calculation of the 50/50 split for Development Costs and will be included in the reconciliation, invoicing and payment made pursuant to Section 3.4.

4.7 Pharmacovigilance; Global Safety Database.

(a) <u>Safety Data Exchange Agreement</u>. The Parties shall amend that certain existing Safety Data Exchange Agreement ("**SDEA**") as needed but in no event later than [...***...] days following the Amendment Effective Date to reflect the reallocation of activities contemplated hereunder and in the Co-Commercialization Agreement(s), which amendment shall be sufficient to enable each Party and its Affiliates to comply with its legal and regulatory obligations. Such amendment may include procedures

regarding the receipt, investigation, recordation, communication, and exchange (as between the Parties), and regulatory submission of, adverse event reports, exposure during pregnancy reports, and any other information concerning the safety of the Licensed Products. In the event that there are any Gilead Combination Products under development, the Parties shall similarly amend the SDEA and/or enter into a separate SDEA to cover such product(s).

(b) <u>Global Safety Database</u>. As between Galapagos and Gilead, Gilead shall be responsible for establishing, holding and maintaining the global safety database for the Licensed Compound, Licensed Products and Gilead Combination Products, at Gilead's sole cost and expense.

4.8 <u>Product Withdrawals and Recalls</u>. If (a) any Regulatory Authority threatens, initiates or advises any action to remove any Licensed Product from the market in the Territory or requires or advises Galapagos, Gilead, or any of their respective Affiliates or Sublicensees to distribute a "Dear Doctor" letter or its equivalent regarding use of such Licensed Product in the Territory, or (b) Gilead determines that an event, incident, or circumstance has occurred that may result in the need for a recall or market withdrawal in the Territory, then in each case ((a) or (b)) Galapagos or Gilead, as applicable, shall, to the extent practicable, notify the other Party of such event or determination immediately, and in any event within [...***...] (or sooner if required by law) after such Party becomes aware of the event or makes such determination. Gilead shall, to the extent practicable, endeavor to discuss and agree with Galapagos upon whether to recall or withdraw the Licensed Product in the Territory; *provided, however, that* if such discussion is not practicable or if the Parties fail to agree within an appropriate time period (recognizing the exigencies of the situation), then Gilead shall decide whether to recall or withdraw such Licensed Product in the Territory. Subject to Section 5.5(a), Gilead shall be responsible, at its sole expense, for conducting any recalls or taking such other necessary remedial action with respect to Licensed Products in the Territory, except to the extent that the recall or withdrawal is attributable to the negligence, breach or intentional misconduct of Galapagos or any of its Affiliates or subcontractors, in which event Galapagos shall bear such costs to the extent of its or its Affiliate's or subcontractor's responsibility.

ARTICLE 5

COMMERCIALIZATION; MEDICAL AFFAIRS

5.1 Commercialization Responsibility and Diligence.

(a) <u>Licensed Territory</u>. Gilead shall be solely responsible for all Commercialization activities relating to the Licensed Products and Gilead Combination Products in the Licensed Territory. Gilead shall use Commercially Reasonable Efforts to Commercialize the first Licensed Product in each of the Major Markets in the Licensed Territory for each of the Target Indications. Notwithstanding anything to the contrary in this Agreement, the immediately foregoing obligation with respect to Commercially Reasonable Efforts in the Major Markets in the Licensed Territory and the Target Indications shall be Gilead's sole diligence obligation with respect to the Commercialization of Licensed Products and Gilead Combination Products, and Gilead shall have the sole right, but not any obligation, to Commercialize other Licensed Products and Gilead Combination Products in other countries or for other indications in the Licensed Territory, and no diligence obligation of Gilead shall arise under this Agreement with respect to Commercialization of Gilead Combination Products.

(b) <u>Shared Territory</u>.

(i) Unless otherwise agreed upon by the Parties in writing, with respect to the Shared Territory, the following shall apply:

(A) <u>Co-Commercialization Selling Party</u>. For the Licensed Product, Gilead will be the Co-Commercialization Selling Party for the United Kingdom and Germany, and Galapagos will be the Co-Commercialization Selling Party for France, Spain, Italy and the Benelux Countries. Logistics providers may be employed in the EU5 Countries, including to perform customer service, receive orders, receive shipments of Licensed Product from Gilead, and conduct other in-country distribution activities in the countries, as determined by the applicable Co-Commercialization Selling Party, *provided* that the Co-Commercialization Selling Party shall take into account any concerns of the Co-Commercialization Non-Selling Party with respect to the identity of any such logistics provider.

(B) Co-Commercialization Roles.

(1) Unless and until termination of the Benelux Co-Commercialization Agreement for the Benelux Countries, Galapagos will have the sole right, under the oversight of the Shared Territory JCC and the other Committees under this Agreement, to Commercialize the Co-Commercialization Products in the Benelux Countries in accordance with the Benelux Co-Commercialization Agreement, *provided* that Gilead will retain responsibility for list price setting for the Benelux Countries.

(2) Unless and until termination of the EU5 Countries Co-Commercialization Agreement with respect to a country, the Parties shall conduct, under the oversight of the Shared Territory JCC and the other Committees under this Agreement, the Commercialization of the Co-Commercialization Products in the EU5 Countries in accordance with the Co-Commercialization Agreement for the EU Countries jointly through the Joint Teams.

(3) For the EU5 Countries:

i. the Co-Commercialization Selling Party will be the lead Party with respect to Rheumatology Indications;

ii. the Co-Commercialization Non-Selling Party will be the lead Party with respect to Gastro Indications;

iii. the lead Party for each indication will lead for sales, marketing, Medical Affairs Activities and Market Access Activities for such indication, with the support of the other Party and subject to the applicable guidance of the Shared Territory JCC. The Parties will discuss in good faith the allocation of specific externally facing Market Access Activities with respect to the Gastro Indications to the Co-Commercialization Selling Party if required for regulatory reasons or to avoid adversely impacting the Commercialization of the Co-Commercialization Product; and

iv. the Co-Commercialization Non-Selling Party in each country will cooperate with and support the Co-Commercialization Selling Party in such country through the Joint Teams, and the non-lead Party for each indication will cooperate with and support the lead Party for such indication through the Joint Teams;

in each case as will be further set forth in the Co-Commercialization Agreement for the EU5 Countries. The Parties shall agree in writing or specify in the applicable Co-Commercialization Agreement the allocation of Co-Commercialization Activities in the EU5 Countries for any Other Co-Commercialization Indications. The goal of the Parties is to achieve approximately a 50%/50% split in terms of each Party's contribution to Co-Commercialization for the EU5 Countries. Therefore, if the Licensed Product is not approved for [...***...], then the

Parties will discuss in good faith a reallocation of the Parties' respective Commercialization activities in the EU5 Countries with the goal of achieving such 50%/50% split. The lead Party shall be the sole Party engaging in Details or other sales force promotional activities with respect to the indications for which it is the lead Party, unless otherwise agreed by the Parties in writing.

(4) Both Parties, under the direction of the Shared Territory JCC, and in accordance with the terms and conditions of this Agreement, will participate in the planning and conduct of the Co-Commercialization activities as and to the extent set forth in this <u>Article 5</u> and the Co-Commercialization Agreements (as applicable). Each Party shall use Commercially Reasonable Efforts to conduct Co-Commercialization activities for the Licensed Products or applicable Gilead Combination Products in the Shared Territory in accordance with the Shared Territory Commercialization Plan and Budget.

(5) <u>Transitional Period</u>. In the event that the applicable Co-Commercialization Agreement is not in place for a country as of the Amendment Effective Date, then during the period from the Amendment Effective Date through the effective date of the applicable Co-Commercialization Agreement, (x) Galapagos may engage in internal Co-Commercialization activities, including hiring, but shall not engage in any other Co-Commercialization activities with respect to such country unless expressly agreed by Gilead in writing, and (y) Gilead will continue to progress its Co-Commercialization activities with respect to such country in consultation with Galapagos and shall consider in good faith Galapagos' views with respect to such Co-Commercialization activities; *provided, however*, that in each case of (x) and (y), the Co-Commercialization Costs associated therewith will be reimbursable only pursuant to the Shared Territory Commercialization Plan and Budget (which may [...***...] if agreed by the Parties) and solely to the extent consistent therewith.

(6) Galapagos will use its reasonable best efforts to ramp up its Co-Commercialization capabilities, including in order to have a physical presence in [...***...] and including a general manager or equivalent local responsible person, as promptly as practicable.

(C) In the event that for a given country in the EU5 Countries, Galapagos desires to cease its Co-Commercialization role in that country, Galapagos shall notify Gilead in writing of such decision at least [...***...] months prior to the anticipated launch of the Licensed Product in such country, in which case effective at the end of such notice period, (i) such country shall be considered in the Licensed Territory and (ii) the Parties shall agree upon appropriate transitional arrangements with respect to any activities being conducted by Galapagos in such country. Thereafter, any decision by Galapagos to not perform the activities assigned to it in the Shared Territory shall require the prior written approval of Gilead and written agreement on appropriate transitional arrangements.

5.2 Commercialization of Licensed Products.

(a) Global Commercialization Strategy for Co-Commercialization Products.

(i) For each Co-Commercialization Product, the key Commercialization principles (among other things) will be set forth in a written summary of the global Commercialization strategy for such Co-Commercialization Product, which strategy will be reviewed by the JCC and approved by the JSC (each, a "**Global Commercialization Strategy**"). Gilead and Galapagos shall jointly prepare the initial draft of such Global Commercialization Strategy, and the JCC shall review and submit such proposed Global Commercialization Strategy to the JSC for approval. Thereafter, annually, Gilead and Galapagos shall jointly prepare updates to the Global Commercialization Strategy, and the JCC shall review such updates and submit proposed updates to the JSC for approval.

(ii) The JCC shall establish a global pricing strategy ("Global Pricing Strategy") for the Licensed Products. The Parties shall prepare, and the JCC shall review and submit to the JSC for approval, such initial Global Pricing Strategy and any updates thereto. The JCC shall review the Global Pricing Strategy on at least an annual basis.

(iii) Notwithstanding anything to the contrary in this Agreement or the Co-Commercialization Agreements, the Parties shall comply with Competition Law Guidelines.

(iv) The Shared Territory JCC will implement the Global Pricing Strategy as follows:

(A) with respect to the EU5 Countries, the Shared Territory JCC will, in accordance with Section 2.6, formulate a regional pricing strategy based on the Global Pricing Strategy, including establishing the public list price and net price parameters for the applicable Licensed Product or Gilead Combination Product in each applicable country, and considering requests from the Joint Teams for approval to deviate from such Shared Territory JCC's pricing parameters; and

(B) with respect to the Benelux Countries, making non-binding recommendations with respect to the public list price parameters (such public list price to be determined solely by Gilead) and net price parameters (such net price to be determined solely by Galapagos) for the applicable Licensed Product or Gilead Combination Product in each applicable country.

(b) Co-Commercialization of Licensed Products and Gilead Combination Products in the Shared Territory.

(i) Subject to oversight of the Shared Territory JCC, the JCC and the JSC, Galapagos shall be solely responsible for Co-Commercialization of the Licensed Products and Gilead Combination Products in the Benelux Countries. Gilead shall be responsible for matters relating to the public list price; otherwise Galapagos has the sole right to set net prices for the sale of Licensed Products and Gilead Combination Products in the Benelux Countries.

(ii) Subject to the oversight of the Shared Territory JCC, the Parties will jointly conduct Co-Commercialization in the EU5 Countries through the Shared Territory JCC and the Joint Teams.

(iii) The Parties shall enter into the Benelux Co-Commercialization Agreement and the EU5 Countries Co-Commercialization Agreement within [...***...] days after the Amendment Effective Date on reasonable and customary terms for similar agreements of this type consistent with this Agreement.

(iv) The Parties acknowledge and agree that, if there are any Licensed Products or Gilead Combination Products (other than the Initial Co-Commercialization Product) (each, a "**New Product**"), that are Developed hereunder following the Amendment Effective Date, the Parties shall either add the New Product to the Co-Commercialization Agreements or enter into separate Co-Commercialization Agreements with respect to such New Product.

(v) Shared Territory Commercialization Plan and Budget.

(A) As further described in this <u>Section 5.2(b)(v</u>), the tactics and strategy for the Commercialization of each Licensed Product and Gilead Combination Product in the Shared Territory shall be described in a comprehensive plan and budget (each such plan and budget, a **"Shared Territory Commercialization Plan and Budget**") that describes [...***...]. The Shared Territory

Commercialization Plan and Budget shall (i) include a detailed description of the Co-Commercialization activities to be undertaken in the Shared Territory with respect to the Licensed Product or, as applicable, any Gilead Combination Product during the following calendar year, (ii) allocate the responsibilities of the Parties for the Co-Commercialization activities under the Shared Territory Commercialization Plan and Budget, and (iii) set forth the amounts budgeted for the Joint Commercialization Costs in the Shared Territory. For clarity, the Shared Territory Commercialization Plan and Budget, with Galapagos taking the lead on the portion relating to the Benelux Countries and both Parties to coordinate regarding the portion relating to the EU5 Countries. The Shared Territory JCC will review the Shared Territory Commercialization Plan and Budget and submit to the JSC for approval no later than [...***...] days after the Amendment Effective Date. The same process shall apply with respect to any updates to the Shared Territory Commercialization Plan and Budget. Each Shared Territory Commercialization Plan and Budget. Shared Territory Commercialization Plan and Budget. Each Shared Territory Commercialization Plan and Budget shall be consistent with the requirements of the Global Commercialization Strategy may be updated from time to time.

(B) The Parties shall propose updates to each Shared Territory Commercialization Plan and Budget [...***...] and shall provide to the Shared Territory JCC such proposed updates for review, feedback and submission to the JSC for approval.

(vi) <u>Conflict of Terms</u>. Except as otherwise provided in this Agreement, in the event of a conflict or inconsistency between the terms of any Co-Commercialization Agreement, or Shared Territory Commercialization Plan and Budget and those of this Agreement, the terms of this Agreement shall govern and control.

(vii) <u>Advertising and Promotional Materials</u>. Gilead shall have sole responsibility for preparing all global Promotional Materials and Promotional Materials for the Licensed Territory. The Parties shall jointly prepare all Promotional Materials for the Shared Territory; *provided* that (A) such Promotional Materials for the Shared Territory shall be consistent with the global Promotional Materials prepared by Gilead, and (B) the applicable Joint Team in each EU5 Country or Galapagos in the Benelux Countries may adapt any such Shared Territory Promotional Materials for local use; *provided further* that any local Promotional Materials shall be consistent with the Global Commercialization Strategy. Deviations from the Shared Territory Promotional Materials by the Joint Team in each EU5 Country or Galapagos in the Benelux Countries are permitted (1) for translation purposes, (2) to the extent required by Applicable Law or (3) to the extent reasonably necessary for successful Commercialization in the applicable country or as approved by the Shared Territory JCC. All Promotional Materials shall be subject to Gilead's promotional material review process.

(viii) <u>Corporate Compliance</u>. Gilead shall maintain during the Co-Commercialization Term its corporate compliance program as existing as of the Amendment Effective Date or a variant thereof. Gilead shall provide to Galapagos its Business Compliance Policies within [...***...] Business Days after the Amendment Execution Date, and Galapagos shall be responsible for complying with such policies in connection with its activities under this Agreement and the Ancillary Agreements, including being responsible for its compliance trainings and training certifications, monitoring and enforcement. Galapagos shall establish and maintain during the Co-Commercialization Term a corporate compliance program, including at least one (1) full-time employee whose sole area of responsibility is compliance and who primarily reports to someone not in any commercial function and is responsible for ensuring that all employees of Galapagos' business conduct rules and regulations, which, subject to the foregoing, shall be consistent with Gilead's Business Compliance Policies. Each compliance program shall, at a minimum, provide for: (A) a compliance committee or other appropriate body with responsibility for operation of the compliance program, (B) a

periodic risk assessment that guides development of policies, training and monitoring activities, (C) appropriate corporate compliance policies, (D) regular compliance training and communication to applicable employees as selected on a risk-based approach, (E) auditing or monitoring or other risk-evaluation processes for applicable activities and (F) mechanisms, compliant with all Applicable Laws, to receive complaints or questions and investigate and remediate potential noncompliance, including a disciplinary component to handle compliance violations. The Parties shall, from and after January 1, 2020, abide by the International Federation of Pharmaceutical Manufacturers Code of Conduct, the European Federation of Pharmaceutical Industries and Associations Code on the Promotion of Prescription-Only Medicines to, and Interactions with, Healthcare Professionals where applicable, and applicable local country codes and guidelines with respect to the Co-Commercialization of the Licensed Product in the Shared Territory.

(ix)

(A) <u>Maintenance of Records</u>. Each Party and its Affiliates shall maintain, or cause to be maintained, complete and accurate books and records in accordance with Applicable Law and in sufficient detail to enable verification of the performance of its Co-Commercialization and other Commercialization activities pursuant to this Agreement. Such records shall be retained by the applicable Party and its Affiliates for at least the longer of (a) [...***...] years following creation and (b) such period as required by Applicable Laws.

(B) <u>Compliance Audit</u>. Gilead shall have the right, during normal business hours (or at such other times as the Parties may mutually agree), upon reasonable prior notice to Galapagos from time to time during the Co-Commercialization Term, to inspect and audit any books, records and accounts or other information of Galapagos relating to its Co-Commercialization and other Commercialization activities pursuant to this Agreement or any Co-Commercialization Agreement, as may be necessary: (i) enable Gilead to monitor and confirm compliance by Galapagos with Applicable Laws and other compliance obligations under this Agreement and the Co-Commercialization Agreements and (ii) for Gilead to comply with Applicable Laws in connection with its obligations under the Shared Territory Commercialization Plan and Budget. Galapagos shall provide Gilead with reasonable access to its facilities and personnel in connection with the foregoing, and the method Gilead uses to perform any such audit or inspection shall be at the sole discretion of Gilead. For the avoidance of doubt, nothing in this <u>Section 5.2</u> shall limit the rights or obligation of the Parties or their Affiliates to conduct audits under the terms of <u>Section 8.14</u> of this Agreement.

(C) Without limiting the indemnification, termination, dispute resolution and other rights of either Party hereunder, in the event that Gilead has a good faith belief that there has been or is reasonably likely to be a breach by Galapagos of the compliance provisions of this Agreement or any Co-Commercialization Agreement and desires to have a discussion regarding same, then upon Gilead's request, the Parties shall promptly convene a meeting of appropriate representatives from each Party within [...***...] Business Days after such request, which may at the request of Gilead be required to include either or both of each Party's general counsel or chief compliance officer. At such meeting, the Parties' representatives shall agree in writing upon a plan to rectify the situation and Galapagos shall take such action as required under the plan.

(x) Each Party shall notify the other Party of any Compliance Breach or Compliance Finding. In the case of any Compliance Breach or Compliance Finding, the Parties shall promptly meet to discuss, and Galapagos shall commence remediation of such Compliance Breach or Compliance Finding. If Galapagos does not cure such Compliance Breach or Compliance Finding within [...***...] days after the conclusion of such meeting, then Gilead may deliver to Galapagos a notice specifying the applicable Compliance Breach or Compliance Finding and the Co-Commercialization

activities the subject of such Compliance Breach or Compliance Finding that Gilead desires that Galapagos suspend ("**Suspension Notice**"), and Galapagos shall suspend its conduct of any such activities. Upon delivery of a Suspension Notice, the Parties shall work to develop a mutually acceptable remediation plan to address the applicable Compliance Breach or Compliance Finding. If the Parties have not agreed on a remediation plan within [...***...] days after delivery of the Suspension Notice (such agreement, not to be unreasonably withheld or delayed), or if at any time Galapagos is not using its reasonable best efforts to implement the remediation plan adopted by the Parties, then Gilead may deliver to Galapagos a notice that it is requiring Galapagos to permanently suspend its activities referenced in the Suspension Notice. In such case, Galapagos shall permanently suspend such activities and Gilead and Galapagos shall reasonably cooperate to transition to Gilead Galapagos' Co-Commercialization activities referenced in the Suspension Notice or country so as to minimize disruption to the Commercialization activities, including sales of the Licensed Product or Gilead Combination Product. In all cases, Galapagos shall withdraw its personnel or employees involved in Commercialization from such Co-Commercialization activities in a professional manner. "**Compliance Finding**" shall mean [...***...]. "**Compliance Breach**" shall mean [...***...].

(xi) <u>Costs; Authority over Personnel</u>. Each Party shall be responsible for all costs and expenses in connection with their respective personnel or employees involved in Commercialization and other personnel (which, for clarity, shall be included in Joint Commercialization Costs solely to the extent included in the relevant FTE Costs), including salaries, incentive compensation, travel expenses and other expenses, providing benefits, deducting federal, state and local payroll taxes, Federal Insurance Contribution Act taxes, unemployment insurance taxes, and any similar taxes and paying workers' compensation premiums, unemployment insurance contributions and any other payments required by Applicable Law to be made on behalf of employees. Nothing in this Agreement or any Co-Commercialization Agreement shall be construed to conclude that any of a Party's personnel or employees involved in Commercialization, or any other agents or employees of such Party or its Affiliates are agents or employees of the other Party or its Affiliates or subject to the other Party's direction and control. Each Party shall have sole authority over the terms and conditions of employment of its personnel or employees involved in Commercialization and other employees, including their selection, management, compensation (including incentive plans) and discharge.

(xii) <u>(Sub)contracting</u>. Notwithstanding anything in this Agreement to the contrary, each Party may use (sub)contractors without the consent of the other Party for its Co-Commercialization activities; *provided*, *however*, that (A) Galapagos may not engage a contract sales force unless expressly provided in the Shared Territory Commercialization Plan and Budget but may use individual contractors as Sales Representatives as reasonably necessary, and (B) Galapagos' brand manager must be an employee of Galapagos. Each Party may delegate its obligations under the Co-Commercialization Agreements to an Affiliate.

5.3 <u>Commercialization Costs</u>. Gilead shall be solely responsible for all Commercialization Costs incurred by or on behalf of Gilead in the Commercialization of Licensed Products and Gilead Combination Products in the Licensed Territory. Galapagos and Gilead shall share all Joint Commercialization Costs with respect to Licensed Products and Gilead Combination Products in the Shared Territory, which shall be included in the calculation of Operating Profit (or Loss) under this Agreement and calculated in accordance with <u>Section 8.9(a)(ii)</u>; provided, however, that for Gilead Combination Products, the Parties shall agree [...***...].

5.4 <u>Commercialization Reports</u>. Gilead shall keep the JCC informed regarding the progress and results of Commercialization activities for Licensed Products and Gilead Combination Products, as applicable, in the Territory. Each Party shall keep the JCC informed regarding the progress and results of Commercialization activities for Licensed Products and applicable Gilead Combination Products in the Shared Territory, including an annual review of results versus goals (as such goals are set forth in the Shared Territory Commercialization Plan and Budget).

5.5 <u>Sales and Distribution</u>. The "**Co-Commercialization Selling Party**" shall mean (x) Galapagos in France, Italy and Spain and the Benelux Countries, and (y) Gilead in all other countries, in each case except as otherwise agreed by the Parties in writing or the Co-Commercialization Selling Party has changed by operation of the terms of this Agreement.

(a) Subject to <u>Section 5.5(c</u>), with respect to and in each country in the Shared Territory (i) the Co-Commercialization Selling Party shall be responsible for handling all physical distribution, returns, recalls, order processing, invoicing and collection, booking of sales, inventory and receivables for Co-Commercialization Products, and (ii) the Co-Commercialization Non-Selling Party shall not accept orders for Co-Commercialization Products or make sales for its own account or for the Co-Commercialization Selling Party's account, and if the Co-Commercialization Non-Selling Party receives any order for any Co-Commercialization Products in the Territory, it shall refer such orders to the Co-Commercialization Selling Party for acceptance or rejection.

(b) In the EU5 Countries under the supervision of the Shared Territory JCC (and Shared Territory pricing committee), (i) the Parties, through the Joint Teams, will be responsible for the operational aspects of negotiating, managing and carrying out national, regional and local (including individual account or hospital) contracting arrangements and account management, and (ii) the Co-Commercialization Selling Party shall negotiate and execute contracts in the applicable country.

(c) In the Benelux Countries, (i) Galapagos have the sole right to determine the applicable net price (for which the net price parameters recommended by the Shared Territory JCC shall not be binding), (ii) Gilead will have the sole right to determine the applicable list price (for which the list price parameters recommended by the Shared Territory JCC shall not be binding), and (iii) subject to <u>Section 5.2(b)(i)</u>, Galapagos will be solely responsible for all aspects of negotiating, managing and carrying out national, regional and local (including individual account or hospital) contracting arrangements and account management, have the front-facing customer role for net price negotiations, and negotiate and execute contracts.

(d) Notwithstanding the foregoing, each Party may make passive sales of the Licensed Products and Gilead Combination Products to customers throughout the European Economic Area and Switzerland. For the purposes of the preceding sentence, "passive sales" shall mean sales made in response to unsolicited requests from individual customers.

(e) From the Amendment Execution Date until the termination of the Co-Commercialization Term for any country in the Shared Territory, Gilead shall not enter into any agreement with any Third Party that would conflict with the rights granted to Galapagos hereunder with respect to such country.

5.6 <u>Scientific and Medical Conferences</u>. The Parties shall coordinate, on behalf of themselves and their Affiliates, regarding their activities at scientific or medical conferences, including regarding which Party's or its Affiliate's or Sublicensee's employees shall staff any promotional and medical information booths that include any Co-Commercialization Product. If both Parties or their respective Affiliates participate in any such conference, then each of the Parties or their respective Affiliates shall maintain its own promotional and medical information booths at such conference unless otherwise agreed between the Parties.

ARTICLE 6

MANUFACTURE AND SUPPLY

6.1 <u>General</u>. Unless otherwise determined by the Parties, at and to the extent requested by Gilead, Galapagos shall, in accordance with customary terms and conditions to be negotiated and agreed between the Parties in a separate supply agreement, supply Licensed Products (as per the Manufacturing process and/or composition as existing at the Effective Date) for some or all of the Development of Licensed Products in the Territory following the Effective Date, and otherwise, Gilead shall be responsible for supply of all Licensed Compounds and Licensed Products for Development and Commercialization in the Territory.

6.2 <u>Manufacturing Technology Transfer.</u> Upon Gilead's request, Galapagos shall transfer to Gilead all Information Controlled by Galapagos during the Term that is necessary or useful to enable the Manufacture of Licensed Products, and not previously transferred to Gilead under this Agreement, by providing copies or samples of relevant documentation, materials and other embodiments of such Information, and by making available its qualified technical personnel on a reasonable basis to consult with Gilead with respect to such Information. Each such Information transfer requested by Gilead ("**Technology Transfer**") shall be commenced within a mutually agreed time following Gilead's request and conducted pursuant to a mutually-agreed technology transfer plan developed by the Parties for the purpose of ensuring the complete and timely transfer of such Information. Gilead will reimburse Galapagos for its out-of-pocket costs incurred in the course of such Technology Transfers, *provided that* such out-of-pocket costs are incurred in accordance with the mutually-agreed technology transfer plan and Galapagos provides an invoice to Gilead evidencing such costs. Gilead shall pay such amounts within [...***...] days of receipt of invoice therefor. Without limiting any other provision in this Agreement, Gilead shall have responsibility for and decision-making authority with respect to all formulation and Manufacturing matters.

6.3 <u>Supply Agreements</u>. The Parties shall in good faith negotiate the terms under which either Party shall supply Licensed Compound and Licensed Products to the other Party for Development activities and such terms shall be set forth in a clinical supply agreement to be entered into between the Parties (the "**Clinical Supply Agreement**").

6.4 <u>Subcontractors</u>; <u>Affiliates</u>. Each Party may perform any of its Manufacturing and supply obligations through one or more Third Parties; <u>provided that</u> (i) such Party remains responsible for the work allocated to, and payment to, such Third Party to the same extent it would if it had done such work itself; (ii) the Third Party undertakes in writing obligations of confidentiality and non-use regarding Confidential Information that are substantially the same as those undertaken by the Parties with respect to Confidential Information pursuant to <u>Article 12</u> hereof; and (iii) the applicable Party uses reasonable efforts to obtain from such Third Party in writing a covenant to assign or exclusively license back (with the right to sublicense) all intellectual property with respect to Licensed Products developed in the course of performing any such Manufacturing to such Party.

6.5 <u>Supply Agreement</u>. The Parties hereby acknowledge and agree that they will enter into separate supply agreements pursuant to which Gilead shall supply Galapagos with Licensed Products and Gilead Combination Products on reasonable and customary terms for France, Italy and Spain and the Benelux Countries respectively (the "**Supply Agreements**"). The Supply Agreements [...***...]. If not entered into on or before the Amendment Effective Date, the Parties shall enter into the Supply Agreements no later than [...***...] days after the Amendment Effective Date.

ARTICLE 7

LICENSES AND EXCLUSIVITY

7.1 <u>Licenses to Gilead</u>. Subject to the terms and conditions of this Agreement, Galapagos hereby grants Gilead, an exclusive, royalty-bearing, sublicensable (solely as permitted in accordance with <u>Section 7.2</u>) license under the Galapagos Technology to Exploit the Licensed Compound and Licensed Product in the Field in the Territory.

(a) <u>Galapagos Retained Rights</u>. Notwithstanding the exclusive license granted to Gilead pursuant to <u>Section 7.1</u>, Galapagos and its Affiliates shall retain, and have the right to sublicense to Third Parties in accordance with <u>Section 7.2(a)</u>, the right under the Galapagos Technology to perform (or to have performed by permitted subcontractors hereunder) the activities assigned to it under this Agreement in accordance with the terms of this Agreement.

(b) Subject to the terms and conditions of this Agreement, Gilead hereby grants Galapagos a non-exclusive, sublicensable (solely as permitted in accordance with <u>Section 7.2</u>), royalty-free, fully-paid license under the Gilead Technology solely to conduct the activities assigned to Galapagos under the Agreement.

7.2 Sublicensing.

(a) Scope of Permissible Sublicensing.

(i) The licenses granted by Galapagos to Gilead under this Agreement may be sublicensed by Gilead without any requirement of consent, *provided that* Gilead shall be liable for any act or omission of any such Sublicensee that is a breach of any of Gilead's obligations under this Agreement as though the same were a breach by Gilead, and Galapagos shall have the right to proceed directly against Gilead with respect to such breach without any obligation to first proceed against such Sublicensee.

(ii) Except with respect to (A) agreements with contract research organizations for performance of Clinical Trials which agreements Galapagos entered into prior to the Execution Date, or (B) agreements for performance of Clinical Trials with any entity set forth on Exhibit H, the (sub)licenses granted by Gilead to Galapagos in <u>Section 7.1(a)</u> may be sublicensed by Galapagos to a subcontractor to perform Galapagos' assigned responsibilities under this Agreement upon prior consent of Gilead (such consent not to be unreasonably withheld, conditioned or delayed) and *provided that* (i) if Gilead fails to notify Galapagos of whether it grants such consent for such subcontractor within [...***...] Business Days of Galapagos' request, then Gilead shall be deemed to have granted such consent with respect to such subcontractor, and (ii) such agreements shall comply with <u>Section 3.8</u>. Galapagos shall be liable for any act or omission of any such Sublicensee that is a breach of any of Galapagos's obligations under this Agreement as though the same were a breach by Galapagos, and Gilead shall have the right to proceed directly against Galapagos with respect to such breach without any obligation to first proceed against such Sublicensee.

(b) <u>Sublicense Agreements</u>. Gilead shall use reasonable efforts to provide that, in each agreement under which it grants a sublicense pursuant to <u>Section 7.2(a)(i)</u> under the license set forth in <u>Section 7.1</u> (each, a "**Sublicense Agreement**"), such Sublicense Agreement requires the Sublicensee to provide the following to Galapagos if this Agreement terminates, and to Gilead if only such Sublicense Agreement terminates: (i) the assignment and transfer of ownership and possession of, or a right of reference to, all Regulatory Materials and Regulatory Approvals Controlled by such Sublicensee with

respect to any Licensed Product (which assignment or right of reference may also be provided directly to Gilead prior to any such termination), but solely to the extent such assignment and transfer, or right of reference, would be required of Gilead under <u>Section 15.6</u>, and (ii) the assignment of, or a freely sublicensable exclusive license to, all intellectual property (including Patents) Controlled by such Sublicensee that covers or embodies a Licensed Product or its respective use, manufacture, sale, or importation and was conceived, discovered, developed or otherwise made by or on behalf of such Sublicensee during the exercise of its rights or fulfillment of its obligations pursuant to such Sublicense Agreement, but solely to the extent such assignment or exclusive license would be required of Gilead under <u>Section 15.6(a)</u>. Each Sublicense Agreement shall be subject to the applicable terms and conditions of this Agreement. For clarity, in the case of any subcontractor, this <u>Section 7.2(b)</u> shall not apply but Gilead shall comply with <u>Section 3.8</u>.

7.3 Distributorships and Co-Promotion Rights.

(a) <u>Distributorships</u>. Gilead shall have the right to appoint its Affiliates, and Gilead and its Affiliates shall have the right, in their sole discretion, to appoint any other Persons, in any country(ies) in the Territory in accordance with Gilead's typical practices for its proprietary products, to distribute, market, and sell Licensed Products in the Territory. If Gilead or any of its Affiliates appoints such a Person and such Person is not an Affiliate of Gilead, that Person shall be a "**Distributor**" for purposes of this Agreement. Any agreement between a Distributor and Gilead or its Affiliates regarding a Licensed Product shall be on commercially reasonable and arm's length terms.

(b) <u>Promotion Rights</u>. For the avoidance of doubt, subject to <u>Section 5.2(b)</u>, (i) Gilead and its Affiliates shall have the right to co-promote the Licensed Products with any other Person(s), or to appoint one or more Third Parties to promote the Licensed Products without Galapagos, in all or any part of the Licensed Territory, and (ii) Gilead and its Affiliates shall have the right to co-promote the Licensed Products with any other Person(s) (in addition to Galapagos), or to appoint one or more Third Parties to promote the Licensed Products with any other Person(s) (in addition to Galapagos), or to appoint one or more Third Parties to promote the Licensed Products without Galapagos, in all or any part of the Shared Territory, to the extent permitted in the Shared Territory Commercialization Plan and Budget.

7.4 <u>Negative Covenant</u>. Each Party covenants that it will not knowingly use or practice any of the other Party's intellectual property rights licensed to it under this Agreement other than for the purposes permitted in the applicable license grant.

7.5 <u>No Implied Licenses</u>. Except as explicitly set forth in this Agreement, neither Party grants to the other Party any license, express or implied, under its intellectual property rights.

7.6 Exclusivity; Non-Compete.

(a) From the Effective Date until [...***...], and subject to the terms of this Agreement, neither Galapagos, Gilead nor any of their respective Affiliates shall, [...***...], outside of the Collaboration conduct any [...***...]. Notwithstanding the foregoing, (i) Gilead or its Affiliates (or Galapagos or its Affiliates, as specifically requested by Gilead in accordance with this Agreement) may [...***...] in all cases that are intended to support the Exploitation of Licensed Products or Gilead Combination Products; and (ii) Gilead and its Affiliates may [...***...].

(b) If, during the period from the Effective Date until [...***...], Galapagos, Gilead or any of their respective Affiliates (the "**7.6** Acquiring Party") acquires or otherwise obtains [...***...] (an "**7.6** Acquired Product") as the result of any license or acquisition from, or merger, acquisition, reorganization, consolidation or combination with, or of, a Third Party or change of control of such Party or any other transaction (each, an "**7.6** Acquisition Transaction", and the Third Party involved in such transaction, the "**7.6** Acquisition Third Party") and, on the date of the completion of such 7.6 Acquisition

Transaction, such Third Party or its Affiliates are [...***...] that, if done by such Party, would violate the restrictions on such Party in Section 7.6(a), then the 7.6 Acquiring Party or such Affiliate will, within [...***...] days after the closing of such 7.6 Acquisition Transaction provide written notice to the other Party that the 7.6 Acquiring Party or such Affiliate has acquired the 7.6 Acquired Product and whether the 7.6 Acquiring Party elects to (A) divest its rights to such 7.6 Acquired Product to the extent violative of Section 7.6(a), (B) cease such [...***...] of such 7.6 Acquired Product to the extent violative of Section 7.6(a) or (C) include such 7.6 Acquired Product [...***...] as if it were a "Licensed Product" for all purposes of this Agreement (provided that, in the case of Galapagos as the 7.6 Acquiring Party, Galapagos shall not have the right to make the election described in this clause (C) without the written agreement of Gilead). If the 7.6 Acquiring Party provides notice as described in clause (A) of the preceding sentence, the 7.6 Acquiring Party and its Affiliates, if applicable, will use [...***...] to divest such rights to such 7.6 Acquired Product within [...***...] after provision of such notice, and if the 7.6 Acquiring Party provides notice as described in clause (B) of the preceding sentence, the 7.6 Acquiring Party, and its Affiliates if applicable, will use [...***...] to cease such [...***...] of such 7.6 Acquired Product as soon as reasonably practicable, giving due consideration to [...***...]. If the 7.6 Acquiring Party or its Affiliates (x) provides notice under clause (A) of the second preceding sentence but is unable to divest such rights to a 7.6 Acquired Product within the [...***...] period specified above despite the use of [...***...] or (y) provides notice under clause (B) of the second preceding sentence but is unable to cease such [...***...] despite the use of [...***...], then in either such case such rights to the 7.6 Acquired Product will be included (or in the case of a 7.6 Acquisition Transaction by Galapagos or its Affiliates, at Gilead's sole option will be included) as if it were a "Licensed Product" under this Agreement with respect to [...***...]. In the case of a 7.6 Acquisition Transaction by Galapagos or its Affiliates, if Gilead does not opt to include the 7.6 Acquired Product as a "Licensed Product" under the previous sentence, then Galapagos shall have the right to [...***...] such 7.6 Acquired Product, notwithstanding anything to the contrary in this Section 7.6.

ARTICLE 8

FINANCIALS

8.1 <u>License Fee</u>. In partial consideration of the license granted by Galapagos hereunder under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall pay to Galapagos a non-refundable, non-creditable license fee of Three Hundred Million Dollars (\$300,000,000) no later than [...***...] Business Days after the Effective Date.

8.2 Licensed Product Milestone Payments; Patent Costs.

(a) In partial consideration of the license granted by Galapagos hereunder under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall make milestone payments to Galapagos based on the first, but not any subsequent achievement by Gilead, its Affiliate or a Sublicensee of the development and regulatory milestone events in each indication as set forth in this <u>Section 8.2(a)</u> for Licensed Products.

	Milestone Payment			
Milestone Event	[***]	Crohn's Disease	Ulcerative Colitis	Other Indication
Dosing of first subject in first Phase 2 Clinical Trial for a Licensed Product	[***]	[***]	\$10,000,000	\$10,000,000
Dosing of first subject in first Phase 3 Clinical Trial for a Licensed Product	[***]	\$50,000,000	\$15,000,000	$[^{***}]$
[***]	[***]	$[^{***}]$	$[^{***}]$	[***]
[***]	[***]	$[^{***}]$	$[^{***}]$	$[^{***}]$
[***]	[***]	$[^{***}]$	$[^{***}]$	$[^{***}]$
[***]	[***]	$[^{***}]$	$[^{***}]$	$[^{***}]$

[...***...] in the immediately foregoing table shall have the defined meaning thereof set forth in <u>Article 1</u>. Each milestone in this <u>Section 8.2(a)</u> shall be paid only once during the Term. The maximum amount of payment to Galapagos pursuant to this <u>Section 8.2(a)</u> shall be seven hundred fifty-five million dollars (\$755,000,000). Gilead shall notify and pay to Galapagos the amounts set forth in this <u>Section 8.2(a)</u> within [...***...] days after the achievement of the applicable milestone event by Gilead, its Affiliate or a Sublicensee. Each such payment shall be made by wire transfer of immediately available funds into an account designated by Galapagos. Each such payment is nonrefundable.

(b) <u>Patent Costs</u>. In addition, in partial consideration of the license granted by Galapagos hereunder under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall be responsible for [...***...] of all Patent Costs incurred in connection with the filing, prosecution and maintenance activities as referred to in <u>Section 9.2(a)</u> for the Galapagos Patents. Galapagos shall invoice Gilead for Gilead's share of such Patent Costs on a quarterly basis and shall include reasonable supporting documentation with respect to such Patent Costs with such invoice; Gilead shall pay any undisputed amounts of such invoices not later than [...***...] days following Gilead's receipt of the applicable invoice.

(c) Licensed Product Sales Milestone Payments.

(i) <u>Events</u>. In partial consideration of the license granted by Galapagos hereunder under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall make each of the sales milestone payments indicated below to Galapagos when aggregate annual Net Sales of all Licensed Products across all indications in the Territory in a given calendar year first reach the dollar values indicated below during the Term.

Aggregate Net Sales of Licensed Products in a Given Calendar Year of	Payment
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

Each milestone in this Section 8.2(c) shall be paid only once during the Term. The maximum total amount of payment to Galapagos pursuant to this Section 8.2(c) shall be six hundred million dollars (\$600,000,000).

(ii) <u>Notice; Payment</u>. Gilead shall notify and pay to Galapagos the amounts set forth in this <u>Section 8.2(c)</u> within [...***...] days after the end of the calendar quarter during which the applicable milestone event has been achieved. Each such payment shall be made by wire transfer of immediately available funds into an account designated by Galapagos. Each such payment is nonrefundable.

8.3 Licensed Product and Gilead Combination Product Royalties.

(a) <u>Licensed Products</u>. In partial consideration of the license granted by Galapagos under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall pay to Galapagos non-refundable royalties on the amount of aggregate Net Sales of Licensed Products in the Licensed Territory in each calendar year, as calculated by multiplying the applicable royalty rates set forth below by the corresponding amount of incremental Net Sales in the Licensed Territory of Licensed Product in such calendar year.

Net Sales of Licensed Products in the Licensed Territory	Royalty Rate
[***]	20%
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	30%

By way of example, and without limitation, if the aggregate Net Sales of a Licensed Product in the Licensed Territory in a particular calendar year is [...***...], the amount of royalties payable under this <u>Section 8.3(a)</u> shall be as follows: [...***...].

(b) <u>Gilead Combination Products</u>. In partial consideration of the license granted by Galapagos under <u>Section 7.1</u> with respect to the Existing Galapagos Patents and the Galapagos Know-How (other than any Collaboration Know-How and Galapagos Foreground Know-How), Gilead shall pay to Galapagos non-refundable royalties on the amount of aggregate Net Sales of Gilead Combination Products in the Territory in each calendar year, as calculated by multiplying [...***...] by the corresponding amount of incremental Net Sales in the Territory of Gilead Combination Products in such calendar year.

Net Sales of Gilead Combination Products in the Territory	Royalty Rate
[***]	20%
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	30%

By way of example, and without limitation, if the aggregate Net Sales of a Gilead Combination Product in the Territory in a particular calendar year is [...***...], and the [...***...] for such Gilead Combination Product is [...***...], then the amount of royalties payable under this <u>Section 8.3(b)</u> shall be as follows: [...***...].

(c) <u>Royalty Term</u>. Royalties under <u>Section 8.3(a)</u> and <u>Section 8.3(b)</u> shall be payable, on a country-by-country basis, on the Net Sales of Licensed Products or Gilead Combination Products, as applicable, beginning on the First Commercial Sale of the first Licensed Product or Gilead Combination Product to launch in such country until [...***...] (the "**Royalty Term**").

(d) <u>Additional Royalty Provisions</u>. The royalties payable under <u>Sections 8.3(a)</u> and <u>8.3(b)</u> will be subject to the following:

(i) <u>Step Down</u>. The operation of the following provisions of this <u>Section 8.3(d)(i)</u> shall apply only with respect to the first such provision to apply to a particular Licensed Product or a Gilead Combination Product in each country in the Territory, so that if the royalty rates have already been reduced to [...***...] of the rates set forth in the applicable table in <u>Section 8.3(a)</u> or <u>Section 8.3(b)</u>, with respect to a particular unit of a product sold to which such reduction applied, no further reduction of such rates shall take place pursuant to this <u>Section 8.3(d)(i)</u>:

(A) [...***...]

(B) [...***...]

(ii) for clarity, the existence of a Valid Claim of a Gilead Patent in a country shall not affect the operation of <u>Section 8.3(c)</u>; and

(iii) [...***...]

(e) <u>Compulsory License</u>. If a Compulsory License is granted to a Third Party with respect to a Licensed Product or Gilead Combination Product, as applicable in any country in the Licensed Territory, and such Third Party actually sells such Licensed Product or Gilead Combination Product in the country under such Compulsory License, with a royalty rate lower than the royalty rate provided by this <u>Section 8.3</u>, then the Parties shall [...***...]. For a Compulsory License regarding a Gilead Combination Product, the Parties' respective share of Net Receipts will be adjusted according to the [...***...] of such Gilead Combination Product.

8.4 <u>Inclusion of Net Sales of Gilead Combination Products for Royalties and Sales Milestones</u>. To the extent [...***...], the [...***...] of such Gilead Combination Product for such country shall be added to the Net Sales of Licensed Products for purposes of calculating the sales milestones, royalty rates and royalties pursuant to <u>Sections 8.2(c)</u> and <u>8.3(a)</u> and such Net Sales corresponding to such [...***...] shall not be counted as Net Sales of Gilead Combination Products for purposes of <u>Section 8.3(b)</u>. Notwithstanding the foregoing, [...***...]. For purposes of this <u>Section 8.4</u>, [...***...] for a Gilead Combination Product shall mean [...***...].

8.5 <u>Royalty Payments and Reports</u>. Within [...***...] Business Days following the end of each calendar quarter, Gilead shall provide to Galapagos a statement setting forth on a country-by-country basis good-faith estimates of the gross sales of Licensed Products or Gilead Combination Products, as applicable, in the Licensed Territory and an estimated calculation of Net Sales in the Licensed Territory with respect to such Licensed Product or Gilead Combination Product, in all cases, reported in United States dollar amounts as determined in accordance with <u>Section 8.12</u>. All amounts payable to Galapagos pursuant to <u>Sections 8.3(a)</u> and <u>8.3(b)</u> shall be paid in U.S. dollars within [...***...] days after the end of each calendar quarter with respect to Net Sales in such calendar quarter. Each payment of royalties due to Galapagos shall be accompanied by a statement, on a country-by-country basis, of the amount of gross sales of Licensed Products and Gilead Combination Products in the Licensed Territory, during the applicable calendar quarter, a calculation of Net Sales in the Licensed Territory with respect to Licensed Territory, during the applicable calendar quarter, a calculation of Net Sales in the Licensed Territory with respect to Licensed Products and Gilead Combination Products showing with reasonable specificity the aggregate deductions from gross sales provided for in the definition of Net Sales during such calendar quarter, and a calculation of the amount of royalty payment due on such sales for such calendar quarter, in all cases, reported in United States dollar amounts as determined in accordance with <u>Section 8.12</u>.

8.6 Payments to Third Parties.

(a) Notwithstanding anything to the contrary herein, Galapagos shall solely be responsible for any payments due under the [...***...] and the [...***...].

(b) If a Third Party has or receives a Patent in any country that Covers the Exploitation of the Licensed Compound or Licensed Product anywhere in the Territory and the Joint Steering Committee determines that Gilead should obtain a license to such Patent as to one or more Licensed Products or Gilead Combination Products in one or more countries for a royalty or other payment to such Third Party (including that any Licensed Product or Gilead Combination Product at issue cannot be reasonably manufactured differently so as to avoid the requirement), Gilead may enter into such a license agreement, subject to the approval of the Joint Steering Committee, and may offset [...***...] percent ([...***...]%) of any such royalties or payments to such Third Parties against any share of Operating Profits or royalties that would otherwise have been due to Galapagos for such Licensed Product or Gilead Combination Product in such country hereunder, up to a maximum of [...***...] percent ([...***...]%) of such Operating Profits or royalties due to Galapagos.

8.7 <u>Following Royalty Term</u>. Upon expiration of the Royalty Term, Gilead's licenses from Galapagos hereunder with respect to Licensed Products and Gilead Combination Products, as applicable, in such country, shall become fully paid-up, perpetual, and irrevocable.

8.8 <u>Veterinary Products</u>. In the event that Gilead enters into an agreement to sublicense rights under this Agreement to, or to otherwise collaborate with, a Third Party for Veterinary Uses, Gilead shall [...***...] with Galapagos any upfront, milestone payments, royalties or other consideration received from such Third Party under such Agreement, and shall pay Galapagos's share of any such payments in accordance with the terms of <u>Section 8.2(c)(ii)</u>. [...***...]. For a Gilead Combination Product sold for Veterinary Use, the Parties' respective share of revenues will be adjusted according to the [...***...] of such Gilead Combination Product.

8.9 Licensed Product Reconciliation of Shared Costs; Profit Sharing.

(a) The terms and conditions of this <u>Section 8.9(a)</u> shall govern the rights and obligations of Galapagos and Gilead with respect to Operating Profits (or Losses) relating to each Licensed Product and Gilead Combination Product in the Shared Territory. For clarity, Galapagos shall have no right to share Operating Profits, and no obligation to bear any Operating Losses, with respect to any Licensed Product or Gilead Combination Product in the Licensed Territory.

(i) <u>Share of Operating Profits and Operating Losses</u>. Galapagos and Gilead shall share all Operating Profits and all Operating Losses (as applicable) for each Licensed Product and Gilead Combination Products for each country in the Shared Territory on the basis of fifty percent (50%) to Gilead and fifty percent (50%) to Galapagos.

(ii) Calculation and Payment.

(A) Within [...***...] Business Days after the end of each calendar quarter beginning with the calendar quarter in which the JSC approves the applicable Shared Territory Commercialization Plan and Budget, each Party shall provide to the other Party (using the form to be attached to the applicable Co-Commercialization Agreement) in reasonable detail of any Costs of Goods Sold for the Shared Territory, its Net Sales and any Joint Commercialization Costs incurred by such Party in such calendar quarter for each Licensed Product and Gilead Combination Product. Any subsequent adjustments made to the figures reported by each Party pursuant to the foregoing sentence will be reflected in the next quarter's profit share calculation and invoice.

(B) Within [...***...] Business Days after the end of each calendar quarter beginning with the calendar quarter in which the First Commercial Sale of a Licensed Product or Gilead Combination Product occurs in the Shared Territory, Gilead shall report to the finance officer designated by Galapagos and the finance officer designated by Gilead (the "Finance Officers") the Parties' Net Sales in the Shared Territory, the Parties' Cost of Goods Sold for the Shared Territory and the Parties' Joint Commercialization Costs incurred in such calendar quarter for each Licensed Product or Gilead Combination Product. Each such report shall specify in reasonable detail (1) for each country in the Shared Territory, the gross amount billed or invoiced for commercial sales of each Licensed Product or Gilead Combination Product, the Net Sales, and all deductions allowed in the calculation of such Net Sales and (2) all expenses included in such Cost of Goods Sold and Joint Commercialization Costs.

(C) If requested by a Party, the other Party shall promptly provide any invoices or other supporting documentation for any payments to a Third Party that individually exceed [...***...] dollars or with respect to which documentation is otherwise reasonably requested. Each report prepared in accordance with this Section shall be prepared using the form to be attached to the applicable Co-Commercialization Agreement.

(D) The Finance Officers shall compare each Party's Joint Commercialization Costs against the budget for the applicable activities assigned to such Party in the Shared Territory Commercialization Plan and Budget. The Finance Officers shall confer and, subject to Section 8.9(a)(ii)(E), agree in writing on a consolidated financial statement setting forth the Operating Profit or Operating Loss for such calendar quarter for such Licensed Product or Gilead Combination Product in the Shared Territory based on the numbers reported by the Parties and calculating each Party's share of such Operating Profit or Operating Loss within [...***...] Business Days after receipt of the consolidated report from Gilead.

(E) Notwithstanding the foregoing, if there is a cost overrun by a Party that, in the aggregate for the then-current calendar year, is greater than [...***...] percent ([...***...]%) of the aggregate amounts budgeted for such Party's activities for the calendar quarter in the Shared Territory Commercialization Plan and Budget, or the reports contain items not in the Shared Territory Commercialization Plan and Budget, or the reports contain items not in the Shared Territory Commercialization Plan and Budget, the Finance Officers shall deduct any amounts in excess of such [...***...] percent ([...***...]%) or related to items not in the Shared Territory Commercialization Plan and Budget (such that the incurring Party bears the excess), and thereafter until the end of the then-current calendar year shall not include any further overrun amounts in such Party's share of the Joint Commercialization Costs.

(1) If a Party reasonably believes that it will exceed its budgeted Joint Commercialization Costs, it shall promptly notify the other Party. The Parties shall use Commercially Reasonable Efforts, as appropriate, to mitigate any cost overrun.

(2) The foregoing shall be without limitation to the Parties' other rights and remedies.

(F) Within [...***...] days after the Operating Profit or Operating Loss are finally determined, (i) Galapagos or Gilead, as applicable, shall make a payment to Gilead or Galapagos respectively, as applicable, so that each of Galapagos and Gilead has been compensated for its respective share of such Operating Profits or has borne its respective share of such Operating Loss in such calendar quarter, and (ii) the Party to receive any such payment shall invoice the other Party for such amounts, *provided* that, where applicable, any such invoices shall be separated into one invoice covering all jurisdictions for which the paying Party is the Co-Commercialization Selling Party and one invoice covering all jurisdictions for which the payment, any undisputed portion of such payment shall be paid in accordance with the foregoing timetable and the remaining, disputed portion shall be paid within [...***...] days after the date on which Galapagos and Gilead, using good faith efforts, resolve the dispute.

(G) In addition, following the Amendment Effective Date, each Party shall consider in good faith other reasonable procedures proposed by the other Party for sharing financial information in order to permit each Party to close its books periodically in a timely manner. For the avoidance of doubt, no cost or expense shall be counted more than once in calculating the Cost of Goods Sold for the Shared Territory or Joint Commercialization Costs, even if such cost or expense falls into more than one of the cost categories that comprise the Cost of Goods Sold for the Shared Territory or Joint Commercialization Costs.

(iii) <u>Consistency with Accounting Treatment</u>. All calculations of Joint Commercialization Costs and Operating Profit and Operating Loss hereunder shall be made in accordance with GAAP, including the provisions thereof regarding expense recognition, as applied by Galapagos and Gilead consistently with their application in their respective financial reporting. For any costs and expenses charged to the other Party that are to be used in the calculation of Operating Profit and Operating Loss, if they include a mark-up on the applicable cost, the Parties will confirm the manner for treating the mark-up in such calculation so as to ensure the intended 50%/50% sharing of the Operating Profit (or Loss). Any amounts paid by one Party to the other Party for services or supply activities in connection with the Co-Commercialization will be submitted by the Party that made such payments and not by the Party providing the applicable supply or service, *provided* that the Party providing the applicable supply or service reports the amount of mark-up included, if any.

(iv) The FTE Rates for certain categories of personnel (including individual contractors) that participate in the Co-Commercialization under a Co-Commercialization Agreement and the method for determining FTE Rates for other personnel shall be set forth in the applicable Co-Commercialization Agreement. Such FTE Rates shall be increased annually by [...***...] percent ([...***...]%), with each annual adjustment effective as of January 1 of each calendar year, with the first such annual adjustment to be made as of January 1, 2020. For the avoidance of doubt, the FTE Rates in any country shall be the same for each Party.

(v) If requested by either Party, the Parties shall work together in good faith to reconcile Net Sales between the Licensed Territory and the Shared Territory arising out of the importation of Licensed Products or Gilead Combination Products into or out of the Shared Territory, to the extent reasonably feasible based on available data (e.g., IQVIA). Each such reconciliation may be requested one (1) time for a calendar year, within [...***...] days after the end of such calendar year. The Parties will reasonably adjust the royalties (for the Licensed Territory) and Operating Profit (or Loss) (for the Shared Territory) for such calendar year to reflect such reconciliation; provided that, at the request of either Party, the Parties will discuss in good faith the extent to which such adjustment would be reasonable given any limitations in available data.

8.10 [Intentionally omitted]

8.11 Taxes

(a) <u>Taxes on Income</u>. Each Party shall be solely responsible for the payment of all taxes imposed on its share of income arising directly or indirectly from the collaborative efforts of the Parties under this Agreement.

(b) <u>Tax Cooperation</u>. The Parties agree to cooperate with one another and use reasonable efforts to avoid or reduce tax withholding or similar obligations in respect of royalties, milestone payments, and other payments made between the Parties under this Agreement. Without limiting the generality of the foregoing, Galapagos shall provide Gilead any tax forms and other information that may be reasonably necessary in order to support its claim of no-withholding or reduced withholding based on an applicable treaty. Each Party shall provide the other with reasonable assistance to enable the recovery, as permitted by Applicable Law, of withholding taxes, value added taxes, or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding tax or value added tax.

(c) <u>Payment of Tax</u>. It is understood and agreed between the Parties that any payments made by Gilead under this Agreement are exclusive of any value added tax ("VAT") or similar tax imposed upon such payments. Where VAT is properly added to a payment made under this Agreement, Gilead will pay the amount of VAT only on receipt of a valid tax invoice issued in accordance with Applicable Law. In addition, to the extent Gilead is required by Applicable Law to deduct and withhold taxes on any payment to Galapagos, Gilead shall pay the amounts of such taxes to the proper Governmental Authority in a timely manner and promptly (within [...***...] days of payment) transmit to Galapagos an official tax certificate or other evidence of such withholding sufficient to enable Galapagos to claim credit for such payment of taxes.

8.12 <u>Foreign Exchange</u>. All payments shall be paid in US Dollars. For purpose of computing such payments, the Net Sales of Licensed Products in countries other than the United States shall be converted into US Dollars in accordance with Gilead's standard practices used in preparing its audited financial statements for the applicable quarter. Gilead's standard worldwide currency conversion methodology on the Effective Date is [...***...]. Gilead shall inform Galapagos of any changes to its standard worldwide currency conversion methodology prior to any such changes becoming effective.

8.13 Late Payments. If a Party does not receive payment of any sum due to it on or before the due date therefor, simple interest shall thereafter accrue on the sum due to such Party from the due date until the date of payment at a per-annum rate of [...***...] percent ([...***...]%) above the prime rate as reported in The Wall Street Journal, Eastern Edition, or the maximum rate allowable by Applicable Law, whichever is less.

8.14 <u>Financial Records</u>; <u>Audits</u>. Each Party and its Affiliates shall use all reasonable efforts to maintain complete and accurate records in sufficient detail to permit the other Party to confirm the accuracy of the amount to be reimbursed, pursuant to this Article 8, with respect to Development Costs, Joint Commercialization Costs, or other amounts to be reimbursed, credited, offset or shared hereunder incurred or generated (as applicable) by such Party's or Affiliate's achievement of sales milestones, royalty payments and other compensation or reimbursement payable under this Agreement. Upon reasonable prior notice, such records shall be open during regular business hours for a period of [...***...] years from the creation of individual records for examination at the auditing Party's expense, and not more often than once each calendar year, by an independent certified public accountant selected by the auditing Party and reasonably acceptable to the audited Party or Affiliate for the sole purpose of verifying for the auditing Party the accuracy of the financial statements or reports or sales milestone notices furnished by the audited Party or Affiliate pursuant to this Agreement or of any payments made, or required to be made, by or to the audited Party or Affiliate to the other pursuant to this Agreement. Any such auditor shall not disclose the audited Party's or Affiliate's confidential information to the auditing Party, but shall, instead, report that there was or was not a discrepancy uncovered by the audit and if such a discrepancy was uncovered, the amount and direction of it. Any amounts shown to be owed but unpaid, or overpaid and in need of reimbursement, shall be paid or refunded (as the case may be) within [...***...] days after the accountant's report, plus interest (as set forth in Section 8.13) from the original due date (unless challenged in good faith by the audited Party, in which case any undisputed portion shall be paid in accordance with the foregoing timetable, any dispute with respect to such challenge shall be resolved in accordance with Article 14, any remaining disputed portion shall be paid within [...***...] days after resolution of the dispute, and interest shall not accrue with respect to the disputed portion during the period of time the dispute is being resolved). The auditing Party shall bear the full cost of such audit unless such audit reveals an overpayment to, or an underpayment by, the audited Party or Affiliate that resulted from a discrepancy in a report that the audited Party or Affiliate provided to the other Party during the applicable audit period, which underpayment or overpayment was more than [...***...] percent ([...***...]%) of the amount set forth in such report, in which case the audited Party or Affiliate shall bear the full cost of such audit. Each Party, at the request of the other Party, shall make available to the other Party the results of any audit performed by the non-requesting Party on such non-requesting Party's Sublicensees hereunder.

8.15 <u>Manner and Place of Payment</u>. All payments owed under this Agreement shall be made by wire transfer in immediately available funds to a bank and account designated in writing by Galapagos or Gilead (as applicable), unless otherwise specified in writing by such Party.

ARTICLE 9

INTELLECTUAL PROPERTY

9.1 <u>Ownership of Inventions</u>. Subject to the remainder of this <u>Section 9.1</u>, ownership of intellectual property developed during the term of the Agreement by either Party alone or by the Parties together will be in accordance with U.S. laws as to inventorship and ownership of intellectual property:

(a) Subject to <u>Section 9.1(b)</u>, as between the Parties, Galapagos shall solely own all right, title and interest in and to all Collaboration Know-How and all Galapagos Foreground Know-How and all Patents arising from any such Collaboration Know-How or Galapagos Foreground Know-How (the "**Collaboration Patents**"), and all right, title and interest in and to all Collaboration Know-How and shall automatically vest solely in Galapagos, and Collaboration Know-How and Galapagos Foreground Know-How shall be deemed Galapagos' Confidential Information. Gilead, for itself and on behalf of its Affiliates

and employees, hereby assigns (and to the extent such assignment can only be made in the future hereby agrees to assign), to Galapagos all right, title and interest in and to Collaboration Know-How. Gilead shall cooperate, and shall cause the foregoing persons and entities to cooperate, with Galapagos to effectuate and perfect the foregoing ownership, including by promptly executing and recording assignments and other documents consistent with such ownership.

(b) As between the Parties, Gilead shall solely own all right, title and interest in and to (i) all Information conceived, discovered, developed or otherwise made in performing the activities under this Agreement (including performance of the Development Plan or Shared Territory Commercialization Plan and Budget), whether solely by one Party or jointly by the Parties, in each case optionally with their Affiliates or any Third Parties or any employees, consultants or agents of any of the foregoing to the extent relating to a Gilead Combination Product (the "Gilead Combination Know-How") and all Patents arising from such Gilead Combination Know-How (the "Gilead Combination Patents") (collectively, "Gilead Combination Technology") and (ii) all Gilead Foreground Know-How and all Patents arising from such Gilead Foreground Know-How and Gilead Foreground Know-How shall be deemed Gilead's Confidential Information. Galapagos shall promptly disclose to Gilead any Gilead Combination Technology conceived, discovered, developed or otherwise made by or on behalf of Galapagos, and shall provide Gilead such documentation regarding same as Gilead may reasonably request. Galapagos, for itself and on behalf of its Affiliates and employees, hereby assigns (and to the extent such assignment can only be made in the future hereby agrees to assign), to Gilead all right, title and interest in and to Gilead Combination Technology (unless already owned by Gilead). Galapagos shall cooperate, and shall cause the foregoing persons and entities to cooperate, with Gilead to effectuate and perfect the foregoing ownership, including by promptly executing and recording assignments and other documents consistent with such ownership.

(c) As between the Parties, the Parties shall jointly own all right, title and interest in and to all Joint Foreground Know-How and all Patents arising from such Joint Foreground Know-How (the "**Joint Patents**"). The rights of the Parties as joint owners shall be determined in accordance with this Agreement and the Applicable Laws of the United States.

(d) This Agreement shall be understood to be a joint research agreement in accordance with 35 U.S.C. §103(c) or 35 U.S.C. §102(c), as applicable, to Develop and Commercialize Licensed Compound or Licensed Products, *provided that* neither Party shall (i) unilaterally invoke the protections of or (ii) be required by this reference to have any Patent take advantage of or become subject to, such §103(c)(3) or 35 U.S.C. §102(c), as applicable, except with the prior written consent of the other Party.

9.2 Prosecution of Patents

(a) <u>Galapagos Patents</u>. Galapagos shall have the first right and authority to prepare, file, prosecute (including any opposition in the European Patent Office but not including the defense of any *Inter Partes* Reviews, Post Grant Reviews, interferences, reissue proceedings, reexaminations and other post-grant proceedings originating in a patent office, which proceedings shall be treated as though an enforcement right subject to <u>Section 9.3</u>) and maintain the Galapagos Patents in any jurisdiction in the Territory using counsel of its choice, unless there is a conflict between said counsel and Gilead, in which case new counsel may be selected who are reasonably acceptable to Gilead. Galapagos shall keep Gilead reasonably informed of all material matters relating to the preparation, filing, prosecution and maintenance of the Galapagos Patents in the Territory (including providing Gilead with copies of all material submissions with the applicable patent office from countries or corresponding authorities within the Territory) and shall reasonably consider in good faith any comments provided by Gilead with respect to such submissions. The Parties shall [...***...] all Patent Costs incurred in connection with the filing,

prosecution and maintenance foregoing activities for the Galapagos Patents without reimbursement by Galapagos. In the event that Gilead elects not to continue sharing such Patent Costs for any Patent within the Galapagos Patents, Gilead shall provide Galapagos with at least [...***...] days written notice thereof, and such Patent shall be removed from the definition of Galapagos Patents under this Agreement and the licenses granted to Gilead and its Affiliates as to such rights shall terminate. In the event that Galapagos elects not to maintain patent protection on any Galapagos Patents, unless such Galapagos Patents are within [...***...] (as identified in <u>Exhibit B</u> as [...***...]) (the [...***...]) Galapagos shall notify Gilead at least [...***...] days before any such Galapagos Patents would become abandoned or otherwise forfeited, and, unless Galapagos has a *bona fide* strategic reason for such election and considers, reasonably and in good faith, all input received from Gilead, (i) Gilead shall have the right to assume preparation, filing, prosecution and maintenance and all related costs of such Galapagos Patents, and (ii) if Gilead so assumes prosecution, then unless it is not possible to assign such Patent to Gilead, Galapagos shall promptly assign to Gilead all right title and interest therein, (iii) the Patent Costs for any Patent assigned to Gilead being the sole responsibility of Gilead in view of such Patent no longer being a Galapagos Patent, (iv) the Patent Costs of any such unassignable Patent continuing to be [...***...] between the Parties, and (v) [...***...].

(b) <u>Gilead Foreground Patents and Gilead Combination Patents</u>. Gilead shall have the sole right and authority to prepare, file, prosecute (including any oppositions, interferences, reissue proceedings, reexaminations and other post-grant proceedings) and maintain the Gilead Foreground Patents and Gilead Combination Patents in any jurisdiction in the Territory using counsel of its choice. Gilead shall be solely responsible for all Patent Costs incurred in connection with the foregoing activities for the Gilead Foreground Patents and Gilead Combination Patents without reimbursement by Galapagos</u>. Gilead shall keep Galapagos reasonably informed of all material matters relating to the preparation, filing, prosecution and maintenance of the Gilead Foreground Patents and Gilead Combination Patents and Gilead Combination Patents in the Major Markets (including providing Gilead with copies of all material correspondence with the applicable patent office from countries or corresponding authorities within the Major Markets) to the extent reasonably related to the Licensed Compound, and shall reasonably consider in good faith any comments provided by Galapagos with respect to such submissions. Galapagos shall bear any costs and expenses it may incur in connection with its review and consultation concerning any such Gilead Foreground Patents and Gilead Combination Patents.

(c) Joint Patents. Galapagos shall have the first right and authority to prepare, file, prosecute (including any opposition in the European Patent Office but not including the defense of any *Inter Partes* Reviews, Post Grant Reviews, interferences, reissue proceedings, reexaminations and other post-grant proceedings originating in a patent office, which proceedings shall be treated as though an enforcement right subject to <u>Section 9.3</u>) and maintain the Joint Patents in any jurisdiction in the Territory using counsel of its choice. The Parties shall [...***...] all Patent Costs incurred in connection with the foregoing activities for the Joint Patents. Galapagos shall keep Gilead reasonably informed of all material matters relating to the preparation, filing, prosecution and maintenance of the Joint Patents in the Major Markets (including providing Gilead with copies of all material correspondence with the applicable patent office from countries or corresponding authorities within the Major Markets) and shall reasonably consider in good faith any comments provided by Gilead with respect to such submissions. In the event that either Party elects not to continue sharing such Patent Costs for any Patent within the Joint Patents, such Party shall notify the other Party, and:

(i) if such notifying Party is Galapagos, (A) Galapagos shall provide Gilead with at least [...***...] days written notification thereof prior to any date of an action needed to maintain material rights in such Patent, (B) Gilead shall have the right, but not the obligation, to assume preparation, filing, prosecution and maintenance and all related Patent Costs of such Joint Patent by written notice to Galapagos not later than [... ***...] days following such written notification from Galapagos, and (C) if Gilead so assumes prosecution, then Galapagos shall promptly assign to Gilead all right title and interest therein; and

(ii) if such notifying Party is Gilead (A) Galapagos shall have the right to assume preparation, filing, prosecution and maintenance and all related Patent Costs of such Joint Patents by written notice to Gilead not later than [...***...] days following such notification from Gilead, and (B) if Galapagos so assumes prosecution, then Gilead shall promptly assign to Gilead all right title and interest therein.

(d) <u>Cooperation in Prosecution</u>. Each Party shall provide the other Party all reasonable notice, assistance and cooperation in the Patent prosecution efforts provided above in this <u>Section 9.2</u>, including providing any necessary powers of attorney and executing any other required documents or instruments for such prosecution.

(e) Patent Term Extensions. Galapagos shall have the sole right, but agrees to consult with Gilead, to apply for and obtain any patent term extension or related extension of rights, including supplementary protection certificates and similar rights, for any Galapagos Patents anywhere in the Territory. Gilead shall have lead responsibility, in consultation with Galapagos, for applying for and obtaining any patent term extension or related extension of rights, including supplementary protection certificates and similar rights, for any Joint Patents, Gilead Foreground Patents and Gilead Combination Patents anywhere in the Territory. If the Parties disagree on the appropriate strategy with respect to such an extension of the Galapagos Patents or Gilead Combination Patents, the disagreement shall be resolved by the JSC. If the JSC is unable to reach consensus on the strategy, [... ***...]. Each Party shall provide reasonable assistance to the other Party in connection with obtaining any such extensions for the Galapagos Patents and Gilead Combination Patents consistent with such strategy. To the extent reasonably and legally required in order to obtain any such extension in a particular country, each Party shall make available to the other a copy of the necessary documentation to enable such other Party to use the same for the purpose of obtaining the extension in such country.

(f) <u>Orange Book Listings</u>. Gilead shall have lead responsibility for making any filing with respect to any Galapagos Patent or any Gilead Combination Patent in connection with the FDA's Orange Book, under the national implementations of Article 10.1(a)(iii) of Directive 2001/EC/83 or other international equivalents. Gilead shall consult with Galapagos regarding the strategy therefor. If the Parties disagree on the appropriate strategy with respect to such a filing, the disagreement shall be resolved by the JSC. If the JSC is unable to reach consensus on the strategy, [...***...]. Each Party shall provide reasonable assistance to the other Party in connection with any such filing.

(g) <u>EU Patent Matters</u>. In addition the Parties shall discuss and agree what patent application filing strategy should be adopted within the European Union, in particular following the coming into force of regulations implementing a unitary patent regime throughout the European Union and whether patents either for individual EU countries or European Patents with Unitary Effect should be opted-in, or opted out (and potentially then opted back in) to the exclusive competence of the Unified Patent Court.

9.3 Infringement by Third Parties.

(a) <u>Notification</u>. If either Party becomes aware of any infringement, threatened infringement, or alleged infringement (i) of any Galapagos Patent, Joint Patent or any Gilead Combination Patent by a Third Party conducting the manufacture, use, marketing, or sale of a product falling within the scope of the exclusive license granted to Gilead under <u>Section 7.1</u>, (ii) within the scope of a Valid Claim under a Gilead Combination Patent, or (iii) as a result of a notification to a Party or its Affiliate pursuant to Sections 505(j) or 355(b)(2) of the FD&C Act (21 U.S.C. § 355(j) and 21 U.S.C. § 355(b)(2)) or a foreign equivalent, of an application for approval of a Generic Product (each, a "**Product Infringement**"), then each Party shall promptly notify the other Party in writing thereof and provide evidence in such Party's possession demonstrating such threatened, alleged or actual infringement or such use.

(b) Enforcement Rights. Without the prior written consent of the other Party, neither Party shall have a right to bring suit or other action to abate an infringement of a Galapagos Patent (excluding the [...***...]), Joint Patent or Gilead Foreground Patent (collectively, the "**Program Patents**"), which infringement is not a Product Infringement. Gilead shall have the first right, but not the obligation, to bring a suit or other action to abate any Product Infringement, under the Program Patents (other than the [...***...]). With respect to the [...***...], Galapagos shall have the sole right, but not the obligation, to bring a suit or other action to abate any Product Infringement. Galapagos shall cooperate fully as may be reasonably requested by Gilead, upon reasonable notice, to maintain such suit or other action, by executing and making available such documents as Gilead may reasonably request, and by performing all other acts which are or may become reasonably necessary to vest in Gilead the right to institute any such suit or other action, including by being joined in such action and using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties. Gilead shall not enter into any settlement or transaction agreement with a Third Party that reduces the scope of or admits invalidity or unenforceability of any Program Patent claims in a manner that will cause material harm to Galapagos, without the prior written consent of Galapagos. Notwithstanding the foregoing, if Gilead does not inform Galapagos that it intends to take such measures with respect to such Product Infringement within [...***...] days after Gilead's receipt of a notice of infringement pursuant to <u>Section 9.3(a)</u>, then Galapagos will have the second right, but not the obligation, to initiate such Product Infringement action or take other measures.

(c) Gilead shall have the sole right, but not the obligation, to bring a suit or other action to abate any infringement, under the Gilead Combination Patents. To the extent such action is a Product Infringement, Galapagos shall cooperate fully as may be reasonably requested by Gilead, upon reasonable notice, to maintain such suit or other action, by executing and making available such documents as Gilead may reasonably request, and by performing all other acts which are or may become reasonably necessary to vest in Gilead the right to institute any such suit or other action, including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties.

(d) <u>Settlement</u>. Without the prior written consent of the other Party, neither Party shall settle any claim, suit or action that it brought under <u>Section 9.3(b)</u> involving Program Patents in any manner that would materially limit the rights or materially expand the obligations of the other Party under this Agreement or the Ancillary Agreements or would expressly admit any liability by the other Party.

(e) Expenses and Recoveries. A Party bringing a claim, suit or action under Section 9.3(b) against any person or entity engaged in Product Infringement of the Galapagos Patents shall be solely responsible for any expenses incurred by such Party as a result of such claim, suit or action, except that where Gilead requests Galapagos to bring suit under the [...***...], and Galapagos agrees to do so, then Gilead shall be responsible for all costs and expenses of any such suit or other action. If such Party recovers monetary damages from such Third Party in such suit or action, such recovery shall be allocated first to the reimbursement of any expenses incurred by the Parties in such litigation, and any remaining amount shall be distributed as follows: [...***...].

9.4 <u>Defense of Program Patents</u>. To the extent any Party receives notice by counterclaim, or otherwise, alleging the invalidity or unenforceability of any Program Patent, it shall bring such fact to the attention of the other Party, including all relevant information related to such claim. The Parties, through the JSC, shall discuss such claim. Where such allegation is made in a *Inter Partes* Review, Post Grant Review, interference, or other patent office proceeding other than an opposition in the European Patent

Office, or where such allegation is made in a counterclaim to a suit or other action brought under <u>Section 9.3</u>, the provisions of <u>Section 9.3</u> as applicable to a Product Infringement shall apply, and in the case of an opposition in the European Patent Office, the provisions of <u>Section 9.2</u> shall apply. In the event Gilead does not elect to defend an action with respect to any Program Patent under this <u>Section 9.4</u>, it shall so notify Galapagos in writing, and Galapagos shall have the right to so defend such action, at Galapagos' expense. Each Party shall provide to the Party defending any such rights under this <u>Section 9.4</u> all reasonable assistance in such enforcement, at such defending Party's request and expense. The defending Party shall keep the other Party regularly informed of the status and progress of such efforts, and shall reasonably consider the other Party's comments on any such efforts.

9.5 <u>Defense of Gilead Combination Patents</u>. To the extent any Party receives notice by counterclaim, or otherwise, alleging the invalidity or unenforceability of any Gilead Combination Patent, it shall bring such fact to the attention of the other Party, including all relevant information related to such claim. The Parties, through the JSC, shall discuss such claim. Gilead shall have the sole right, but not the obligation, to respond to any such challenge. Galapagos shall provide Gilead all reasonable assistance in such enforcement, at Gilead's request and expense. Gilead shall keep Galapagos regularly informed of the status and progress of such efforts.

9.6 Defense of Infringement Actions. During the Term, each Party shall bring to the attention of the other Party all information regarding potential infringement or any claim of infringement of Third Party intellectual property rights in connection with the development, manufacture, use, importation, offer for sale, or sale of Licensed Products in the Territory. The Parties shall discuss such information and decide how to handle such matter, with Gilead having final authority to decide in the event of a disagreement. Subject to <u>Article 11</u>, Gilead shall have the first right to defend any such action that relates to a Licensed Product or Gilead Combination Product, and otherwise, each Party shall be solely responsible for defending any action, suit, or other proceeding brought against it alleging infringement of Third Party (including by being joined to such action if so requested and by using commercially reasonable efforts to obtain any necessary joinder or cooperation in any such suit or other action from any applicable Third Parties) in connection with defense of such action, suit or proceeding, and if an action, suit or other proceeding is brought against both Parties, then Gilead shall be responsible for defending such action, suit or other proceeding. This <u>Section 9.6</u> shall not be interpreted as placing on either Party a duty of inquiry regarding Third Party intellectual property rights.

9.7 <u>Patent Marking</u>. Gilead shall, and shall require its Affiliates and Sublicensees to, use Commercially Reasonable Efforts to mark Licensed Products sold by it hereunder (in a reasonable manner consistent with industry custom and practice) with appropriate patent numbers or indicia to the extent permitted by Applicable Law, in those countries in which such markings or such notices impact recoveries of damages or equitable remedies available with respect to infringements of patents.

9.8 <u>Personnel Obligations</u>. Prior to beginning work under this Agreement relating to any research, preclinical development, Development or Commercialization of a Licensed Compound or Licensed Product, each employee of Gilead or Galapagos or of either Party's respective Affiliates shall be bound by non-disclosure and invention assignment obligations which are consistent with the obligations of Gilead or Galapagos, as appropriate, in this <u>Article 9</u>, to the extent permitted by Applicable Law, including: (a) promptly reporting any invention, discovery, process or other intellectual property right; (b) assigning to Gilead or Galapagos, as appropriate, all of his or her right, title and interest in and to any invention, discovery, process or other intellectual property right; (c) in the case of employees working in the United States, taking actions reasonably necessary to secure patent protection; (d) performing all acts and signing, executing, acknowledging and delivering any and all documents required for effecting the obligations and purposes of this Agreement; and (e) abiding by the obligations of confidentiality and non-use set forth in <u>Article 12</u>. It is understood and agreed that such non-disclosure and invention assignment agreement need not reference or be specific to this Agreement.

9.9 <u>Trademarks</u>. Gilead shall be responsible for the selection, registration, maintenance and defense of all trademarks for use in connection with the sale or marketing of Licensed Products in the Field in the Territory (the "**Marks**"). The fees and expenses incurred in connection therewith for Marks applicable to Licensed Product in the Territory shall be the responsibility of Gilead. All uses of the Marks shall be reviewed by the JCC and shall comply with Applicable Law (including those laws and regulations particularly applying to the proper use and designation of trademarks in the applicable countries). Neither Party shall, without the other Party's prior written consent, use any trademarks or house marks of the other Party (including the other Party's corporate name), or marks confusingly similar thereto, in connection with such Party's marketing or promotion of Licensed Products under this Agreement, except as may be expressly authorized in connection with activities under <u>Section 5.2</u> or any Co-Commercialization Agreement and except to the extent required to comply with Applicable Law, Gilead shall own all Marks with respect to the Licensed Products. In addition to the Marks, each package of a Licensed Product sold in the Shared Territory, if requested by Galapagos and permitted by Applicable Law (including approval by all applicable Regulatory Authorities), shall be labeled with Galapagos's corporate name or logo in a manner determined by the JCC.

9.10 <u>Confirmatory Patent Licenses</u>. Each Party shall, if so requested by the other Party, promptly enter into confirmatory license agreements, in a form consistent with the terms of this Agreement and reasonably acceptable to the Parties, for purposes of recording the licenses granted under this Agreement with patent offices in the Territory. Each Party shall bear its own filing costs and any costs of outside counsel or experts required with respect to such recordations.

9.11 <u>Patent Committee</u>. Promptly (but no later than [...***...] days) after the Effective Date, the Parties will establish a patent committee (the "**Patent Committee**"). The purpose of the Patent Committee is to facilitate the discussion and coordination of (a) strategies regarding intellectual property, and (b) prosecution and maintenance, enforcement and defense matters in accordance with and subject to the terms of this Agreement. The Patent Committee will meet on a quarterly basis or as otherwise agreed by the Parties. All final decisions related to the prosecution and maintenance, enforcement or defense of any Patent will be made by the Party with the right to control such prosecution and maintenance, enforcement or defense, as applicable, as set forth in this <u>Article 9</u>.

ARTICLE 10

REPRESENTATIONS AND WARRANTIES AND COVENANTS

10.1 <u>Mutual Representations and Warranties as of Effective Date</u>. Each Party hereby represents, warrants, and covenants (as applicable) to the other Party as of the Effective Date as follows:

(a) <u>Corporate Existence and Power</u>. It is a company or corporation duly organized, validly existing, and in good standing under the laws of the jurisdiction in which it is incorporated, and has full corporate power and authority and the legal right to own and operate its property and assets and to carry on its business as it is now being conducted and as contemplated in this Agreement, including the right to grant the licenses granted by it hereunder.

(b) <u>Authority and Binding Agreement</u>. Except as set forth in this <u>Section 10.1(b)</u>, (i) it has the corporate power and authority and the legal right to enter into this Agreement and perform its obligations hereunder; (ii) it has taken all necessary corporate action on its part required to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder; and (iii) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, valid, and

binding obligation of such Party that is enforceable against it in accordance with its terms. Notwithstanding the foregoing, Galapagos undertakes (A) to submit a proposal to its shareholders to ratify <u>Section 15.6</u> of this Agreement at Galapagos' next annual shareholders' meeting to be held on 26 April 2016, and (B) subject to having obtained such shareholder approval in accordance with (A), to file such resolution with the clerk's office of the commercial court of Antwerp (division Mechelen), Belgium, in accordance with section 556 of the Belgian Companies Code.

(c) <u>No Conflict</u>. It is not a party to and will not enter into any agreement that would prevent it from granting the rights or exclusivity granted or intended to be granted to the other Party under this Agreement or performing its obligations under this Agreement.

(d) <u>No Debarment</u>. Such Party is not debarred, has not been convicted, and is not subject to debarment or conviction pursuant to Section 306 of the FD&C Act. In the course of the Development of the Licensed Compound, Licensed Products or Gilead Combination Products, such Party has not used prior to the Effective Date and shall not use, during the Term, any employee, agent or independent contractor who has been debarred by any Regulatory Authority, or, to the best of such Party's knowledge, is the subject of debarment proceedings by a Regulatory Authority or has been convicted pursuant to Section 306 of the FD&C Act.

(e) <u>Anti-Corruption</u>. Such Party, its Affiliates and their respective directors, officers, employees, agents or other persons or entities acting on its behalf (all the foregoing collectively "**Representatives**") have conducted and will conduct their businesses in compliance with the Foreign Corrupt Practices Act of 1977, as amended (such act, including the rules and regulations thereunder, the "FCPA"), the UK Bribery Act of 2010 ("**Bribery Act**") and any other applicable anti-corruption laws, rules or regulations (collectively, "**Anti-Corruption Laws**"). Without limiting the generality of the foregoing, such Party represents and warrants that it has and will have necessary procedures in place to prevent bribery and corrupt conduct by it and its Representatives.

10.2 <u>Mutual Representations and Warranties as of Amendment Effective Date</u>. Each Party hereby represents, warrants, and covenants (as applicable) to the other Party as of the Amendment Effective Date as follows:

(a) <u>Corporate Existence and Power</u>. It is a company or corporation duly organized, validly existing, and in good standing under the laws of the jurisdiction in which it is incorporated, and has full corporate power and authority and the legal right to own and operate its property and assets and to carry on its business as it is now being conducted and as contemplated in this Agreement and the Co-Commercialization Agreements, including the right to grant the licenses granted by it hereunder and thereunder.

(b) <u>Authority and Binding Agreement</u>. Except as set forth in this <u>Section 10.2(b)</u>, (i) it has the corporate power and authority and the legal right to enter into this Agreement, and the Co-Commercialization Agreements and perform its obligations hereunder and thereunder; (ii) it has taken all necessary corporate action on its part required to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder; and (iii) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes legal, valid, and binding obligations of such Party that are enforceable against it in accordance with its terms, except to the extent that enforcement of the rights and remedies created hereby is subject to (A) bankruptcy, insolvency, reorganization, moratorium and other similar laws of general application affecting the rights and remedies of creditors or (B) laws governing specific performance, injunctive relief and other equitable remedies.

(c) <u>No Conflict</u>. It is not a party to and will not enter into any agreement that would prevent it from granting the rights or exclusivity granted or intended to be granted to the other Party under this Agreement or the Co-Commercialization Agreements or performing its obligations under this Agreement, or the Co-Commercialization Agreements.

10.3 <u>Representations</u>, <u>Warranties and Covenants by Galapagos</u>. Galapagos hereby represents, warrants and covenants to Gilead, as of the Effective Date, as follows, except as set forth otherwise in the Specific Disclosures:

(a) <u>Title; Encumbrances</u>. Galapagos owns or has a valid right to use the Galapagos Technology existing as of the Effective Date, including the Existing Galapagos Patents, <u>provided, however, that</u> the foregoing shall not constitute a representation or warranty of non-infringement of a Third Party's intellectual property rights. Galapagos has the right to grant the licenses to Gilead as purported to be granted pursuant to this Agreement. Neither Galapagos Patents or Galapagos Know-How to any Third Party that would conflict with the licenses to Gilead as purported to be granted purported to this Agreement.

(b) All Existing Galapagos Patents are listed on <u>Exhibit B</u>. All Existing Galapagos Patents are subsisting and are not invalid or unenforceable, in whole or in part, are being prosecuted in the patent offices indicated on <u>Exhibit B</u> in accordance with Applicable Law, and all applicable fees have been paid on or before the Effective Date. The Existing Galapagos Patents represent all Patents within Galapagos' or its Affiliates' ownership or Control that Galapagos reasonable believes include claims covering the making, using, and composition of matter of the Licensed Compounds or the Licensed Products, or the Exploitation thereof, as of the Effective Date. Galapagos Patents. To the Galapagos' Knowledge, Galapagos and its Affiliates have presented, or will present prior to the pertinent patent office deadlines, all relevant references, documents, or information of which it and the inventors are aware to the relevant patent examiner at the pertinent patent office, in connection with the prosecution of the pending patent applications included in the Existing Galapagos Patents.

(c) To Galapagos' Knowledge, there are no claims, judgments, or settlements against, or amounts with respect thereto, owed by Galapagos or any of its Affiliates relating to the Existing Regulatory Documentation, the Existing Galapagos Patents, or the Galapagos Know-How. No claim or litigation has been brought or threatened in writing by any Person against Galapagos alleging, and Galapagos has no Knowledge of any reasonable basis for any such claim or allegation, whether or not asserted, that (i) the Galapagos Patents or the Galapagos Know-How are invalid or unenforceable, or (ii) the Existing Regulatory Documentation, the Galapagos Patents, or the Galapagos Know-How, or the Development, Manufacture, Commercialization or other Exploitation of the Licensed Compounds or Licensed Products as contemplated herein, does or will violate, infringe, misappropriate or otherwise conflict or interfere with, any Patent or other intellectual property or proprietary right of any Third Party.

(d) To Galapagos's Knowledge, no Person is infringing or threatening to infringe or misappropriating or threatening to misappropriate the Galapagos Patents, the Galapagos Know-How, or the Regulatory Documentation. To Galapagos' Knowledge, there are no activities by Third Parties that would constitute infringement or misappropriation of the Galapagos Technology (in the case of pending claims, evaluating them as if issued).

(e) Each Person who, to Galapagos's Knowledge, has or has had any rights in or to any Existing Galapagos Patents or any Galapagos Know-How, has assigned and has executed an agreement assigning its entire right, title, and interest in and to such Existing Galapagos Patents and Galapagos Know-How to Galapagos. To Galapagos's Knowledge, no current officer, employee, agent, or consultant of Galapagos or any of its Affiliates is in violation of any term of any assignment or other agreement regarding the protection of Patents or other intellectual property or proprietary information of Galapagos or such Affiliate or of any employment contract relating to the relationship of any such Person with Galapagos.

(f) None of the intellectual property rights licensed hereunder by Galapagos to Gilead and existing as of the Effective Date are owned or Controlled in whole or in part by any Third Party.

(g) To Galapagos's Knowledge, all works of authorship and all other materials subject to copyright protection included in Galapagos Know-How are original and were either created by employees of Galapagos or its Affiliates within the scope of their employment or are otherwise works made for hire, or all right, title, and interest in and to such materials have been legally and fully assigned and transferred to Galapagos or such Affiliate, and all rights in all inventions and discoveries, made, developed, or conceived by any employee or independent contractor of Galapagos or any of its Affiliates during the course of their employment (or other retention) by Galapagos or such Affiliate, and relating to or included in Galapagos Know-How or that are the subject of one or more Existing Galapagos Patents have been or will be assigned in writing to Galapagos or such Affiliate.

(h) Galapagos has obtained the right (including under any Patents and other intellectual property rights) to use all Information and all other materials (including any formulations and manufacturing processes and procedures) developed or delivered by any Third Party under any agreements between Galapagos and any such Third Party with respect to the Licensed Compound or Licensed Product, and Galapagos has the rights under each such agreement to transfer such Information or other materials to Gilead and its designees and to grant Gilead the right to use such know-how or other materials in the Development or Commercialization of the Licensed Compounds or the Licensed Products without restriction.

(i) With respect to those portions of the Galapagos Know-How the confidentiality of which is material to the Exploitation of Licensed Products or Gilead Combination Products, such portions of the Galapagos Know-How have been kept confidential or have been disclosed to Third Parties only under terms of confidentiality. To Galapagos' Knowledge, and its Affiliates, no breach of such confidentiality has been committed by any Third Party.

(j) <u>No Proceedings</u>. There are no pending, and to Galapagos' Knowledge there are no threatened, actions, claims, demands, suits, proceedings, arbitrations, grievances, citations, summonses, subpoenas, inquiries or investigations of any nature, civil, criminal, regulatory or otherwise, in law or in equity, against Galapagos or any of its Affiliates or, to the knowledge of Galapagos, pending or threatened against any Third Party, in each case involving the Galapagos Technology, or relating to the transactions contemplated by this Agreement.

(k) <u>No Misappropriation</u>. The conception and reduction to practice of any inventions and the use or development of any other Information within the Galapagos Technology have not constituted or involved the misappropriation of trade secrets or other rights or property of any Third Party.

(1) <u>No Other Agreements</u>. Other than [...***...], Galapagos has not entered into (a) any agreement with a Third Party pursuant to which Galapagos grants or has granted rights to such Third Party to Commercialize any product containing a Licensed Compound, nor (b) any agreement with a Third Party pursuant to which Galapagos may be obligated to pay a royalty or other consideration with respect to sales of a Licensed Product or Gilead Combination Product, nor (c) any agreement that is material to, conflicts with, or restricts in any material manner, the rights licensed by Galapagos to Gilead hereunder.

(m) To Galapagos's Knowledge, Galapagos has provided or made available to Gilead, prior to the Effective Date, true, complete, and correct copies of (i) the file wrapper and other documents and materials relating to the prosecution, defense, maintenance, validity, and enforceability of the Galapagos Patents in the Major Markets (other than such file wrapper and other documents and materials that are reasonably available for download from publicly available electronic databases); and (ii) all material adverse information with respect to the safety and efficacy of the Licensed Compound known to Galapagos.

(n) Galapagos has no Knowledge of any scientific or technical facts or circumstances that have not been disclosed to Gilead, and that would, in Galapagos' reasonable estimation, adversely affect the scientific, therapeutic, or commercial potential of the Licensed Compounds or Licensed Products. Galapagos has no Knowledge of anything that has not been disclosed to Gilead, and that, in Galapagos' reasonable estimation, could adversely affect the acceptance, or the subsequent approval, by any Regulatory Authority of any filing, application or request for Regulatory Approval.

(o) To Galapagos's Knowledge, Galapagos and its Affiliates and licensees have generated, prepared, maintained, and retained all Regulatory Documentation that is required to be maintained or retained pursuant to and in accordance with Applicable Law, and all such information is true, complete and correct and what it purports to be.

(p) To Galapagos's Knowledge, Galapagos and its Affiliates have conducted, and their respective contractors licensees and consultants have conducted, all Development of the Licensed Compounds or the Licensed Products that they have conducted prior to the Effective Date in accordance with Applicable Law. To Galapagos's Knowledge, Galapagos has conducted, and has caused its licensees, contractors and consultants to conduct, any and all pre-clinical and clinical studies related to the Licensed Compounds and Licensed Products in accordance with Applicable Law. To Galapagos' Knowledge, Galapagos and its Affiliates and licensees have employed (and, with respect to such tests and studies that Galapagos will perform, will employ) Persons with appropriate education, knowledge and experience to conduct and to oversee the conduct of the pre-clinical and clinical studies with respect to the Licensed Compounds and Licensed Products.

(q) To Galapagos's Knowledge, neither Galapagos nor any of its Affiliates or licensees, nor any of its or their respective officers, employees, or agents has made an untrue statement of material fact or fraudulent statement to the FDA or any other Regulatory Authority with respect to the Development of the Licensed Compounds or the Licensed Products, failed to disclose a material fact required to be disclosed to the FDA or any other Regulatory Authority with respect to the Development of the Licensed Compounds or the Licensed Products, or committed an act, made a statement, or failed to make a statement with respect to the Development of the Licensed Compounds or the Licensed Products that could reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities", set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto or any analogous laws or policies in the Territory.

(r) Except with respect to [...***...], to Galapagos's Knowledge, there are no royalties or other amounts based on sales of Licensed Compounds or Licensed Products, which royalties or other amounts will be required to be paid to a Third Party as a result of the Development or Commercialization of the Licensed Compounds or Licensed Products that arise out of any agreement to which Galapagos or any of its Affiliates is a party.

(s) The inventions claimed in the Existing Galapagos Patents (i) were not conceived or made in connection with any research activities funded, in whole or in part, by the federal government of the United States or any agency thereof, (ii) are not a "subject invention" as that term is described in 35 U.S.C. Section 201(f), and (iii) are not otherwise subject to the provisions of the Bayh-Dole Act.

(t) Neither the execution, delivery and performance by Galapagos of this Agreement and the Subscription Agreement, nor the consummation by Galapagos of the transactions contemplated hereby or thereby, will, other than with respect to the HSR Act, article 566 of the Belgian Companies Code (solely in relation to ratification of the Change of Control provisions hereof), the Belgian Act of May 2, 2007 on the disclosure of major shareholdings in issuers whose shares are admitted to trading on a regulated market or the Belgian Royal Decree of February 2, 2008 on the disclosure of major shareholdings, require Galapagos to (i) obtain any consent or authorization of, or (ii) give any notice to, or make any filing or registration with, any Governmental Authority or other Person.

10.4 <u>No Transfer of Title</u>. Galapagos covenants and agrees that from the Effective Date until the expiration of the Term, neither it nor its Affiliates shall enter into any agreement with any Third Party, whether written or oral, with respect to, or otherwise assign, transfer, license, or convey its right, title or interest in or to, the Galapagos Technology, in each case, that is in conflict with the rights granted by Galapagos to Gilead under this Agreement or that would prevent Galapagos from performing its obligations under this Agreement or prevent Gilead from exercising its rights hereunder. In consideration of the foregoing covenant, Galapagos hereby grants to Gilead a fully paid up exclusive, sublicensable (solely as permitted in accordance with <u>Section 7.2</u>) license under all Information, if any, other than Galapagos Know-How, which Information is Controlled as of the Effective Date or during the Term by Galapagos or its Affiliate(s) and is reasonably necessary or useful to Exploit Licensed Compound or Licensed Products or Gilead Combination Products in the Field in the Territory.

10.5 <u>Disclaimer</u>. Galapagos makes no representations, warranties or covenants except as expressly set forth in this <u>Article 10</u> concerning the Galapagos Technology.

10.6 <u>No Other Representations or Warranties</u>. EXCEPT AS EXPRESSLY STATED IN THIS AGREEMENT OR THE CO-COMMERCIALIZATION AGREEMENTS, NO REPRESENTATIONS OR WARRANTIES WHATSOEVER, WHETHER EXPRESS OR IMPLIED, INCLUDING WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT, OR NON-MISAPPROPRIATION OF THIRD PARTY INTELLECTUAL PROPERTY RIGHTS, IS MADE OR GIVEN BY OR ON BEHALF OF A PARTY. EXCEPT AS EXPRESSLY STATED IN THIS AGREEMENT, ALL REPRESENTATIONS AND WARRANTIES, WHETHER ARISING BY OPERATION OF LAW OR OTHERWISE, ARE HEREBY EXPRESSLY EXCLUDED.

ARTICLE 11

INDEMNIFICATION

11.1 <u>Indemnification by Galapagos</u>. Galapagos shall defend, indemnify, and hold Gilead, its Affiliates, subcontractors, Sublicensees and Distributors, and its and their respective officers, directors, employees, and agents (the "**Gilead Indemnitees**") harmless from and against any and all damages or other amounts payable to a Third Party claimant, as well as any reasonably incurred attorneys' fees and costs of litigation incurred by such Gilead Indemnitees (collectively, "**Gilead Damages**"), all to the extent resulting from claims, suits, proceedings or causes of action brought by or on behalf of such Third Party ("**Gilead Claims**") against such Gilead Indemnitee that arise from or are based on:

(a) (i) a breach of any of Galapagos's representations, warranties and obligations under this Agreement or any Co-Commercialization Agreement; (ii) the willful misconduct or grossly negligent acts of Galapagos, its Affiliates, subcontractors or Sublicensees (excluding Gilead, its Affiliates, and Sublicensees as licensees or Sublicensees of Galapagos hereunder), or the officers, directors, employees, or agents of Galapagos or its Affiliates, subcontractors, or such Sublicensees; or (iii) any violation of Applicable Law by Galapagos, its Affiliates, subcontractors or Sublicensees (excluding Gilead, its Affiliates, and Sublicensees as licensees or Sublicensees or Sublicensees of Galapagos hereunder), or the officers, directors, employees, or agents of Galapagos or its Affiliates, and Sublicensees as licensees or Sublicensees or Sublicensees of Galapagos hereunder), or the officers, directors, employees, or agents of Galapagos or its Affiliates, contractors or such Sublicensees; excluding, in each case ((i), (ii) and (iii)), any damages or other amounts to the extent that Gilead has an obligation to indemnify any Galapagos Indemnitee pursuant to Sections 11.2(b) though (d); and

(b) (i) the Exploitation by or on behalf of Galapagos or its Affiliates, subcontractors, licensees or Sublicensees (excluding such conduct by or on behalf of Gilead, its Affiliates and Sublicensees as licensees or Sublicensees of Galapagos hereunder) of any Licensed Product or Licensed Compound (A) prior to the Effective Date, and (B) for the benefit of any Terminated Region following the applicable effective date of termination; or (ii) the exercise or use by or on behalf of Galapagos, its Affiliates, subcontractors, licensees, or Sublicensees (excluding such exercise by Gilead, its Affiliates, and Sublicensees as licensees of Galapagos hereunder) of rights under any license or right of reference, or in or to any Regulatory Materials, Regulatory Approvals, Marks or other Information, in each case granted, transferred or made available by or on behalf of Gilead or any of its Affiliates to Galapagos following or in connection with termination of this Agreement with respect to any Terminated Region(s), including pursuant to any post-termination transition agreement.

11.2 Indemnification by Gilead. Gilead shall defend, indemnify, and hold Galapagos, its Affiliates, subcontractors, distributors, licensees and Sublicensees, and each of their respective officers, directors, employees, and agents, (the "Galapagos Indemnitees") harmless from and against any and all damages or other amounts payable to a Third Party claimant, as well as any reasonably incurred attorneys' fees and costs of litigation incurred by such Galapagos Indemnitees (collectively, "Galapagos Damages"), all to the extent resulting from any claims, suits, proceedings or causes of action brought by such Third Party (collectively, "Galapagos Claims") against such Galapagos Indemnitee that arise from or are based on: (a) the Exploitation of the Licensed Compound, Licensed Product or Gilead Combination Product by Gilead or its Affiliates, subcontractors, Distributors or Sublicensees in the Territory, but excluding the Shared Program Activities; (b) a breach of any of Gilead's representations, warranties, and obligations under the Agreement or any Co-Commercialization Agreement; (c) the willful misconduct or grossly negligent acts of Gilead or its Affiliates, subcontractors, Distributors, or Sublicensees; or (d) any violation of Applicable Law by Gilead, its Affiliates, subcontractors, Distributors, or Sublicensees; or sagents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractors, employees, or agents of Gilead or its Affiliates, subcontractor

11.3 Indemnification Procedures. The Party claiming indemnity under this <u>Article 11</u> (the "Indemnified Party") shall give written notice to the Party from whom indemnity is being sought (the "Indemnifying Party") promptly after learning of the claim, suit, proceeding or cause of action for which indemnity is being sought ("Claim"). The Indemnifying Party's obligation to defend, indemnify, and hold harmless pursuant to <u>Section 11.1</u> or <u>11.2</u>, as applicable, shall be reduced to the extent the Indemnifying Party assume the defense of any Claim for which indemnity is being sought by giving written notice to the Indemnified Party within [...***...] days after receipt of the notice of the Claim. The assumption of defense of the Claim shall not be construed as an acknowledgment that the Indemnifying Party is liable to indemnified Party's claim for indemnification. The Indemnified Party shall provide the Indemnifying Party of any defenses it may assert against the Indemnified Party's claim for indemnification. The Indemnified Party shall provide the Indemnifying Party with reasonable assistance, at the Indemnifying Party's expense, in connection with the defense. The Indemnified Party may participate in and monitor such defense with counsel of its own choosing at its sole expense; *provided, however, that* the Indemnifying Party shall have the right to assume and conduct the defense of the Claim with counsel of its choice. The Indemnifying Party shall not settle any Claim without the prior written consent of the Indemnified Party, not to be unreasonably withheld, conditioned or delayed,

unless the settlement involves only the payment of money. The Indemnified Party shall not settle any such Claim without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld, conditioned or delayed. If the Indemnifying Party does not assume and conduct the defense of the Claim as provided above, (a) the Indemnified Party may defend against, and consent to the entry of any judgment or enter into any settlement with respect to the Claim in any manner the Indemnified Party may deem reasonably appropriate (and the Indemnified Party need not consult with, or obtain any consent from, the Indemnifying Party in connection therewith), and (b) the Indemnified Party reserves any right it may have under this <u>Article 11</u> to obtain indemnification from the Indemnifying Party.

11.4 <u>Certain Third Party Claims Related to Licensed Products and Gilead Combination Products in the Shared Territory</u>. The Parties shall share in any Shared Program Damages. With respect to any Shared Program Damages incurred by a Party (or any of its Indemnified Persons) during the Term, such Shared Program Damages shall be deemed to constitute (and shall be included in) Joint Commercialization Costs. After the Term, any Shared Program Damages that accrued during the Term shall continue to be shared with fifty percent (50%) borne by Gilead and fifty percent (50%) borne by Galapagos. If either Party receives notice of a Third Party claim that arises from or is based on any Shared Program Activities, such Party shall inform the other Party in writing as soon as reasonably practicable, and the Parties shall discuss a strategy on how to defend against such Third Party claim.

11.5 <u>Limitation of Liability</u>. NEITHER PARTY SHALL BE LIABLE TO THE OTHER FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, PUNITIVE, OR INDIRECT DAMAGES ARISING FROM OR RELATING TO ANY BREACH OF THIS AGREEMENT OR ANY TORT CLAIMS ARISING HEREUNDER, REGARDLESS OF ANY NOTICE OF THE POSSIBILITY OF SUCH DAMAGES. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS <u>SECTION 11.5</u> IS INTENDED TO OR SHALL LIMIT OR RESTRICT (A) THE INDEMNIFICATION RIGHTS OR OBLIGATIONS OF ANY PARTY UNDER <u>SECTION 11.1</u>, 11.2 OR 11.4, (B) DAMAGES AVAILABLE FOR A PARTY'S BREACH OF ITS CONFIDENTIALITY OBLIGATIONS UNDER <u>ARTICLE 12</u>, OR (C) DAMAGES AVAILABLE IN THE CASE OF A PARTY'S FRAUD, GROSS NEGLIGENCE OR INTENTIONAL MISCONDUCT.

11.6 <u>Insurance</u>. Each Party shall maintain in full force and effect during the term of this Agreement insurance required by law in the country where work is performed. Without limiting the foregoing each Party shall maintain in full force and effect during the term of this Agreement, either reasonable self-insurance with the ability to cover the liabilities of such Party that could reasonably occur in view of the activities of such Party under this Agreement and the Ancillary Agreements, or insurance policies with the following insurance coverages, with limits of liability not less than those specified below:

(a) Commercial General Liability with minimum limits of [...***...] each occurrence, [...***...] General Aggregate, and [...***...] Products/Completed Operations Aggregate, including coverage for premises liability, personal and advertising injury, products and completed operations liability, clinical trial liability, contractual liability, and broad form property damage. Each policy shall name the other party as an additional insured as respects liability arising from premises rented or owned and liability arising from all ongoing operations. Such insurance may be provided on a claims-made basis, however, such insurance shall have a retroactive date prior to the date that any work will be performed pursuant to the Agreement, and shall be maintained (or shall have an extended reporting period) of at least [...***...] years after the termination of this Agreement. The use of primary and excess limits to achieve the total required limits is acceptable.

(b) Workers' Compensation insurance in compliance by the local law requirements of the state/jurisdiction in which the work is to be performed and Employer's Liability insurance in amounts not less than [...***...] Bodily Injury by Accident-Each Accident, [...***...] Bodily Injury by Disease-Policy Limit, and [...***...] Bodily Injury by Disease-Each Employee. Where permitted by law, such policies shall contain a waiver of the insurer's subrogation rights against the other Party.

(c) All insurance programs provided are required to be maintained hereunder shall be from insurers having an A.M. Best rating of [... ***...] or better, or its equivalent.

(d) To the extent requested by the other Party, each Party shall provide the other with an original certificate of insurance evidencing that (i) all such insurance coverages are in effect, and (ii) none of the required policies of insurance shall be terminated or canceled by insurers except upon at least [...***...] calendar days written notice to the other Party. Any failure to maintain the insurance coverage required by this Section shall be a material breach which may be cured only by restoring such coverage retroactive to the date of lapse of the prior coverage. Nothing contained in this Section is intended, or shall be construed, to limit either Party's indemnity obligations.

(e) Automobile Liability insurance for bodily injury, property damage and automobile contractual liability covering all owned, hired and non-owned autos with a combined single limit of liability for each accident of not less than [...***...].

ARTICLE 12

CONFIDENTIALITY

12.1 <u>Confidential Information</u>. Except to the extent expressly authorized by this Agreement or otherwise agreed in writing by the Parties, each Party agrees that, during the Term and for [...***...] years thereafter, it shall, and shall cause its Affiliates, to keep confidential and not publish or otherwise disclose to any Third Party, and not use for any purpose other than as provided for in this Agreement or any Ancillary Agreement, any Confidential Information of the other Party or any of its Affiliates, *provided that* each Party and its Affiliates may disclose the Confidential Information of the other Party or any of its Affiliates, *provided that* each Party and its Affiliates may disclose the Confidential Information of the other Party or its Affiliates to the receiving Party's and its Affiliates' officers, directors, employees, agents and advisors who in each case are bound by commercially reasonable obligations of confidentiality with respect to the use and disclosure of such Confidential Information. Notwithstanding the foregoing, Confidential Information of a Party or its Affiliate shall exclude that portion of such information or materials that the receiving Party's Affiliate) can demonstrate by competent written proof:

(a) was already known to the receiving Party or its Affiliate, other than under an obligation of confidentiality, at the time of disclosure by the other Party;

(b) was generally available to the public or otherwise part of the public domain at the time of its disclosure to the receiving Party;

(c) became generally available to the public or otherwise part of the public domain after its disclosure and other than through any wrongful act, fault, or negligence of the receiving Party;

(d) is subsequently disclosed to the receiving Party or its Affiliate by a Third Party without obligations of confidentiality with respect thereto; or

(e) is independently discovered or developed by the receiving Party or its Affiliate without the aid, application, or use of Confidential Information;

provided that specific disclosures made under this Agreement shall not be deemed to be subject to any of the foregoing exceptions merely because they are embraced by general disclosures in the public knowledge or literature or in the possession of the receiving Party, and any combination of features disclosed under this Agreement shall not be deemed subject to the above exceptions merely because individual features are in the public knowledge or literature or in the possession of the receiving Party. The Parties acknowledge that Confidential Information has been provided by the Parties to each other prior to the Effective Date pursuant to the Existing Confidentiality Agreement and the Subscription Agreement, including the terms and conditions thereof. The Parties agree that as of the Effective Date, all such Confidential Information shall be protected by the terms and conditions of this Agreement, which shall replace those of such Existing Confidentiality Agreement.

12.2 <u>Authorized Disclosure of Confidential Information</u>. Notwithstanding <u>Section 12.1</u>, each Party may disclose Confidential Information to the extent such disclosure is reasonably necessary in the following situations:

(a) filing or prosecuting Galapagos Patents, Gilead Foreground Patents, Gilead Combination Patents and Joint Patents in accordance with <u>Article 9</u> with the consent of the other Party, such consent not to be unreasonably withheld, delayed or conditioned;

(b) regulatory filings and other filings with Governmental Authorities (including Regulatory Authorities), including filings with the SEC or FDA, with respect to a Licensed Product as permitted hereunder, *provided that* such disclosure is, in the opinion of outside counsel required;

(c) responding to a valid order of a court of competent jurisdiction or other competent authority; *provided that* the receiving Party shall, to the extent reasonably practicable under the circumstances, first have given to the disclosing Party notice and a reasonable opportunity to quash the order or obtain a protective order requiring that the Confidential Information be held in confidence or used only for the purpose for which the order was issued; and *provided further that* if such order is not quashed or a protective order is not obtained, the Confidential Information disclosed shall be limited to the information that is legally required to be disclosed;

(d) complying with Applicable Law;

(e) disclosure to its Affiliates and Third Parties only on a need-to-know basis and solely in connection with the performance by the disclosing Party of its obligations or the exercise of its rights under this Agreement or any Co-Commercialization Agreement (including with respect to Development, Manufacturing and Commercialization of Licensed Products); *provided that* each disclosee, prior to any such disclosure, must be bound by obligations of confidentiality and non-use at least as equivalent in scope as those set forth in <u>Sections 12.1</u> and <u>12.2</u>;

(f) with prior notice to the other Party as permitted by Applicable Law, disclosure of the material terms of this Agreement or any Ancillary Agreement to any *bona fide* potential or actual investor, investment banker, acquirer, merger partner or other potential or actual financial partner; *provided that* each disclosee must be bound by obligations of confidentiality and non-use at least as restrictive as those set forth in <u>Sections 12.1</u> and 12.2 prior to any such disclosure, except that, where the disclosee is an investor, investment banker or financial partner, such disclosee shall only need to be bound by commercially reasonable confidential terms; and

(g) disclosure of any Collaboration results or status reports (including data from any Clinical Trials) by Gilead, or in the case of Galapagos, to any *bona fide* potential or actual investor, investment banker, acquirer, merger partner or other *bona fide* potential or actual financial partner with the consent of Gilead, such consent not to be unreasonably withheld, delayed or conditioned; *provided that*

each disclosee must be bound by obligations of confidentiality and non-use at least as equivalent in scope as those set forth in <u>Sections 12.1</u> and <u>12.2</u> prior to any such disclosure, except that, where the disclosee is an investor, investment banker or financial partner, such disclosee shall only need to be bound by commercially reasonable confidential terms. Gilead shall not withhold its approval to such disclosure by Galapagos of any information that, in the opinion of its outside counsel, is required by Applicable Law to be so disclosed.

Notwithstanding the foregoing, in the event that a Party is required to make a disclosure of the other Party's Confidential Information pursuant to <u>Sections 12.2(a)</u>, <u>12.2(b)</u> or <u>12.2(d)</u>, it will, except where impracticable, give reasonable advance notice to the other Party of such disclosure and use reasonable efforts to secure confidential treatment of such information. In any event, the Parties agree to take all reasonable action to avoid disclosure of Confidential Information hereunder.

12.3 Terms of Agreement.

(a) The Parties agree that the material terms of the License and Collaboration Agreement and this Agreement and any Ancillary Agreements are the Confidential Information of both Parties, subject to the special authorized disclosure provisions set forth in <u>Section 12.2</u> and this <u>Section 12.3</u>.

(b) If either Party or any of its Affiliates desires to make a press release or other similar public announcement concerning the terms of the License and Collaboration Agreement, the First Amendment and this Agreement or any Co-Commercialization Agreement or Galapagos or any of its Affiliates desires to make a press release or other similar public announcement concern any activities under the License and Collaboration Agreement, the First Amendment and this Agreement, in either case, such Party shall give reasonable prior advance notice of the proposed text of such press release or announcement to the other Party for its prior review and approval (except as otherwise provided herein), such approval not to be unreasonably withheld, conditioned or delayed, except that, subject to <u>Section 12.3(c)</u>, in the case of a press release or filings with a Governmental Authority required by Applicable Law, such Party shall provide the other Party with such advance notice as it reasonably can and shall not be required to obtain approval therefor. The other Party shall provide its comments, if any, within [...***...] Business Days after receiving the press release for review. Neither Party shall withhold its approval to disclosure by the other Party of any information that, in the opinion of its outside counsel, is required by Applicable Law to be disclosed. Neither Party shall be required to seek the permission of the other Party to repeat any information regarding the terms of the License and Collaboration Agreement, the First Amendment and this Agreement that have already been publicly disclosed by such Party or such Party's Affiliates, or by the other Party or any of its Affiliates, in accordance with this <u>Section 12.3; provided that</u> such information remains accurate as of such repeat.

(c) The Parties acknowledge that either or both Parties may be obligated to make a filing (including to file a copy of this Agreement and the Subscription Agreement) with the SEC or other Governmental Authorities. Each Party shall be entitled to make such a required filing, *provided that* it shall (i) agree (such agreement not to be unreasonably withheld, conditioned or delayed) with the other Party in advance regarding the form of redacted copy of the License and Collaboration Agreement, the First Amendment and this Agreement and the Subscription Agreement to be so filed (the "**Redacted Agreements**"), (ii) request, and use commercially reasonable efforts consistent with Applicable Laws to obtain, confidential treatment of all terms redacted from the License and Collaboration Agreement, the First Amendment and this Agreement and the Subscription Agreement, as reflected in the Redacted Agreement, for a period of at least [...***...] years, (iii) promptly deliver to the other Party any written correspondence received by it or its representatives from such Governmental Authority with respect to such confidential treatment request and promptly advise the other Party of any other material communications between it or

its representatives with such Governmental Authority with respect to such confidential treatment request, (iv) upon the written request of the other Party, if legally justifiable, request an appropriate extension of the term of the confidential treatment period, and (v) if such Governmental Authority requests any changes to the redactions set forth in the Redacted Agreement, use commercially reasonable efforts consistent with Applicable Laws to support the redactions in the Redacted Agreement as originally filed and not agree to any changes to the Redacted Agreement without, to the extent practical, first discussing such changes with the other Party and taking the other Party's comments into consideration when deciding whether to agree to such changes. Each Party shall be responsible for its own legal and other external costs in connection with any such filing, registration or notification.

12.4 <u>Public Disclosures of Data</u>. Neither Galapagos nor any of its Affiliates shall, except as may be required by Applicable Law, publicly disclose data or results of Clinical Trials or Nonclinical Studies that have not already been publicly disclosed with respect to any Licensed Product (whether conducted prior to or during the Term of this Agreement), except as provided in this <u>Section 12.4</u>.

(a) <u>Press Releases</u>. The Parties shall coordinate to issue a joint press release covering the top line results of all material Clinical Trials or Nonclinical Studies. Gilead shall have the final authority to decide when and if any such press release shall be made and, subject to the following sentence, as to the content thereof, in all cases subject to the ability of Galapagos to make any disclosure required by Applicable Law. Gilead shall provide a copy of any such proposed press release to Galapagos at least [...***...] Business Days, or such shorter period as Gilead may require in order to comply with Applicable Law, prior to the date of release thereof and consider in good faith any comments provided by Galapagos, provided that Galapagos shall keep confidential and not disclose the information contained in such proposed press release prior to the release thereof by Gilead.

(b) <u>Publications</u>. Any oral presentation, abstract or poster of any such data and results at scientific and medical conferences or publications of such data and results in peer reviewed journals (collectively, "**Publications**") shall be made only pursuant to this <u>Section 12.4(b)</u>. Gilead shall control the dissemination of all Publications and shall have the final authority to decide which Publications are made. Gilead shall provide to Galapagos a copy of all proposed oral presentations, abstracts or posters at least [...***...] Business Days prior to the date of submission thereof and a copy of all other Publications at least [...***...] Business Days prior to the date of submission thereof, and in each case consider in good faith any comments provided by Galapagos. Galapagos shall have a right to propose to Gilead that a Publication be made, and [...***...].

ARTICLE 13

TERM AND TERMINATION

13.1 <u>Term</u>. The License and Collaboration Agreement became effective as of the Effective Date and remain in effect through the Amendment Effective Date ("**Original Term**"), and this Agreement shall become effective as of the Amendment Effective Date and, unless earlier terminated pursuant to this <u>Article 13</u>, shall expire (a) in the Licensed Territory, on a country-by-country basis at the end of the Royalty Term in such country, and (b) in the Shared Territory, on a country-by-country basis, at such time as a Generic Product is first sold in such country (the "**Amended Agreement Term**", and the Original Term and the Amended Agreement Term, collectively, the "**Term**").

13.2 [Intentionally Omitted]

13.3 Termination Rights of each Party.

(a) Termination by Gilead.

(i) <u>At Will</u>. Subject to <u>Section 13.3(a)(ii)</u>, at any time after [...***...] (the "**Earliest Termination Date**"), Gilead shall have the right to terminate this Agreement upon (Y) if prior to the date of submission of the first Marketing Authorization Application for a Licensed Product, [...***...] days prior written notice, and (Z) if on or after the date of submission of the first Marketing Authorization Application for a Licensed Product, [...***...] days prior written notice, (such termination in each case, a "**Termination at Will**").

(ii) [...***...] Gilead shall have the right to terminate this Agreement on a Licensed Product-by-Licensed Product, indication-byindication, or country-by-country basis upon [...***...] days prior written notice to Galapagos if [...***...], that [...***...] or [...***...], including [...***...], render [...***...] of the [...***...] of the [...***...] of the [...***...] for Gilead (collectively, [...***...]).

(b) Termination by Galapagos.

(i) <u>Termination for IP Challenge</u>. Galapagos shall have the right to terminate this Agreement in its entirety upon [...***...] days' written notice to Gilead in the event that [...***...] (a "**Patent Challenge**"); [...***...].

13.4 Termination by Either Party for Breach or Insolvency.

(a) <u>Breach</u>. Subject to <u>Section 13.4(b)</u>, Galapagos shall have the right to terminate this Agreement in its entirety or with respect to any country upon written notice to Gilead if Gilead materially breaches its obligations under this Agreement with respect to such country <u>provided that</u> the materiality of such breach shall be measured in relation to the entire Territory, and, after receiving written notice from Galapagos identifying such material breach by Gilead in reasonable detail, fails to cure such material breach within [...***...] days from the date of such notice (or within [... ***...] days from the date of such notice in the event such material breach is solely based upon Gilead's failure to pay any amounts due Galapagos hereunder). Subject to <u>Section 13.4(b)</u> and <u>13.4(c)</u>, Gilead shall have the right to terminate this Agreement in its entirety or with respect to a country upon written notice to Galapagos if Galapagos materially breaches its obligations under this Agreement with respect to such country and, after receiving written notice from Gilead identifying such material breach by Galapagos in reasonable detail of its obligations under this Agreement, fails to cure such material breach within [...***...] days from the date of such notice in the event such material breach by Galapagos in reasonable detail of its obligations under this Agreement, fails to cure such material breach within [...***...] days from the date of such notice in the event such material breach by Galapagos in reasonable detail of its obligations under this Agreement, fails to cure such material breach within [...***...] days from the date of such notice in the event such material breach is solely based upon Galapagos' failure to pay any amounts due Gilead hereunder).

(b) <u>Disputed Breach</u>. If the alleged breaching Party disputes in good faith the existence or materiality of a breach specified in a notice provided by the other Party in accordance with <u>Section 13.4(a)</u>, and such alleged breaching Party provides the other Party notice of such dispute within such [...***...] day or [...***...] day period, as applicable, then the non-breaching Party shall not have the right to terminate this Agreement under <u>Section 13.4(a)</u> unless and until an arbitrator, in accordance with <u>Article 14</u>, has determined that the alleged breaching Party has materially breached the Agreement and that such Party fails to cure such breach within [...***...] days following such arbitrator's decision (except to the extent such breach involves the failure to make a payment when due, which breach must be cured within [...***...] days following such arbitrator's decision). It is understood and agreed that during the pendency of such dispute, all of the terms and conditions of this Agreement shall remain in effect.

(c) [...***...].

(d) <u>Insolvency</u>. If, at any time during the Term (i) a case is commenced by or against either Party under Title 11, United States Code, as amended, or analogous provisions of Applicable Law outside the United States (the "**Bankruptcy Code**") and, in the event of an involuntary case under the Bankruptcy Code, such case is not dismissed within [...***...] days after the commencement thereof, (ii) either Party files for or is subject to the institution of bankruptcy, liquidation or receivership proceedings (other than a case under the Bankruptcy Code), (iii) either Party assigns all or a substantial portion of its assets for the benefit of creditors, (iv) a receiver or custodian is appointed for either Party's business, or (v) a substantial portion of either Party's business is subject to attachment or similar process; then, in any such case ((i), (ii), (iii), (iv) or (v)), the other Party may terminate this Agreement upon written notice to the extent permitted under Applicable Law.

13.5 <u>Effects of Termination of the Agreement</u>. Upon termination of this Agreement with respect to one or more countries of the Territory or in its entirety (the "**Terminated Regions**"; with the entire Territory being the Terminated Regions in the event of termination of this Agreement in its entirety), the following shall apply with respect to the Terminated Regions (in addition to any other rights and obligations under this <u>Article 13</u> or otherwise under this Agreement with respect to such termination):

(a) Licenses. The licenses granted in <u>Article 7</u> shall terminate with respect to the Terminated Regions, except that limited license rights shall remain in effect in the Terminated Region(s) solely for the limited purpose of allowing Gilead to Develop or Manufacture Licensed Product(s) (including any intermediate thereof or any active pharmaceutical ingredient or other material contained therein) in the Terminated Region(s) for sale or distribution thereof in any country which has not been terminated. Subject to the foregoing, effective upon the effective date of termination of this Agreement with respect to any Terminated Region, Gilead hereby grants to Galapagos, effective only upon such termination, an exclusive, perpetual, irrevocable, royalty-bearing license, with the right to grant multiple tiers of sublicenses, under (i) the Gilead Foreground Know-How and any Patent claiming Gilead Foreground Know-How, (ii) the Joint Patents, and (iii) any other Patents Controlled by Gilead that, absent a license, as such Reversion Product as it exists in the applicable Terminated Region, (the "**Gilead Background Patents**") as such Patents and interests in such Patents exist as of the effective date of terminated Regions. For clarity, the foregoing license under Gilead Background Patents extends solely to those elements of a Reversion Product that incorporate additional technology that would infringe a Gilead Background Patent. The foregoing license shall [...***...].

(b) <u>Marks</u>. Gilead shall assign to Galapagos all right, title and interest in and to those Marks used exclusively with Reversion Products and used exclusively in the Terminated Regions (excluding any such Marks that include, in whole or part, any corporate name or logo of Gilead or its Affiliate or Sublicensee).

(c) <u>Regulatory Materials</u>. Gilead shall grant to Galapagos a right of reference under all Regulatory Materials and Regulatory Approvals for Reversion Products in the Terminated Regions that are Controlled by Gilead or its Affiliates or Sublicensees, unless and until assigned to Galapagos pursuant to any Transition Agreement.

(d) Conduct During Termination Notice Period.

(i) Following any notice of termination permitted under this <u>Article 13</u> other than any termination pursuant to <u>Section 13.4</u>, during any applicable termination notice period (the applicable "**Termination Notice Period**"), each Party shall continue to perform all of its obligations under this Agreement, including performing all activities allocated to it pursuant to the Development Plan and Shared Territory Commercialization Plan and Budget, respectively, then in effect in accordance with the terms and conditions of this Agreement. In such circumstances, each Party shall also continue to bear its share of all Development Costs and Joint Commercialization Costs in each case incurred during the Termination Notice Period.

(ii) During the applicable Termination Notice Period, neither Party shall [...***...].

(e) <u>Transition Agreement</u>. In connection with the termination of this Agreement in its entirety or with respect to one or more countries, the Parties shall enter into a written agreement (the "**Transition Agreement**") that would effectuate the terms and conditions of this <u>Section 13.5(e)</u> and would include other reasonable terms and conditions, including terms allocating costs and expenses, describing the Parties' indemnification obligations, setting forth the Parties' obligations with respect to unauthorized sales, and setting forth other coordination obligations. If, despite such efforts, the Parties are unable to agree upon such terms and conditions within [...***...] days from the effective date of the termination, either Party may refer the dispute for resolution by arbitration in accordance with <u>Section 14.2</u>, and the arbitrator shall have the authority to require the Parties to execute a Transition Agreement in the form approved by the arbitrator.

(i) <u>Transition Assistance</u>. The Transition Agreement shall require Gilead to, [...***...], provide reasonable consultation and assistance for a period of no more than [...***...] days for the purpose of disclosing and providing to Galapagos, all [...***...] that is relevant to the Reversion Products and the applicable Terminated Regions, and, at Galapagos' request, all then-existing commercial arrangements to the extent relating solely and specifically to the Reversion Products and the applicable Terminated Regions that Gilead is able, using reasonable commercial efforts, to disclose and provide to Galapagos, in each case, to the extent reasonably necessary or useful for Galapagos to commence or continue researching, Developing, Manufacturing or Commercializing the Reversion Products with respect to the applicable Terminated Regions. The foregoing shall include assigning, upon request of Galapagos, any agreements with Third Party suppliers or vendors to the extent they solely and specifically cover the supply or sale of Reversion Products in applicable Terminated Regions. If any such contract between Gilead and a Third Party is not assignable to Galapagos (whether by such contract's terms or because such contract does not relate specifically to Reversion Products or the Terminated Regions) but is otherwise reasonably necessary or useful for Galapagos to commence or continue researching, Developing, Manufacturing, or Commercializing Reversion Products with respect to the Terminated Regions, then Gilead shall reasonably cooperate with Galapagos in Galapagos' efforts to obtain from such Third Party the assignment of such contract or of that portion of such contract that solely relates to researching, Developing, Manufacturing, or Commercializing Reversion Products with respect to the Terminated Regions. Unless and until the necessary Third Party Manufacturing agreements are assigned to Galapagos pursuant to the preceding sentences, or if Gilead Manufactures the Reversion Products itself (and thus there is no contract to assign), the Transition Agreement shall require Gilead to supply such bulk finished Reversion Product, as applicable, to Galapagos for a reasonable period (not to exceed [...***...] months) to enable Galapagos to establish an alternate, validated source of supply for the applicable Reversion Products. The cost to Galapagos for such supply shall be the cost of goods for such Reversion Products calculated in accordance with industry standards (excluding [...***...]) plus [...***...] percent [...***...]%.

(ii) <u>Regulatory Materials</u>. The Transition Agreement shall require Gilead to transfer and assign to Galapagos all Regulatory Materials and Regulatory Approvals in and for the benefit of the Terminated Regions solely relating to Reversion Products that are owned by Gilead or its Affiliates. The Transition Agreement shall contain terms governing the coordination of the Party's ongoing regulatory responsibilities with respect to Licensed Products.

(f) <u>Third-Party Agreements</u>. To the extent that any payments would be owed by Gilead to any Third Parties (including royalties, milestones and other amounts) under any Third Party agreements that are applicable to the grant to Galapagos of any (sub)license, right of reference or other right provided in this <u>Section 13.5</u> or the Transition Agreement, or that are applicable to the exercise by Galapagos or any of its Affiliates or Sublicensees of any sublicense or other right with respect thereto, Gilead shall notify Galapagos of the existence and anticipated amounts of such payments and Galapagos shall have the right either to decline such (sub)license, right of reference or other right provided in this <u>Section 13.5</u> or the Transition Agreement or to take the same, in which case Galapagos agrees to comply with any obligations under such agreements of Gilead that apply to Galapagos and of which Galapagos was informed by Gilead and to make such payments.

(g) Remaining Inventories.

(i) Gilead shall be entitled, during the [...***...] days following termination of this Agreement, to finish any work-in-progress and to sell in the Terminated Regions any inventory of Licensed Product that remains on hand as of the effective date of the termination. Gilead shall pay Galapagos the royalties applicable to such sales in accordance with the terms and conditions of this Agreement.

(ii) At any time within [...***...] days after the effective date of termination with respect to any Terminated Region(s), Galapagos shall have the right, upon written notification to Gilead, to purchase from Gilead any or all of the inventory of Reversion Products held by Gilead with respect to such Terminated Region(s) as of the date of such notice solely for distribution in the Terminated Region(s) and not for distribution in other countries (that are not committed to be supplied to any Third Party or Sublicensee as of such date) at a price equal to the cost of goods for such Reversion Products calculated in accordance with industry standards (excluding [...***...]) plus [...***...] percent [...***...]%.

13.6 <u>Other Remedies</u>. Termination or expiration of this Agreement for any reason shall not release either Party from any liability or obligation that already has accrued prior to such expiration or termination, nor affect the survival of any provision hereof to the extent it is expressly stated to survive such termination. Termination or expiration of this Agreement for any reason shall not constitute a waiver or release of, or otherwise be deemed to prejudice or adversely affect, any rights, remedies or claims, whether for damages or otherwise, that a Party may have hereunder or that may arise out of or in connection with such termination or expiration.

13.7 <u>Rights in Bankruptcy</u>. All rights and licenses granted under or pursuant to this Agreement by Galapagos and Gilead are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the U.S. Bankruptcy Code, licenses of right to "intellectual property" as defined under Section 101 of the U.S. Bankruptcy Code. The Parties agree that each Party, as licensee of certain rights under this Agreement, shall retain and may fully exercise all of its rights and elections under the U.S. Bankrupt Code. The Parties further agree that, in the event of the commencement of a bankruptcy proceeding by or against a Party (such Party, the "**Bankrupt Party**") under the U.S. Bankruptcy Code, the other Party shall be entitled to a complete duplicate of (or complete access to, as appropriate) any intellectual property licensed to such other Party and all embodiments of such intellectual property, which, if not already in such other Party's possession, shall be promptly delivered to it (a) upon any such commencement of a bankruptcy proceeding upon such other Party's written request therefor, unless the Bankrupt Party elects to continue to perform all of its obligations under this Agreement or (b) if not delivered under clause (a), following the rejection of this Agreement by the Bankrupt Party upon written request therefor by the other Party. The Parties acknowledge and agree that of the milestones and royalties to be paid pursuant to <u>Article 8</u>, only the sales milestones contained in <u>Section 8.3</u> (c) and the royalties contained in <u>Section 8.3</u> shall constitute royalties within the meaning of Bankrupt Code § 365(n) with respect to the licenses of intellectual property hereunder.

13.8 <u>Survival</u>. Termination or expiration of this Agreement shall not affect rights or obligations of the Parties under this Agreement that have accrued prior to the date of termination or expiration of this Agreement. Notwithstanding anything to the contrary, the following provisions shall survive and apply after expiration or termination of this Agreement in its entirety: <u>Article 1, Article 12, Article 14, Article 15, Sections 3.7, 4.8, 8.5, 8.7, 8.9, 8.11, 8.12, 8.13, 8.14, 8.15, 9.1, 10.5, 10.6, 13.5, 13.6, 13.7, and 13.8. In addition, the other applicable provisions of <u>Article 8</u> shall survive such expiration or termination of this Agreement in its entirety to the extent required to make final reimbursements, reconciliations or other payments incurred or accrued prior to the date of termination or expiration. For any surviving provisions requiring action or decision by a Committee or an Executive Officer, each Party will appoint representatives to act as its Committee members or Executive Officer, as applicable. All provisions not surviving in accordance with the foregoing shall terminate upon expiration or termination of this Agreement and be of no further force and effect. If this Agreement is terminated with respect to one or more Terminated Regions but not in its entirety, then following such termination the foregoing provisions of this Agreement shall remain in effect with respect to the Terminated Regions (to the extent they would survive and apply in the event the Agreement expires or is terminated in its entirety), and all provisions not surviving in accordance with the foregoing shall provisions not surviving in accordance with the foregoing shall remain of this Agreement expires or is terminated in its entirety), and all provisions not surviving in accordance with the foregoing shall terminate upon termination of this Agreement with respect to the applicable Terminated Region(s) and be of no further force and effect (and for the avoidance of doubt all provisions of this Agreement shall remain </u>

ARTICLE 14

DISPUTE RESOLUTION

14.1 Disputes. It is the objective of the Parties to establish procedures to facilitate the resolution of disputes arising under this Agreement in an expedient manner by mutual cooperation and without resort to litigation. In the event of any disputes, controversies or differences which may arise between the Parties out of or in relation to or in connection with this Agreement (including disputes relating to the matters that can be referred to the Executive Officers pursuant to Section 2.4(b) but excluding any other disputes arising from a Committee), including any alleged failure to perform, or breach, of this Agreement, or any issue relating to the interpretation or application of this Agreement, then upon the request of either Party by written notice, the Parties agree to meet and discuss in good faith a possible resolution thereof, which good faith efforts shall include at least one in-person meeting between the Executive Officers of each Party. If the matter is not resolved within [...***...] days following the written request for discussions, either Party may then invoke the provisions of Section 14.2 or Section 14.9, as appropriate, *provided, however, that* the provisions of Section 14.2 shall not be invoked and rather Galapagos shall have the right to invoke its decision-making authority if the dispute relates to the preparation, filing, prosecution or maintenance of the Galapagos Patents pursuant to Section 9.2(a). For the avoidance of doubt, any disputes, controversies or differences arising from a Committee pursuant to <u>Article 2</u> shall be resolved solely in accordance with <u>Section 2.4</u>.

14.2 <u>Arbitration</u>. Any dispute, controversy, difference or claim which may arise between the Parties and not from a Committee, out of or in relation to or in connection with the License and Collaboration Agreement or this Agreement (including arising out of or relating to the validity, construction, interpretation, enforceability, breach, performance, application or termination of this Agreement) that is not resolved pursuant to <u>Section 14.1</u>, except for a dispute, claim or controversy under <u>Section 14.9</u> or as otherwise noted in <u>Section 14.1</u>, shall be settled by binding arbitration in accordance

with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators, one each chosen by the respective Parties and the third chosen by mutual agreement of the first two, and otherwise in accordance with the ICC Rules. The arbitrators shall have significant experience and shall have expertise in licensing and partnering agreements in the pharmaceutical and biotechnology industries. Either Party, following the end of the [...***...] day period referenced in <u>Section 14.1</u>, may refer such issue to arbitration by submitting a written notice of such request to the other Party. The place of arbitration shall be New York and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither Party shall be required to give general discovery of documents, but may be required only to produce specific, identified documents which are relevant to the dispute. The Parties shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Parties, and shall be governed by the terms and conditions hereof, including the limitation on damages set forth in <u>Section 11.5</u>. The Parties agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of the State of New York applicable to the commencement of a lawsuit shall apply to the commencement of arbitration under this <u>Article 14</u>. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each Party. All proceedings and decisions of the arbitrators shall be deemed Confidential Information of each of the Parties, and shall be subject to <u>Article 12</u>.

14.3 <u>Governing Law</u>. This Agreement shall be governed by and construed under the substantive laws of the State of New York, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of this Agreement to the substantive law of another jurisdiction.

14.4 <u>Award</u>. Any award to be paid by one Party to the other Party as determined by the arbitrator as set forth above under <u>Section 14.2</u> shall be promptly paid in U.S. dollars free of any tax, deduction or offset; and any costs, fees or taxes incident to enforcing the award shall, to the maximum extent permitted by law, be charged against the Party resisting enforcement. Each Party agrees to abide by the award rendered in any arbitration conducted pursuant to this <u>Article 14</u>, and agrees that, subject to the U.S. Federal Arbitration Act, 9 U.S.C. §§ 1-16, judgment may be entered upon the final award in the U.S. Federal District Court for the Southern District of New York and that other courts may award full faith and credit to such judgment in order to enforce such award.

14.5 <u>Injunctive Relief; Remedy for Breach of Exclusivity</u>. Nothing in this <u>Article 14</u> will preclude either Party from seeking equitable relief or interim or provisional relief from a court of competent jurisdiction, including a temporary restraining order, preliminary injunction or other interim equitable relief, concerning a dispute either prior to or during any arbitration if necessary to protect the interests of such Party or to preserve the status quo pending the arbitration proceeding. Therefore, in addition to its rights and remedies otherwise available at law, including the recovery of damages for breach of this Agreement, upon an adequate showing of material breach, and without further proof of irreparable harm other than this acknowledgement, such non-breaching Party shall be entitled to seek (a) immediate equitable relief, specifically including, but not limited to, both interim and permanent restraining orders and injunctions, and (b) such other and further equitable relief as the court may deem proper under the circumstances. For the avoidance of doubt, nothing in this <u>Section 14.5</u> shall otherwise limit a breaching Party's opportunity to cure a material breach as permitted in accordance with <u>Section 13.4</u>.

14.6 <u>Confidentiality</u>. The arbitration proceeding shall be confidential and the arbitrator shall issue appropriate protective orders to safeguard each Party's Confidential Information. Except as required by law, no Party shall make (or instruct the arbitrator to make) any public announcement with respect to the proceedings or decision of the arbitrator without prior written consent of the other Party. The existence of any dispute submitted to arbitration, and the award, shall be kept in confidence by the Parties and the arbitrator, except as required in connection with the enforcement of such award or as otherwise required by Applicable Law.

14.7 <u>Survivability</u>. Any duty to arbitrate under this Agreement shall remain in effect and be enforceable after termination of this Agreement for any reason.

14.8 <u>Jurisdiction</u>. For the purposes of this <u>Article 14</u>, the Parties acknowledge their diversity (Gilead having its principal place of business in the State of California and Galapagos having its principal place of business in Belgium), and except as provided in <u>Section 14.9</u>, agree to accept the jurisdiction of any United States District Court located in New York for the purposes of enforcing or appealing any awards entered pursuant to this <u>Article 14</u> and for enforcing the agreements reflected in this <u>Article 14</u>.

14.9 <u>Patent and Trademark Disputes</u>. Notwithstanding <u>Section 14.2</u>, any dispute, controversy or claim relating to the scope, validity, enforceability or infringement of any Galapagos Patents, Gilead Patents, Gilead Combination Patents, Joint Patents or Marks covering the manufacture, use, importation, offer for sale or sale of Licensed Products or Gilead Combination Products shall be submitted to a court of competent jurisdiction in the country in which such patent or trademark rights were granted or arose.

ARTICLE 15

MISCELLANEOUS

15.1 [Intentionally Omitted]

15.2 Entire Agreement; Amendment. This Agreement, including the Exhibits hereto, and the Ancillary Agreements set forth the complete, final and exclusive agreement and all the covenants, promises, agreements, warranties, representations, conditions and understandings between the Parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings between the Parties existing as of the Effective Date with respect to the subject matter hereof, including the Materials Transfer Agreement and Existing Confidentiality Agreement. In the event of any inconsistency between any plan hereunder (including the Development Plan or Shared Territory Commercialization Plan and Budget) and this Agreement, the terms of this Agreement shall prevail. There are no covenants, promises, agreements, warranties, representations, conditions or understandings, either oral or written, between the Parties other than as are set forth herein and therein. No subsequent alteration, amendment, change or addition to this Agreement shall be binding upon the Parties unless reduced to writing and signed by an authorized officer of each Party. The Parties hereby agree and acknowledge that this Agreement amends and restates the License and Collaboration Agreement as amended by the First Amendment in its entirety, and the License and Collaboration Agreement as amended by the First Amendment.

15.3 Force Majeure. Both Parties shall be excused from the performance of their obligations under this Agreement to the extent that such performance is prevented or delayed by force majeure and the nonperforming Party promptly provides notice of the prevention to the other Party. Such excuse shall be continued so long as the condition constituting force majeure continues and the nonperforming Party takes reasonable efforts to remove the condition. For purposes of this Agreement, force majeure shall mean conditions beyond the control of the Parties, including an act of God, war, civil commotion, terrorist act, labor strike or lock-out, epidemic, failure or default of public utilities or common carriers, destruction of production facilities or materials by fire, earthquake, storm or like catastrophe, and failure of plant or machinery (*provided that* such failure could not have been prevented by the exercise of skill, diligence, and prudence that would be reasonably and ordinarily expected from a skilled and experienced person engaged in the same type of undertaking under the same or similar circumstances). Notwithstanding the foregoing, a Party shall not be excused from making payments owed hereunder because of a force majeure affecting such Party.

15.4 Notices. Any notice required or permitted to be given under this Agreement shall be in writing, shall specifically refer to this Agreement, and shall be addressed to the appropriate Party at the address specified below or such other address as may be specified by such Party in writing in accordance with this Section 15.4, and shall be deemed to have been given for all purposes (a) when received, if hand-delivered, sent by a reputable international expedited delivery service (with receipt confirmed) or facsimile (with transmission confirmed), or (b) [...***...] Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. This Section 15.4 is not intended to govern the day-to-day business communications necessary between the Parties in performing their obligations under the terms of this Agreement.

If to Galapagos:	Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: Chief Executive Officer Fax: [****]
With a copy to (which shall not constitute notice):	Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: Legal Department Fax: [****]
With a copy to (which shall not constitute notice):	Baker & McKenzie LLP 452 Fifth Avenue New York, New York 10018 Attention: Olivia Tyrrell Oren Livne Fax: +1 212-310-1818
If to Gilead:	Gilead Biopharmaceutics Ireland UC c/o Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 USA Attention: Alliance Management
With a copy to (which shall not constitute notice):	Gilead Biopharmaceutics Ireland UC c/o Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 USA Attention: General Counsel
	75

With a copy to (which shall not constitute notice):

Covington & Burling LLP Salesforce Tower 415 Mission Street, Suite 5400 San Francisco, CA 94105-2533 Attention: Amy L. Toro, Esq. Fax: [...****...]

15.5 <u>No Strict Construction; Headings</u>. This Agreement has been prepared jointly and shall not be strictly construed against either Party. Ambiguities, if any, in this Agreement shall not be construed against any Party, irrespective of which Party may be deemed to have authored the ambiguous provision. The headings of each Article and Section in this Agreement have been inserted for convenience of reference only and are not intended to limit or expand on the meaning of the language contained in the particular Article or Section.

15.6 Assignment; Industry Transaction; Acquired Programs.

(a) Neither Party may assign or transfer this Agreement or any rights or obligations hereunder without the prior written consent of the other, except that a Party may make such an assignment without the other Party's consent to an Affiliate or to a successor to substantially all of the business to which this Agreement relates, whether in a merger, sale of stock, sale of assets, reorganization or other transaction. Any permitted successor or assignee of rights or obligations hereunder shall, in a writing to the other Party, expressly assume performance of such rights or obligations (and in any event, any Party assigning this Agreement to an Affiliate shall remain bound by the terms and conditions hereof). Any permitted assignment shall be binding on the successors of the assigning Party. Any assignment or attempted assignment by either Party in violation of the terms of this <u>Section 15.6</u> shall be null, void and of no legal effect.

(b) Notwithstanding anything to the contrary herein, (i) no material, Information, Patent, Regulatory Materials or Regulatory Approvals Controlled by a Drug Company transaction counterparty of Galapagos or any of such counterparty's Affiliates prior to an Industry Transaction of Galapagos shall be deemed Controlled by Galapagos for purposes of this Agreement after such Industry Transaction. For the purposes of this Agreement, (A) "**Industry Transaction**" of a Party shall mean that (1) such Party shall be controlled directly or indirectly by, or merged into, a Drug Company where such Drug Company is the surviving entity, or (2) any sale, license or other transfer (in one transaction or a series of related transactions) of all or substantially all of such Party's assets or that portion of such Party's business pertaining to the subject matter of this Agreement shall have occurred to a Drug Company, and (B) "**Drug Company**" shall mean any entity that conducts research and development in the biotechnology or pharmaceutical industry or develops or commercializes therapeutic or diagnostic products.

(c) In addition, upon an Industry Transaction of Galapagos, Gilead shall have the right to (i) terminate Galapagos's Co-Commercialization rights for any and all Licensed Products and Gilead Combination Products following such Industry Transaction and the Operating Profit and Operating Loss sharing pursuant to <u>Section 8.7</u> in favor of royalties pursuant to <u>Section 8.3</u>, by providing written notice to Galapagos; and (ii) disband the JSC, JDC, JCC, Shared Territory JCC, Joint Team, any internal Gilead committee involvement of Galapagos and the Patent Committee, and terminate the activities thereof and thereafter undertake all activities assigned by this Agreement thereto solely and exclusively by itself, *provided, however*, in each case that Gilead shall have no right to exercise such rights under (i) and/or (ii) if the Industry Transaction of Galapagos is with respect to a Drug Company with a market capitalization that does not exceed [...***...] percent ([...***...]%) of Galapagos' market capitalization, as measured immediately before the public announcement of such Industry Transaction.

(d) Acquired Programs.

(i) Notwithstanding anything to the contrary in this Agreement, in the event of an acquisition of a Party or its business after the Effective Date by a Third Party (an "Acquirer") whether by merger, asset purchase or otherwise and such Acquirer controls any program(s) that but for this <u>Section 15.6(d)</u>, would violate <u>Section 7.6(a)</u> (each such program, a "Competing Program"), then the Acquirer and any Affiliate of the Acquirer that becomes an Affiliate of the acquired Party as a result of such acquisition (but excluding the acquired Party), shall not be subject to <u>Section 7.6(a)</u> as to any such Competing Programs in existence prior to the closing date of such acquisition, or for the subsequent development and commercialization of such Competing Programs (including new products from any such Competing Programs); <u>provided, however, that</u> no Information or Patents of the other Party are used by or on behalf of the Acquirer of the acquired Party (or any Affiliate of such Acquirer) in more than a *de minimis* fashion in connection with such subsequent development and commercialization of any Competing Programs.

(ii) Reserved.

15.7 <u>Performance by Affiliates</u>. Subject to the limitations of <u>Section 7.2</u>, each Party may discharge any obligations and exercise any right hereunder through any of its Affiliates. Each Party hereby guarantees the performance by its Affiliates of such Party's obligations under this Agreement, and shall cause its Affiliates to comply with the provisions of this Agreement in connection with such performance. Any breach by a Party's Affiliate of any of such Party's obligations under this Agreement shall be deemed a breach by such Party, and the other Party may proceed directly against such Party without any obligation to first proceed against such Party's Affiliate.

15.8 <u>Compliance with Applicable Law</u>. Each Party shall comply with Applicable Law in the course of performing its obligations or exercising its rights pursuant to this Agreement.

15.9 <u>Further Actions</u>. Each Party agrees to execute, acknowledge and deliver such further instruments, and to do all such other acts, as may be necessary or appropriate in order to carry out the purposes and intent of this Agreement.

15.10 <u>Severability</u>. If any one or more of the provisions of this Agreement is held to be invalid or unenforceable by an arbitrator or by any court of competent jurisdiction from which no appeal can be or is taken, the provision shall be considered severed from this Agreement and shall not serve to invalidate any remaining provisions hereof. The Parties shall make a good faith effort to replace any invalid or unenforceable provision with a valid and enforceable one such that the objectives contemplated by the Parties when entering into this Agreement may be realized.

15.11 <u>No Waiver</u>. Any delay in enforcing a Party's rights under this Agreement or any waiver as to a particular default or other matter shall not constitute a waiver of such Party's rights to the future enforcement of its rights under this Agreement, except with respect to an express written and signed waiver relating to a particular matter for a particular period of time.

15.12 <u>Independent Contractors</u>. Each Party shall act solely as an independent contractor, and nothing in this Agreement shall be construed to give either Party the power or authority to act for, bind, or commit the other Party in any way. Nothing herein shall be construed to create the relationship of partners, principal and agent, or joint-venture partners between the Parties. Additionally, the Parties expressly agree that this Agreement is not intended to be treated as a partnership for Taxation purposes and that the Parties will reflect such agreement in all relevant Tax filings.

15.13 <u>Counterparts</u>. This Agreement may be executed in one (1) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Signature pages of this Amendment may be exchanged by email or in pdf or other electronic means without affecting the validity thereof.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have executed this Agreement in duplicate originals by their duly authorized officers as of the Amendment Effective Date.

GALAPAGOS NV

By:

Name: Title: GILEAD BIOPHARMACEUTICS IRELAND UC

Name:	
Title:	

By:

By:

Name: Title:

Signature Page to Amended and Restated License and Collaboration Agreement

EXHIBIT A

Subscription Agreement

[...***...(14 pages omitted)]

EXHIBIT B

Existing Galapagos Patents

[...***...(13 pages omitted)]

Exhibit B - Page 1

EXHIBIT C

Initial JSC Representatives

For Galapagos: [...***...] For Gilead: [...***...]

Exhibit C - Page 1

EXHIBIT D

Initial Development Plan

[...***...(six pages omitted)]

Exhibit D - Page 1

EXHIBIT E

[Intentionally Omitted]

EXHIBIT F

Access Territory

[...***...(2 pages omitted)]

Exhibit F - Page 1

EXHIBIT G

[Intentionally Omitted]

EXHIBIT H

Pre-Consented CROs

[...***...]

Exhibit H - Page 1

EXHIBIT I

[Intentionally Omitted]

EXHIBIT J

[...***...] of Gilead Combination Products

[...***...]

Exhibit J - Page 1

EXHIBIT K

Specific Disclosures

[...***...]

Exhibit K - Page 1

EXHIBIT L

Assignment

[...***...]

Exhibit L - Page 1

EXHIBIT M

Clinical Studies Currently in 2019 JSC Approved Budget (Active)

[...***...]

Exhibit M - Page 1

July 14, 2019

SUBSCRIPTION AGREEMENT relating to ordinary shares in Galapagos NV

between

Galapagos NV

as Issuer

and

Gilead Therapeutics A1 Unlimited Company

as Investor

Subscription Agreement

Between: (1) Galapagos NV, a limited liability company ("naamloze vennootschap" / "société anonyme") organised and existing under Belgian law, with registered office at Generaal De Wittelaan L11 A3, 2800 Mechelen, Belgium, registered with the register of legal entities (Antwerp) under number 0466.460.429;

represented for the purposes of this Agreement by Onno van de Stolpe (Director and Chief Executive Officer) and Bart Filius (Chief Financial Officer and Chief Operating Officer);

hereinafter referred to as the "Issuer";

And: (2) Gilead Therapeutics A1 Unlimited Company, an unlimited liability company formed under the laws of Ireland , with registered office at 70 Sir John Rogerson's Quay, Dublin 2, Ireland;

represented for the purposes of this Agreement by David Cadogan (Director) and Padraig Clancy (Director);

hereinafter referred to as the "Investor";

The parties sub (1) and (2) above are hereinafter referred to as the "Parties" and each individually as a "Party".

Whereas:

- (A) The Issuer, a Belgian limited liability company listed on the regulated markets of Euronext Brussels and Amsterdam and the NASDAQ Stock Market, is a clinical-stage biotechnology company specialised in the discovery and development of small molecule medicines with novel modes of action.
- (B) The Investor is an indirect wholly-owned subsidiary of Gilead Sciences, Inc., a U.S. corporation listed on the NASDAQ Stock Market and a research-based biopharmaceutical company focused on the discovery, development, and commercialisation of innovative medicines ("**Parent Investor**").
- (C) The Issuer and Parent Investor are, simultaneously with the execution and delivery of this Agreement, entering into an option, license and collaboration agreement (the "Option, License and Collaboration Agreement") pursuant to which the Issuer will discover, research, and develop molecules and products, and Parent Investor will have an option to participate in the development and commercialisation of molecules and products, in each case, on the terms and conditions set forth in such agreement.
- (D) The Issuer and Gilead Biopharmaceutics Ireland UC, an Affiliate of the Investor, are, simultaneously with the execution and delivery of this Agreement, entering into an amendment to the License and Collaboration Agreement between the Issuer and Gilead Biopharmaceutics Ireland UC, dated as of December 16, 2015, on the terms and conditions as determined in such amendment (the "License and Collaboration Agreement Amendment").
- (E) The Investor wishes to invest an aggregate amount of cash equal to the Subscription Amount (as defined below) in the share capital of the Issuer through a private placement of Ordinary Shares (as defined below). Thereto the Issuer intends to increase its share capital and issue premium with an aggregate amount equal to the Subscription Amount, by means of a capital increase approved by the Issuer's board of directors within the framework of the authorised capital.

- (F) The Investor has agreed to subscribe to such capital increase to the extent and on the terms and subject to the conditions set forth in this Agreement.
- (G) Promptly, and in any event within sixty (60) days following the Closing, the Board of Directors shall convene the Issuer EGM (as defined below) at which the shareholders of the Issuer will be asked to approve, among other things, the issuance to the Investor of the Initial Warrants.

It is agreed as follows:

1 Definitions and Interpretation

1.1 Definitions

The following terms and expressions that are not defined elsewhere in this Agreement have the following meaning in this Agreement:

Acting in Concert means, when used in relation to a person or entity, acting in concert (in onderling overleg handelende personen / personnes agissant de concert) in the sense of Article 3, §1, 5° of the Belgian Act of 1 April 2007 regarding public takeover bids, or Article 1, §2, 5° of the Belgian Royal Decree of 27 April 2007 regarding public takeover bids.

Affiliate means, with respect to a particular person or entity, any person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such person or entity, for so long as such control exists, regardless of whether such person or entity is or becomes an Affiliate on or after the Date of this Agreement. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, or by contract or otherwise.

Antitrust Approvals has the meaning given to it in the Option, License and Collaboration Agreement.

Agreement means this subscription agreement and its exhibits.

Belgian Companies and Associations Code means the Belgian Companies and Associations Code of 23 March 2019, as amended from time to time, and the rules and regulations promulgated thereunder.

Belgian Companies Code means the Belgian Companies Code of 7 May 1999, as amended from time to time, and the rules and regulations promulgated thereunder.

Blocked Account has the meaning given to it in Article 2.6.2(i).

Board Designee Proposal has the meaning given to it in 7.2.1.

Board of Directors means the Issuer's board of directors (raad van bestuur / conseil d'administration), <u>provided</u>, however, that if the Issuer adopts the two-tier board system (duaal bestuur / administration duale) pursuant to the Belgian Companies and Associations Code, references to the "Board of Directors" shall mean a reference to the Issuer's supervisory board (raad van toezicht / conseil de surveillance), and references to "director" (bestuurder/administrateur) and member of the "Board of Directors" shall mean a reference to a member of the Issuer's supervisory board.

Business Day means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States, (c) a bank or other public holiday in Brussels, Belgium, (d) a bank or other public holiday in Ireland or (e) the period commencing on December 25th and ending on January 1st (inclusive).

Page 3 of 75

Capital Increase has the meaning given to it in Article 2.1.

Closing means the date on which the Capital Increase is effected, which date is set forth in Article 2.6.1 below.

Code means the United States Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder.

Date of this Agreement means the date on which this Agreement is entered into.

Equity Security means (i) any share representing the share capital of the Issuer, and (ii) any other security, financial instrument, certificate and other right (including options, futures, swaps and other derivatives) issued or, with respect to options, futures, swaps and other derivatives, contracted by the Issuer and representing, being exercisable, convertible or exchangeable into or for, or otherwise providing a right to acquire, directly or indirectly, any of the Equity Securities referred to in (i).

EU Market Abuse Regulation means Regulation (EU) No 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse (market abuse regulation) and repealing Directive 2003/6/EC of the European Parliament and of the Council and Commission Directives 2003/124/EC, 2003/125/EC and 2004/72/EC.

FSMA has the meaning given to it in Article 6.1.5.

Fully Diluted Basis means, as of any date of determination, all issued and outstanding shares of capital stock of the Issuer (assuming that each share continues to have one vote) and all shares of capital stock that would be issued assuming the exercise of all options, warrants and other rights to acquire new shares of capital stock of the Issuer, and assuming the exercise of all securities convertible or exchangeable into new shares of capital stock of the Issuer, in each case whether or not then exercisable.

Initial Lock-up Period has the meaning given to it in Article 6.2.1.

Initial Warrant Proposal has the meaning given to it in 7.2.1.

Initial Warrant means the warrants named "Initial Gilead Warrant A" and "Initial Gilead Warrant B", substantially in the form attached hereto as Exhibit A, to subscribe for Ordinary Shares.

Investor Board Designees means the individuals to be identified by the Investor from time to time in accordance with the provisions of this Agreement, initially one of whom shall be Parent Investor's Chief Executive Officer and the other of whom shall be Parent Investor's Chief Scientific Officer (provided that if either such person is unable to so serve, his or her replacement shall be designated in writing by the Investor), to serve on the Board of Directors.

Issuer EGM has the meaning given to it in Article 7.2.1.

Issuer EGM Proposals has the meaning given to it in Article 7.2.1.

License and Collaboration Agreement means the license and collaboration agreement between the Issuer and Gilead Biopharmaceutics Ireland UC, dated as of December 16, 2015.

License and Collaboration Agreement Amendment has the meaning given to it in Recital (D).

Lift of the Standstill has the meaning given to it in Article 6.1.3.

Lift of Standstill Notification has the meaning given to it in Article 6.1.3.

Page 4 of 75

Lock-up Period has the meaning given to it in Article 6.2.1.

Option, License and Collaboration Agreement has the meaning given to it in Recital (C) of this Agreement.

Ordinary Shares means new shares of the only existing class in the capital of the Issuer.

Parent Investor has the meaning given to it in Recital (B) of this Agreement.

Standstill has the meaning given to it in Article 6.1.2 of this Agreement.

Standstill Limit has the meaning given to it in Article 6.1.2(i) of this Agreement.

Standstill Period has the meaning given to it in Article 6.1.1 of this Agreement.

Subscription Amount means the amount, denominated in Euros, equal to the product of the price per Subscription Share stated in Article 2.4.2 <u>multiplied by</u> the number of Subscription Shares to be issued by the Issuer to the Investor in accordance with Article 2.4.3.

Subscription Shares has the meaning given to it in Article 2.2 of this Agreement.

Subsequent Issuer EGM has the meaning given to it in Article 7.2.2.

Subsequent Issuer EGM Proposals has the meaning given to it in Article 7.2.2.

Subsequent Lock-up Period has the meaning given to it in Article 6.2.1.

Subsequent Lock-up Period Threshold has the meaning given to it in Article 6.2.1.

Subsequent Warrant means the warrant, named "Subsequent Gilead Warrant B", substantially in the form attached hereto as Exhibit B, to subscribe for Ordinary Shares.

Subsidiary means, when used in relation to an entity (for the purpose of this definition, the "reference entity"), an entity in which the reference entity directly or indirectly owns, beneficially or of record, (a) an amount of voting securities or other interests in such entity that is sufficient to enable the reference entity to elect at least a majority of the members of such entity's board of directors or other governing body, or (b) at least 50% of the outstanding equity or financial interests of such entity.

Warrants means the Initial Warrants and the Subsequent Warrant.

1.2 Interpretation

- **1.2.1** The titles and headings included in this Agreement are for convenience only and shall not be taken into account in the interpretation of the provisions of this Agreement.
- **1.2.2** The words "herein", "hereof", "hereunder", "hereby", "hereto", "herewith" and words of similar import shall refer to this Agreement as a whole and not to any particular Article, paragraph or other subdivision.
- **1.2.3** All periods of time set out in this Agreement shall be calculated from midnight to midnight local time in Brussels, Belgium. They shall start on the day following the day on which the event triggering the relevant period of time has occurred. The expiration date shall be included in the period of time. If the expiration date is not a Business Day, it shall be postponed until the next Business Day. Unless otherwise provided herein, all periods of time shall be calculated in calendar days. All periods of time consisting of a number of months (or years) shall be calculated from the day in the month (or year) when the triggering event has occurred until the eve of the same day in the following month(s) (or year(s)) ("van de zoveelste tot de dag vóór de zoveelste" / "de quantième à veille de quantième").

Page 5 of 75

- **1.2.4** References to any statute, regulation or statutory provision shall be deemed to include reference to any statute, regulation or statutory instrument which amends, extends, consolidates or replaces the same (or shall have done so) and to any other regulation, statutory instrument or other subordinate legislation made thereunder or pursuant thereto, provided that no such reference shall include any amendment, extension or replacement of the same with retrospective effect.
- **1.2.5** The original version of this Agreement has been made in English. Should this Agreement be translated in whole or in part into another language (if at all), the original English version shall prevail between the Parties hereto to the fullest extent possible and permitted by Belgian law. Notwithstanding the foregoing, Belgian legal concepts which are expressed in English language terms, are to be interpreted in accordance with the Belgian legal terms to which they refer, and the use herein of Dutch words in this Agreement as translation for certain words or concepts shall be conclusive in the determination of the relevant legal concept under Belgian law of the words or concepts that are so translated herein.

2 Capital Increase and Share Subscription

- **2.1** The Issuer agrees to proceed with a capital increase decided upon by a unanimous vote of the Board of Directors within the framework of the authorised capital, with the cancellation of the preferential subscription rights of the shareholders to the benefit of the Investor in an aggregate amount equal to the Subscription Amount (share capital and issue premium) (the "**Capital Increase**"), on the date of the Closing.
- **2.2** The Investor agrees to subscribe to the Capital Increase for the full amount of the Subscription Amount, in return for which the Investor shall receive the aggregate number of Ordinary Shares to be determined in accordance with Article 2.4.3 (the "**Subscription Shares**").
- **2.3** The Investor has been advised that it is receiving "Restricted Securities" (as defined in Issuer's amended and restated deposit agreement related to American Depositary Shares dated on or about 23 April 2015 (as amended and restated from time to time)) and agrees not to deposit the Ordinary Shares into such deposit agreement unless and until permitted to deposit such Ordinary Shares in accordance with the terms of such deposit agreement.

2.4 Capital Increase and issuance of Subscription Shares

- **2.4.1** The Capital Increase shall be decided upon by a unanimous vote of the Board of Directors within the framework of the authorised capital, with the cancellation of the preferential subscription rights of the shareholders to the benefit of the Investor, it being understood that the Board of Directors will have approved the special report required under Articles 596 and 598 of the Belgian Companies Code for the issuance of shares with cancellation of the preferential subscription right of the shareholders to the benefit of one or more persons who are not employees of the company or one of its subsidiaries.
- 2.4.2 Price per Subscription Share

The price per Subscription Share shall be equal to €140.59.

Page 6 of 75

2.4.3 Number of Subscription Shares

The number of Subscription Shares to be issued by the Issuer to the Investor following the Capital Increase shall be equal to the number of Ordinary Shares sufficient to bring Investor's aggregate ownership percentage in the Issuer (together with the shares held by the Investor, the Parent Investor or any of their Affiliates) on Closing to 20.1% on a Fully Diluted Basis (rounded up to the nearest whole share) (provided that if the Investor, the Parent Investor, any of the Affiliates of the Investor or Parent Investor, or any party Acting in Concert with the Investor, the Parent Investor, or any of the Affiliates of the Investor or Parent Investor, acquire any existing shares of the Issuer between the Date of this Agreement and Closing, to the extent such acquisition is permitted by (i) applicable law, (ii) the standstill provisions of Article 6.1 and (iii) the other provisions of this Agreement, such shares shall not be taken into account to calculate the number of Subscription Shares to be issued by the Issuer to the Investor at the Closing and for which the Investor shall subscribe, and the acquisition of such shares shall not reduce the number of Subscription Shares to be issued as if no such shares were acquired).

For illustration purposes, based on the total number of shares of the Issuer on a Fully Diluted Basis on the Date of this Agreement and the total number of shares of the Issuer held by the Investor on the Date of this Agreement, the number of Subscription Shares issuable to the Investor pursuant to the provisions of this Agreement as of the Date of this Agreement would amount to 6,828,985.

2.5 Rights attaching to the Subscription Shares

- **2.5.1** Without prejudice to Article 2.5.2, the Subscription Shares shall be identical in all respects (including the right to share in the dividends over the accounting year 2019, in any profits (including profits carried forward and reserves) and in any dividends declared as from their issue) to the existing shares of the Issuer.
- **2.5.2** The Issuer shall obtain a listing of the Subscription Shares on the regulated markets of Euronext Brussels and Amsterdam in accordance with Article 7.

2.6 Closing

2.6.1 The Closing shall occur on the latest of (x) the fifth Business Day following the receipt of all necessary Antitrust Approvals (as defined in the Option, License and Collaboration Agreement), (y) the Effective Date (as defined in the Option, License and Collaboration Agreement) and (z) the effective date of the License and Collaboration Agreement Amendment. In the event that the Option, License and Collaboration Agreement terminates for any reason prior to the Closing, then this Agreement shall simultaneously and automatically terminate. The Investor's obligation to effect the Closing shall be subject to the condition precedent that the Issuer shall have delivered to the Investor at the Closing a certificate, signed by a duly authorised officer of the Issuer, stating on behalf of the Issuer that (i) the representations and warranties of the Issuer in Article 4 hereof are true and correct as of the Closing and (ii) that the Issuer has complied in all material respects with its covenants hereunder.

2.6.2 On Closing:

(i) the full Subscription Amount shall be available on the Euro-denominated blocked account in the Issuer's name with KBC Bank NV ([***]) (the "**Blocked Account**"). To this

Page 7 of 75

end, the Investor or its designee shall instruct or cause the instruction of the wire of the Subscription Amount in Euros to the Blocked Account two (2) Business Days prior to Closing, it being understood that any bank charges, costs and expenses relating to this payment shall be borne by the Investor or its Affiliates;

- (ii) the Capital Increase shall be effected; and
- (iii) the Subscription Shares shall be issued in dematerialised form.

3 Representations and Warranties by the Investor

The Investor represents and warrants to the Issuer on the Date of this Agreement and at the Closing (except for those representations and warranties made as of a particular date, in which case the Investor represents and warrants to the Issuer as follows as of such date) that:

- **3.1** Validity of the Agreement. This Agreement has been duly authorised and executed by the Investor and constitutes a valid and legally binding obligation of the Investor.
- **3.2 Consents**. Subject to the receipt of all necessary Antitrust Approvals, all necessary consents, authorisations, notifications, actions or other things required to be taken, fulfilled or done by the Investor in accordance with applicable law (including without limitation the obtaining of any consent or license or the making of any filing or registration) for the subscription of the Subscription Shares pursuant to this Agreement, the carrying out of the other transactions contemplated by this Agreement as of the Closing or the compliance by the Investor with the terms of this Agreement as of the Closing have been obtained or made and are, or will on Closing be, in full force and effect.
- **3.3 Shareholding.** The Investor is a Subsidiary of the Parent Investor. As of the Date of this Agreement, the Investor owns 6,760,701 of the outstanding shares of the Issuer, and no other Equity Security. As of the Date of this Agreement, none of the Investor, the Parent Investor or any of their Subsidiaries owns any shares of the Issuer or other Equity Securities, other than the shares described in the prior sentence. Without taking into account the Warrants or the shares issuable (but not yet issued) thereunder, at any time until and including the Closing, the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, and any party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor, will not, directly or indirectly, own, or have the right to acquire under Equity Securities, more than 29.9% of the then issued and outstanding voting securities of the Issuer , assuming (x) the exercise, conversion or exchange of any Equity Securities held by any of them at that time that are exercisable, convertible or exchangeable for shares of the Issuer at such time, and (y) the issuance of the Subscription Shares to the Investor (the resulting number of securities rounded down).
- **3.4** Irish tax status of Investor: Investor is, for Irish tax purposes, resident in Ireland and subject to the Irish ordinary corporate income tax regime on its worldwide income (subject to an allocation of taxing rights to other jurisdictions based on Double Tax Treaties). Hence, Investor is, for Irish tax purposes, not considered to have only a branch in Ireland. Investor is not considered as tax resident in any other jurisdiction.

Page 8 of 75

4 Representations and Warranties of the Issuer

The Issuer represents and warrants to the Investor on the Date of this Agreement and at the Closing (except for those representations and warranties made as of a particular date, in which case the Issuer represents and warrants to the Investor as follows as of such date) that:

- **4.1 Incorporation and Authority**. It is duly incorporated and validly existing and in good standing under the laws of Belgium, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents. The Issuer has the requisite corporate power and authority to enter into and to consummate the actions contemplated by this Agreement, and otherwise to carry out its obligations hereunder. The Issuer's execution of this Agreement has been duly authorised by all necessary corporate action on the part of the Issuer, and no further corporate action, and no additional approval or ratification of its Board of Directors or general meeting of shareholders (other than (a) the unanimous approval by the Board of Directors before the notary public to make use of the authorised capital in respect of the Capital Increase, which the Issuer has committed to obtain pursuant to Article 2.1, (b) the preparation of the relevant special reports in connection herewith as required by applicable Belgian company law rules, and (c) the actions, reports and approvals in relation to the Issuer EGM and Subsequent Issuer EGM. Proposals and Subsequent Issuer EGM Proposals, all as contemplated by Article 7.2) is required by the Issuer. This Agreement has been (or upon delivery will have been) duly executed by the Issuer and is, or when delivered in accordance with the terms hereof, will constitute the legal, valid and binding obligation of the Issuer enforceable against the Issuer in accordance with its terms.
- 4.2 Validity of Shares and Warrants and Absence of Breach. The Subscription Shares and the shares of the Issuer issuable pursuant to the Warrants, when issued and paid for in accordance with this Agreement, will be validly and duly issued and fully paid shares of the only class of shares of the Issuer in accordance with the applicable provisions of the Issuer's organisational documents and Belgian law. The Warrants, when issued in accordance with this Agreement, will be validly and duly issued warrants of the Issuer in accordance with the applicable provisions of the Issuer's organisational documents and Belgian law, provided that the issuance of the Warrants to the Investor will require the approval of the Issuer's shareholders at a shareholders meeting. The Subscription Shares, the Warrants and the shares of the Issuer issuable pursuant to the Warrants will be free and clear of all liens, pledges, encumbrances, mortgages, security interests, or easement or transfer restrictions of any nature whatsoever (other than those that find their origin solely with the Investor and save for the transfer restrictions referred to in this Agreement), provided that the issuance of the Warrants to the Investor will require the approval of the Issuer's shareholders meeting. None of (i) the execution and delivery of this Agreement, or (ii) the issuance of the Subscription Shares, the Warrants or the shares of the Issuer's organisational documents or any law, regulation or stock exchange rule, or give rise to the activation of any material rights of third parties under any agreement, law, rule or regulation binding on the Issuer or any of its subsidiaries, provided that (a) the necessary Antitrust Approvals have been obtained, and (b) the issuance of the Warrants to the Investor will require the approval of the Issuer's shareholders meeting.
- **4.3 Consents**. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under Belgian law or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the carrying out by the Issuer of the actions contemplated by this Agreement or the compliance by the Issuer with the terms of this Agreement will, save as otherwise set forth in this Agreement in Article 7, be in full force and effect.

Page 9 of 75

- **4.4 No Material Change in Issuer's Business.** On the Date of this Agreement, there has been no material change in the Issuer's business since its 2018 Annual Report which would be reasonably likely to have a material adverse effect with regard to the Issuer's consolidated earnings and equity, save as previously publicly disclosed.
- **4.5 Brokers and Finders.** No person will have, as a result of the transactions contemplated by this Agreement, any right, interest or claim against or upon the Investor for any commission, fee or other compensation relating to the Subscription Shares or the Warrants.

5 Antitrust Approvals.

The actions contemplated to take place on the Closing pursuant to this Agreement are subject to the receipt of the necessary Antitrust Approvals as contemplated by the Option, License and Collaboration Agreement. If the necessary Antitrust Approvals have not been obtained by the Outside Date (as defined by the Option, License and Collaboration Agreement), this Agreement shall be automatically terminated without further action required by either Party.

6 Post-closing commitments and rights of the Investor

6.1 Standstill

- **6.1.1** The standstill obligation, as set out in this Article 6.1, will take effect as of the Date of this Agreement and will terminate on the earlier of (x) the date that is ten (10) years following the date on which the Closing occurs and (y) the termination of this Agreement if the Closing does not occur (the "**Standstill Period**").
- **6.1.2** During the Standstill Period, the Investor, the Parent Investor or any of their Affiliates, shall not:
 - (i) without the express written consent of the Issuer, directly or indirectly acquire any additional Equity Securities (other than the Warrants) of the Issuer, if after giving effect to such acquisition the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor would (without taking into account the Warrants or the shares issuable (but not yet issued) thereunder owned by them at that time) together in the aggregate directly or indirectly own or have the right to acquire more than 29.9% of the then issued and outstanding voting securities of the Issuer (assuming the exercise, conversion or exchange of any Equity Securities held by any of them at any time (other than the Warrants) that are exercisable, convertible or exchangeable into or for shares of the Issuer at such time) (the resulting number of securities rounded down) (the "Standstill Limit");
 - (ii) directly or indirectly encourage or support a tender, exchange or other offer or proposal by a third party;
 - (iii) propose (a) any merger, consolidation, business combination, tender or exchange offer, purchase of the Issuer's assets or businesses, or similar transaction involving the Issuer or (b) any recapitalization, restructuring, liquidation or other extraordinary transaction with respect to the Issuer (it being understood that the Investor's Chief Executive

Page 10 of 75

Officer may contact the Issuer's Chief Executive Officer on a non-public and non-committal basis to gauge the Issuer's Chief Executive Officer's views on the Issuer's potential interest in any such matter described in clause (a) or (b)); or

- (iv) directly or indirectly (a) submit matters to, request that matters be submitted to, or request the convening of, a general meeting of the shareholders of the Issuer, or (b) solicit proxies or consents, or become a participant in a solicitation in relation to matters submitted to a general meeting of the shareholders of the Issuer, in each case of (a) and (b) without or against the recommendation or support by the Board of Directors except that Investor may solicit proxies or consents and may become a participant in a solicitation in connection with any proposal that would adversely affect its rights under this Agreement, the Warrants, or the Option, License and Collaboration Agreement or as a shareholder of the Issuer, and may also make a proposal pursuant to Article 7.3.1, provided that the provisions of this Article 6.1.2(iv) shall automatically cease to apply when the Investor ceases to have the right to appoint Investor Board Designees pursuant to Article 7.3;
- (v) (a) make public statements with respect to (save if legally obliged to) or, (b) with the actual knowledge of the Parent Investor's executive officers, provide assistance to, commit to, or discuss or enter into any agreement or arrangement with any party to do, any of the foregoing prohibited actions provided that in relation to prohibited actions in subsection (ii) that have been committed without the actual knowledge of the Parent Investor's executive officers, the Investor and Parent Investor shall promptly terminate and unwind such actions upon written request of the Issuer.

(Article 6.1.2 (i) through (v) together, the "Standstill").

6.1.3 In the event that (i) the Issuer has received a non-public binding or non-binding offer, prior to the end of the Standstill Period, by a bona fide third party, other than the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor, regarding a bona fide potential takeover bid on the Issuer (including any tender, exchange or other offer or proposal to acquire a majority of the outstanding equity securities of the Issuer or all or a substantial part of its consolidated assets (except where the disposal of such substantial part of its assets would not adversely affect the Issuer's ability to comply with its obligations under the Option, Licence and Collaboration Agreement in a material respect), and such offer is publicly supported or recommended by the Board of Directors, (ii) the Issuer determines to commence, prior to the end of the Standstill Period, a process to seek a potential sale of the Issuer or all or a substantial part of its consolidated assets (except where the disposal of such substantial part of its assets would not adversely affect the Issuer's ability to comply with its obligations under the Option, Licence and Collaboration Agreement in a material respect), (iii) a bid other than from the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any Affiliates of the Investor of the Parent Investor, is announced to take over the Issuer pursuant to article 7 or 8, §1, §2 and §3 of the Belgian Royal Decree of 27 April 2007 on public takeover bids, or (iv) any bona fide person or entity (other than the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor) publicly discloses any plans, determined as serious by the Board of Directors to

Page 11 of 75

further pursue it, to make such a bid, then the Standstill obligations shall automatically cease to apply effective upon the occurrence of such approach, process, announcement or public disclosure (the "**Lift of the Standstill**") and the Issuer shall notify the Investor of such offer, process or announcement as promptly as practicable and in any event no later than one (1) Business Day after such offer, process or announcement (the "**Lift of Standstill Notification**"). Upon the Lift of the Standstill, the Standstill obligation of the Investor shall be lifted in order to level the playing field and grant it equal opportunities to prepare a takeover bid or take other permitted actions, such that the Investor shall not in any respect be disadvantaged or limited relative to any other bidder for the Issuer. It being understood and agreed by the Investor that the Issuer shall not be required to specify in the Lift of Standstill Notification the identity of any involved third party or any other specifics of the terms of such proposed transaction.

- **6.1.4** The Investor explicitly acknowledges that the information contained in the Lift of Standstill Notification may, prior to the public announcement of such information, constitute material, non-public information of the Issuer and inside information within the meaning of the EU Market Abuse Regulation. When receiving the Lift of Standstill Notification, the Investor shall take all appropriate measures to ensure the confidentiality of such information.
- **6.1.5** It is expressly acknowledged by the Parties that the Issuer may not be in a position to notify the Investor of a takeover bid beforehand, such as for example in the event of a non-solicited takeover bid immediately filed with the Financial Services and Markets Authority (the "FSMA") without prior approach.
- **6.1.6** For the avoidance of doubt, the Investor and its Affiliates may acquire, including on the open market or in privately negotiated purchases, any additional Equity Securities of the Issuer, but only up to the Standstill Limit.

6.2 Lock-up

6.2.1 During the period running from the Date of this Agreement through the earlier of (x) the date that is two (2) years following the date on which the Closing occurs and (y) the termination of this Agreement if the Closing does not occur (the "Initial Lock-up Period"), the Investor and Parent Investor shall not, and shall cause their Affiliates not to, without the prior consent of the Issuer, transfer, sell or otherwise dispose of any Equity Securities held by the Investor, the Parent Investor and their Affiliates, as applicable, (other than transfers, sales or dispositions permitted pursuant to Article 6.2.4) and during the period running from the date that is two (2) years following the date on which the Closing occurs through the date that is five (5) years following the date on which the Closing occurs (the "Subsequent Lock-up Period" and, together with the Initial Lock-up Period, as applicable, the "Lock-up Period"), the Investor and Parent Investor shall not, and shall cause their Affiliates not to, without the prior consent of the Issuer, transfer, sell or otherwise dispose of any Equity Securities held by the Investor, and their Affiliates, as applicable, the "Lock-up Period"), the Investor and Parent Investor shall not, and shall cause their Affiliates not to, without the prior consent of the Issuer, transfer, sell or otherwise dispose of any Equity Securities held by the Investor, the Parent Investor and their Affiliates, as applicable, to the extent that after such transfer, sale or disposal the Investor, the Parent Investor and any of their Affiliates would own less than 20.1% of the then issued and outstanding voting securities of the Issuer (on a non-diluted basis) (other than transfers, sales or dispositions permitted pursuant to Article 6.2.4) (the "Subsequent Lock-up Period Threshold"); provided, however, that the Lock-up Period shall automatically terminate in the event that (i) the Issuer has received a non-public offer, prior to the end of the Lock-up Period, by a bona fide third party

Page 12 of 75

fide potential takeover bid on the Issuer (including any tender, exchange or other offer or proposal to acquire a majority of the outstanding shares of the Issuer or all or a substantial part of its consolidated assets (except where the disposal of such substantial part of its assets would not adversely affect the Issuer's ability to comply with its obligations under the Option, License and Collaboration Agreement in any material respect), and such offer is publicly supported or recommended by the Board of Directors, (ii) the Issuer determines to commence, prior to the end of the Lock-Up Period, a process to seek a potential sale of the Issuer or all or a substantial part of its consolidated assets (except where the disposal of such substantial part of its assets would not adversely affect the Issuer's ability to comply with its obligations under the Option, Licence and Collaboration Agreement in a material respect), (iii) a bid other than from the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor) publicly disc

- **6.2.2** Upon expiry of the Initial Lock-up Period (up to the Subsequent Lock-up Period Threshold) or the Subsequent Lock-up Period, as applicable, the Investor, the Parent Investor and any of their Affiliates may, after notifying the Issuer of their intent to do so, transfer, sell or otherwise dispose of the shares of the Issuer, taking into account that:
 - (i) when instructing a bank to sell the shares of the Issuer (other than in a transaction described in subsections (ii) through (iv) below) the bank shall be instructed to use reasonable efforts to effect such sale of the shares of the Issuer in a manner reasonably intended to minimize disturbances of the share price of the Issuer, as quoted on Euronext Brussels and Amsterdam or on the NASDAQ Stock Market;
 - (ii) <u>when selling the shares of the Issuer on the open market</u> the Investor shall be permitted to sell shares of the Issuer on the open market at a daily volume not to exceed 20% of the average daily volume of shares of the Issuer as traded on the relevant market on which the sale is effected for the previous thirty (30) trading days, as quoted on Euronext Brussels and Amsterdam;
 - (iii) when selling the shares of the Issuer through a privately negotiated transaction the transaction shall not be subject to the limitations in this Article 6.2.2 if the transaction would not be reported in Euronext's consolidated tape on the date on which the transaction is agreed by the parties; or
 - (iv) <u>when selling the shares of the Issuer in a "block trade"</u> the block trade shall not be subject to the limitations in this Article 6.2.2 if it is the only sale by the Investor on the date of the block trade.
- **6.2.3** After the expiry of the Initial Lock-up Period, the Issuer shall, to the extent legally permitted, provide notice to the Investor as promptly as reasonably practicable in the event that the Issuer becomes aware that any person or entity is interested in purchasing or selling shares of the Issuer in a "block trade".

Page 13 of 75

- **6.2.4** The following transfers, sales or divestment of Equity Securities shall be permitted and not be subject to the restrictions set out in Article 6.2.1:
 - (i) any transfer, sale or other divestment of Equity Securities by the Investor to any of its Affiliates, provided that (a) the obligations of the Investor, the Parent Investor or any of their Affiliates pursuant to this Agreement and the Option, License and Collaboration Agreement remain unaffected by the proposed transfer, sale or divestment, (b) the transferee agrees in writing to the Issuer to be bound by the restrictions set out in Article 6.2.1 and 6.2.2 in relation to the Equity Securities it received and the other obligations of the Investor in relation to the Equity Securities under this Agreement, and (c) the relevant Equity Securities will be re-transferred to the Investor or the Parent Investor immediately prior to the transferee ceasing to be an Affiliate of the Investor or the Parent Investor; and
 - (ii) any transfer pursuant to a stock lending that is agreed to by the parties to facilitate the cashless exercise of stock based incentive plans that have been put in place for employees, officers, directors or consultants of the Issuer or its Subsidiaries, provided that upon expiry of the stock lending and the re-transfer of the relevant shares of the Issuer to the Investor, such shares are again subject to the restrictions set out in Article 6.2.1 and 6.2.2 for the remainder of the term.

6.3 Anti-Dilution

If at any time between the Closing and the date that is ten (10) years following the date on which the Closing occurs the Issuer offers new shares of the Issuer or any new securities convertible into or exercisable or exchangeable for new shares of the Issuer to any person or entity other than Investor or its Affiliates (other than to current or future employees, consultants, directors and/or officers of the Issuer or its Affiliates in the form of equity-based incentive or compensation or the exercise or vesting of such incentive or compensation), the Issuer shall permit the Investor to invest in such offering or, at the option of the Issuer, a concurrent offering, on the same terms as the other investors in such offering (and at a price per share or security equal to the one paid by the other investors in such offering), for an aggregate amount of up to (x) the Investor's aggregate percentage ownership of shares of the Issuer on a Fully Diluted Basis immediately prior to such offering <u>multiplied</u> (and rounded up afterwards) by (y) the aggregate gross consideration to be received by the Issuer in such offering; provided that, without taking into account the Warrants or the shares issuable (but not yet issued) thereunder, the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor shall not directly or indirectly own or have the right to acquire under Equity Securities more than 29.9% of the then issued and outstanding voting securities of the Issuer (assuming the exercise, conversion or exchange of any Equity Securities held by any of them at any given time that are exercisable, convertible or exchangeable into or for shares of the Issuer at such time) (the resulting number of securities rounded down).

6.4 Information and remedy

6.4.1 As from the Date of this Agreement until the expiry of the Standstill Period, the Investor shall, upon written request from the Issuer, provide the Issuer within five (5) Business Days, the

Page 14 of 75

aggregate number of Equity Securities owned by the Investor, the Parent Investor and their Affiliates to the extent calculable (it being understood that the Warrants can be described without calculating a number of shares issuable thereunder). As from the Date of this Agreement until the expiry of the Standstill Period, the Issuer shall provide the Investor with notice of any exercise or conversion of any options, warrants, convertible securities or other right to acquire, directly or indirectly, any of the new shares of the Issuer by any person or entity (other than the Investor, the Parent Investor and their Affiliates) within five (5) Business Days after such exercise or conversion.

- **6.4.2** In the event of a material breach of the provisions of Articles 6.1 (other than Article 6.1.4) or 6.2 by the Investor (provided that a breach of the notice requirements of Article 6.2.2 shall not constitute a material breach), the Parent Investor or any of their Affiliates and, provided that such breach can be cured, such breach is not cured within five (5) Business Days after such breach has been notified in writing to the Investor, the following rights of the Investor shall be suspended until such breach has been cured:
 - (i) the rights of the Investor pursuant to Articles 6.3 and 7 (other than pursuant to Article 7.4); and
 - (ii) the right to exercise the Warrants.

7 Covenants by the Issuer

- 7.1 To the extent legally required, the Issuer shall prepare a listing prospectus and shall use reasonable efforts to obtain a listing on the regulated markets of Euronext Brussels and Amsterdam for the Subscription Shares, within ninety (90) days following the Closing. In such case, the effective listing will be subject to regulatory approval of the listing prospectus. To the extent that a listing prospectus is not legally required to obtain a listing on the regulated markets of Euronext Brussels and Amsterdam for the Subscription Shares, the Issuer shall cause the listing of the Subscription Shares as soon as practicable after the Closing, and in any event no later than five (5) Business Days after the Closing.
- 7.2 General Meetings of the Issuer.
 - 7.2.1 Promptly, and in any event within sixty (60) days following the Closing, the Board of Directors shall convene an extraordinary general meeting of the shareholders of the Issuer (the "Issuer EGM"; provided that, if the attendance quorum for any Issuer EGM Proposal (as defined below) (if any) is not achieved at such extraordinary general meeting of the shareholders, the Board of Directors shall, as soon as practicable and in any event no later than forty (40) days following such extraordinary general shareholders' meeting, convene a second extraordinary general meeting of the shareholders, the agenda of which shall include the relevant Issuer EGM Proposal, at which no attendance quorum will apply for the relevant Issuer EGM Proposal, and which shall constitute the "Issuer EGM" with respect to the relevant Issuer EGM Proposal), at which the shareholders of the Issuer will be asked to approve (i) the appointment to the Board of Directors, including such other candidate directors as shall be necessary to appoint the Investor Board Designees and to comply with the gender diversity rules under Belgian company law and to maintain the foreign private issuer status of the Issuer (the "Board Designee Proposal"), (ii) the issuance to the Investor of the Initial Warrants (the "Initial Warrant Proposal"), and (iii) the authorization to the Board of Directors, valid for a period of five (5) years from the date of publication of the authorization in the Annexes to the

Page 15 of 75

Belgian State Gazette, to increase the share capital of the Issuer in one or several times with an amount up to 20% of the share capital at the time of the convening of the Issuer EGM, which capital increases may be achieved by the issuance of shares, convertible bonds and/or warrants exercisable by contributions in cash or in kind, with or without issuance premium, and which authorization will explicitly authorize the Board of Directors to restrict or cancel the shareholders' preferential subscription rights, even if such restriction or cancellation is made for the benefit of one or more specific persons other than the employees of the Issuer or its subsidiaries (including the Investor and its Affiliates) (the "**Authorized Capital Proposal**" and, collectively with the Board Designee Proposal and the Initial Warrant Proposal, the "**Issuer EGM Proposals**"). The Board of Directors shall use reasonable efforts, including to support and defend the Issuer EGM Proposals and to recommend that the Issuer's shareholders approve the Issuer EGM Proposals, to cause the Issuer EGM Proposals to be approved.

- 7.2.2 Between the 57 month and 59 month anniversary of the Closing, the Board of Directors shall convene an extraordinary general meeting of the shareholders of the Issuer (the "Subsequent Issuer EGM"; provided that, if the attendance quorum for any Subsequent Issuer EGM Proposal (as defined below) (if any) is not achieved at such extraordinary general meeting of the shareholders, the Board of Directors shall, as soon as practicable and in any event no later than thirty five (35) days after such extraordinary general meeting of the shareholders and in any event no later than the expiry of the sixty (60) month anniversary of the Closing, convene a second extraordinary general meeting of the shareholders, the agenda of which shall include the relevant Subsequent Issuer EGM Proposal, at which no attendance quorum will apply for the relevant Subsequent Issuer EGM Proposal, and which shall constitute the "Subsequent Issuer EGM" with respect to the relevant Subsequent Issuer EGM Proposal), at which the shareholders of the Issuer will be asked to approve (i) the issuance to the Investor of the Subsequent Warrant (the "Subsequent Warrant Proposal") and (ii) the authorization to the Board of Directors, valid for a period of five (5) years from the date of publication of the authorization in the Annexes to the Belgian State Gazette, to increase the share capital of the Issuer in one or several times with an amount up to 20% of the share capital at the time of the convening of the Subsequent Issuer EGM, which capital increases may be achieved by the issuance of shares, convertible bonds and/or warrants exercisable by contributions in cash or in kind, with or without issuance premium, and which authorization will explicitly authorize the Board of Directors to restrict or cancel the shareholders' preferential subscription rights, even if such restriction or cancellation is made for the benefit of one or more specific persons other than the employees of the Issuer or its subsidiaries (including the Investor and its Affiliates) (together with the Subsequent Warrant Proposal, the "Subsequent Issuer EGM Proposals"). The Board of Directors shall use reasonable efforts, including to support and defend the Subsequent Issuer EGM Proposals and to recommend that the Issuer's shareholders approve the Subsequent Issuer EGM Proposals, to cause the Subsequent Issuer EGM Proposals to be approved.
- **7.2.3** In the event that any Issuer EGM Proposal or Subsequent Issuer EGM Proposal or a proposal of the Investor pursuant to Article 7.3.1 fails to receive shareholder approval at the Issuer EGM or the Subsequent Issuer EGM, as applicable, or, as relevant, a subsequent general shareholders' meeting of the Issuer, other than as a result of the failure by the Investor or its Affiliates to vote with all their voting securities in favour of such Issuer EGM Proposal, Subsequent Issuer EGM Proposal or such proposal of the Investor pursuant to Article 7.3.1 to the extent permitted by applicable law, without limiting any other remedies that may be available, the following rules shall apply:

Page 16 of 75

- (i) in the event that either or both of the Investor Board Designees to be appointed in connection with the Board Designee Proposal or a proposal of the Investor pursuant to Article 7.3.1 are not approved by the Issuer EGM or, as relevant, a subsequent general shareholders' meeting of the Issuer, other than as a result of the failure by the Investor or its Affiliates to vote with all of their voting securities in favour of such Issuer EGM Proposal or such proposal of the Investor pursuant to Article 7.3.1 to the extent permitted by applicable law, the Investor Board Designee(s) not appointed will be invited to the Board of Directors as observers who, as much as legally possible, (x) shall have the right to attend and participate in the meetings of the Board of Directors), and (y) shall have the same duties as if they were a director taking into account the foregoing, in each case of (x) and (y) subject to the provisions of Article 7.3 which shall apply mutatis mutandis to the aforementioned observers.
- (ii) In the event that the Initial Warrant Proposal and/or the Subsequent Warrant Proposal is not approved by the Issuer EGM or the Subsequent Issuer EGM, as applicable, other than as a result of the failure by the Investor or its Affiliates to vote with all of their voting securities in favour of such Issuer EGM Proposal or Subsequent Issuer EGM Proposal to the extent permitted by applicable law, the following provisions shall apply:
 - (a) The Parties shall collaborate to structure and organise the right to subscribe for Ordinary Shares as reflected by the relevant Warrants via a contractual subscription or option agreement or otherwise, allowing, to the greatest extent legally possible, the Investor to subscribe for or acquire the number of shares of the Issuer it would otherwise be able to subscribe for or acquire upon exercise of the relevant Warrants, whether through the Issuer's authorised capital or otherwise.
 - (b) As long as the Issuer's Board of Directors shall have the power to issue Ordinary Shares within the framework of the authorised capital, the authorised capital shall to the greatest extent legally possible be used with priority for the purpose of the issuance of shares to the Investor as provided for in sub-paragraph (a) of paragraph (ii) of this Article 7.2.3, without prejudice to the right of the Issuer to use the authorised capital for equity-based incentives or compensation for current or future employees, consultants, directors and/or officers of the Issuer or its Affiliates.
 - (c) From the Closing until the earlier of (x) the date that the Initial Warrants are issued and, (y) the date that the Investor, the Parent Investor and any of their Affiliates together for the first time own 25.1% of the actually issued and outstanding shares of the Issuer (rounded down to the nearest whole share) or more, unless the Issuer has provided the Investor with the right to subscribe for new shares of authorised capital that would result in the Investor, the Parent Investor and their Affiliates in the aggregate owning 25.1% of the actually issued and outstanding shares of the Issuer (rounded down to the nearest whole share) or more at a price per share no higher than the price per share that it would

Page 17 of 75

have been entitled to subscribe for or acquire pursuant to the Initial Warrant and the Investor has declined to accept such right or has failed to pay the purchase price for such new shares at the closing of their issuance, any of the following proposals shall not be approved by or submitted by the Board of Directors at its own initiative to the general shareholders' meeting of the Issuer without the prior written consent of the Investor:

- (I) any proposal to amend, repeal or otherwise modify any provision of the Issuer's articles of association that would be reasonably expected to adversely affect the interests of the Investor, the Parent Investor or any of their Affiliates (other than to the extent legally required);
- (II) any proposal to decrease the Issuer's share capital (other than in relation to issuing Equity Securities to the Investor or its Affiliates) or repurchase, redeem or reacquire any Equity Securities;
- (III) any proposal to modify the rights of any Issuer securities in a manner adverse to the Investor, the Parent Investor or any of their Affiliates or to cancel or limit any preferential subscription rights of the Investor;
- (IV) any proposal to dissolve, liquidate or wind-up the business and affairs of the Issuer;
- (V) any proposal to consummate, or enter into any agreement to consummate, a merger or de-merger of the Issuer or similar transaction involving the Issuer; or
- (VI) any proposal to transfer or dispose of all or substantially all of the assets and liabilities of the Issuer.

The above provisions of this sub-paragraph (c) (x) shall not apply to any amendment to the Issuer's articles of association, any capital increase or issue of Equity Securities, or other operation, decision or proposal within the framework of any equity-based incentive or compensation for current or future employees, consultants, directors and/or officers of the Issuer or its Affiliates and (y) shall be without prejudice to the ability of the Board of Directors to submit such proposals or agenda items to the general meeting of the shareholders of the Issuer that the Board of Directors is legally required to so submit at the request of shareholders of the Issuer.

Notwithstanding the foregoing, any time the Board of Directors determines to make use of its authorised capital until the earlier of (x) the date that the Initial Warrants are issued and (y) the date that the Investor, the Parent Investor and any of their Affiliates together for the first time own 25.1% of the actually issued and outstanding shares of the Issuer (rounded down to the nearest whole share) or more, the Board of Directors should first make use of its authorised capital to issue shares (provided that the Investor subscribes for such shares) in order to allow for the Investor to subscribe for the shares that it would have been entitled to subscribe for or acquire pursuant to the Initial Warrant at a price per share no higher than the price per share that it would have been entitled to subscribe for or acquire pursuant to the Initial Warrant.

Page 18 of 75

- (iii) The provisions of paragraph (ii) of this Article 7.2.3 shall be without prejudice to the right of the Investor to acquire shares of the Issuer in the open market up to, and within the limitations of, the Standstill Limit. Subject to the foregoing paragraphs of this Article 7.2.3, the Board of Directors shall take all actions necessary to include any such unapproved Issuer EGM Proposal, Subsequent Issuer EGM Proposal or proposal of the Investor pursuant to Article 7.3.1, as applicable, on the agenda of an extraordinary general meeting to be convened as soon as possible following the date of the Issuer EGM or the Subsequent Issuer EGM, as applicable. If the attendance quorum for any such unapproved Issuer EGM Proposal, Subsequent Issuer EGM Proposal or (to the extent applicable) proposal of the Investor pursuant to Article 7.3.1 (if any) is not achieved at such extraordinary general meeting of the shareholders, the Board of Directors shall, as soon as practicable and in any event no later than thirty five (35) days following such extraordinary general shareholders' meeting, convene a second extraordinary general meeting of the shareholders, the agenda of which shall include the relevant Issuer EGM Proposals, Subsequent Issuer EGM Proposals or proposal of the Investor pursuant to Article 7.3.1, at which no attendance quorum will apply for the relevant Issuer EGM Proposals, Subsequent Issuer EGM Proposals or proposal of the Investor pursuant to Article 7.3.1. The Board of Directors shall use reasonable efforts, including to support and defend such unapproved Issuer EGM Proposal, Subsequent Issuer EGM Proposal and proposal of the Investor pursuant to Article 7.3.1, and to recommend that the Issuer's shareholders approve such unapproved Issuer EGM Proposal, Subsequent Issuer EGM Proposal and proposal of the Investor pursuant to Article 7.3.1, to cause such unapproved Issuer EGM Proposal, Subsequent Issuer EGM Proposal and proposal of the Investor pursuant to Article 7.3.1, to be approved. The Board of Directors shall continue to re-submit proposals pursuant to this Article 7.2.3(iii), to subsequent annual general meetings of the shareholders of the Issuer or extraordinary general meetings of the shareholders held at the same time, until such proposals are approved by the Issuer's shareholders. If the attendance quorum for any such proposal (if any) is not achieved at such extraordinary general meeting of the shareholders, the Board of Directors shall, as soon as practicable, convene a second extraordinary general meeting of the shareholders, the agenda of which shall include the relevant proposals, at which no attendance quorum will apply for the relevant proposals.
- **7.2.4** The Issuer shall (i) begin preparing the legally required reports in connection with the issue of the Subscription Shares and the Warrants in accordance with the terms of this Agreement immediately following the Date of this Agreement, (ii) use reasonable efforts to obtain the approval or "nihil obstat" in relation to such reports by the FSMA (to the extent legally required), (iii) provide the Investor and its legal counsel with the ability to review and comment on such reports reasonably in advance prior their submission to the FSMA, and shall take into account the reasonable comments of the Investor and its legal counsel in relation to the drafting of such reports, the presentation of the issue of the Subscription Shares and the Warrants and such reports to the FSMA, and (iv) keep the Issuer and its legal counsel reasonably informed of the comments from the FSMA in relation to the reports and presentations.

7.3 Investor Board Designees

7.3.1 Subject to the occurrence of the Closing, until the expiration of the Collaboration Term (as defined in the Option, License and Collaboration Agreement) (or the termination of the Option,

Page 19 of 75

License and Collaboration Agreement, if earlier (other than if the Issuer breaches the Option, License and Collaboration Agreement and the Option, License and Collaboration Agreement is terminated by the Parent Investor as a result)), Investor will have the right to have two (2) Investor Board Designees appointed to the Board of Directors. From the date of the expiration of the Collaboration Term (as defined in the Option, License and Collaboration Agreement) (or the termination of the Option, License and Collaboration Agreement, if earlier (other than if the Issuer breaches the Option, License and Collaboration Agreement and the Option, License and Collaboration Agreement is terminated by the Parent Investor as a result)) until the date that the Investor, the Parent Investor, any of the Affiliates of the Investor or the Parent Investor, or any other party Acting in Concert with the Investor, the Parent Investor or any of the Affiliates of the Investor or the Parent Investor, cease to own, directly or indirectly, at least 20% of the then outstanding shares of the Issuer on a non-diluted basis (the resulting number of shares rounded up), Investor will have the right to have one (1) Investor Board Designee appointed to the Board of Directors, and shall then cause one (1) of its two (2) Investor Board Designees to resign. From the date that the Investor will no longer have the right to have one (1) Investor Board Designees appointed to the Board of Directors as aforementioned, it shall cause its Investor Board Designee to resign. Subject to the foregoing, the Board of Directors shall use reasonable efforts, including to convene a general meeting of shareholders of the Issuer as soon as practicable at which the shareholders of the Issuer will be asked to approve the appointment to the Board of Directors of the Investor Board Designees, to support and defend the appointment to the Board of Directors of the Investor Board Designees and to recommend that the Issuer's shareholders approve the appointment to the Board of Directors of the Investor Board Designees, in order to cause the appointment to the Board of Directors of the Investor Board Designees. In particular, and without prejudice to the foregoing, upon the termination of the board mandate of any Investor Board Designee (for whatever cause), at the option of Investor:

- (i) the Board of Directors shall as soon as practicably possible co-opt to the Board of Directors a replacement of the Investor Board Designee proposed by the Investor, and shall use reasonable efforts to cause the confirmation of the co-optation at the next general meeting of shareholders of the Issuer, or
- (ii) the Board of Directors shall as soon as practicably possible convene a general meeting of shareholders of the Issuer at which the shareholders of the Issuer will be asked to approve the appointment to the Board of Directors of a replacement of the Investor Board Designee proposed by the Investor; the Board of Directors shall use reasonable efforts to cause the appointment to the Board of Directors of the replacement.
- **7.3.2** The Parties agree that the composition of the Issuer's Board of Directors will need to take into account the relevant rules, regulations and corporate governance requirements in relation to gender diversity, and that the Board of Directors may not have a majority of the members of the Board of Directors being U.S. citizens or residents.
- 7.3.3 Investor Board Designees are required to act as any other member of the Board of Directors with regard to discretion and confidentiality.
- **7.3.4** Without prejudice to the rules and principles regarding conflicts of interest as provided for by applicable law or the provisions of the Issuer's articles of association or corporate governance charter from time to time, the Investor Board Designees shall be deemed to have a conflict of interest in relation to any transaction or decision by the Issuer or any of its Affiliates in respect

Page 20 of 75

of which Investor, the Parent Investor or any of their Affiliates would have a conflict of interest of a material financial or commercial nature if they were themselves a member of the Board of Directors (provided that the mere existence of this Agreement, the Warrants, the Option, License and Collaboration Agreement and the License and Collaboration Agreement (as amended from time to time) and the rights of the Investor, the Parent Investor or any of their Affiliates thereunder are not a sufficient justification to invoke such a conflict of interest), it being understood that any transaction or decision by the Issuer or any of its Affiliates with or in relation to the Investor, the Parent Investor or any of their Affiliates, including pursuant to or in relation to this Agreement, the Warrants, the Option, License and Collaboration Agreement and the License and Collaboration Agreement (as amended from time to time) is deemed to constitute such a conflict of interest. In the event the aforementioned matter is submitted to the Board of Directors for deliberation or approval, the Investor Board Designees shall inform the other members of the Board of Directors timely in advance that the matter shall be deemed to create a conflict of interest, and subsequently refrain from further participation in the deliberation and decision making process in relation to such matters, provided that the Investor Board Designees shall be permitted to participate in discussions among the Board of Directors prior to the deliberation and decision making process in relation to such matters.

- 7.4 During the Standstill Period and during any period that the Issuer and the Investor, the Parent Investor, or any of the Affiliates of the Investor or the Parent Investor were to be affiliates or intermediaries of each other (within the meaning of the Belgian Royal Decree of 27 April 2007 on public takeover bids) or were to Act in Concert, the Issuer shall not, without the express written consent of the Investor, directly or indirectly (including through affiliates or intermediaries within the meaning of the Belgian Royal Decree of 27 April 2007 on public takeover bids or parties Acting in Concert with the Issuer or such affiliates or intermediaries) acquire any voting securities of the Issuer, or take any other action, if after giving effect to such acquisition or the taking of such other action the Investor, the Parent Investor, or any of the Affiliates of the Investor or the Parent Investor, would directly or indirectly (including through affiliates or intermediaries) acquire any voting securities or intermediaries) of the Investor or the Parent Investor, would directly or indirectly (including through affiliates or intermediaries) acquires or intermediaries within the meaning of the Belgian Royal Decree of 27 April 2007 on take over bids or any other party Acting in Concert with the Investor or such affiliates or intermediaries) own, or were to continue to own, more than 29.9% of the then issued and outstanding voting securities of the Issuer on a non-diluted basis (the resulting number of securities rounded down) or would otherwise oblige the Investor, the Parent Investor, or any of the Affiliates or the Parent Investor, to launch a public takeover bid on securities of the Issuer.
- **7.5** From and after the date hereof until the date that the Investor, together with its Affiliates, no longer owns 10% or more of the outstanding shares of the Issuer, the Issuer shall, upon reasonable notice by the Investor, provide access (for not more often than once per calendar year) to the Investor to its financial books to facilitate compliance with the Investor's accounting or financial reporting requirements.
- **7.6** Nothing in this Agreement restricts the Issuer to consider and proceed with any business combination, divestment, acquisition, financing, licensing or other business or commercial transaction (whether in whole or in part), except as agreed in relation to or pursuant to the Option, License and Collaboration Agreement or the License and Collaboration Agreement (as amended from time to time).

Page 21 of 75

7.7 Upon adoption of the new Belgian Companies and Association Code the two-tier board system (duaal bestuur / administration duale) shall be continued. In addition, each share shall have one vote.

8 Survival

The respective provisions contained in Articles 7, 8, 9, 15 and 18 hereof shall remain in full force and effect, regardless of the end of the Standstill Period or of the Lock-up Period.

9 Taxes – Expenses

- **9.1** The Issuer shall pay any stamp, issue, registration, documentary or other taxes and duties, including interest and penalties, payable in Belgium on or in connection with the issuance of the Subscription Shares or the signing of this Agreement. The Issuer shall also pay all costs associated with the listing of the Subscription Shares on the regulated markets of Euronext Brussels and Amsterdam in accordance with Article 7.
- **9.2** Each Party shall bear its own costs and expenses (including legal and other advisory fees) incurred in connection with the preparation of this Agreement, and all related agreements and transactions.
- 9.3 US Federal Income Tax Information Reporting
 - **9.3.1** For each taxable year in which Investor holds shares of Issuer, Issuer shall determine whether Issuer or any of Issuer's Subsidiaries was a "passive foreign investment company," as defined in section 1297(a) of the Code (a "**PFIC**") for such taxable year and, if Issuer determines that Issuer or any of Issuer's Subsidiaries was a PFIC for such taxable year, (a) Issuer shall, no later than thirty (30) days from the date of the close of Issuer's taxable year, notify Investor of such determination and (b) Issuer shall provide Investor with adequate information in Issuer's possession (at Investor's expense) in order for Parent Investor, in consultation with Investor and Issuer, to complete its U.S. Internal Revenue Service Form 8621 with respect to Issuer or such Subsidiary; provided that if Parent Investor intends to elect to treat Issuer and/or any Subsidiary as a "qualified electing fund," as defined in section 1295 of the Code, Investor shall notify Issuer of such intent, and Issuer shall provide Investor (at Investor's expense) with PFIC Annual Information Statements.
 - **9.3.2** For each taxable year in which Investor holds shares of Issuer, Investor shall determine whether Issuer or any of Issuer's Subsidiaries is a "controlled foreign corporation," within the meaning of as defined in section 957 of the Code (a "**CFC**") for such taxable year with respect to Parent Investor and, if Investor determines that Issuer or any of Issuer's Subsidiaries was a CFC for such taxable year with respect to Parent Investor, (a) Issuer shall, no later than thirty (30) days from the close of Issuer's taxable year, notify Issuer of such determination and (b) Issuer shall provide Investor with adequate information in Issuer's possession (at Investor's expense) in order for Investor, in consultation with Issuer, to reasonably determine any amounts required to be included pursuant to sections 951(a) and 951A of the Code in the gross income of Parent Investor as defined in section 951(b) of the Code and to comply with Parent Investor's filing obligations under the Code, including, but not limited to, completing the U.S. Internal Revenue Service Form 5471 with respect to Issuer or any such Subsidiaries, all of such information to be issued in an annual statement.

Page 22 of 75

- **9.4** The Parties agree to reasonably cooperate with one another and use reasonable efforts to mitigate or reduce tax withholding or similar obligations in respect of payments made by Investor to Issuer under this Agreement where an exemption or reduction of such withholding or similar obligation is available under the applicable legislation, including double tax treaties. Without limiting the generality of the foregoing, Issuer shall provide Investor at Investor's expense any tax forms and other information in Issuer's possession that may be reasonably requested by Parent Investor in order for it to prepare its U.S. tax filings. Each Party shall provide the other with reasonable assistance to enable the recovery, as permitted by Applicable Law (as defined in the Option, License and Collaboration Agreement), of withholding taxes, value added taxes, or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding tax or value added tax.
- **9.5** Except as would have an adverse effect on Issuer or any of Issuer's Subsidiaries and subject to the sole consent of Issuer, such consent not to be unreasonably withheld, Issuer agrees to reasonably cooperate with Investor and use reasonable efforts (i) to provide Investor with such information as is reasonably requested by Investor to permit Investor to make the determinations under Section 9.3.2 of this Agreement, and (ii) to avoid or reduce any amounts required to be included pursuant to sections 951(a) and 951A of the Code in the gross income of any "United States shareholder", as defined in section 951(b) of the Code, including, without limitation, by filing any elections reasonably requested by Investor under US Treasury Regulations section 301.7701-3 with respect to any Subsidiary.

10 No Assignment

Except with the prior written consent of the other Party, neither of the Parties hereto shall be entitled to transfer or assign any of its rights or obligations under this Agreement, provided, however, that the Investor may freely assign its rights and obligations to any of its Affiliates.

11 Entire Agreement

This Agreement contains the entire agreement between the Parties in respect of its subject matter. It replaces and annuls all prior agreements, communications, offers, proposals or correspondence, oral or written, exchanged or concluded between the Parties relating to the same subject matter, including the Subscription Agreement, dated as of December 16, 2015, by and between the Issuer and Gilead Biopharmaceutics Ireland UC (provided, however, that such replacement and annulment of the aforementioned Subscription Agreement of December 16, 2015, shall be without prejudice to any rights accrued to any party under the latter agreement up to the date of this Agreement).

12 Further Assurances

Subject to Article 5, each of the Parties shall use reasonable efforts to take all actions and do all things necessary, proper or advisable to consummate the transactions contemplated by this Agreement and the issuance of the Warrants on the terms of this Agreement and the Warrants.

13 Severability

13.1 If any provision in this Agreement is held to be illegal, invalid or unenforceable, in whole or in part, under any applicable law, then such provision or part of it shall be deemed not to form part of this Agreement, and the legality, validity or enforceability of the remainder of this Agreement shall not be affected.

Page 23 of 75

13.2 In such case, each Party shall use reasonable efforts to immediately negotiate in good faith a valid replacement provision that is as close as possible to the original intention of the Parties and has the same or as similar as possible economic effect.

14 Limitation of Liability

NEITHER PARTY SHALL BE LIABLE TO THE OTHER FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, PUNITIVE, OR INDIRECT DAMAGES ARISING FROM OR RELATING TO ANY BREACH OF THIS AGREEMENT OR ANY TORT CLAIMS ARISING HEREUNDER, REGARDLESS OF ANY NOTICE OF THE POSSIBILITY OF SUCH DAMAGES. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS ARTICLE 14 IS INTENDED TO OR SHALL LIMIT OR RESTRICT (A) DAMAGES AVAILABLE FOR A PARTY'S BREACH OF ITS CONFIDENTIALITY OBLIGATIONS UNDER ARTICLE 13 OF THE OPTION, LICENSE AND COLLABORATION AGREEMENT, OR (B) DAMAGES AVAILABLE IN THE CASE OF A PARTY'S FRAUD, GROSS NEGLIGENCE OR INTENTIONAL MISCONDUCT.

15 Notices

Any notice required or permitted to be given under this Agreement shall be in writing, shall specifically refer to this Agreement, and shall be addressed to the appropriate Party at the address specified below or such other address as may be specified by such Party in writing in accordance with this Article 15, and shall be deemed to have been given for all purposes (a) when received, if hand-delivered, sent by a reputable international expedited delivery service or sent by facsimile (with transmission confirmed), or (b) five (5) Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. This Article 15 is not intended to govern the day-to-day business communications necessary between the Parties in performing their obligations under the terms of this Agreement.

If to Galapagos NV:

With a copy to (which shall not constitute notice):

If to Gilead Therapeutics A1 Unlimited Company:

Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: Chief Executive Officer Fax: [***]

Galapagos NV Generaal De Wittelaan L11 A3 2800 Mechelen Belgium Attention: General Counsel Fax: [***]

Gilead Therapeutics A1 Unlimited Company

c/o Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 USA Attention: President and Chief Operating Officer Facsimile: [***]

Gilead Therapeutics A1 Unlimited Company c/o Gilead Sciences, Inc. 333 Lakeside Drive

Page 24 of 75

With copies to (which shall not constitute notice):

Foster City, CA 94404 USA Attention: General Counsel Facsimile: [***]

Eubelius Louizalaan 99 Av. Louise BE-1050 Brussels Attention: Marieke Wyckaert; Joris De Wolf Facsimile: [***]

Skadden, Arps, Slate, Meagher & Flom LLP 4 Times Square New York, NY 10036 USA Attention: Stephen F. Arcano Facsimile: [***]

Skadden, Arps, Slate, Meagher & Flom LLP 500 Boylston Street Boston, Massachusetts 02116 USA Attention: Graham Robinson Facsimile: [***]

16 Counterparts

This Agreement may be signed in counterparts, in the number of originals stated hereinafter on the signature page. When taken together, the counterparts signed by all Parties shall constitute one and the same instrument. Signature pages of this Agreement may be exchanged by email or in pdf or other electronic means without affecting the validity thereof.

17 Specific Enforcement

Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement shall in any way limit the ability of a Party to seek or obtain from any court of competent jurisdiction any remedies available at law or in equity (including injunctive relief) to enforce any covenant or agreement of the other Party hereunder.

18 Governing Law and Jurisdiction

18.1 Governing Law

This agreement and any non-contractual obligations arising out of or in connection with it shall be governed by and construed in accordance with Belgian law.

18.2 Jurisdiction

18.2.1 It is the objective of the Parties to establish procedures to facilitate the resolution of disputes arising under this Agreement in an expedient manner by mutual cooperation and without resort to litigation. In the event of any disputes, controversies or differences which may arise between the Parties out of or in relation to or in connection with this Agreement, including any alleged failure to perform, or breach, of this Agreement, or any issue relating to the interpretation or application of this Agreement, then upon the request of either Party by written notice, the Parties agree to meet and discuss in good faith a possible resolution thereof, which good faith efforts shall include at least one in-person meeting between the Executive Officers (as defined

Page 25 of 75

in the Option, License and Collaboration Agreement) of each Party. If the matter is not resolved within thirty (30) days following the written request for discussions, either Party may then invoke the provisions of Article 18.2.2.

- 18.2.2 Any dispute, controversy, difference or claim which may arise between the Parties out of or in relation to or in connection with this Agreement (including arising out of or relating to the validity, construction, interpretation, enforceability, breach, performance, application or termination of this Agreement) that is not resolved pursuant to Article 18.2.1, shall be settled by binding arbitration in accordance with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators, one each chosen by the respective Parties and the third chosen by mutual agreement of the first two, and otherwise in accordance with the ICC Rules. The arbitrators shall have significant experience and shall have expertise in Belgian corporate law. Either Party, following the end of the thirty (30) day period referenced in Article 18.2.1, may refer such issue to arbitration by submitting a written notice of such request to the other Party. The place of arbitration shall be New York and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither Party shall be required to give general discovery of documents, but may be required only to produce specific, identified documents which are relevant to the dispute. The Parties shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Parties, and shall be governed by the terms and conditions hereof, including the limitation on damages set forth in Article 14. The Parties agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of Belgian law applicable to the commencement of a lawsuit shall apply to the commencement of arbitration under this Article 18.2.2. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each Party. All proceedings and decisions of the arbitrators shall be deemed Confidential Information (as defined in the Option, License and Collaboration Agreement) of each of the Parties, and shall be subject to Article 13 of the Option, License and Collaboration Agreement.
- **18.2.3** Articles 15.4 through 15.6 of the Option, License and Collaboration Agreement are incorporated herein by reference and shall apply mutatis mutandis to this Agreement.

[Signature page follows.]

Page 26 of 75

Done as of the date first written above in 2 originals, each Party acknowledging receipt of its own original.

Galapagos NV:

/s/ Onno van de Stolpe

Name: Onno van de Stolpe

Title: Director and Chief Executive Officer

/s/ Bart Filius

Name: Bart Filius

Title: Chief Financial Officer and Chief Operating Officer

[Signature Page to Subscription Agreement]

Gilead Therapeutics A1 Unlimited Company

/s/ David Cadogan Name: David Cadogan

Title: Director /s/ Padraig Clancy Name: Padraig Clancy

Director Title:

[Signature Page to Subscription Agreement]

Exhibit A

Form of Initial Warrants

[Attached]

INITIAL GILEAD WARRANT A

–issued by–

GALAPAGOS NV

-dated-

[date] 2019

TABLE OF CONTENT

Articles 1.	5	Certain Definitions and Interpretation	Page 2
	1.1. 1.2. 1.3. 1.4. 1.5.	Certain definitions Headings Meaning of references Fractional value of Shares Language	2 4 4 5 5
2.		Issuance, Nature and Form of the Warrant	5
	2.1. 2.2. 2.3. 2.4.	Issuance and nature Registered form Transferability of the Warrant No listing of the Warrant	5 5 5 6
3.		Term of the Warrant	6
4.		Shares	6
	4.1. 4.2. 4.3.	Number of Shares issuable upon an exercise of the Warrant Nature and form of the Shares Listing of the Shares	6 7 7
5.		Exercise Price	7
6.		Specific Conditions	7
7.		Exercise of the Warrant	8
	7.1. 7.2. 7.3. 7.4.	Exercise Notice Payment of the Exercise Price Issue and delivery of the Shares Allocation of the Exercise Price	8 8 9 9
8.		Adjustments	9
1	8.1. 8.2. 8.3.	General Adjustments for Share Reorganisations Adjustments for mergers and de-mergers	9 10 10
9.		Representations and Warranties	11
	9.1. 9.2.	Representations and Warranties of the Company Representations and Warranties of the holder of the Warrant	11 12
10.		Miscellaneous	12
	10.1. 10.2. 10.3. 10.4. 10.5. 10.6.	Binding nature of the Conditions Severability Specific Enforcement Costs and expenses Governing law and jurisdiction Notices	12 12 12 13 13 13

i

GALAPAGOS NV

Limited Liability Company

Generaal De Wittelaan L11 A3, 2800 Mechelen (Belgium) Register of Legal Persons VAT BE 0466.460.429 (Antwerp, division Mechelen)

INITIAL GILEAD WARRANT A

RECITALS

On 14 July 2019, Galapagos NV (hereafter further referred to as the "**Company**"), entered into a Subscription Agreement (as defined below) with Gilead Therapeutics A1 Unlimited Company (hereafter further referred to as the "**Investor**"). The Investor is an indirect wholly-owned subsidiary of Gilead Sciences, Inc. (hereafter further referred to as the "**Parent Investor**"), a U.S. corporation listed on the NASDAQ Stock Market and a researchbased biopharmaceutical company focused on the discovery, development, and commercialisation of innovative medicines. Simultaneously with the execution of the Subscription Agreement, the Company and the Parent Investor also entered into an Option, License and Collaboration Agreement (as defined below). Pursuant to the Option, License and Collaboration Agreement, the Company agreed to discover, research, and develop molecules and products, and Parent Investor agreed to have an option to participate in the development and commercialisation of molecules and products, in each case, on the terms and conditions set forth in such agreement. Pursuant to the Subscription Agreement, the Investor agreed to make an investment into the share capital of the Company. The investment was effected on [date]. As part of the overall agreement between the Company, the Investor and the Parent Investor, the Subscription Agreement also provided for the issuance to the Investor of a number of warrants. The present terms and conditions (hereinafter referred to as the "**Conditions**") contain the issue and exercise conditions of the Initial Gilead Warrant A issued by the Company, as contemplated by the Subscription Agreement. The issue of the Warrants is exclusively reserved to the Investor, in consideration of the subscription for shares and entry into the collaboration contemplated by the agreements referred to above.

1. CERTAIN DEFINITIONS AND INTERPRETATION

1.1. Certain definitions

In these Conditions, the following words and expressions that are not defined elsewhere in these Conditions shall have the following meanings, save where the context requires otherwise:

"Acting in Concert" means, when used in relation to a person or entity, acting in concert (in onderling overleg handelende personen / personnes agissant de concert) in the sense of Article 3, §1, 5° of the Belgian Act of 1 April 2007 regarding public takeover bids, or Article 1, §2, 5° of the Belgian Royal Decree of 27 April 2007 regarding public takeover bids.

"Affiliate" means, when used with respect to a person or entity, any person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such person or entity, for so long as such control exists, regardless of whether such person or entity is or becomes an Affiliate on or after the date of the Subscription Agreement. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, or by contract or otherwise.

"Belgian Companies and Associations Code" means the Belgian Companies and Associations Code of 23 March 2019, as amended from time to time, and the rules and regulations promulgated thereunder.

"Belgian Companies Code" means the Belgian Companies Code of 7 May 1999, as amended from time to time, and the rules and regulations promulgated thereunder.

"**Business Day**" means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States, (c) a bank or other public holiday in Brussels, Belgium, (d) a bank or other public holiday in Ireland or (e) the period commencing on December 25th and ending on January 1st (inclusive).

"**Company**" means Galapagos NV/SA, a corporation (naamloze vennootschap / société anonyme) organized and existing under the laws of Belgium, with registered office at Generaal De Wittelaan L11 A3, 2800 Mechelen, Belgium, and registered with the Register of Legal Persons under enterprise number 0466.460.429 (Antwerp, division Mechelen).

"Conditions" means the present terms and conditions of the Warrants.

"Equity Security" means (a) any Share, and (b) any other security, financial instrument, certificate and other right (including options, futures, swaps and other derivatives) issued or, with respect to options, futures, swaps and other derivatives, contracted by the Company and representing, being exercisable, convertible or exchangeable into or for, or otherwise providing a right to acquire, directly or indirectly, any of the Equity Securities referred to in (a).

"Exercise Account" has the meaning as defined in Article 7.2(b).

"Exercise Date" has the meaning as defined in Article 7.1(b).

"Exercise Notice" has the meaning as defined in Article 7.1(a).

"Exercise Price" means the exercise price of the Warrant, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, as determined pursuant to Article 5.

"Expiry Date" has the meaning as defined in Article 3(a).

"Gilead Warrant" means each of the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"**Initial Gilead Warrant**" means each of the warrants (inschrijvingsrechten or warrants / droits de souscription) issued by the Company on the Issue Date to the Investor, and named, respectively, the "Initial Gilead Warrant A" and "Initial Gilead Warrant B".

"**Investor**" means Gilead Therapeutics A1 Unlimited Company, an unlimited liability company formed under the laws of Ireland, registered with Ireland's Companies Registration Office under number 615395.

"Issue Date" means the date on which the Warrant has been issued by the extraordinary general shareholders' meeting of the Company, i.e. [date].

"**Notice**" has the meaning given to it in Article 10.6.

"**Option, License and Collaboration Agreement**" means the option, license and collaboration agreement, dated 14 July 2019, by and between the Company and the Parent Investor.

"Parent Investor" means Gilead Sciences, Inc., a corporation incorporated under the laws of Delaware.

"**Reference Date**" means [date], being the date on which the Subscription Shares were issued by the Company and were subscribed for by the Investor pursuant to the Subscription Agreement.

"**Reference Exercise Price**" means a price, per Share subscribed for upon an exercise of the Warrant in relation to such Shares, that shall be equal to EUR 140.59, being the issue price, on a per Share basis, that was paid by the Investor with respect to the Subscription Shares that were issued to the Investor on the Reference Date pursuant to the Subscription Agreement.

"Share" means any share (aandeel / action) outstanding from time to time representing the Company's share capital.

"Share Reorganisation" has the meaning given to it in Article 8.2(a).

"Subscription Agreement" means the subscription agreement, dated 14 July 2019, by and between the Company and the Investor, in relation to the subscription by the Investor for the Subscription Shares and the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"**Subscription Shares**" means the [6,828,985] Shares that were issued by the Company and subscribed for by the Investor on the Reference Date pursuant to the Subscription Agreement.

"Subsequent Gilead Warrant B" means the warrant (inschrijvingsrecht or warrant / droit de souscription), named "Subsequent Gilead Warrant B" to be issued by the Company to the Investor pursuant to the Subscription Agreement.

"Succession Transaction" has the meaning given to it in Article 8.3(a).

"Successor Company" has the meaning given to it in Article 8.3(a).

"Term" means the term of the Warrant as referred to in Article 3.

"Warrant" means the Initial Gilead Warrant, named "Initial Gilead Warrant A.

"Warrant Limit" has the meaning given to it in Article 4.1(a).

1.2. Headings

Headings and the table of contents used in these Conditions are for convenience purposes only and shall not affect the construction or interpretation of these Conditions.

1.3. Meaning of references

Unless the context does not so permit, or save where specifically indicated otherwise:

- references to Articles are to Articles in these Conditions, and references to sub-Articles or paragraphs are to sub-Articles or paragraphs of the Article in which such references appear;
- (b) the words "herein", "hereof", "hereunder", "hereby", "hereto", "herewith" and words of similar import shall refer to these Conditions as a whole and not to any particular Article, paragraph or other subdivision;
- (c) references to the word "include" or "including" (or any similar term) are not to be construed as implying any limitation, and general words introduced by the word "other" (or any similar term) shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things;
- (d) any reference to "writing" or "written" includes any method of reproducing words or text in a legible and non-transitory form but, for the avoidance of doubt, shall not include e-mail;
- (e) references to any statute, regulation or statutory provision shall be deemed to include reference to any statute, regulation or statutory instrument which amends, extends, consolidates or replaces the same (or shall have done so) and to any other regulation, statutory instrument or other subordinate legislation made thereunder or pursuant thereto, provided that no such reference shall include any amendment, extension or replacement of the same with retrospective effect;

(f) all periods of time set out herein shall be calculated from midnight to midnight local time in Brussels, Belgium. They shall start on the day following the day on which the event triggering the relevant period of time has occurred. The expiration date shall be included in the period of time. If the expiration date is not a Business Day, it shall be postponed until the next Business Day. Unless otherwise provided herein, all periods of time shall be calculated in calendar days. All periods of time consisting of a number of months (or years) shall be calculated from the day in the month (or year) when the triggering event has occurred until the eve of the same day in the following month(s) (or year(s)) ("van de zoveelste tot de dag vóór de zoveelste" / "de quantième à veille de quantième").

1.4. Fractional value of Shares

For the purpose of these Conditions, the fractional value (fractiewaarde / pair comptable) of the Company's Shares from time to time shall be determined as a fraction, (a) the numerator of which is the amount of the Company's share capital at that time, and (b) the denominator of which is the aggregate number of actually issued and outstanding Shares of the Company at that time.

1.5. Language

The Conditions have been prepared in English and a Dutch translation will be prepared. In the case of discrepancies between the English and the Dutch version, the English version shall prevail between the parties hereto to the fullest extent possible and permitted by Belgian law. Notwithstanding the foregoing, Belgian legal concepts which are expressed in English language terms, are to be interpreted in accordance with the Belgian legal terms to which they refer, and the use herein of Dutch and/or French words in these Conditions as translation for certain words or concepts shall be conclusive in the determination of the relevant legal concept under Belgian law of the words or concepts that are so translated herein.

2. ISSUANCE, NATURE AND FORM OF THE WARRANT

2.1. Issuance and nature

- (a) The Warrant has been issued, without any additional consideration being due by the Investor or any of its Affiliates, pursuant to a resolution of the extraordinary general shareholders' meeting of the Company held on the Issue Date, with dis-application of the statutory preferential subscription rights of the shareholders of the Company for the benefit of the Investor.
- (b) Subject to, and in accordance with, the terms and conditions set forth in these Conditions, the Warrant confers the right (but not the obligation) on the holder thereof to subscribe, upon any exercise of the Warrant, for a number of new Shares to be issued by the Company.
- (c) Except as otherwise provided under Belgian law, the holder of the Warrant is no shareholder of the Company solely by virtue of holding the Warrant, and therefore does not have the rights of a shareholder in relation to the Shares to be issued or delivered to the holder of the Warrant upon an exercise of the Warrant until the exercise of the Warrant and the issue or delivery of the relevant Shares.

2.2. Registered form

The Warrant is in registered form. In accordance with applicable law, the Warrant is recorded in a warrant register book, which is kept at the registered office of the Company. The Warrant cannot be converted into a bearer instrument or in dematerialized form.

2.3. Transferability of the Warrant

The Warrant shall be transferrable in the same manner and in accordance with the same rules as those that apply (mutatis mutandis) to the Subscription Shares pursuant to the Subscription Agreement. Transfers of the Warrant that do not comply with this Article 2.3 are not enforceable vis-à-vis the Company.

2.4. No listing of the Warrant

The Warrant shall not be listed at any time on a securities exchange, regulated market or similar securities market.

3. TERM OF THE WARRANT

- (a) The Warrant has a term (the "**Term**") starting as of the Issue Date and ending on 11:59 p.m. on the date which falls one (1) year after the Issue Date (the "**Expiry Date**").
- (b) The Warrant automatically lapses and becomes invalid (vervallen) by operation of law on 11:59 p.m. of the Expiry Date.
- (c) Subject to and in accordance with the terms and conditions set forth in these Conditions, the Warrant can be exercised at one or several occasions at any time during the Term.

4. SHARES

4.1. Number of Shares issuable upon an exercise of the Warrant

- (a) Subject to the terms and conditions set forth in these Conditions, the Warrant entitles the holder thereof to subscribe, during the entire Term of the Warrant, upon each exercise of the Warrant, for a maximum number of Shares (the "Warrant Limit") that is, in the aggregate with respect to each exercise of the Warrant, sufficient to bring the number of Shares owned by the Investor, the Parent Investor and any of their Affiliates to 25.1% of the actually issued and outstanding Shares immediately after the issue of the Shares that are to be issued upon the relevant exercise of the Warrant (rounded down to the nearest whole Share).
- (b) Notwithstanding the provisions of paragraph (a), if and as long as the person or entity exercising the Warrant is not the Investor, the Parent Investor or an Affiliate of the Investor or Parent Investor, the Warrant Limit shall be equal to one (1) Share. The provisions of paragraph (a) of this Article 4.1 will again apply when the person or entity exercising the Warrant is the Investor, the Parent Investor or any Affiliate of the Investor.
- (c) The Warrant can be exercised at one or several occasions during the entire Term, but not more than once per period of three (3) months; provided that such limitation on the frequency of exercising the Warrant shall not apply within a given three (3) month period in which there has been already an exercise of the Warrant to the extent that within such three (3) month period there has been a material development regarding the Company or the trading of Shares or if the Company or any other person or entity (other than the Investor, the Parent Investor, any of the Affiliates of the Investor or Parent Investor, or any party Acting in Concert with the Investor, the Parent Investor or any Affiliate of the Investor or the Parent Investor) provides notice that it intends to convene, requests to convene, or convenes, a meeting of shareholders. On each occasion the Warrant is exercised, the number of Shares that the holder of the Warrant will be entitled to subscribe for, will, in the aggregate with respect to such exercise of the Warrant, be limited to the then applicable Warrant Limit. The Warrant shall remain outstanding for the remaining duration of the Term even if exercised for a number of Shares that is equal to the then applicable Warrant Limit.
- (d) The Warrant can only be exercised for a whole number of Shares, and not with respect to fractions of Shares.

4.2. Nature and form of the Shares

- (a) Each new Share to be issued by the Company upon each exercise of the Warrant shall have the same rights and benefits as, and rank pari passu in all respects including as to entitlement to dividends and other distributions, with the existing and outstanding Shares at the moment of their issue and will be entitled to dividends and other distributions in respect of which the relevant record date or due date falls on or after the date of their issue.
- (b) Notwithstanding any other provision of these Conditions, upon each exercise of the Warrant, the Company shall have the right, to be determined by it in its discretion, to deliver a number of existing Shares in lieu of (all or a portion of) the new Shares that would otherwise need to be issued upon such exercise of the Warrant, <u>provided</u> that (i) such existing Shares confer the rights referred to in paragraph (a) and (ii) the delivery of such number of existing Shares (together with any new Shares issued upon such exercise of the Warrant) results in the Investor, the Parent Investor and any of their Affiliates owning the same percentage of Shares of the actually issued and outstanding Shares of the Company immediately after the issue of the Shares that had to be issued upon such exercise of the Warrant, rounded down to the nearest whole Share, that they would otherwise own if only new Shares were to have been issued upon such exercise of the Warrant. The holder of the Warrant cannot be obliged to pay a price per Share for the delivery of existing Shares that is higher than the applicable Exercise Price.
- (c) The Shares to be delivered upon each exercise of the Warrant shall be delivered in registered form.

4.3. Listing of the Shares

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) legally requires a listing prospectus, the Company shall use reasonable efforts to obtain such admission within ninety (90) days following the issue of such Shares. In such event, the effective admission to listing will be subject to regulatory approval of the listing prospectus.

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) does not legally require a listing prospectus, the Company shall cause such admission as soon as practicable after the issue of such Shares, and in any event no later than five (5) Business Days after the issue of such Shares.

5. EXERCISE PRICE

The Exercise Price of the Warrant shall, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, be equal to the Reference Exercise Price.

6. SPECIFIC CONDITIONS

(a) Upon subscription for the Warrant and when exercising the Warrant, the holder of the Warrant will comply with (i) the Belgian Act of 2 May 2007 regarding the disclosure of important participations in issuers of which shares are admitted to trading on a regulated market and miscellaneous provisions as amended from time to time, and the rules and regulations promulgated thereunder, and (ii) insider trading and/or dealings or transactions in the securities of the Company, including notably Regulation (EU) No 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse (market abuse regulation) and repealing Directive 2003/6/EC of the European Parliament and of the Council and Commission Directives 2003/124/EC, 2003/125/EC and 2004/72/EC, and the relevant regulations, directives and other rules promulgated thereunder, as well as similar rules and regulations elsewhere in other jurisdictions.

(b) During the Term, the Investor shall, upon written request from the Company, provide the Company within five (5) Business Days, the aggregate number of Equity Securities owned by the Investor, the Parent Investor and their Affiliates to the extent calculable (it being understood that the Warrants can be described without calculating a number of shares issuable thereunder). During the Term, the Company shall provide the Investor with notice of any exercise or conversion of any options, warrants, convertible securities or other right to acquire, directly or indirectly, any of the new shares of the Company by any person or entity (other than the Investor, the Parent Investor and their Affiliates) within five (5) Business Days after such exercise or conversion.

7. EXERCISE OF THE WARRANT

7.1. Exercise Notice

- (a) The Warrant can only be exercised by means of a written notice to the Company (the "**Exercise Notice**"). The Exercise Notice must be served on the Company in accordance with the provisions of Article 10.6.
- (b) The date on which the Exercise Notice with respect to a specific exercise of the Warrant shall have been served (or be deemed served) on the Company pursuant to Article 10.6 shall be the exercise date of the Warrant with respect to that exercise (the "Exercise Date").
- (c) The Exercise Notice must state (i) the number of Shares with respect to which the Warrant is exercised at that occasion, and (ii) the applicable aggregate Exercise Price, as determined in accordance with the provisions of these Conditions. The Exercise Date should fall within the Term.
- (d) If the number of Shares with respect to which the Warrant is exercised as indicated in the Exercise Notice exceeds the then applicable Warrant Limit, the Warrant shall be deemed exercised with respect to the number of Shares just below the then applicable Warrant Limit only. The Company shall as soon as practical, and in any event before the expiry of the term within which the relevant Shares are to be issued to the holder of the Warrant pursuant to Article 7.3 of these Conditions, notify the holder of the Warrant thereof in accordance with the provisions of Article 10.6, which notification shall include the reasoned calculation of the then applicable Warrant Limit.
- (e) Upon receipt of the Exercise Notice, the Company may request the holder of the Warrant in writing to provide to the Company with such further declarations and documents, which are necessary to comply with all applicable legal and regulatory provisions in connection with the exercise of the Warrant and the issue or delivery of the Shares resulting therefrom, including pursuant to Article 6(b) of these Conditions.

7.2. Payment of the Exercise Price

- (a) Upon each exercise of the Warrant, the applicable aggregate Exercise Price must be paid by means of a payment in cash.
- (b) The aggregate Exercise Price shall be paid to the Euro-denominated blocked account in the Company's name with KBC Bank NV ([***]) (the "Exercise Account").
- (c) The amount of the applicable aggregate Exercise Price must be paid by means of a wire transfer of such amount in immediately available funds in euro to the Exercise Account.
- (d) If the applicable aggregate Exercise Price is not paid in accordance with paragraph (c) within a term of ten (10) Business Days following the date on which the Company shall have notified the holder of the Warrant of the details of the Exercise Account in accordance with paragraph (b), the Warrant shall be deemed not to have been exercised, without prejudice to the right of the holder of the Warrant to exercise the Warrant at later occasions until the Expiry Date subject to and in accordance with the terms and conditions set forth in these Conditions.

7.3. Issue and delivery of the Shares

- (a) The Company shall only be obliged to issue Shares upon an exercise of the Warrant provided that (i) the relevant Exercise Notice has been made in accordance with Article 7.1, and (ii) the applicable aggregate Exercise Price has been paid in accordance with the provisions of Article 7.2. Subject to the foregoing, the Company shall issue or deliver the relevant Shares as soon as practicable, but in any event no later than the later of (x) six (6) calendar days after the Exercise Date, and (y) the day other than (a) a Saturday or a Sunday or (b) a bank or other public holiday in Belgium following the receipt of the applicable aggregate Exercise Price on the Exercise Account.
- (b) The Company shall take all steps and carry out all formalities that shall be required by virtue of these Conditions, the Company's articles of association and applicable law in order to issue the new Shares upon an exercise of the Warrant (without prejudice, however, to the right of the Company to deliver existing Shares in accordance with the provisions of Article 4.2(b)).
- (c) In accordance with applicable law, upon each exercise of the Warrant, the capital increase and issue of new Shares resulting therefrom (as relevant) shall be formally recorded before a notary public by one authorised representative of the Company.

7.4. Allocation of the Exercise Price

- (a) Each time upon an exercise of the Warrant and the issue of new Shares pursuant to these Conditions, the applicable aggregate Exercise Price shall be allocated to the share capital of the Company. If the applicable Exercise Price per Share issued is greater than the fractional value of the existing Shares immediately prior to the capital increase, then the applicable aggregate Exercise Price shall be allocated in such a manner that per Share issued (i) a part of the applicable aggregate Exercise Price equal to the fractional value of the existing Shares immediately prior to the capital increase shall be booked as share capital, and (ii) the balance of the applicable aggregate Exercise Price shall be booked as issue premium. Such issue premium shall be accounted for on the liabilities side of the Company's balance sheet as net equity. The account on which the issue premium shall be booked shall, like the share capital, serve as the guarantee for third parties and, save for the possibility of a capitalisation of those reserves, can only be reduced on the basis of a valid resolution of the general shareholders' meeting passed in the manner required for an amendment to the Company's articles of association.
- (b) Following the issue of new Shares and the capital increase resulting therefrom, each of the Shares (existing and new) shall represent the same fraction of the Company's share capital.

8. ADJUSTMENTS

8.1. General

(a) Notwithstanding Article 501, paragraph 1 of the Belgian Companies Code (or its successor provision Article 7:71, §1 under the Belgian Companies and Associations Code), and with the exception of the issue of any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde), of any securities with voting rights other than Shares or of any Equity Securities in relation to any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde) or any securities with voting rights other than Shares and any reclassification of existing Shares (which shall not be permitted pursuant to this Article 8.1(a), the Company may proceed with all actions that it deems appropriate in relation to its capital, its articles of association, its financial condition, even if such actions lead to a reduction of the benefits allocated to the Warrant, including but not limited to, mergers or acquisitions, capital increases or reductions

(including those subject to conditions precedent), the incorporation of reserves into the capital with or without the issue of new Shares, the issue of dividends or other distributions, the issue of other Equity Securities and the amendment of arrangements or provisions relating to the distribution of profits or liquidation proceeds, provided, however, that (i) the terms of the Warrant may not be amended without Investor's written consent, and (ii) that Shares issued or issuable under the Warrant shall not be treated differently (had they already been issued at that time) than other Shares already issued. If the rights of the holder of the Warrant are affected by an action or transaction permitted by the immediately preceding sentence, the holder of the Warrant will not be entitled to a change of the Exercise Price, the Reference Exercise Price or the Warrant Limit, an amendment to the Conditions or any other form of compensation (financial or otherwise) unless (i) specifically provided for in Articles 8.2 and 8.3 of these Conditions and/or (ii) such action or transaction was undertaken with the primary purpose of adversely affecting the rights or value of the Warrant.

(b) The provisions of this Article 8 are without prejudice to the provisions of the Subscription Agreement (such as the anti-dilution protection).

8.2. Adjustments for Share Reorganisations

- (a) If at any time as of the Issue Date up to the Expiry Date there is a change of the fractional value of the Shares as a result of a consolidation (or reverse stock split), a subdivision (or stock split) or otherwise, in each such case without increase or reduction of the Company's share capital (each such transaction a "Share Reorganisation"), the Warrant Limit shall not be affected (since it is expressed as a percentage of the actually issued and outstanding Shares). In addition, the Exercise Price of the Warrant (i.e. on a per Share basis) shall be divided by a fraction, (A) the numerator of which is equal to the fractional value of the outstanding Shares of the Company immediately before to the Share Reorganisation, and (B) the denominator of which is equal to the fractional value of the outstanding Shares of the Company immediately after the Share Reorganisation.
- (b) Any adjustment made pursuant to paragraph (a) of this Article 8.2 shall become effective immediately after the effective date of the relevant Share Reorganisation that gives rise to such adjustment. The Company shall notify the holder of the Warrant of such adjustment by written notice as soon as practicable after the effective date of the Share Reorganisation concerned.

8.3. Adjustments for mergers and de-mergers

(a) If at any time as of the Issue Date up to the Expiry Date there is a merger (fusie / fusion) of the Company with or into another legal person or entity whereby the Company is not the surviving entity or a de-merger (splitsing / scission) in whole or in part, whereby in each of these cases the Shares of the Company are exchanged into, or the shareholders of the Company receive, shares, other securities, cash or other property of one or more other legal persons or entities (each such legal person or entity a "Successor Company") as a result of such merger or de-merger (each such transaction a "Succession Transaction"), then the Warrant shall be replaced by (or, if the Company survives, to the holder of the Warrant shall be issued) one or more warrants to be issued by each of the respective Successor Companies that give right to such respective shares, other securities, cash or other property that the holder of the Warrant concerned had been entitled to receive if the Warrant had been exercised in full up to the Warrant Limit immediately before to the occurrence of the Succession Transaction concerned but after giving effect to dilution upon the exercise or conversion of all rights to acquire Shares that are exercised or converted in connection with such transaction. The number of warrants to be so issued in replacement of the Warrant (or, if the Company survives, to be so issued), as well as the terms and conditions of such warrants (including the Exercise Price and the Warrant Limit) will need to be, as a whole, equivalent to the terms and conditions of the Warrant and have economically substantially the same effect for the holder of the Warrant concerned.

- (b) An adjustment made pursuant to paragraph (a) of this Article 8.3 shall become effective upon, and subject to, the completion of the Succession Transaction that gives rise to such adjustment. The Company (or the respective Successor Companies, as the case may be) shall notify the holder of the Warrant of such adjustment as soon as practicable after the completion of the Succession Transaction concerned.
- (c) In the case of any merger or de-merger of the Company as contemplated by paragraph (a) of this Article 8.3, the Company must procure that the successor or acquiring persons or entities shall expressly assume the due observance and performance of the covenants and obligations set out in the Conditions.

9. REPRESENTATIONS AND WARRANTIES

9.1. Representations and Warranties of the Company

Upon each exercise of the Warrant, the Company shall be deemed to represent and warrant to the holder of the Warrant on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of Belgium, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Validity of Shares and Absence of Breach. Each Share to be issued or delivered by the Company upon such exercise of the Warrant, and, with respect to new Shares only, as of when issued and paid for in accordance with these Conditions, are validly and duly issued and fully paid ordinary shares of the Company in accordance with the applicable provisions of the Company's organisational documents and Belgian law, having the same fractional value (fractiewaarde) as the then existing Shares, and free and clear of all liens, pledges, encumbrances, mortgages, security interests, or easement or transfer restrictions of any nature whatsoever (other than those that find their origin solely with the holder of the Warrant and save for the transfer restrictions referred to in the Subscription Agreement, as the case may be). The issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant will not result in a breach of, default under any material agreement to which the Company is a party or the Company's organisational documents or any law, regulation or stock exchange rule, or give rise to the activation of any material rights of third parties under any agreement, law, rule or regulation binding on the Company or any of its subsidiaries.
- (c) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under Belgian law or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Company with the terms of these Conditions and the Warrant will, save as otherwise set forth in these Conditions in Article 4.3, be in full force and effect.
- (d) Brokers and Finders. No person will have, as a result of the exercise of the Warrant, any right, interest or claim against or upon the holder of the Warrant for any commission, fee or other compensation relating to the Shares to be issued or delivered by the Company upon such exercise of the Warrant.

9.2. Representations and Warranties of the holder of the Warrant

Upon each exercise of the Warrant, the holder of the Warrant shall be deemed to represent and warrant to the Company on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of its jurisdiction of incorporation, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under the law applicable to its jurisdiction of organisation or incorporation, or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Warrant holder with the terms of these Conditions and the Warrant will, save as otherwise set forth in those Conditions in Article 4.3, be in full force and effect.
- (c) Information. Without taking into account the Warrants or the shares issuable (but not yet issued) thereunder, except to give effect to the issue and delivery of the Shares to be issued or delivered by the Company to the Investor upon the particular exercise of the Warrant that occurs on the date that this representation and warranty is made, the Investor, the Parent Investor, and their respective Affiliates do not, directly or indirectly, own, or have the right to acquire, voting securities of the Company in excess of the Warrant Limit (assuming the exercise, conversion or exchange of any Equity Securities (other than the Warrants except as described in this Article 9.2(c)) held by any of them at that time that are exercisable, convertible or exchangeable into or for shares of the Company at that time) (the resulting number of securities rounded down).

10. MISCELLANEOUS

10.1. Binding nature of the Conditions

In the case of subscription for the Warrant, the subscriber shall be bound by, and deemed to have accepted, the present Conditions. In the event of a transfer of the Warrant (or any right thereto), the acquirer or transferee shall be bound by, and deemed to have accepted, the present Conditions.

10.2. Severability

Whenever possible, the provisions of the Conditions shall be interpreted in such a manner that they are valid and enforceable under the applicable legislation.

If any provision in these Conditions is held to be illegal, invalid or unenforceable, in whole or in part, under any applicable law, then such provision or part of it shall be deemed not to form part of these Conditions, and the legality, validity or enforceability of the remainder of these Conditions shall not be affected.

In that event, the illegal, invalid or non-enforceable provision or part thereof is automatically replaced with the legal, valid and enforceable provision that is the closest to the original provision or part thereof as regards content, bearing and intention.

10.3. Specific Enforcement

Notwithstanding anything in these Conditions to the contrary, nothing in these Conditions shall in any way limit the ability of the Company and the holder of the Warrant to seek or obtain from any court of competent jurisdiction any remedies available at law or in equity (including injunctive relief) to enforce any covenant or agreement of the holder of the Warrant respectively the Company hereunder.

10.4. Costs and expenses

The Company shall pay any stamp, issue, registration, documentary or other taxes and duties, including interest and penalties, payable in Belgium on or in connection with the issue or delivery of the Shares upon each exercise of the Warrant. The Company shall also pay all costs associated with the listing of the relevant Shares on the regulated markets of Euronext Brussels and Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time).

10.5. Governing law and jurisdiction

The Conditions and the Warrant and any non-contractual obligations arising out of or in connection with each of them are governed by, and are to be construed in accordance with, Belgian law.

Any dispute, controversy, difference or claim which may arise between the Company and the holder of the Warrant out of or in relation to or in connection with the Conditions or the Warrant (including arising out of or relating to the validity, construction, interpretation, enforceability, breach, performance, application, exercise, expiry or termination of the Conditions or the Warrant), shall be settled by binding arbitration in accordance with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators, one chosen by the holder of the Warrant, one chosen by the Company and the third chosen by mutual agreement of the first two, and otherwise in accordance with the ICC Rules. The arbitrators shall have significant experience and shall have expertise in Belgian corporate law. Each of the Company and the holder of the Warrant may refer such dispute to arbitration by submitting a written notice of such request to the holder of the Warrant respectively the Company. The place of arbitration shall be New York and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither the Company nor the holder of the Warrant shall be required to give general discovery of documents, but may be required only to produce specific, identified documents that are relevant to the dispute. Each of the Company and the holder of the Warrant shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Company and the holder of the Warrant, and shall be governed by the terms and conditions hereof and the limitation on damages set forth in Article 13 of the Subscription Agreement. The parties hereto agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of Belgian law applicable to the commencement of a lawsuit shall apply to the commencement of arbitration. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each of the Company and the holder of the Warrant. All proceedings and decisions of the arbitrators shall be deemed Confidential Information (as set out in the Subscription Agreement) of each of the Company and the holder of the Warrant, and shall be subject to Article 13 of the Option, License and Collaboration Agreement.

10.6. Notices

Any notice, notification, demand or other communication ("**notice**") to be given under these Conditions shall be in writing, shall specifically refer to these Conditions, and shall be addressed to the appropriate party at the address specified below or such other address as may be specified by such party in writing in accordance with this Article 10.6, and shall be deemed to have been given for all purposes (i) when received, if handdelivered, sent by a reputable international expedited delivery service or sent by facsimile (with transmission confirmed), or (ii) five (5) Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. The current details for notices are:

(a) if to the Company: the address of the Company's registered office, with the notice made for the attention of the General Counsel of the Company, or the address for notices to the Company pursuant to the Subscription Agreement.



(b) if to the holder of the Warrant: to such holder's address as set out in the warrant register book, or the address for notices to such party (as the case may be) pursuant to the Subscription Agreement.

* * *

INITIAL GILEAD WARRANT B

-issued by-

GALAPAGOS NV

-dated-

[date] 2019

TABLE OF CONTENT

Article 1.	s	Certain Definitions and Interpretation	Page 2
	1.1. 1.2. 1.3. 1.4. 1.5.	Certain definitions Headings Meaning of references Fractional value of Shares Language	2 4 4 5 5
2.		Issuance, Nature and Form of the Warrant	5
	2.1. 2.2. 2.3. 2.4.	Issuance and nature Registered form Transferability of the Warrant No listing of the Warrant	5 5 6 6
3.		Term of the Warrant	6
4.		Shares	6
	4.1. 4.2. 4.3.	Number of Shares issuable upon an exercise of the Warrant Nature and form of the Shares Listing of the Shares	6 7 7
5.		Exercise Price	7
6.		Specific Conditions	7
7.		Exercise of the Warrant	8
	7.1. 7.2. 7.3. 7.4.	Exercise Notice Payment of the Exercise Price Issue and delivery of the Shares Allocation of the Exercise Price	8 8 9 9
8.		Adjustments	9
	8.1. 8.2. 8.3.	General Adjustments for Share Reorganisations Adjustments for mergers and de-mergers	9 10 10
9.		Representations and Warranties	11
	9.1. 9.2.	Representations and Warranties of the Company Representations and Warranties of the holder of the Warrant	11 12
10.		Miscellaneous	12
	10.1. 10.2. 10.3. 10.4. 10.5. 10.6.	Binding nature of the Conditions Severability Specific Enforcement Costs and expenses Governing law and jurisdiction Notices	12 12 13 13 13 13 13

i

GALAPAGOS NV

Limited Liability Company

Generaal De Wittelaan L11 A3, 2800 Mechelen (Belgium) Register of Legal Persons VAT BE 0466.460.429 (Antwerp, division Mechelen)

INITIAL GILEAD WARRANT B

RECITALS

On 14 July 2019, Galapagos NV (hereafter further referred to as the "**Company**"), entered into a Subscription Agreement (as defined below) with Gilead Therapeutics A1 Unlimited Company (hereafter further referred to as the "**Investor**"). The Investor is an indirect wholly-owned subsidiary of Gilead Sciences, Inc. (hereafter further referred to as the "**Parent Investor**"), a U.S. corporation listed on the NASDAQ Stock Market and a researchbased biopharmaceutical company focused on the discovery, development, and commercialisation of innovative medicines. Simultaneously with the execution of the Subscription Agreement, the Company and the Parent Investor also entered into an Option, License and Collaboration Agreement (as defined below). Pursuant to the Option, License and Collaboration Agreement, the Company agreed to discover, research, and develop molecules and products, and Parent Investor agreed to have an option to participate in the development and commercialisation of molecules and products, in each case, on the terms and conditions set forth in such agreement. Pursuant to the Subscription Agreement, the Investor agreed to make an investment into the share capital of the Company. The investment was effected on [*date*]. As part of the overall agreement between the Company, the Investor and the Parent Investor, the Subscription Agreement also provided for the issuance to the Investor of a number of warrants. The present terms and conditions (hereinafter referred to as the "**Conditions**") contain the issue and exercise conditions of the Initial Gilead Warrant B issued by the Company, as contemplated by the Subscription Agreement. The issue of the Warrants is exclusively reserved to the Investor, in consideration of the subscription for shares and entry into the collaboration contemplated by the agreements referred to above.

1. CERTAIN DEFINITIONS AND INTERPRETATION

1.1. Certain definitions

In these Conditions, the following words and expressions that are not defined elsewhere in these Conditions shall have the following meanings, save where the context requires otherwise:

"Acting in Concert" means, when used in relation to a person or entity, acting in concert (*in onderling overleg handelende personen / personnes agissant de concert*) in the sense of Article 3, §1, 5° of the Belgian Act of 1 April 2007 regarding public takeover bids, or Article 1, §2, 5° of the Belgian Royal Decree of 27 April 2007 regarding public takeover bids.

"Affiliate" means, when used with respect to a person or entity, any person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such person or entity, for so long as such control exists, regardless of whether such person or entity is or becomes an Affiliate on or after the date of the Subscription Agreement. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, or by contract or otherwise.

"Belgian Companies and Associations Code" means the Belgian Companies and Associations Code of 23 March 2019, as amended from time to time, and the rules and regulations promulgated thereunder.

"Belgian Companies Code" means the Belgian Companies Code of 7 May 1999, as amended from time to time, and the rules and regulations promulgated thereunder.

"**Business Day**" means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States, (c) a bank or other public holiday in Brussels, Belgium, (d) a bank or other public holiday in Ireland or (e) the period commencing on December 25th and ending on January 1st (inclusive).

"**Company**" means Galapagos NV/SA, a corporation (*naamloze vennootschap / société anonyme*) organized and existing under the laws of Belgium, with registered office at Generaal De Wittelaan L11 A3, 2800 Mechelen, Belgium, and registered with the Register of Legal Persons under enterprise number 0466.460.429 (Antwerp, division Mechelen).

"Conditions" means the present terms and conditions of the Warrants.

"Equity Security" means (a) any Share, and (b) any other security, financial instrument, certificate and other right (including options, futures, swaps and other derivatives) issued or, with respect to options, futures, swaps and other derivatives, contracted by the Company and representing, being exercisable, convertible or exchangeable into or for, or otherwise providing a right to acquire, directly or indirectly, any of the Equity Securities referred to in (a).

"Exercise Account" has the meaning as defined in Article 7.2(b).

"Exercise Date" has the meaning as defined in Article 7.1(b).

"Exercise Notice" has the meaning as defined in Article 7.1(a).

"Exercise Price" means the exercise price of the Warrant, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, as determined pursuant to Article 5.

"Expiry Date" has the meaning as defined in Article 3(a).

"Gilead Warrant" means each of the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"**Initial Gilead Warrant**" means each of the warrants (*inschrijvingsrechten* or *warrants / droits de souscription*) issued by the Company on the Issue Date to the Investor, and named, respectively, the "Initial Gilead Warrant A" and "Initial Gilead Warrant B".

"**Investor**" means Gilead Therapeutics A1 Unlimited Company, an unlimited liability company formed under the laws of Ireland, registered with Ireland's Companies Registration Office under number 615395.

"Issue Date" means the date on which the Warrant has been issued by the extraordinary general shareholders' meeting of the Company, *i.e.* [date].

"Notice" has the meaning given to it in Article 10.6.

"**Option, License and Collaboration Agreement**" means the option, license and collaboration agreement, dated 14 July 2019, by and between the Company and the Parent Investor.

"Parent Investor" means Gilead Sciences, Inc., a corporation incorporated under the laws of Delaware.

"**Reference Date**" means [*date*], being the date on which the Subscription Shares were issued by the Company and were subscribed for by the Investor pursuant to the Subscription Agreement.

"**Reference Exercise Price**" means a price, per Share subscribed for upon an exercise of the Warrant in relation to such Shares, that shall be equal to the greater of (i) 120% multiplied by the arithmetic mean of the daily volume weighted average trading price of the Company's Shares as traded on Euronext

Brussels and Euronext Amsterdam (or such other regulated markets on which the Company's Shares will be trading at that time) on each of the trading days during the period of 30 calendar days ending on the calendar day immediately preceding the date of the Exercise Notice with respect to such exercise, and (ii) EUR 140.59, being the issue price, on a per Share basis, that was paid by the Investor with respect to the Subscription Shares that were issued to the Investor on the Reference Date pursuant to the Subscription Agreement.

"Share" means any share (aandeel / action) outstanding from time to time representing the Company's share capital.

"Share Reorganisation" has the meaning given to it in Article 8.2(a).

"**Subscription Agreement**" means the subscription agreement, dated 14 July 2019, by and between the Company and the Investor, in relation to the subscription by the Investor for the Subscription Shares and the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"Subscription Shares" means the [6,828,985] Shares that were issued by the Company and subscribed for by the Investor on the Reference Date pursuant to the Subscription Agreement.

"Subsequent Gilead Warrant B" means the warrant (*inschrijvingsrecht* or *warrant / droit de souscription*), named "Subsequent Gilead Warrant B" to be issued by the Company to the Investor pursuant to the Subscription Agreement.

"Succession Transaction" has the meaning given to it in Article 8.3(a).

"**Successor Company**" has the meaning given to it in Article 8.3(a).

"Term" means the term of the Warrant as referred to in Article 3.

"Warrant" means the Initial Gilead Warrant, named "Initial Gilead Warrant B".

"Warrant Limit" has the meaning given to it in Article 4.1(a).

1.2. Headings

Headings and the table of contents used in these Conditions are for convenience purposes only and shall not affect the construction or interpretation of these Conditions.

1.3. Meaning of references

Unless the context does not so permit, or save where specifically indicated otherwise:

- (a) references to Articles are to Articles in these Conditions, and references to sub-Articles or paragraphs are to sub-Articles or paragraphs of the Article in which such references appear;
- (b) the words "herein", "hereof", "hereunder", "hereby", "hereto", "herewith" and words of similar import shall refer to these Conditions as a whole and not to any particular Article, paragraph or other subdivision;
- (c) references to the word "include" or "including" (or any similar term) are not to be construed as implying any limitation, and general words introduced by the word "other" (or any similar term) shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things;
- (d) any reference to "writing" or "written" includes any method of reproducing words or text in a legible and non-transitory form but, for the avoidance of doubt, shall not include e-mail;
- (e) references to any statute, regulation or statutory provision shall be deemed to include reference to any statute, regulation or statutory instrument which amends, extends,

consolidates or replaces the same (or shall have done so) and to any other regulation, statutory instrument or other subordinate legislation made thereunder or pursuant thereto, provided that no such reference shall include any amendment, extension or replacement of the same with retrospective effect;

(f) all periods of time set out herein shall be calculated from midnight to midnight local time in Brussels, Belgium. They shall start on the day following the day on which the event triggering the relevant period of time has occurred. The expiration date shall be included in the period of time. If the expiration date is not a Business Day, it shall be postponed until the next Business Day. Unless otherwise provided herein, all periods of time shall be calculated in calendar days. All periods of time consisting of a number of months (or years) shall be calculated from the day in the month (or year) when the triggering event has occurred until the eve of the same day in the following month(s) (or year(s)) ("van de zoveelste tot de dag vóór de zoveelste" / "de quantième à veille de quantième").

1.4. Fractional value of Shares

For the purpose of these Conditions, the fractional value (*fractiewaarde / pair comptable*) of the Company's Shares from time to time shall be determined as a fraction, (a) the numerator of which is the amount of the Company's share capital at that time, and (b) the denominator of which is the aggregate number of actually issued and outstanding Shares of the Company at that time.

1.5. Language

The Conditions have been prepared in English and a Dutch translation will be prepared. In the case of discrepancies between the English and the Dutch version, the English version shall prevail between the parties hereto to the fullest extent possible and permitted by Belgian law. Notwithstanding the foregoing, Belgian legal concepts which are expressed in English language terms, are to be interpreted in accordance with the Belgian legal terms to which they refer, and the use herein of Dutch and/or French words in these Conditions as translation for certain words or concepts shall be conclusive in the determination of the relevant legal concept under Belgian law of the words or concepts that are so translated herein.

2. ISSUANCE, NATURE AND FORM OF THE WARRANT

2.1. Issuance and nature

- (a) The Warrant has been issued, without any additional consideration being due by the Investor or any of its Affiliates, pursuant to a resolution of the extraordinary general shareholders' meeting of the Company held on the Issue Date, with dis-application of the statutory preferential subscription rights of the shareholders of the Company for the benefit of the Investor.
- (b) Subject to, and in accordance with, the terms and conditions set forth in these Conditions, the Warrant confers the right (but not the obligation) on the holder thereof to subscribe, upon any exercise of the Warrant, for a number of new Shares to be issued by the Company.
- (c) Except as otherwise provided under Belgian law, the holder of the Warrant is no shareholder of the Company solely by virtue of holding the Warrant, and therefore does not have the rights of a shareholder in relation to the Shares to be issued or delivered to the holder of the Warrant upon an exercise of the Warrant until the exercise of the Warrant and the issue or delivery of the relevant Shares.

2.2. Registered form

The Warrant is in registered form. In accordance with applicable law, the Warrant is recorded in a warrant register book, which is kept at the registered office of the Company. The Warrant cannot be converted into a bearer instrument or in dematerialized form.

2.3. Transferability of the Warrant

The Warrant shall be transferrable in the same manner and in accordance with the same rules as those that apply (mutatis mutandis) to the Subscription Shares pursuant to the Subscription Agreement. Transfers of the Warrant that do not comply with this Article 2.3 are not enforceable vis-à-vis the Company.

2.4. No listing of the Warrant

The Warrant shall not be listed at any time on a securities exchange, regulated market or similar securities market.

3. TERM OF THE WARRANT

- (a) The Warrant has a term (the "**Term**") starting as of the Issue Date and ending on 11:59 p.m. on the date which falls five (5) years after the Reference Date (the "**Expiry Date**").
- (b) The Warrant automatically lapses and becomes invalid (vervallen) by operation of law on 11:59 p.m. of the Expiry Date.
- (c) Subject to and in accordance with the terms and conditions set forth in these Conditions, the Warrant can be exercised at one or several occasions at any time during the Term.

4. SHARES

4.1. Number of Shares issuable upon an exercise of the Warrant

- (a) Subject to the terms and conditions set forth in these Conditions, the Warrant entitles the holder thereof to subscribe, during the entire Term of the Warrant, upon each exercise of the Warrant, for a maximum number of Shares (the "Warrant Limit") that is, in the aggregate with respect to each exercise of the Warrant, sufficient to bring the number of Shares owned by the Investor, the Parent Investor and any of their Affiliates and any other party Acting in Concert with the Investor, the Parent Investor or any of their Affiliates to 29.9% of the actually issued and outstanding Shares immediately after the issue of the Shares that are to be issued upon the relevant exercise of the Warrant (rounded down to the nearest whole Share).
- (b) Notwithstanding the provisions of paragraph (a), if and as long as the person or entity exercising the Warrant is not the Investor, the Parent Investor or an Affiliate of the Investor or Parent Investor, the Warrant Limit shall be equal to one (1) Share. The provisions of paragraph (a) of this Article 4.1 will again apply when the person or entity exercising the Warrant is the Investor, the Parent Investor or any Affiliate of the Investor.
- (c) The Warrant can be exercised at one or several occasions during the entire Term, but not more than once per period of three (3) months; provided that such limitation on the frequency of exercising the Warrant shall not apply within a given three (3) month period in which there has been already an exercise of the Warrant to the extent that within such three (3) month period there has been a material development regarding the Company or the trading of Shares or if the Company or any other person or entity (other than the Investor, the Parent Investor, any of the Affiliates of the Investor or Parent Investor, or any party Acting in Concert with the Investor, the Parent Investor or any Affiliate of the Investor or the Parent Investor) provides notice that it intends to convene, requests to convene, or convenes, a meeting of shareholders. On each occasion the Warrant is exercised, the number of Shares that the holder of the Warrant will be entitled to subscribe for, will, in the aggregate with respect to such exercise of the Warrant, be limited to the then applicable Warrant Limit. The Warrant shall remain outstanding for the remaining duration of the Term even if exercised for a number of Shares that is equal to the then applicable Warrant Limit.

(d) The Warrant can only be exercised for a whole number of Shares, and not with respect to fractions of Shares.

4.2. Nature and form of the Shares

- (a) Each new Share to be issued by the Company upon each exercise of the Warrant shall have the same rights and benefits as, and rank pari passu in all respects including as to entitlement to dividends and other distributions, with the existing and outstanding Shares at the moment of their issue and will be entitled to dividends and other distributions in respect of which the relevant record date or due date falls on or after the date of their issue.
- (b) Notwithstanding any other provision of these Conditions, upon each exercise of the Warrant, the Company shall have the right, to be determined by it in its discretion, to deliver a number of existing Shares in lieu of (all or a portion of) the new Shares that would otherwise need to be issued upon such exercise of the Warrant, <u>provided</u> that (i) such existing Shares confer the rights referred to in paragraph (a) and (ii) the delivery of such number of existing Shares (together with any new Shares issued upon such exercise of the Warrant) results in the Investor, the Parent Investor and any of their Affiliates and any other party Acting in Concert with the Investor, the Parent Investor or any of their Affiliates owning the same percentage of Shares of the actually issued and outstanding Shares of the Company immediately after the issue of the Shares that had to be issued upon such exercise of the Warrant, rounded down to the nearest whole Share, that they would otherwise own if only new Shares were to have been issued upon such exercise of the Warrant. The holder of the Warrant cannot be obliged to pay a price per Share for the delivery of existing Shares that is higher than the applicable Exercise Price.
- (c) The Shares to be delivered upon each exercise of the Warrant shall be delivered in registered form.

4.3. Listing of the Shares

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) legally requires a listing prospectus, the Company shall use reasonable efforts to obtain such admission within ninety (90) days following the issue of such Shares. In such event, the effective admission to listing will be subject to regulatory approval of the listing prospectus.

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) does not legally require a listing prospectus, the Company shall cause such admission as soon as practicable after the issue of such Shares, and in any event no later than five (5) Business Days after the issue of such Shares.

5. EXERCISE PRICE

The Exercise Price of the Warrant shall, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, be equal to the Reference Exercise Price.

6. SPECIFIC CONDITIONS

(a) Upon subscription for the Warrant and when exercising the Warrant, the holder of the Warrant will comply with (i) the Belgian Act of 2 May 2007 regarding the disclosure of important participations in issuers of which shares are admitted to trading on a regulated market and miscellaneous provisions as amended from time to time, and the rules and regulations promulgated thereunder, and (ii) insider trading and/or dealings or transactions in the securities of the Company, including notably Regulation (EU) No 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse (market abuse regulation) and

repealing Directive 2003/6/EC of the European Parliament and of the Council and Commission Directives 2003/124/EC, 2003/125/EC and 2004/72/EC, and the relevant regulations, directives and other rules promulgated thereunder, as well as similar rules and regulations elsewhere in other jurisdictions.

(b) During the Term, the Investor shall, upon written request from the Company, provide the Company within five (5) Business Days, the aggregate number of Equity Securities owned by the Investor, the Parent Investor and their Affiliates to the extent calculable (it being understood that the Warrants can be described without calculating a number of shares issuable thereunder). During the Term, the Company shall provide the Investor with notice of any exercise or conversion of any options, warrants, convertible securities or other right to acquire, directly or indirectly, any of the new shares of the Company by any person or entity (other than the Investor, the Parent Investor and their Affiliates) within five (5) Business Days after such exercise or conversion.

7. EXERCISE OF THE WARRANT

7.1. Exercise Notice

- (a) The Warrant can only be exercised by means of a written notice to the Company (the "**Exercise Notice**"). The Exercise Notice must be served on the Company in accordance with the provisions of Article 10.6.
- (b) The date on which the Exercise Notice with respect to a specific exercise of the Warrant shall have been served (or be deemed served) on the Company pursuant to Article 10.6 shall be the exercise date of the Warrant with respect to that exercise (the "Exercise Date").
- (c) The Exercise Notice must state (i) the number of Shares with respect to which the Warrant is exercised at that occasion, and (ii) the applicable aggregate Exercise Price, as determined in accordance with the provisions of these Conditions. The Exercise Date should fall within the Term.
- (d) If the number of Shares with respect to which the Warrant is exercised as indicated in the Exercise Notice exceeds the then applicable Warrant Limit, the Warrant shall be deemed exercised with respect to the number of Shares just below the then applicable Warrant Limit only. The Company shall as soon as practical, and in any event before the expiry of the term within which the relevant Shares are to be issued to the holder of the Warrant pursuant to Article 7.3 of these Conditions, notify the holder of the Warrant thereof in accordance with the provisions of Article 10.6, which notification shall include the reasoned calculation of the then applicable Warrant Limit.
- (e) Upon receipt of the Exercise Notice, the Company may request the holder of the Warrant in writing to provide to the Company with such further declarations and documents, which are necessary to comply with all applicable legal and regulatory provisions in connection with the exercise of the Warrant and the issue or delivery of the Shares resulting therefrom, including pursuant to Article 6(b) of these Conditions.

7.2. Payment of the Exercise Price

- (a) Upon each exercise of the Warrant, the applicable aggregate Exercise Price must be paid by means of a payment in cash.
- (b) Following the receipt of the Exercise Notice, the Company shall, as promptly as practicable and in any event no later than one (1) Business Day after the Exercise Date, notify the holder of the Warrant of the account onto which the applicable aggregate Exercise Price must be paid (the "Exercise Account").

- (c) The amount of the applicable aggregate Exercise Price must be paid by means of a wire transfer of such amount in immediately available funds in euro to the Exercise Account.
- (d) If the applicable aggregate Exercise Price is not paid in accordance with paragraph (c) within a term of ten (10) Business Days following the date on which the Company shall have notified the holder of the Warrant of the details of the Exercise Account in accordance with paragraph (b), the Warrant shall be deemed not to have been exercised, without prejudice to the right of the holder of the Warrant to exercise the Warrant at later occasions until the Expiry Date subject to and in accordance with the terms and conditions set forth in these Conditions.

7.3. Issue and delivery of the Shares

- (a) The Company shall only be obliged to issue Shares upon an exercise of the Warrant provided that (i) the relevant Exercise Notice has been made in accordance with Article 7.1, and (ii) the applicable aggregate Exercise Price has been paid in accordance with the provisions of Article 7.2. Subject to the foregoing, the Company shall issue or deliver the relevant Shares as soon as practicable, but in any event no later than the later of (x) six (6) calendar days after the Exercise Date, and (y) the day other than (a) a Saturday or a Sunday or (b) a bank or other public holiday in Belgium following the receipt of the applicable aggregate Exercise Price on the Exercise Account.
- (b) The Company shall take all steps and carry out all formalities that shall be required by virtue of these Conditions, the Company's articles of association and applicable law in order to issue the new Shares upon an exercise of the Warrant (without prejudice, however, to the right of the Company to deliver existing Shares in accordance with the provisions of Article 4.2(b)).
- (c) In accordance with applicable law, upon each exercise of the Warrant, the capital increase and issue of new Shares resulting therefrom (as relevant) shall be formally recorded before a notary public by one authorised representative of the Company.

7.4. Allocation of the Exercise Price

- (a) Each time upon an exercise of the Warrant and the issue of new Shares pursuant to these Conditions, the applicable aggregate Exercise Price shall be allocated to the share capital of the Company. If the applicable Exercise Price per Share issued is greater than the fractional value of the existing Shares immediately prior to the capital increase, then the applicable aggregate Exercise Price shall be allocated in such a manner that per Share issued (i) a part of the applicable aggregate Exercise Price equal to the fractional value of the existing Shares immediately prior to the capital increase shall be booked as share capital, and (ii) the balance of the applicable aggregate Exercise Price shall be booked as issue premium. Such issue premium shall be accounted for on the liabilities side of the Company's balance sheet as net equity. The account on which the issue premium shall be booked shall, like the share capital, serve as the guarantee for third parties and, save for the possibility of a capitalisation of those reserves, can only be reduced on the basis of a valid resolution of the general shareholders' meeting passed in the manner required for an amendment to the Company's articles of association.
- (b) Following the issue of new Shares and the capital increase resulting therefrom, each of the Shares (existing and new) shall represent the same fraction of the Company's share capital.

8. ADJUSTMENTS

8.1. General

(a) Notwithstanding Article 501, paragraph 1 of the Belgian Companies Code (or its successor provision Article 7:71, §1 under the Belgian Companies and Associations Code), and with the exception of the issue of any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde), of any securities with voting rights

other than Shares or of any Equity Securities in relation to any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde) or any securities with voting rights other than Shares and any reclassification of existing Shares (which shall not be permitted pursuant to this Article 8.1(a), the Company may proceed with all actions that it deems appropriate in relation to its capital, its articles of association, its financial condition, even if such actions lead to a reduction of the benefits allocated to the Warrant, including but not limited to, mergers or acquisitions, capital increases or reductions (including those subject to conditions precedent), the incorporation of reserves into the capital with or without the issue of new Shares, the issue of dividends or other distributions, the issue of other Equity Securities and the amendment of arrangements or provisions relating to the distribution of profits or liquidation proceeds, provided, however, that (i) the terms of the Warrant may not be amended without Investor's written consent, and (ii) that Shares issued or issuable under the Warrant shall not be treated differently (had they already been issued at that time) than other Shares already issued. If the rights of the holder of the Warrant are affected by an action or transaction permitted by the immediately preceding sentence, the holder of the Warrant will not be entitled to a change of the Exercise Price, the Reference Exercise Price or the Warrant Limit, an amendment to the Conditions or any other form of compensation (financial or otherwise) unless (i) specifically provided for in Articles 8.2 and 8.3 of these Conditions and/or (ii) such action or transaction was undertaken with the primary purpose of adversely affecting the rights or value of the Warrant.

(b) The provisions of this Article 8 are without prejudice to the provisions of the Subscription Agreement (such as the anti-dilution protection).

8.2. Adjustments for Share Reorganisations

- (a) If at any time as of the Issue Date up to the Expiry Date there is a change of the fractional value of the Shares as a result of a consolidation (or reverse stock split), a subdivision (or stock split) or otherwise, in each such case without increase or reduction of the Company's share capital (each such transaction a "Share Reorganisation"), the Warrant Limit shall not be affected (since it is expressed as a percentage of the actually issued and outstanding Shares). In addition, the Exercise Price of the Warrant (i.e. on a per Share basis) shall be divided by a fraction, (A) the numerator of which is equal to the fractional value of the outstanding Shares of the Company immediately before to the Share Reorganisation, and (B) the denominator of which is equal to the fractional value of the outstanding Shares of the Company immediately after the Share Reorganisation.
- (b) Any adjustment made pursuant to paragraph (a) of this Article 8.2 shall become effective immediately after the effective date of the relevant Share Reorganisation that gives rise to such adjustment. The Company shall notify the holder of the Warrant of such adjustment by written notice as soon as practicable after the effective date of the Share Reorganisation concerned.

8.3. Adjustments for mergers and de-mergers

(a) If at any time as of the Issue Date up to the Expiry Date there is a merger (fusie / fusion) of the Company with or into another legal person or entity whereby the Company is not the surviving entity or a de-merger (splitsing / scission) in whole or in part, whereby in each of these cases the Shares of the Company are exchanged into, or the shareholders of the Company receive, shares, other securities, cash or other property of one or more other legal persons or entities (each such legal person or entity a "Successor Company") as a result of such merger or de-merger (each such transaction a "Succession Transaction"), then the Warrant shall be replaced by (or, if the Company survives, to the holder of the Warrant shall be issued) one or more warrants to be issued by each of the respective Successor Companies that give right to such respective shares, other securities, cash or other property that the holder of the Warrant concerned had been entitled to receive if the Warrant had been exercised in full up to the

Warrant Limit immediately before to the occurrence of the Succession Transaction concerned but after giving effect to dilution upon the exercise or conversion of all rights to acquire Shares that are exercised or converted in connection with such transaction. The number of warrants to be so issued in replacement of the Warrant (or, if the Company survives, to be so issued), as well as the terms and conditions of such warrants (including the Exercise Price and the Warrant Limit) will need to be, as a whole, equivalent to the terms and conditions of the Warrant and have economically substantially the same effect for the holder of the Warrant concerned.

- (b) An adjustment made pursuant to paragraph (a) of this Article 8.3 shall become effective upon, and subject to, the completion of the Succession Transaction that gives rise to such adjustment. The Company (or the respective Successor Companies, as the case may be) shall notify the holder of the Warrant of such adjustment as soon as practicable after the completion of the Succession Transaction concerned.
- (c) In the case of any merger or de-merger of the Company as contemplated by paragraph (a) of this Article 8.3, the Company must procure that the successor or acquiring persons or entities shall expressly assume the due observance and performance of the covenants and obligations set out in the Conditions.

9. REPRESENTATIONS AND WARRANTIES

9.1. Representations and Warranties of the Company

Upon each exercise of the Warrant, the Company shall be deemed to represent and warrant to the holder of the Warrant on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of Belgium, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Validity of Shares and Absence of Breach. Each Share to be issued or delivered by the Company upon such exercise of the Warrant, and, with respect to new Shares only, as of when issued and paid for in accordance with these Conditions, are validly and duly issued and fully paid ordinary shares of the Company in accordance with the applicable provisions of the Company's organisational documents and Belgian law, having the same fractional value (fractiewaarde) as the then existing Shares, and free and clear of all liens, pledges, encumbrances, mortgages, security interests, or easement or transfer restrictions of any nature whatsoever (other than those that find their origin solely with the holder of the Warrant and save for the transfer restrictions referred to in the Subscription Agreement, as the case may be). The issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant will not result in a breach of, default under any material agreement to which the Company is a party or the Company's organisational documents or any law, regulation or stock exchange rule, or give rise to the activation of any material rights of third parties under any agreement, law, rule or regulation binding on the Company or any of its subsidiaries.
- (c) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under Belgian law or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Company with the terms of these Conditions and the Warrant will, save as otherwise set forth in these Conditions in Article 4.3, be in full force and effect.

(d) **Brokers and Finders**. No person will have, as a result of the exercise of the Warrant, any right, interest or claim against or upon the holder of the Warrant for any commission, fee or other compensation relating to the Shares to be issued or delivered by the Company upon such exercise of the Warrant.

9.2. Representations and Warranties of the holder of the Warrant

Upon each exercise of the Warrant, the holder of the Warrant shall be deemed to represent and warrant to the Company on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of its jurisdiction of incorporation, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under the law applicable to its jurisdiction of organisation or incorporation, or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Warrant holder with the terms of these Conditions and the Warrant will, save as otherwise set forth in those Conditions in Article 4.3, be in full force and effect.
- (c) Information. Without taking into account the Warrants or the shares issuable (but not yet issued) thereunder, except to give effect to the issue and delivery of the Shares to be issued or delivered by the Company to the Investor upon the particular exercise of the Warrant that occurs on the date that this representation and warranty is made, the Investor, the Parent Investor, and their respective Affiliates do not, directly or indirectly, own, or have the right to acquire, voting securities of the Company in excess of the Warrant Limit (assuming the exercise, conversion or exchange of any Equity Securities (other than the Warrants except as described in this Article 9.2(c)) held by any of them at that time that are exercisable, convertible or exchangeable into or for shares of the Company at that time) (the resulting number of securities rounded down).

10. MISCELLANEOUS

10.1. Binding nature of the Conditions

In the case of subscription for the Warrant, the subscriber shall be bound by, and deemed to have accepted, the present Conditions. In the event of a transfer of the Warrant (or any right thereto), the acquirer or transferee shall be bound by, and deemed to have accepted, the present Conditions.

10.2. Severability

Whenever possible, the provisions of the Conditions shall be interpreted in such a manner that they are valid and enforceable under the applicable legislation.

If any provision in these Conditions is held to be illegal, invalid or unenforceable, in whole or in part, under any applicable law, then such provision or part of it shall be deemed not to form part of these Conditions, and the legality, validity or enforceability of the remainder of these Conditions shall not be affected.

In that event, the illegal, invalid or non-enforceable provision or part thereof is automatically replaced with the legal, valid and enforceable provision that is the closest to the original provision or part thereof as regards content, bearing and intention.

10.3. Specific Enforcement

Notwithstanding anything in these Conditions to the contrary, nothing in these Conditions shall in any way limit the ability of the Company and the holder of the Warrant to seek or obtain from any court of competent jurisdiction any remedies available at law or in equity (including injunctive relief) to enforce any covenant or agreement of the holder of the Warrant respectively the Company hereunder.

10.4. Costs and expenses

The Company shall pay any stamp, issue, registration, documentary or other taxes and duties, including interest and penalties, payable in Belgium on or in connection with the issue or delivery of the Shares upon each exercise of the Warrant. The Company shall also pay all costs associated with the listing of the relevant Shares on the regulated markets of Euronext Brussels and Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time).

10.5. Governing law and jurisdiction

The Conditions and the Warrant and any non-contractual obligations arising out of or in connection with each of them are governed by, and are to be construed in accordance with, Belgian law.

Any dispute, controversy, difference or claim which may arise between the Company and the holder of the Warrant out of or in relation to or in connection with the Conditions or the Warrant (including arising out of or relating to the validity, construction, interpretation, enforceability, breach, performance, application, exercise, expiry or termination of the Conditions or the Warrant), shall be settled by binding arbitration in accordance with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators, one chosen by the holder of the Warrant, one chosen by the Company and the third chosen by mutual agreement of the first two, and otherwise in accordance with the ICC Rules. The arbitrators shall have significant experience and shall have expertise in Belgian corporate law. Each of the Company and the holder of the Warrant may refer such dispute to arbitration by submitting a written notice of such request to the holder of the Warrant respectively the Company. The place of arbitration shall be New York and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither the Company nor the holder of the Warrant shall be required to give general discovery of documents, but may be required only to produce specific, identified documents that are relevant to the dispute. Each of the Company and the holder of the Warrant shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Company and the holder of the Warrant, and shall be governed by the terms and conditions hereof and the limitation on damages set forth in Article 13 of the Subscription Agreement. The parties hereto agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of Belgian law applicable to the commencement of a lawsuit shall apply to the commencement of arbitration. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each of the Company and the holder of the Warrant. All proceedings and decisions of the arbitrators shall be deemed Confidential Information (as set out in the Subscription Agreement) of each of the Company and the holder of the Warrant, and shall be subject to Article 13 of the Option, License and Collaboration Agreement.

10.6. Notices

Any notice, notification, demand or other communication ("**notice**") to be given under these Conditions shall be in writing, shall specifically refer to these Conditions, and shall be addressed to the appropriate party at the address specified below or such other address as may be specified by such party in writing in accordance with this Article 10.6, and shall be deemed to have been given for all purposes (i) when received, if handdelivered, sent by a reputable international expedited delivery service or sent by facsimile (with transmission confirmed), or (ii) five (5) Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. The current details for notices are:

- (a) if to the Company: the address of the Company's registered office, with the notice made for the attention of the General Counsel of the Company, or the address for notices to the Company pursuant to the Subscription Agreement.
- (b) if to the holder of the Warrant: to such holder's address as set out in the warrant register book, or the address for notices to such party (as the case may be) pursuant to the Subscription Agreement.

* * 14

*

Exhibit B

Form of Subsequent Warrant

[Attached]

SUBSEQUENT GILEAD WARRANT B

-issued by-

GALAPAGOS NV

-dated-

[date] 2019

TABLE OF CONTENT

Articles 1.	5	Certain Definitions and Interpretation	Page 2
	1.1. 1.2. 1.3. 1.4. 1.5.	Certain definitions Headings Meaning of references Fractional value of Shares Language	2 4 4 5 5
2.		Issuance, Nature and Form of the Warrant	5
	2.1. 2.2. 2.3. 2.4.	Issuance and nature Registered form Transferability of the Warrant No listing of the Warrant	5 5 6 6
3.		Term of the Warrant	6
4.		Shares	6
	4.1. 4.2. 4.3.	Number of Shares issuable upon an exercise of the Warrant Nature and form of the Shares Listing of the Shares	6 7 7
5.		Exercise Price	7
6.		Specific Conditions	7
7.		Exercise of the Warrant	8
	7.1. 7.2. 7.3. 7.4.	Exercise Notice Payment of the Exercise Price Issue and delivery of the Shares Allocation of the Exercise Price	8 8 9 9
8.		Adjustments	9
1	8.1. 8.2. 8.3.	General Adjustments for Share Reorganisations Adjustments for mergers and de-mergers	9 10 10
9.		Representations and Warranties	11
	9.1. 9.2.	Representations and Warranties of the Company Representations and Warranties of the holder of the Warrant	11 12
10.		Miscellaneous	12
	10.1. 10.2. 10.3. 10.4. 10.5. 10.6.	Binding nature of the Conditions Severability Specific Enforcement Costs and expenses Governing law and jurisdiction Notices	12 12 13 13 13 13 13

i

GALAPAGOS NV

Limited Liability Company

Generaal De Wittelaan L11 A3, 2800 Mechelen (Belgium) Register of Legal Persons VAT BE 0466.460.429 (Antwerp, division Mechelen)

SUBSEQUENT GILEAD WARRANT B

RECITALS

On 14 July 2019, Galapagos NV (hereafter further referred to as the "**Company**"), entered into a Subscription Agreement (as defined below) with Gilead Therapeutics A1 Unlimited Company (hereafter further referred to as the "**Investor**"). The Investor is an indirect wholly-owned subsidiary of Gilead Sciences, Inc. (hereafter further referred to as the "**Parent Investor**"), a U.S. corporation listed on the NASDAQ Stock Market and a researchbased biopharmaceutical company focused on the discovery, development, and commercialisation of innovative medicines. Simultaneously with the execution of the Subscription Agreement, the Company and the Parent Investor also entered into an Option, License and Collaboration Agreement (as defined below). Pursuant to the Option, License and Collaboration Agreement, the Company agreed to discover, research, and develop molecules and products, and Parent Investor agreed to have an option to participate in the development and commercialisation of molecules and products, in each case, on the terms and conditions set forth in such agreement. Pursuant to the Subscription Agreement, the Investor agreed to make an investment into the share capital of the Company. The investment was effected on [*date*]. As part of the overall agreement between the Company, the Investor and the Parent Investor, the Subscription Agreement also provided for the issuance to the Investor of a number of warrants. The present terms and conditions (hereinafter referred to as the "**Conditions**") contain the issue and exercise conditions of the Subscription of the subscription for shares and entry into the collaboration contemplated by the agreements referred to above.

1. CERTAIN DEFINITIONS AND INTERPRETATION

1.1. Certain definitions

In these Conditions, the following words and expressions that are not defined elsewhere in these Conditions shall have the following meanings, save where the context requires otherwise:

"Acting in Concert" means, when used in relation to a person or entity, acting in concert (*in onderling overleg handelende personen / personnes agissant de concert*) in the sense of Article 3, §1, 5° of the Belgian Act of 1 April 2007 regarding public takeover bids, or Article 1, §2, 5° of the Belgian Royal Decree of 27 April 2007 regarding public takeover bids.

"Affiliate" means, when used with respect to a person or entity, any person, corporation, partnership, or other entity that controls, is controlled by or is under common control with such person or entity, for so long as such control exists, regardless of whether such person or entity is or becomes an Affiliate on or after the date of the Subscription Agreement. For the purposes of this definition, the word "control" (including, with correlative meaning, the terms "controlled by" or "under the common control with") means the actual power, either directly or indirectly through one or more intermediaries, to direct or cause the direction of the management and policies of such entity, whether by the ownership of more than fifty percent (50%) of the voting stock of such entity, or by contract or otherwise.

"Belgian Companies and Associations Code" means the Belgian Companies and Associations Code of 23 March 2019, as amended from time to time, and the rules and regulations promulgated thereunder.

"Belgian Companies Code" means the Belgian Companies Code of 7 May 1999, as amended from time to time, and the rules and regulations promulgated thereunder.

"**Business Day**" means a day other than (a) a Saturday or a Sunday, (b) a bank or other public holiday in California, United States, (c) a bank or other public holiday in Brussels, Belgium, (d) a bank or other public holiday in Ireland or (e) the period commencing on December 25th and ending on January 1st (inclusive).

"**Company**" means Galapagos NV/SA, a corporation (*naamloze vennootschap / société anonyme*) organized and existing under the laws of Belgium, with registered office at Generaal De Wittelaan L11 A3, 2800 Mechelen, Belgium, and registered with the Register of Legal Persons under enterprise number 0466.460.429 (Antwerp, division Mechelen).

"Conditions" means the present terms and conditions of the Warrants.

"Equity Security" means (a) any Share, and (b) any other security, financial instrument, certificate and other right (including options, futures, swaps and other derivatives) issued or, with respect to options, futures, swaps and other derivatives, contracted by the Company and representing, being exercisable, convertible or exchangeable into or for, or otherwise providing a right to acquire, directly or indirectly, any of the Equity Securities referred to in (a).

"Exercise Account" has the meaning as defined in Article 7.2(b).

"Exercise Date" has the meaning as defined in Article 7.1(b).

"Exercise Notice" has the meaning as defined in Article 7.1(a).

"Exercise Price" means the exercise price of the Warrant, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, as determined pursuant to Article 5.

"Expiry Date" has the meaning as defined in Article 3(a).

"Gilead Warrant" means each of the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"**Initial Gilead Warrant**" means each of the warrants (*inschrijvingsrechten* or *warrants / droits de souscription*) issued by the Company on the Issue Date to the Investor, and named, respectively, the "Initial Gilead Warrant A" and "Initial Gilead Warrant B".

"**Investor**" means Gilead Therapeutics A1 Unlimited Company, an unlimited liability company formed under the laws of Ireland, registered with Ireland's Companies Registration Office under number 615395.

"Issue Date" means the date on which the Warrant has been issued by the extraordinary general shareholders' meeting of the Company, *i.e.* [date].

"**Notice**" has the meaning given to it in Article 10.6.

"**Option, License and Collaboration Agreement**" means the option, license and collaboration agreement, dated 14 July 2019, by and between the Company and the Parent Investor.

"Parent Investor" means Gilead Sciences, Inc., a corporation incorporated under the laws of Delaware.

"**Reference Date**" means [*date*], being the date on which the Subscription Shares were issued by the Company and were subscribed for by the Investor pursuant to the Subscription Agreement.

"**Reference Exercise Price**" means a price, per Share subscribed for upon an exercise of the Warrant in relation to such Shares, that shall be equal to the greater of (i) 120% multiplied by the arithmetic mean of the daily volume weighted average trading price of the Company's Shares as traded on Euronext

Brussels and Euronext Amsterdam (or such other regulated markets on which the Company's Shares will be trading at that time) on each of the trading days during the period of 30 calendar days ending on the calendar day immediately preceding the date of the Exercise Notice with respect to such exercise, and (ii) EUR 140.59, being the issue price, on a per Share basis, that was paid by the Investor with respect to the Subscription Shares that were issued to the Investor on the Reference Date pursuant to the Subscription Agreement.

"Share" means any share (aandeel / action) outstanding from time to time representing the Company's share capital.

"Share Reorganisation" has the meaning given to it in Article 8.2(a).

"**Subscription Agreement**" means the subscription agreement, dated 14 July 2019, by and between the Company and the Investor, in relation to the subscription by the Investor for the Subscription Shares and the Initial Gilead Warrants and Subsequent Gilead Warrant B.

"Subscription Shares" means the [6,828,985] Shares that were issued by the Company and subscribed for by the Investor on the Reference Date pursuant to the Subscription Agreement.

"**Subsequent Gilead Warrant B**" means the warrant (*inschrijvingsrecht* or *warrant / droit de souscription*), named "Subsequent Gilead Warrant B" issued by the Company on the Issue Date to the Investor.

"Succession Transaction" has the meaning given to it in Article 8.3(a).

"Successor Company" has the meaning given to it in Article 8.3(a).

"Term" means the term of the Warrant as referred to in Article 3.

"Warrant" means the Subsequent Gilead Warrant B.

"Warrant Limit" has the meaning given to it in Article 4.1(a).

1.2. Headings

Headings and the table of contents used in these Conditions are for convenience purposes only and shall not affect the construction or interpretation of these Conditions.

1.3. Meaning of references

Unless the context does not so permit, or save where specifically indicated otherwise:

- (a) references to Articles are to Articles in these Conditions, and references to sub-Articles or paragraphs are to sub-Articles or paragraphs of the Article in which such references appear;
- (b) the words "herein", "hereof", "hereunder", "hereby", "hereto", "herewith" and words of similar import shall refer to these Conditions as a whole and not to any particular Article, paragraph or other subdivision;
- (c) references to the word "include" or "including" (or any similar term) are not to be construed as implying any limitation, and general words introduced by the word "other" (or any similar term) shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things;
- (d) any reference to "writing" or "written" includes any method of reproducing words or text in a legible and non-transitory form but, for the avoidance of doubt, shall not include e-mail;
- (e) references to any statute, regulation or statutory provision shall be deemed to include reference to any statute, regulation or statutory instrument which amends, extends,

consolidates or replaces the same (or shall have done so) and to any other regulation, statutory instrument or other subordinate legislation made thereunder or pursuant thereto, provided that no such reference shall include any amendment, extension or replacement of the same with retrospective effect;

(f) all periods of time set out herein shall be calculated from midnight to midnight local time in Brussels, Belgium. They shall start on the day following the day on which the event triggering the relevant period of time has occurred. The expiration date shall be included in the period of time. If the expiration date is not a Business Day, it shall be postponed until the next Business Day. Unless otherwise provided herein, all periods of time shall be calculated in calendar days. All periods of time consisting of a number of months (or years) shall be calculated from the day in the month (or year) when the triggering event has occurred until the eve of the same day in the following month(s) (or year(s)) ("van de zoveelste tot de dag vóór de zoveelste" / "de quantième à veille de quantième").

1.4. Fractional value of Shares

For the purpose of these Conditions, the fractional value (fractiewaarde / pair comptable) of the Company's Shares from time to time shall be determined as a fraction, (a) the numerator of which is the amount of the Company's share capital at that time, and (b) the denominator of which is the aggregate number of actually issued and outstanding Shares of the Company at that time.

1.5. Language

The Conditions have been prepared in English and a Dutch translation will be prepared. In the case of discrepancies between the English and the Dutch version, the English version shall prevail between the parties hereto to the fullest extent possible and permitted by Belgian law. Notwithstanding the foregoing, Belgian legal concepts which are expressed in English language terms, are to be interpreted in accordance with the Belgian legal terms to which they refer, and the use herein of Dutch and/or French words in these Conditions as translation for certain words or concepts shall be conclusive in the determination of the relevant legal concept under Belgian law of the words or concepts that are so translated herein.

2. ISSUANCE, NATURE AND FORM OF THE WARRANT

2.1. Issuance and nature

- (a) The Warrant has been issued, without any additional consideration being due by the Investor or any of its Affiliates, pursuant to a resolution of the extraordinary general shareholders' meeting of the Company held on the Issue Date, with dis-application of the statutory preferential subscription rights of the shareholders of the Company for the benefit of the Investor.
- (b) Subject to, and in accordance with, the terms and conditions set forth in these Conditions, the Warrant confers the right (but not the obligation) on the holder thereof to subscribe, upon any exercise of the Warrant, for a number of new Shares to be issued by the Company.
- (c) Except as otherwise provided under Belgian law, the holder of the Warrant is no shareholder of the Company solely by virtue of holding the Warrant, and therefore does not have the rights of a shareholder in relation to the Shares to be issued or delivered to the holder of the Warrant upon an exercise of the Warrant until the exercise of the Warrant and the issue or delivery of the relevant Shares.

2.2. Registered form

The Warrant is in registered form. In accordance with applicable law, the Warrant is recorded in a warrant register book, which is kept at the registered office of the Company. The Warrant cannot be converted into a bearer instrument or in dematerialized form.

2.3. Transferability of the Warrant

The Warrant shall be transferrable in the same manner and in accordance with the same rules as those that apply (mutatis mutandis) to the Subscription Shares pursuant to the Subscription Agreement. Transfers of the Warrant that do not comply with this Article 2.3 are not enforceable vis-à-vis the Company.

2.4. No listing of the Warrant

The Warrant shall not be listed at any time on a securities exchange, regulated market or similar securities market.

3. TERM OF THE WARRANT

- (a) The Warrant has a term (the "**Term**") starting as of the Issue Date and ending on 11:59 p.m. of the day (the "**Expiry Date**") that is the earlier of five (5) years after (i) the fifth anniversary of the Reference Date and (ii) the Issue Date.
- (b) The Warrant automatically lapses and becomes invalid (vervallen) by operation of law on 11:59 p.m. of the Expiry Date.
- (c) Subject to and in accordance with the terms and conditions set forth in these Conditions, the Warrant can be exercised at one or several occasions at any time during the Term.

4. SHARES

4.1. Number of Shares issuable upon an exercise of the Warrant

- (a) Subject to the terms and conditions set forth in these Conditions, the Warrant entitles the holder thereof to subscribe, during the entire Term of the Warrant, upon each exercise of the Warrant, for a maximum number of Shares (the "Warrant Limit") that is, in the aggregate with respect to each exercise of the Warrant, sufficient to bring the number of Shares owned by the Investor, the Parent Investor and any of their Affiliates and any other party Acting in Concert with the Investor, the Parent Investor or any of their Affiliates to 29.9% of the actually issued and outstanding Shares immediately after the issue of the Shares that are to be issued upon the relevant exercise of the Warrant (rounded down to the nearest whole Share).
- (b) Notwithstanding the provisions of paragraph (a), if and as long as the person or entity exercising the Warrant is not the Investor, the Parent Investor or an Affiliate of the Investor or Parent Investor, the Warrant Limit shall be equal to one (1) Share. The provisions of paragraph (a) of this Article 4.1 will again apply when the person or entity exercising the Warrant is the Investor, the Parent Investor or any Affiliate of the Investor.
- (c) The Warrant can be exercised at one or several occasions during the entire Term, but not more than once per period of three (3) months; provided that such limitation on the frequency of exercising the Warrant shall not apply within a given three (3) month period in which there has been already an exercise of the Warrant to the extent that within such three (3) month period there has been a material development regarding the Company or the trading of Shares or if the Company or any other person or entity (other than the Investor, the Parent Investor, any of the Affiliates of the Investor or Parent Investor, or any party Acting in Concert with the Investor, the Parent Investor or the Parent Investor) provides notice that it intends to convene, requests to convene, or convenes, a meeting of shareholders. On each occasion the Warrant is exercised, the number of Shares that the holder of the Warrant will be entitled to subscribe for, will, in the aggregate with respect to such exercise of the Warrant, be limited to the then applicable Warrant Limit. The Warrant shall remain outstanding for the remaining duration of the Term even if exercised for a number of Shares that is equal to the then applicable Warrant Limit.

(d) The Warrant can only be exercised for a whole number of Shares, and not with respect to fractions of Shares.

4.2. Nature and form of the Shares

- (a) Each new Share to be issued by the Company upon each exercise of the Warrant shall have the same rights and benefits as, and rank pari passu in all respects including as to entitlement to dividends and other distributions, with the existing and outstanding Shares at the moment of their issue and will be entitled to dividends and other distributions in respect of which the relevant record date or due date falls on or after the date of their issue.
- (b) Notwithstanding any other provision of these Conditions, upon each exercise of the Warrant, the Company shall have the right, to be determined by it in its discretion, to deliver a number of existing Shares in lieu of (all or a portion of) the new Shares that would otherwise need to be issued upon such exercise of the Warrant, <u>provided</u> that (i) such existing Shares confer the rights referred to in paragraph (a) and (ii) the delivery of such number of existing Shares (together with any new Shares issued upon such exercise of the Warrant) results in the Investor, the Parent Investor and any of their Affiliates and any other party Acting in Concert with the Investor, the Parent Investor or any of their Affiliates owning the same percentage of Shares of the actually issued and outstanding Shares of the Company immediately after the issue of the Shares that had to be issued upon such exercise of the Warrant, rounded down to the nearest whole Share, that they would otherwise own if only new Shares were to have been issued upon such exercise of the Warrant. The holder of the Warrant cannot be obliged to pay a price per Share for the delivery of existing Shares that is higher than the applicable Exercise Price.
- (c) The Shares to be delivered upon each exercise of the Warrant shall be delivered in registered form.

4.3. Listing of the Shares

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) legally requires a listing prospectus, the Company shall use reasonable efforts to obtain such admission within ninety (90) days following the issue of such Shares. In such event, the effective admission to listing will be subject to regulatory approval of the listing prospectus.

If the admission of the Shares that are to be issued upon an exercise of the Warrant to trading on the regulated markets of Euronext Brussels and Euronext Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time) does not legally require a listing prospectus, the Company shall cause such admission as soon as practicable after the issue of such Shares, and in any event no later than five (5) Business Days after the issue of such Shares.

5. EXERCISE PRICE

The Exercise Price of the Warrant shall, per Share that shall be subscribed for upon an exercise of the Warrant in relation to such Shares, be equal to the Reference Exercise Price.

6. SPECIFIC CONDITIONS

(a) Upon subscription for the Warrant and when exercising the Warrant, the holder of the Warrant will comply with (i) the Belgian Act of 2 May 2007 regarding the disclosure of important participations in issuers of which shares are admitted to trading on a regulated market and miscellaneous provisions as amended from time to time, and the rules and regulations promulgated thereunder, and (ii) insider trading and/or dealings or transactions in the securities of the Company, including notably Regulation (EU) No 596/2014 of the European Parliament and of the Council of 16 April 2014 on market abuse (market abuse regulation) and

repealing Directive 2003/6/EC of the European Parliament and of the Council and Commission Directives 2003/124/EC, 2003/125/EC and 2004/72/EC, and the relevant regulations, directives and other rules promulgated thereunder, as well as similar rules and regulations elsewhere in other jurisdictions.

(b) During the Term, the Investor shall, upon written request from the Company, provide the Company within five (5) Business Days, the aggregate number of Equity Securities owned by the Investor, the Parent Investor and their Affiliates to the extent calculable (it being understood that the Warrants can be described without calculating a number of shares issuable thereunder). During the Term, the Company shall provide the Investor with notice of any exercise or conversion of any options, warrants, convertible securities or other right to acquire, directly or indirectly, any of the new shares of the Company by any person or entity (other than the Investor, the Parent Investor and their Affiliates) within five (5) Business Days after such exercise or conversion.

7. EXERCISE OF THE WARRANT

7.1. Exercise Notice

- (a) The Warrant can only be exercised by means of a written notice to the Company (the "**Exercise Notice**"). The Exercise Notice must be served on the Company in accordance with the provisions of Article 10.6.
- (b) The date on which the Exercise Notice with respect to a specific exercise of the Warrant shall have been served (or be deemed served) on the Company pursuant to Article 10.6 shall be the exercise date of the Warrant with respect to that exercise (the "Exercise Date").
- (c) The Exercise Notice must state (i) the number of Shares with respect to which the Warrant is exercised at that occasion, and (ii) the applicable aggregate Exercise Price, as determined in accordance with the provisions of these Conditions. The Exercise Date should fall within the Term.
- (d) If the number of Shares with respect to which the Warrant is exercised as indicated in the Exercise Notice exceeds the then applicable Warrant Limit, the Warrant shall be deemed exercised with respect to the number of Shares just below the then applicable Warrant Limit only. The Company shall as soon as practical, and in any event before the expiry of the term within which the relevant Shares are to be issued to the holder of the Warrant pursuant to Article 7.3 of these Conditions, notify the holder of the Warrant thereof in accordance with the provisions of Article 10.6, which notification shall include the reasoned calculation of the then applicable Warrant Limit.
- (e) Upon receipt of the Exercise Notice, the Company may request the holder of the Warrant in writing to provide to the Company with such further declarations and documents, which are necessary to comply with all applicable legal and regulatory provisions in connection with the exercise of the Warrant and the issue or delivery of the Shares resulting therefrom, including pursuant to Article 6(b) of these Conditions.

7.2. Payment of the Exercise Price

- (a) Upon each exercise of the Warrant, the applicable aggregate Exercise Price must be paid by means of a payment in cash.
- (b) Following the receipt of the Exercise Notice, the Company shall, as promptly as practicable and in any event no later than one (1) Business Day after the Exercise Date, notify the holder of the Warrant of the account onto which the applicable aggregate Exercise Price must be paid (the "Exercise Account").

- (c) The amount of the applicable aggregate Exercise Price must be paid by means of a wire transfer of such amount in immediately available funds in euro to the Exercise Account.
- (d) If the applicable aggregate Exercise Price is not paid in accordance with paragraph (c) within a term of ten (10) Business Days following the date on which the Company shall have notified the holder of the Warrant of the details of the Exercise Account in accordance with paragraph (b), the Warrant shall be deemed not to have been exercised, without prejudice to the right of the holder of the Warrant to exercise the Warrant at later occasions until the Expiry Date subject to and in accordance with the terms and conditions set forth in these Conditions.

7.3. Issue and delivery of the Shares

- (a) The Company shall only be obliged to issue Shares upon an exercise of the Warrant provided that (i) the relevant Exercise Notice has been made in accordance with Article 7.1, and (ii) the applicable aggregate Exercise Price has been paid in accordance with the provisions of Article 7.2. Subject to the foregoing, the Company shall issue or deliver the relevant Shares as soon as practicable, but in any event no later than the later of (x) six (6) calendar days after the Exercise Date, and (y) the day other than (a) a Saturday or a Sunday or (b) a bank or other public holiday in Belgium following the receipt of the applicable aggregate Exercise Price on the Exercise Account.
- (b) The Company shall take all steps and carry out all formalities that shall be required by virtue of these Conditions, the Company's articles of association and applicable law in order to issue the new Shares upon an exercise of the Warrant (without prejudice, however, to the right of the Company to deliver existing Shares in accordance with the provisions of Article 4.2(b)).
- (c) In accordance with applicable law, upon each exercise of the Warrant, the capital increase and issue of new Shares resulting therefrom (as relevant) shall be formally recorded before a notary public by one authorised representative of the Company.

7.4. Allocation of the Exercise Price

- (a) Each time upon an exercise of the Warrant and the issue of new Shares pursuant to these Conditions, the applicable aggregate Exercise Price shall be allocated to the share capital of the Company. If the applicable Exercise Price per Share issued is greater than the fractional value of the existing Shares immediately prior to the capital increase, then the applicable aggregate Exercise Price shall be allocated in such a manner that per Share issued (i) a part of the applicable aggregate Exercise Price equal to the fractional value of the existing Shares immediately prior to the capital increase shall be booked as share capital, and (ii) the balance of the applicable aggregate Exercise Price shall be booked as issue premium. Such issue premium shall be accounted for on the liabilities side of the Company's balance sheet as net equity. The account on which the issue premium shall be booked shall, like the share capital, serve as the guarantee for third parties and, save for the possibility of a capitalisation of those reserves, can only be reduced on the basis of a valid resolution of the general shareholders' meeting passed in the manner required for an amendment to the Company's articles of association.
- (b) Following the issue of new Shares and the capital increase resulting therefrom, each of the Shares (existing and new) shall represent the same fraction of the Company's share capital.

8. ADJUSTMENTS

8.1. General

(a) Notwithstanding Article 501, paragraph 1 of the Belgian Companies Code (or its successor provision Article 7:71, §1 under the Belgian Companies and Associations Code), and with the exception of the issue of any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde), of any securities with voting rights

other than Shares or of any Equity Securities in relation to any Shares with a different fractional value (fractiewaarde) than the then existing Shares or with a par value (nominale waarde) or any securities with voting rights other than Shares and any reclassification of existing Shares (which shall not be permitted pursuant to this Article 8.1(a), the Company may proceed with all actions that it deems appropriate in relation to its capital, its articles of association, its financial condition, even if such actions lead to a reduction of the benefits allocated to the Warrant, including but not limited to, mergers or acquisitions, capital increases or reductions (including those subject to conditions precedent), the incorporation of reserves into the capital with or without the issue of new Shares, the issue of dividends or other distributions, the issue of other Equity Securities and the amendment of arrangements or provisions relating to the distribution of profits or liquidation proceeds, provided, however, that (i) the terms of the Warrant may not be amended without Investor's written consent, and (ii) that Shares issued or issuable under the Warrant shall not be treated differently (had they already been issued at that time) than other Shares already issued. If the rights of the holder of the Warrant are affected by an action or transaction permitted by the immediately preceding sentence, the holder of the Warrant will not be entitled to a change of the Exercise Price, the Reference Exercise Price or the Warrant Limit, an amendment to the Conditions or any other form of compensation (financial or otherwise) unless (i) specifically provided for in Articles 8.2 and 8.3 of these Conditions and/or (ii) such action or transaction was undertaken with the primary purpose of adversely affecting the rights or value of the Warrant.

(b) The provisions of this Article 8 are without prejudice to the provisions of the Subscription Agreement (such as the anti-dilution protection).

8.2. Adjustments for Share Reorganisations

- (a) If at any time as of the Issue Date up to the Expiry Date there is a change of the fractional value of the Shares as a result of a consolidation (or reverse stock split), a subdivision (or stock split) or otherwise, in each such case without increase or reduction of the Company's share capital (each such transaction a "Share Reorganisation"), the Warrant Limit shall not be affected (since it is expressed as a percentage of the actually issued and outstanding Shares). In addition, the Exercise Price of the Warrant (i.e. on a per Share basis) shall be divided by a fraction, (A) the numerator of which is equal to the fractional value of the outstanding Shares of the Company immediately before to the Share Reorganisation, and (B) the denominator of which is equal to the fractional value of the outstanding Shares of the Company immediately after the Share Reorganisation.
- (b) Any adjustment made pursuant to paragraph (a) of this Article 8.2 shall become effective immediately after the effective date of the relevant Share Reorganisation that gives rise to such adjustment. The Company shall notify the holder of the Warrant of such adjustment by written notice as soon as practicable after the effective date of the Share Reorganisation concerned.

8.3. Adjustments for mergers and de-mergers

(a) If at any time as of the Issue Date up to the Expiry Date there is a merger (fusie / fusion) of the Company with or into another legal person or entity whereby the Company is not the surviving entity or a de-merger (splitsing / scission) in whole or in part, whereby in each of these cases the Shares of the Company are exchanged into, or the shareholders of the Company receive, shares, other securities, cash or other property of one or more other legal persons or entities (each such legal person or entity a "Successor Company") as a result of such merger or de-merger (each such transaction a "Succession Transaction"), then the Warrant shall be replaced by (or, if the Company survives, to the holder of the Warrant shall be issued) one or more warrants to be issued by each of the respective Successor Companies that give right to such respective shares, other securities, cash or other property that the holder of the Warrant concerned had been entitled to receive if the Warrant had been exercised in full up to the

Warrant Limit immediately before to the occurrence of the Succession Transaction concerned but after giving effect to dilution upon the exercise or conversion of all rights to acquire Shares that are exercised or converted in connection with such transaction. The number of warrants to be so issued in replacement of the Warrant (or, if the Company survives, to be so issued), as well as the terms and conditions of such warrants (including the Exercise Price and the Warrant Limit) will need to be, as a whole, equivalent to the terms and conditions of the Warrant and have economically substantially the same effect for the holder of the Warrant concerned.

- (b) An adjustment made pursuant to paragraph (a) of this Article 8.3 shall become effective upon, and subject to, the completion of the Succession Transaction that gives rise to such adjustment. The Company (or the respective Successor Companies, as the case may be) shall notify the holder of the Warrant of such adjustment as soon as practicable after the completion of the Succession Transaction concerned.
- (c) In the case of any merger or de-merger of the Company as contemplated by paragraph (a) of this Article 8.3, the Company must procure that the successor or acquiring persons or entities shall expressly assume the due observance and performance of the covenants and obligations set out in the Conditions.

9. REPRESENTATIONS AND WARRANTIES

9.1. Representations and Warranties of the Company

Upon each exercise of the Warrant, the Company shall be deemed to represent and warrant to the holder of the Warrant on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of Belgium, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Validity of Shares and Absence of Breach. Each Share to be issued or delivered by the Company upon such exercise of the Warrant, and, with respect to new Shares only, as of when issued and paid for in accordance with these Conditions, are validly and duly issued and fully paid ordinary shares of the Company in accordance with the applicable provisions of the Company's organisational documents and Belgian law, having the same fractional value (fractiewaarde) as the then existing Shares, and free and clear of all liens, pledges, encumbrances, mortgages, security interests, or easement or transfer restrictions of any nature whatsoever (other than those that find their origin solely with the holder of the Warrant and save for the transfer restrictions referred to in the Subscription Agreement, as the case may be). The issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant will not result in a breach of, default under any material agreement to which the Company is a party or the Company's organisational documents or any law, regulation or stock exchange rule, or give rise to the activation of any material rights of third parties under any agreement, law, rule or regulation binding on the Company or any of its subsidiaries.
- (c) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under Belgian law or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Company with the terms of these Conditions and the Warrant will, save as otherwise set forth in these Conditions in Article 4.3, be in full force and effect.

(d) **Brokers and Finders**. No person will have, as a result of the exercise of the Warrant, any right, interest or claim against or upon the holder of the Warrant for any commission, fee or other compensation relating to the Shares to be issued or delivered by the Company upon such exercise of the Warrant.

9.2. Representations and Warranties of the holder of the Warrant

Upon each exercise of the Warrant, the holder of the Warrant shall be deemed to represent and warrant to the Company on the date of the issue or delivery of the Shares to be issued or delivered by the Company upon such exercise of the Warrant:

- (a) **Incorporation**. It is duly incorporated and validly existing and in good standing under the laws of its jurisdiction of incorporation, with full power and authority to conduct its business and is not in violation of any of the provisions of its organisational documents.
- (b) Consents. All necessary consents, authorisations, notification, actions or things required to be taken, fulfilled or done under the law applicable to its jurisdiction of organisation or incorporation, or any of the Antitrust Laws (as defined in the Option, License and Collaboration Agreement) (including, without limitation, the obtaining of any consent or license or the making of any filing or registration) for the exercise of the Warrant, the actions contemplated by these Conditions and the Warrant or the compliance by the Warrant holder with the terms of these Conditions and the Warrant will, save as otherwise set forth in those Conditions in Article 4.3, be in full force and effect.
- (c) Information. Without taking into account the Warrants or the shares issuable (but not yet issued) thereunder, except to give effect to the issue and delivery of the Shares to be issued or delivered by the Company to the Investor upon the particular exercise of the Warrant that occurs on the date that this representation and warranty is made, the Investor, the Parent Investor, and their respective Affiliates do not, directly or indirectly, own, or have the right to acquire, voting securities of the Company in excess of the Warrant Limit (assuming the exercise, conversion or exchange of any Equity Securities (other than the Warrants except as described in this Article 9.2(c)) held by any of them at that time that are exercisable, convertible or exchangeable into or for shares of the Company at that time) (the resulting number of securities rounded down).

10. MISCELLANEOUS

10.1. Binding nature of the Conditions

In the case of subscription for the Warrant, the subscriber shall be bound by, and deemed to have accepted, the present Conditions. In the event of a transfer of the Warrant (or any right thereto), the acquirer or transferee shall be bound by, and deemed to have accepted, the present Conditions.

10.2. Severability

Whenever possible, the provisions of the Conditions shall be interpreted in such a manner that they are valid and enforceable under the applicable legislation.

If any provision in these Conditions is held to be illegal, invalid or unenforceable, in whole or in part, under any applicable law, then such provision or part of it shall be deemed not to form part of these Conditions, and the legality, validity or enforceability of the remainder of these Conditions shall not be affected.

In that event, the illegal, invalid or non-enforceable provision or part thereof is automatically replaced with the legal, valid and enforceable provision that is the closest to the original provision or part thereof as regards content, bearing and intention.

10.3. Specific Enforcement

Notwithstanding anything in these Conditions to the contrary, nothing in these Conditions shall in any way limit the ability of the Company and the holder of the Warrant to seek or obtain from any court of competent jurisdiction any remedies available at law or in equity (including injunctive relief) to enforce any covenant or agreement of the holder of the Warrant respectively the Company hereunder.

10.4. Costs and expenses

The Company shall pay any stamp, issue, registration, documentary or other taxes and duties, including interest and penalties, payable in Belgium on or in connection with the issue or delivery of the Shares upon each exercise of the Warrant. The Company shall also pay all costs associated with the listing of the relevant Shares on the regulated markets of Euronext Brussels and Amsterdam (and such other regulated markets on which the Company's Shares will be trading at that time).

10.5. Governing law and jurisdiction

The Conditions and the Warrant and any non-contractual obligations arising out of or in connection with each of them are governed by, and are to be construed in accordance with, Belgian law.

Any dispute, controversy, difference or claim which may arise between the Company and the holder of the Warrant out of or in relation to or in connection with the Conditions or the Warrant (including arising out of or relating to the validity, construction, interpretation, enforceability, breach, performance, application, exercise, expiry or termination of the Conditions or the Warrant), shall be settled by binding arbitration in accordance with the applicable rules of the International Chamber of Commerce ("ICC Rules") by three (3) arbitrators, one chosen by the holder of the Warrant, one chosen by the Company and the third chosen by mutual agreement of the first two, and otherwise in accordance with the ICC Rules. The arbitrators shall have significant experience and shall have expertise in Belgian corporate law. Each of the Company and the holder of the Warrant may refer such dispute to arbitration by submitting a written notice of such request to the holder of the Warrant respectively the Company. The place of arbitration shall be New York and the language (including all testimony, evidence and written documentation) shall be English. The arbitrators shall establish procedures to facilitate and complete such arbitration as soon and efficiently as practicable. Unless the arbitrators expressly determine otherwise, neither the Company nor the holder of the Warrant shall be required to give general discovery of documents, but may be required only to produce specific, identified documents that are relevant to the dispute. Each of the Company and the holder of the Warrant shall have the right to be represented by counsel. Any judgment or award rendered by the arbitrators shall be final and binding on the Company and the holder of the Warrant, and shall be governed by the terms and conditions hereof and the limitation on damages set forth in Article 13 of the Subscription Agreement. The parties hereto agree that such a judgment or award may be enforced in any court of competent jurisdiction. The statute of limitations of Belgian law applicable to the commencement of a lawsuit shall apply to the commencement of arbitration. The arbitrators shall determine the allocation of costs and expenses and attorneys' fees in the arbitration to be borne by each of the Company and the holder of the Warrant. All proceedings and decisions of the arbitrators shall be deemed Confidential Information (as set out in the Subscription Agreement) of each of the Company and the holder of the Warrant, and shall be subject to Article 13 of the Option, License and Collaboration Agreement.

10.6. Notices

Any notice, notification, demand or other communication ("**notice**") to be given under these Conditions shall be in writing, shall specifically refer to these Conditions, and shall be addressed to the appropriate party at the address specified below or such other address as may be specified by such party in writing in accordance with this Article 10.6, and shall be deemed to have been given for all purposes (i) when received, if hand-delivered, sent by a reputable international expedited delivery service or sent by facsimile (with transmission confirmed), or (ii) five (5) Business Days after mailing, if mailed by first class certified or registered mail, postage prepaid, return receipt requested. Any notice delivered by facsimile

shall be confirmed by a hard copy delivered by a reputable international expedited delivery service as soon as practicable thereafter. The current details for notices are:

- (a) if to the Company: the address of the Company's registered office, with the notice made for the attention of the General Counsel of the Company, or the address for notices to the Company pursuant to the Subscription Agreement.
- (b) if to the holder of the Warrant: to such holder's address as set out in the warrant register book, or the address for notices to such party (as the case may be) pursuant to the Subscription Agreement.

* * *